FORD MOTOR CO(F)

10-Q

Quarterly report pursuant to sections 13 or 15(d) Filed on 05/07/2010 Filed Period 03/31/2010





UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM 10-Q

[A]	SECURITIES EXCHANGE ACT OF 1934	X IS(u) OF THE
	For the quarterly period ended March 31, 2010	
[]	TRANSITION REPORT PURSUANT TO SECTION 13 O SECURITIES EXCHANGE ACT OF 1934	R 15(d) OF THE
	For the transition period from	to
	Commission File Number: 1-3950	
		OTOR COMPANY gistrant as specified in its charter)
	Delaware (State of Incorporation)	38-0549190 (IRS Employer Identification No.)
	One American Road, Dearborn, Michigan (Address of principal executive offices)	48126 (Zip Code)
	(Registrant's telep	(313) 322-3000 none number, including area code)
durin	Indicate by check mark whether the registrant (1) has filed all rep g the preceding 12 months (or for such shorter period that the registements for the past 90 days. Yes \square No \square	orts required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 strant was required to file such reports), and (2) has been subject to such filing
requii		conically and posted on its corporate Web site, if any, every Interactive Data File S-T ($\S232.405$ of this chapter) during the preceding 12 months (or for such shorter es \boxtimes No \square
comp	any. See the definitions of "large accelerated filer," "accelerated	d filer, an accelerated filer, a non-accelerated filer, or a smaller reporting filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large Smaller reporting company □
1	Indicate by check mark whether the registrant is a shell company	as defined in Rule 12b-2 of the Exchange Act). Yes \square No \square
	As of April 29, 2010, the registrant had outstanding 3,335,728,63	shares of Common Stock and 70,852,076 shares of Class B Stock.
	Exhibit inde	a located on page number 73.

CONSOLIDATED STATEMENT OF OPERATIONS For the Periods Ended March 31, 2010 and 2009 (in millions, except per share amounts)

Sales and revenues	First Q 2010 (unau	2009
Automotive sales	\$28,894	\$20.980
Financial Services revenues	2,672	3,410
Total sales and revenues		24,390
Costs and expenses		
Automotive cost of sales	25,139	21,413
Selling, administrative and other expenses	3,089	3,692
Interest expense	1,701	1,921
Financial Services provision for credit and insurance losses	(41)	
Total costs and expenses	29,888	27,428
Automotive interest income and other non-operating income/(expense), net (Note 10)	189	1,352
Financial Services other income/(loss), net (Note 10)	126	113
Equity in net income/(loss) of affiliated companies	142	(89)
Income/(Loss) before income taxes	2,135	(1,662)
Provision for/(Benefit from) income taxes	50	(227)
Income/(Loss) from continuing operations	2,085	(1,435)
Income/(Loss) from discontinued operations	•	•
Net income/(loss)	2,085	(1,435)
Less: Income/(Loss) attributable to noncontrolling interests	•	(8)
Net income/(loss) attributable to Ford Motor Company	\$ 2,085	\$(1,427)
NET INCOME/(LOSS) ATTRIBUTABLE TO FORD MOTOR COMPANY		
Income/(Loss) from continuing operations	\$ 2,085	\$(1,427)
Income/(Loss) from discontinued operations	<u>•</u>	
Net income/(loss)	\$ 2,085	\$(1,427)
AMOUNTS PER SHARE ATTRIBUTABLE TO FORD MOTOR COMPANY COMMON AND CLASS B STOCK (Note 13) Basic income/(loss)		
Income/(Loss) from continuing operations	\$ 0.62	\$ (0.60)
Income/(Loss) from discontinued operations	•	• (0.00)
Net income/(loss)	\$ 0.62	\$ (0.60)
Diluted income/(loss)		
Income/(Loss) from continuing operations	\$ 0.50	\$ (0.60)
Income/(Loss) from discontinued operations		•
Net income/(loss)	\$ 0.50	\$ (0.60)

SECTOR STATEMENT OF OPERATIONS For the Periods Ended March 31, 2010 and 2009 (in millions, except per share amounts)

	First 2010 (unau	Quarter 2009 dited)
AUTOMOTIVE		
Sales	\$ 28,894	\$ 20,980
Costs and expenses	25 120	21 412
Cost of sales Selling, administrative and other expenses	25,139 2,220	21,413
<u>. </u>		
Total costs and expenses	27,359	23,422
Operating income/(loss)	1,535	(2,442)
Interest expense	542	469
Interest income and other non-operating income/(expense), net (Note 10)	189	1,352
Equity in net income/(loss) of affiliated companies	138	49
Income/(Loss) before income taxes • Automotive	1,320	(1,510)
income/Loss) before income taxes • Automotive	1,320	(1,510)
FINANCIAL SERVICES		
Revenues	2,672	3,410
Costs and expenses	_,~	2,120
Interest expense	1,159	1,452
Depreciation	660	1,435
Operating and other expenses	209	248
Provision for credit and insurance losses	(41)	402
Total costs and expenses	1,987	3,537
Other income/(loss), net (Note 10)	126	113
Equity in net income/(loss) of affiliated companies	4	(138)
Income/(Loss) before income taxes • Financial Services	815	(152)
TOTAL COMPANY		
Income/(Loss) before income taxes	2.135	(1,662)
Provision for/(Benefit from) income taxes	50	(227)
Income/(Loss) from continuing operations	2,085	(1,435)
Income/(Loss) from discontinued operations	•	•
Net income/(loss)	2,085	(1.435)
Less: Income/(Loss) attributable to noncontrolling interests	•	(8)
Net income/(loss) attributable to Ford Motor Company	\$ 2,085	\$ (1,427)
NET INCOME/(LOSS) ATTRIBUTABLE TO FORD MOTOR COMPANY		
Income/(Loss) from continuing operations	\$ 2,085	\$ (1,427)
Income/(Loss) from discontinued operations		<u> </u>
Net income/(loss)	\$ 2,085	\$ (1,427)
AMOUNTS PER SHARE ATTRIBUTABLE TO FORD MOTOR COMPANY COMMON AND CLASS B STOCK (Note 13)		
Basic income/(loss)		
Income/(Loss) from continuing operations	\$ 0.62	\$ (0.60)
Income/(Loss) from discontinued operations	•	•
Net income/(loss)	\$ 0.62	\$ (0.60)
Diluted income/(loss)		
Income/(Loss) from continuing operations	\$ 0.50	\$ (0.60)
Income/(Loss) from discontinued operations	•	•
Net income/(loss)	\$ 0.50	\$ (0.60)

CONSOLIDATED BALANCE SHEET (in millions)

	March 31, 2010		December 31, 2009
A CONTINU	(uı	naudited)	
ASSETS Cash and cash equivalents	\$	24.356	\$ 20.894
Marketable securities	Ψ	21,883	21,387
Finance receivables, net (Note 4)		73,837	76,996
Other receivables, net		6,925	7,257
Net investment in operating leases		15.818	17.270
Inventories (Note 5)		6,292	5,041
Equity in net assets of affiliated companies		2,544	2,428
Net property		22,826	22,637
Deferred income taxes		2,861	3,479
Goodwill and other net intangible assets (Note 7)		192	208
Assets of held-for-sale operations (Note 12)		8,076	7,618
Other assets		6,358	6,825
Total assets	\$	191,968	\$ 192,040
LIABILITIES			
Pavables	\$	15,611	\$ 14.301
Accrued liabilities and deferred revenue	Þ	44.445	46.144
Debt (Note 9)		130,105	131,635
Deferred income taxes		1,600	2,421
Liabilities of held-for-sale operations (Note 12)		5,644	5,321
Total liabilities		197,405	199.822
Total natinues		197,403	199,822
EQUITY			
Capital stock			
Common Stock, par value \$0.01 per share (3,345 million shares issued) Class B Stock, par value \$0.01 per share (71 million shares issued)		33	33
Capital in excess of par value of stock		17,382	16,786
Accumulated other comprehensive income/(loss)		(11,199)	(10,864)
Treasury stock		(178)	(177)
Retained earnings/(Accumulated deficit)		(11,514)	(13,599)
Total equity/(deficit) attributable to Ford Motor Company (Note 17)		(5,475)	(7,820)
Equity/(Deficit) attributable to noncontrolling interests (Note 17)		38	38
Total equity/(deficit) (Note 17)		(5,437)	(7,782)
Total liabilities and equity	\$	191,968	\$ 192,040

The following table includes assets to be used to settle liabilities of the consolidated variable interest entities ("VIEs"). These assets and liabilities are included in the consolidated balance sheet above. See Note 6 for additional information on our VIEs.

ASSETS		
Cash and cash equivalents	\$ 5,117	\$ 4,922
Finance receivables, net	54,471	57,353
Other receivables, net	25	34
Net investment in operating leases	10,765	10,246
Inventories	22	106
Net property	31	154
Other assets	39	56
LIABILITIES		
Payables	22	23
Accrued liabilities and deferred revenue	504	560
Debt	47.929	46.167

SECTOR BALANCE SHEET (in millions)

	March 31, 2010		Dec	ember 31, 2009
	(ur	naudited)		
ASSETS Automotive				
Cash and cash equivalents	\$	12,801	\$	9,762
Marketable securities		12,491		15,169
Total cash and marketable securities		25,292		24,931
Receivables, net		3,332		3,378
Inventories (Note 5) Deferred income taxes		6,292 493		5,041 479
Other current assets		2,797		2,832
Current receivable from Financial Services		2,834		2,568
Total current assets		41,040		39,229
Equity in net assets of affiliated companies		2,420		2,307
Net property Deferred income taxes		22,655 5,600		22,455 5,660
Goodwill and other net intangible assets (Note 7)		183		199
Assets of held-for-sale operations (Note 12)		8,076		7,618
Other assets		1,726		1,650
Non-current receivable from Financial Services		256		50.110
Total Automotive assets		81,956		79,118
Financial Services Cash and cash equivalents		11,555		11,132
Marketable securities		9,824		6,864
Finance receivables, net (Note 4)		77,439		80,885
Net investment in operating leases		13,780		15,062
Equity in net assets of affiliated companies Goodwill and other net intangible assets (Note 7)		124		121
Other assets		4,330		5,039
Total Financial Services assets		117,061		119,112
Intersector elimination		(3,532)		(3,224)
Total assets	\$	195,485	\$	195,006
LIABILITIES				
Automotive				
Trade payables	\$	11,898	\$	10,599
Other payables		2,480		2,466
Accrued liabilities and deferred revenue Deferred income taxes		17,642 2,949		18,138 3,091
Debt payable within one year (Note 9)		5,009		1,638
Total current liabilities		39,978		35,932
Long-term debt (Note 9)		29,242		31,972
Other liabilities		22,337		23,132
Deferred income taxes Liabilities of held-for-sale operations (Note 12)		524 5,644		561 5,321
Total Automotive liabilities		97,725		96,918
Financial Services		91,123		90,918
Payables		1,233		1,236
Debt (Note 9)		96,286		98,671
Deferred income taxes		1,644		1,735
Other liabilities and deferred income Payable to Automotive		4,476 3,090		4,884 2,568
Total Financial Services liabilities		106,729		109,094
Intersector elimination		(3,532)		(3,224)
Total liabilities		200,922		202,788
EQUITY				
Capital stock		22		22
Common Stock, par value \$0.01 per share (3,345 million shares issued) Class B Stock, par value \$0.01 per share (71 million shares issued)		33		33
Capital in excess of par value of stock		17,382		16,786
Accumulated other comprehensive income/(loss)		(11,199)		(10,864)
Treasury stock		(178)		(177)
Retained earnings/(Accumulated deficit)		(11,514)		(13,599)
Total equity/(deficit) attributable to Ford Motor Company (Note 17)		(5,475)		(7,820)
Equity/(Deficit) attributable to noncontrolling interests (Note 17)		(5,437)		(7,782)
Total equity/(deficit) (Note 17) Total liabilities and equity	<u>¢</u>	195,485	\$	195,006
Town nations and equity	Ψ	173,403	Ψ	193,000

CONDENSED CONSOLIDATED STATEMENT OF CASH FLOWS For the Periods Ended March 31, 2010 and 2009 (in millions)

		First Quarter		
		2010 (unaud		2009
				audited)
Cash flows from operating activities of continuing operations		2 402		1015
Net cash (used in)/provided by operating activities	\$	2,683	\$	4,016
Cash flows from investing activities of continuing operations				
Capital expenditures		(1.068)		(1,069)
Acquisitions of retail and other finance receivables and operating leases		(6,979)		(6,032)
Collections of retail and other finance receivables and operating leases		9,602		10,047
Purchases of securities		(18,341)		(22,002)
Sales and maturities of securities		17,987		19,071
Settlements of derivatives		46		1,163
Proceeds from sale of businesses		•		166
Cash change due to deconsolidation of joint ventures		•		(343)
Other		(80)		(336)
Net cash (used in)/provided by investing activities		1,167		665
Cash flows from financing activities of continuing operations				
Sales of Common Stock		530		•
Changes in short-term debt		(1,042)		(3,869)
Proceeds from issuance of other debt		8,827		15,410
Principal payments on other debt		(8,506)		(16,355)
Other		79		(87)
Net cash (used in)/provided by financing activities		(112)		(4,901)
Effect of exchange rate changes on cash		(276)		(310)
Cumulative correction of Financial Services prior period error		•		(630)
Net increase/(decrease) in cash and cash equivalents from continuing operations		3,462		(1,160)
Cash flows from discontinued operations				
Cash flows from operating activities of discontinued operations		•		•
Cash flows from investing activities of discontinued operations		•		•
Cash flows from financing activities of discontinued operations		<u> </u>		<u>.</u>
Net increase/(decrease) in cash and cash equivalents	<u>\$</u>	3,462	\$	(1,160)
Cash and cash equivalents at January 1	\$	20,894	\$	21,804
Cash and cash equivalents of discontinued/held-for-sale operations at January 1	Ψ	•	Ψ	•
Net increase/(decrease) in cash and cash equivalents		3,462		(1,160)
Less: cash and cash equivalents of discontinued/held-for-sale operations at March 31		•		(-,-30)
Cash and cash equivalents at March 31	\$	24,356	\$	20,644
Cush und Cush equituelles at Francis of	<u> </u>	24,330	Ψ	20,077

CONDENSED SECTOR STATEMENT OF CASH FLOWS For the Periods Ended March 31, 2010 and 2009 (in millions)

	First Qu	arter 2010	First Quarter 2009				
	Automotive	Financial Services		Financial Services			
	(una	udited)	(unaudi	ited)			
Cash flows from operating activities of continuing operations	Φ (0	2 0 1400	¢ (2.410)	¢ 1.011			
Net cash (used in)/provided by operating activities	\$ 60	3 \$ 1,488	\$ (2,410)	\$ 1,911			
Cash flows from investing activities of continuing operations							
Capital expenditures	(1,06	4) (4	(1,064)	(5)			
Acquisitions of retail and other finance receivables and operating leases	(1,00	• (6,979		(6.032)			
Collections of retail and other finance receivables and operating leases		• 9,673		10.124			
Net (acquisitions)/collections of wholesale receivables		• 521		4,438			
Purchases of securities	(9,10			(5,544)			
Sales and maturities of securities	11,91			5,854			
Settlements of derivatives	(12			921			
Proceeds from sale of businesses	`	• •	1	165			
Cash change due to deconsolidation of joint ventures		• •	(343)	•			
Other	(7) (73		(9)			
Net cash (used in)/provided by investing activities	1.61		(5,652)	9,912			
Not cash (used my provided by investing activities	1,01	0 337	(3,032)),)12			
Cash flows from financing activities of continuing operations							
Sales of Common Stock	53	0 •	•	•			
Changes in short-term debt	26		359	(4,228)			
Proceeds from issuance of other debt	31			5,272			
Principal payments on other debt	(11			(15,285)			
Other	11			(15)			
Net cash (used in)/provided by financing activities	1.10			(14,256)			
Net easi (used in)/provided by infaileing activities	1,10	0 (1,737) 10,273	(14,230)			
Effect of exchange rate changes on cash	(8	0) (196	(102)	(208)			
Net change in intersector receivables/payables and other liabilities	(20			590			
Cumulative correction of prior period error	,	•	•	(630)			
Net increase/(decrease) in cash and cash equivalents from continuing operations	3,03	9 423	1,521	(2,681)			
reconcenses (decrease) in cash and cash equivalents from continuing operations	3,03	, 123	1,521	(2,001)			
Cash flows from discontinued operations							
Cash flows from operating activities of discontinued operations		•	•	•			
Cash flows from investing activities of discontinued operations		•	•	•			
Cash flows from financing activities of discontinued operations		•	•	•			
Net increase/(decrease) in cash and cash equivalents	\$ 3,03	9 \$ 423	\$ 1,521	\$ (2,681)			
· · · · · · · · · · · · · · · · · · ·							
Cash and cash equivalents at January 1	\$ 9.76	2 \$ 11.132	\$ 6.132	\$ 15.672			
Cash and cash equivalents of discontinued/held-for-sale operations at January 1	φ ,,,,	• 11,132	• 0,132	• 13,072			
Net increase/(decrease) in cash and cash equivalents	3.03	9 423	1.521	(2,681)			
Less: cash and cash equivalents of discontinued/held-for-sale operations at March 31	3,03	• •	1,521	(2,001)			
Cash and cash equivalents at March 31	\$ 12,80	1 \$ 11,555	\$ 7,653	\$ 12.991			
Cash and Cash equivalents at Iviai Ch 31	9 12,80	<u> </u>	φ 7,033	Ψ 12,791			

CONSOLIDATED STATEMENT OF COMPREHENSIVE INCOME For the Periods Ended March 31, 2010 and 2009 (in millions)

	First Quarter			
	2010			2009
		(unau	dited)	
Net income/(loss)	\$	2,085	\$	(1,435)
Other comprehensive income/(loss), net of tax:				
Foreign currency translation		(489)		(447)
Net gain/(loss) on derivative instruments		(1)		(87)
Employee benefit-related		157		(5)
Net holding gain/(loss)		<u>(2)</u>		<u>(1</u>)
Total other comprehensive income/(loss), net of tax		(335)		(540)
Comprehensive income/(loss)		1,750		(1,975)
Less: Comprehensive income/(loss) attributable to noncontrolling interests (Note 17)		<u>·</u>		<u>(8)</u>
Comprehensive income/(loss) attributable to Ford Motor Company	\$	1,750	\$	(1,967)

FORD MOTOR COMPANY AND SUBSIDIARIES NOTES TO THE FINANCIAL STATEMENTS

Footnote		Page
Note 1	Principles of Presentation and Consolidation	10
Note 2	Fair Value Measurements	13
Note 3	Cash and Restricted Cash	21
Note 4	Finance Receivables - Financial Services Sector	22
Note 5	Inventories	22
Note 6	Variable Interest Entities	22
Note 7	Goodwill and Other Net Intangible Assets	26
Note 8	Retirement Benefits	27
Note 9	Debt and Commitments	28
Note 10	Other Income/(Loss)	36
Note 11	Income Taxes	36
Note 12	Held-For-Sale Operations, Discontinued Operations, Other Dispositions, and Acquisitions	36
Note 13	Amounts Per Share Attributable to Ford Motor Company Common and Class B Stock	38
Note 14	Derivative Financial Instruments and Hedging Activities	39
Note 15	Segment Information	44
Note 16	Commitments	44
Note 17	Equity/(Deficit) Attributable to Ford Motor Company and Noncontrolling Interests	46

NOTE 1. PRINCIPLES OF PRESENTATION AND CONSOLIDATION

Our financial statements are presented in accordance with generally accepted accounting principles ("GAAP") in the United States for interim financial information, and instructions to the Quarterly Report on Form 10-Q and Rule 10-01 of Regulation S-X. We show certain of our financial statements on both a consolidated and a sector basis for our Automotive and Financial Services sectors. All intercompany items and transactions have been eliminated in both the consolidated and sector basis financial statements. Reconciliations of certain line items are explained below in this Note, where the presentation of these intercompany eliminations or consolidated adjustments differs between the consolidated and sector financial statements.

In the opinion of management, these unaudited financial statements reflect a fair statement of the results of operations and financial condition of Ford Motor Company and its consolidated subsidiaries and consolidated VIEs of which we are the primary beneficiary for the periods and at the dates presented. The operating results for interim periods are not necessarily indicative of results that may be expected for any other interim period or for the full year. Reference should be made to the financial statements contained in our Annual Report on Form 10-K for the year ended December 31, 2009, updated by the revised disclosures in Exhibit 99 to our Current Report on Form 8-K dated May 7, 2010 showing retrospective application of the new accounting standard on VIE consolidation effective January 1, 2010 ("2009 Form 10-K Report"). For purposes of this report, "Ford," the "Company," "we," "our," "us" or similar references mean Ford Motor Company and our consolidated subsidiaries and our consolidated VIEs of which we are the primary beneficiary, unless the context requires otherwise. All held-for-sale assets and liabilities are excluded from the footnotes unless otherwise noted. See Note 12 for details of held-for-sale operations.

In the first quarter of 2009, our wholly-owned subsidiary Ford Motor Credit Company LLC ("Ford Credit") recorded a \$630 million cumulative adjustment to correct for the overstatement of Financial Services sector cash and cash equivalents and certain accounts payable that originated in prior periods. The impact on previously-issued annual and interim financial statements was not material.

Adoption of New Accounting Standards

Fair Value Measurements. We adopted the new accounting standard on fair value measurements on January 1, 2010 which both requires new disclosures and clarifies existing disclosure requirements. The standard requires a greater level of disaggregated information in the fair value hierarchy and expands disclosures about valuation techniques and inputs to measure fair value. Refer to Note 2 for further information regarding our fair value measurements.

Transfers of Financial Assets. On January 1, 2010, we adopted the new accounting standard related to transfers of financial assets. The standard provides greater transparency about transfers of financial assets and a company's continuing involvement in the transferred financial assets. The standard also removes the concept of a qualifying special-purpose entity from U.S. GAAP and changes the requirements for derecognizing financial assets. The new accounting standard did not have a material impact on our financial condition, results of operations, or financial disclosures.

Variable Interest Entities. On January 1, 2010, we adopted the new accounting standard on VIEs. The standard requires ongoing assessments of whether an entity is the primary beneficiary of a VIE, and enhances the disclosures about an entity's involvement with a VIE. This standard requires the consolidation of a VIE if an entity has both (i) the power to direct the activities of the VIE, and (ii) the obligation to absorb losses or the right to receive residual returns that could potentially be significant to the VIE.

In applying this new standard, we deconsolidated certain entities. These entities were primarily Automotive joint ventures previously consolidated due to contractual agreements that resulted in a disproportionate relationship between our voting interest in these entities and our exposure to the economic risks and potential rewards of the entities. We held a majority of the variable interests in the VIEs, and therefore, were deemed to be the primary beneficiary. We did not, however, possess the power to direct the activities of the VIEs that most significantly impacted the VIEs' economic performance. Due to the absence of this power, adoption of the new standard resulted in the deconsolidation of the majority of these Automotive joint ventures. The most significant Automotive joint ventures deconsolidated were Ford Otomotiv Sanayi Anonim Sirketi ("Ford Otosan") and AutoAlliance, Inc. ("AAI"). Ford Otosan is a joint venture between Ford (41% partner), Koc Group of Turkey (41% partner), and public investors (18%). AAI is a joint venture between Ford (50% partner) and Mazda Motor Corporation ("Mazda") (50%) in North America. We concluded in each case that the power to direct the activities that most significantly impact the entity's economic performance were shared equally among unrelated parties. As a result, we account for the ownership in each of these joint ventures as equity method investments.

The new accounting standard did not result in any deconsolidation or consolidation of new entities within our Financial Services sector.

NOTE 1. PRINCIPLES OF PRESENTATION AND CONSOLIDATION (Continued)

Refer to Note 6 for further information regarding our VIEs. We have retrospectively applied this new accounting standard and revised our prior year financial statements herein accordingly.

The following table sets forth selected financial data as the data would have appeared had we applied the new consolidation standard for the first quarter of 2009, compared to the originally reported amount in our Quarterly Report on Form 10-Q for the period ended March 31, 2009 (dollar amounts in millions, except for per share amounts). As noted, 2009 data herein have been adjusted to reflect the new accounting standard on VIE consolidation.

	First Quarter 2009					
		Revised		As Originally evised Reported		fect of nange
SUMMARY OF OPERATIONS Total Company			'			
Sales and revenues	\$	24,390	\$	24,778	\$	(388)
Income/(Loss) before income taxes	\$	(1,662)	\$	(1,620)	\$	(42)
Provision for/(Benefit from) income taxes		(227)		(204)		(23)
Income/(Loss) from continuing operations		(1,435)		(1,416)		(19)
Income/(Loss) from discontinued operations		<u> </u>		<u> </u>		<u> </u>
Net income/(loss)		(1,435)		(1,416)		(19)
Less: Income/(Loss) attributable to noncontrolling interests		(8)		<u>11</u>		(19)
Net income/(loss) attributable to Ford Motor Company	\$	(1,427)	\$	(1,427)	\$	
Automotive Sector						
Sales	\$	20,980	\$	21,368	\$	(388)
Operating income/(loss)		(2,442)		(2,338)		(104)
Income/(Loss) before income taxes		(1,510)		(1,468)		(42)
Amounts Per Share Attributable to Ford Motor Company Common and Class B Stock						
Basic:	\$	(0, (0)	¢.	(0, (0)	\$	
Income/(Loss) from continuing operations Income/(Loss) from discontinued operations	Þ	(0.60)	\$	(0.60)	Þ	•
, ,	Ф	(0.60)	¢	(0.60)	c	
Net income/(loss)	<u> </u>	(0.60)	<u>э</u>	(0.60)	<u>»</u>	
Diluted:	_		_			
Income/(Loss) from continuing operations	\$	(0.60)	\$	(0.60)	\$	•
Income/(Loss) from discontinued operations		<u> </u>	_	<u> </u>		·
Net income/(loss)	\$	(0.60)	\$	(0.60)	\$:

Reconciliations between Consolidated and Sector Financial Statements

Deferred Tax Assets and Liabilities. The difference between the total assets and total liabilities as presented in our sector balance sheet and consolidated balance sheet is the result of netting of deferred income tax assets and liabilities. The reconciliation between total sector and consolidated balance sheets is as follows (in millions):

	rch 31, 2010	D	ecember 31, 2009
Sector balance sheet presentation of deferred income tax assets:	 		
Automotive sector current deferred income tax assets	\$ 493	\$	479
Automotive sector non-current deferred income tax assets	5,600		5,660
Financial Services sector deferred income tax assets*	 285		306
Total	6,378		6,445
Reclassification for netting of deferred income taxes	(3,517)		(2,966)
Consolidated balance sheet presentation of deferred income tax assets	\$ 2,861	\$	3,479
Sector balance sheet presentation of deferred income tax liabilities:			
Automotive sector current deferred income tax liabilities	\$ 2,949	\$	3,091
Automotive sector non-current deferred income tax liabilities	524		561
Financial Services sector deferred income tax liabilities	 1,644		1,735
Total	5,117		5,387
Reclassification for netting of deferred income taxes	(3,517)		(2,966)
Consolidated balance sheet presentation of deferred income tax liabilities	\$ 1,600	\$	2,421

^{*} Financial Services deferred income tax assets are included in Financial Services other assets on our sector balance sheet.

NOTE 1. PRINCIPLES OF PRESENTATION AND CONSOLIDATION (Continued)

Debt Reduction Actions

Automotive Acquisition of Financial Services Debt. During 2008 and 2009, we issued 159,913,115 shares of Ford Common Stock through an equity distribution agreement and used the proceeds of \$1 billion to purchase Ford Credit debt and related interest of \$20 million. We recognized a gain on extinguishment of debt of \$68 million on the transactions, recorded in Automotive interest income and other non-operating income/(expense), net. As of March 31, 2010, approximately \$349 million of the debt purchased has matured (\$214 million and \$135 million matured in the first quarter of 2010 and 2009, respectively), and \$267 million was repurchased (during the third quarter of 2009) from us by Ford Credit.

On our consolidated balance sheet, we net the remaining debt purchased by us with the outstanding debt of Ford Credit, reducing our consolidated marketable securities and debt balances by \$432 million and \$646 million at March 31, 2010 and December 31, 2009, respectively. On our sector balance sheet, the acquisition is reported separately as *Automotive marketable securities* and *Financial Services debt* as it has not been retired or cancelled by Ford Credit.

Financial Services Acquisition of Automotive Debt. During the first quarter of 2009, the Financial Services sector acquired \$2.2 billion principal amount of Automotive secured term loan for an aggregate cost of \$1.1 billion (including transaction costs). This transaction settled on March 27, 2009, following which Ford Credit distributed the secured term loan to its immediate parent, Ford Holdings LLC ("Ford Holdings"), whereupon the debt was forgiven. As a result, we recorded a gain on extinguishment of debt in the amount of \$1.1 billion, net of transaction costs, in Automotive interest income and other non-operating income/(expense), net. At March 31, 2010, the debt acquired is no longer outstanding.

See the table below for the reconciliation between total sector and consolidated cash flows.

NOTE 1. PRINCIPLES OF PRESENTATION AND CONSOLIDATION (Continued)

Sector to Consolidated Cash Flow Reconciliation. We present certain cash flows from the wholesale receivables, finance receivables and the debt reduction actions differently on our sector and consolidated statements of cash flows. The reconciliation between total sector and consolidated cash flows is as follows (in millions):

Automotive cash flows from operating activities of continuing operations Financial Services cash flows from operating activities of continuing operations Total sector cash flows from operating activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Automotive cash flows from operating activities of continuing operations Financial Services cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) (521) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisitio		First (<u> Juarter</u>
Financial Services cash flows from operating activities of continuing operations Total sector cash flows from operating activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Consolidated cash flows from operating activities of continuing operations Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from investing activities of continuing operations Financial Services from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Automotive s		2010	2009
Total sector cash flows from operating activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Automotive cash flows from operating activities of continuing operations Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Finance receivables (b) Finance receivables (c) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Financial Services sector acquisition of Continuing operations Financial Services cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclas	Automotive cash flows from operating activities of continuing operations	\$ 603	\$ (2,410)
Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Consolidated cash flows from operating activities of continuing operations Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Finance receivables (b) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Financial Services sector acquisition of Continuing operations Financial Services cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) 4 1,054	Financial Services cash flows from operating activities of continuing operations	1,488	1,911
Wholesale receivables (a) Finance receivables (b) Consolidated cash flows from operating activities of continuing operations Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Financial Services sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Financial Services sector acquisition of Automotive sector debt (d) Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Automotive cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Total sector cash flows from operating activities of continuing operations	2,091	(499)
Finance receivables (b) Consolidated cash flows from operating activities of continuing operations Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Consolidated cash flows from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automoti	Reclassifications from investing to operating cash flows:		
Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Financial Services sector debt (d) Automotive			
Automotive cash flows from investing activities of continuing operations Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) (71) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Reclassifications from investing to financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Finance receivables (b)	71	77
Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Automotive cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Automotive sector debt (d) Automotive sector acquisition of Automotive sector debt (d) Automotive sector acquisition of Automotive sector debt (d)	Consolidated cash flows from operating activities of continuing operations	\$ 2,683	\$ 4,016
Financial Services cash flows from investing activities of continuing operations Total sector cash flows from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) Finance receivables (b) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Automotive cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Financial Services sector debt (d) Automotive sector acquisition of Automotive sector debt (d) Automotive sector acquisition of Automotive sector debt (d) Automotive sector acquisition of Automotive sector debt (d)			
Total sector cash flows from investing activities of continuing operations Reclassifications from investing to operating cash flows: Wholesale receivables (a) (521) (4,438) Finance receivables (b) (71) (77) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) (214) (134) Financial Services sector acquisition of Automotive sector debt (d) • 1,054 Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Automotive cash flows from investing activities of continuing operations	\$ 1,616	\$ (5,652)
Reclassifications from investing to operating cash flows: Wholesale receivables (a) (521) (4,438) Finance receivables (b) (71) (77) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) (214) (134) Financial Services sector acquisition of Automotive sector debt (d) • 1,054 Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Financial Services cash flows from investing activities of continuing operations	357	9,912
Reclassifications from investing to operating cash flows: Wholesale receivables (a) (521) (4,438) Finance receivables (b) (71) (77) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) (214) (134) Financial Services sector acquisition of Automotive sector debt (d) • 1,054 Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Total sector cash flows from investing activities of continuing operations	1,973	4,260
Finance receivables (b) (71) (77) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) (214) (134) Financial Services sector acquisition of Automotive sector debt (d) • 1,054 Consolidated cash flows from investing activities of continuing operations \$1,167 \$665 Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)		ĺ	,
Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Automotive sector acquisition of Automotive sector debt (d) (214) (134) (134) (14,256) (1,434) (14,256) (326) (3,981) (317)		(521)	(4,438)
Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) 1,054 Consolidated cash flows from investing activities of continuing operations		(71)	(77)
Financial Services sector acquisition of Automotive sector debt (d) Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) 1,054 \$ 1,167 \$ 665 (1,434) (14,256) (3,981) Financial Services rector acquisition of Financial Services sector debt (c) 134 134 Financial Services sector acquisition of Automotive sector debt (d)			
Consolidated cash flows from investing activities of continuing operations Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) 1,167 10,275 11,108 10,275 (3,981) 134 Financial Services sector acquisition of Automotive sector debt (d) 134 134		(214)	
Automotive cash flows from financing activities of continuing operations Financial Services cash flows from financing activities of continuing operations Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) 11,108 \$ 10,275 (1,434) (14,256) (3,981) (3,981) (3,981) (3,981)	1		1,054
Financial Services cash flows from financing activities of continuing operations (1,434) (14,256) Total sector cash flows from financing activities of continuing operations (326) (3,981) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) • (1,054)	Consolidated cash flows from investing activities of continuing operations	\$ 1,167	<u>\$ 665</u>
Financial Services cash flows from financing activities of continuing operations (1,434) (14,256) Total sector cash flows from financing activities of continuing operations (326) (3,981) Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) 214 134 Financial Services sector acquisition of Automotive sector debt (d) (1,054)			
Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) (326) (3,981) 134 (1,054)			
Total sector cash flows from financing activities of continuing operations Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) (326) (3,981) 134 Financial Services sector acquisition of Automotive sector debt (d) (1,054)	Financial Services cash flows from financing activities of continuing operations	(1,434)	(14,256)
Reclassifications from investing to financing cash flows: Automotive sector acquisition of Financial Services sector debt (c) Financial Services sector acquisition of Automotive sector debt (d) 134 (1,054)			
Financial Services sector acquisition of Automotive sector debt (d) (1,054)			· · · · ·
	Automotive sector acquisition of Financial Services sector debt (c)	214	134
Consolidated cash flows from financing activities of continuing operations $\frac{(112)}{(4,901)}$	Financial Services sector acquisition of Automotive sector debt (d)	•	(1,054)
	Consolidated cash flows from financing activities of continuing operations	\$ (112)	\$ (4,901)

- (a) In addition to the cash flow from vehicles sold by us, the cash flow from wholesale finance receivables (being reclassified from investing to operating) includes financing by Ford Credit of used and non-Ford vehicles. 100% of cash flows from wholesale finance receivables have been reclassified for consolidated presentation as the portion of these cash flows from used and non-Ford vehicles is impracticable to separate.
- (b) Includes cash flows of finance receivables purchased/collected from certain divisions and subsidiaries of the Automotive sector.
- (c) See "Debt Reduction Actions" above for further discussion. Cash inflows related to these transactions are reported as financing activities on the consolidated statement of cash flows and investing activities on the sector statement of cash flows. For first quarter 2009, this includes \$135 million related to the maturity of Ford Credit debt previously acquired by our Automotive sector.
- (d) See "2009 Secured Term Loan Actions" within the Automotive section of Note 9 for further discussion of these transactions. Cash outflows related to these transactions are reported as financing activities on the consolidated statement of cash flows and investing or operating activities on the sector statement of cash flows.

NOTE 2. FAIR VALUE MEASUREMENTS

Certain assets and liabilities are presented on our financial statements at fair value. Assets and liabilities measured at fair value on a recurring basis on our balance sheet include cash equivalents, marketable securities, derivative financial instruments and retained interest in securitized assets. Assets and liabilities measured at fair value on a recurring basis for disclosure only include finance receivables and debt. Fair value of these items are presented together with the related carrying value in Notes 4 and 9, respectively. Assets and liabilities measured at fair value on a recurring basis vary based on specific circumstances such as impairments.

Fair Value Measurements

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The fair value should be based on assumptions that market participants would use, including a consideration of non-performance risk. In determining fair value, we use various valuation methodologies and prioritize the use of observable inputs. We assess the inputs used to measure fair value using a three-tier hierarchy based on the extent to which inputs used in measuring fair value are observable in the market:

- Level 1 inputs include quoted prices for identical instruments and are the most observable.
- Level 2 inputs include quoted prices for similar assets and observable inputs such as interest rates, currency exchange rates and yield curves.
- Level 3 inputs are not observable in the market and include management's judgments about the assumptions market participants would use in pricing the asset or liability.

For instruments measured using Level 3 inputs, a reconciliation of the beginning and ending balances is disclosed.

Valuation Methodologies

Cash, Cash Equivalents and Marketable Securities. Cash and all highly liquid investments with a maturity of 90 days or less at date of purchase are classified as Cash and cash equivalents. Investments in securities with a maturity date greater than 90 days at the date of purchase are classified as Marketable securities. Cash on hand, time deposits, certificates of deposit, and money market accounts are reported at par value, which approximates fair value. For other investment securities, we generally measure fair value based on a market approach using prices obtained from pricing services. We review all pricing data for reasonability and observability of inputs. Pricing methodologies and inputs to valuation models used by the pricing services depend on the security type (i.e., asset class). Where possible, fair values are generated using market inputs including quoted prices (the closing price in an exchange market), bid prices (the price at which a dealer stands ready to purchase) and other market information. For securities that are not actively traded, the pricing services obtain quotes for similar fixed-income securities or utilize matrix pricing, benchmark curves or other factors to determine fair value. In certain cases, when observable pricing data is not available, we estimate the fair value of investment securities based on an income approach using industry standard valuation models and estimates regarding non-performance risk.

Derivative Financial Instruments. Our derivatives are over-the-counter customized derivative transactions and are not exchange traded. We estimate the fair value of these instruments based on an income approach using industry standard valuation models. These models project future cash flows and discount the future amounts to a present value using market-based expectations for interest rates, foreign exchange rates and the contractual terms of the derivative instruments. The discount rate used is the relevant interbank deposit rate (e.g., LIBOR) plus an adjustment for non-performance risk. The adjustment reflects the full credit default swap ("CDS") spread applied to a net exposure, by counterparty (considering effects of collateral). We use our counterparty's CDS spread when we are in a net asset position and our own CDS spread when we are in a net liability position.

In certain cases, market data are not available and we use management judgment to develop assumptions which are used to determine fair value. This includes situations where there is illiquidity for a particular currency or commodity or for longer-dated instruments. Also, for interest rate swaps and cross-currency interest rate swaps used in securitization transactions, the notional amount of the swap is based on actual payments on the securitized contracts. We use management judgment to estimate the timing and amount of the swap cash flows based on historical pre-payment speeds.

Retained Interests in Securitized Assets. We estimate the fair value of retained interests based on income approach using internal valuation models. These models project future cash flows of the monthly collections on the sold finance receivables in excess of amounts needed for payment of the debt and other obligations issued or arising in the securitization transactions. The projected cash flows are discounted to a present value based on market inputs and our own assumptions regarding credit losses, prepayment speed, and discount rate. Our assumptions regarding prepayment speed and credit losses are based on historical performance.

Finance Receivables. We estimate the fair value of finance receivables based on an income approach using internal valuation models. These models project future cash flows of financing contracts incorporating appropriate funding pricing and enhancement requirements. The projected cash flows are discounted to a present value based on market inputs and our own assumptions regarding credit losses, pre-payment speed, and discount rate. Our assumptions regarding prepayment speed and credit losses are based on historical performance.

Debt. We estimate the fair value of debt based on a market approach using quoted market prices or current market rates for similar debt with approximately the same remaining maturities, where possible. Where market prices are not available, we estimate fair value based on an income approach using discounted cash flow models. These models project future cash flows and discount the future amounts to a present value using market-based expectations for interest rates, our own credit risk and the contractual terms of the debt instruments. For asset-backed debt issued in securitization transactions, the principal payments are based on the actual payments on the securitized contracts. We use management judgment to estimate the timing and amount of the debt cash flows based on historical pre-payment speeds.

Input Hierarchy of Items Measured at Fair Value on a Recurring Basis

The following tables summarize the fair values by input hierarchy of items measured at fair value on a recurring basis on our balance sheet (in millions):

	March 31, 2010					
	Level 1	Level 2	Level 3	Total		
Automotive Sector						
Assets						
Cash equivalents – financial instruments (a)						
U.S. government	\$ 1,085		\$ •	\$ 1,085		
Government-sponsored enterprises	•	2,399	•	2,399		
Government – non-U.S.	•	99	•	99		
Corporate debt		2,353	<u> </u>	2,353		
Total cash equivalents – financial instruments	1,085	4,851	•	5,936		
Marketable securities (b)						
U.S. government	5,706	•	•	5,706		
Government-sponsored enterprises	•	3,782	•	3,782		
Corporate debt	•	99	•	99		
Mortgage-backed and other asset-backed	•	114	2	116		
Equity	583		•	583		
Government – non-U.S.	•	750	•	750		
Other liquid investments (c)		1,023		1,023		
Total marketable securities	6,289	5,768	2	12,059		
Derivative financial instruments		= 0		= 0		
Foreign exchange contracts	•	59	•	59		
Commodity contracts	•	4	8	12		
Other – warrants	<u> </u>	<u> </u>	3	3		
Total derivative financial instruments (d)		63	11	74		
Total assets at fair value	\$ 7,374	\$10,682	<u>\$ 13</u>	\$18,069		
Liabilities						
Derivative financial instruments						
Foreign exchange contracts	\$ •	\$ 128	\$ •	\$ 128		
Commodity contracts		29		29		
Total derivative financial instruments (d)		157		157		
Total liabilities at fair value	\$ •	\$ 157	\$ •	\$ 157		

⁽a) "Cash equivalents - financial instruments" in this table excludes time deposits, certificates of deposit, money market accounts, and other cash equivalents reported at par value on our balance sheet totaling \$4.6 billion as of March 31, 2010 for the Automotive sector, which approximates fair value. In addition to these cash equivalents, we also had cash on hand totaling \$2.3 billion as of March 31, 2010. Excludes an investment in Ford Credit debt securities held by the Automotive sector with a carrying value of \$432 million and an estimated fair value of \$439 million as of March 31, 2010; see Note 1 for additional detail.

Includes certificates of deposit and time deposits with a maturity of more than 90 days at date of purchase.

⁽d) See Note 14 for additional information regarding derivative financial instruments.

		March 3	31, 2010	
	Level 1	Level 2	Level 3	Total
Financial Services Sector				
Assets				
Cash equivalents – financial instruments (a)				
U.S. government	\$ 1,179		\$ •	\$ 1,179
Government-sponsored enterprises	•	850	•	850
Government – non-U.S.	•	427	•	427
Corporate debt		300		300
Total cash equivalents – financial instruments	1,179	1,577	•	2,756
Marketable securities				
U.S. government	6,860	•	•	6,860
Government-sponsored enterprises	•	1,996	•	1,996
Corporate debt	•	165	•	165
Mortgage-backed	•	244	•	244
Government – non-U.S.	•	323	•	323
Other liquid investments (b)		236		236
Total marketable securities	6,860	2,964	•	9,824
Derivative financial instruments (c)				
Interest rate contracts	•	1,124	350	1,474
Foreign exchange contracts	•	6	•	6
Cross currency interest rate swap contracts		72	•	72
Total derivative financial instruments		1,202	350	1,552
Retained interest in securitized assets (d)	•	•	13	13
Total assets at fair value	\$ 8,039	\$ 5,743	\$ 363	\$14,145
Liabilities				
Derivative financial instruments (c)				
Interest rate contracts	\$ •	\$ 312	\$ 365	\$ 677
Foreign exchange contracts	•	85	•	85
Cross-currency interest rate swap contracts	•	134	168	302
Total derivative financial instruments		531	533	1,064
Total liabilities at fair value	\$ •	\$ 531	\$ 533	\$ 1,064

⁽a) "Cash equivalents – financial instruments" in this table excludes time deposits, certificates of deposit, money market accounts, and other cash equivalents reported at par value on our balance sheet totaling \$6.7 billion as of March 31, 2010 for the Financial Services sector, which approximates fair value. In addition to these cash equivalents, we also had cash on hand totaling \$2.1 billion as of March 31, 2010 for the Financial Services sector.
(b) Includes certificates of deposit and time deposits with a maturity of more than 90 days at date of purchase.
(c) See Note 14 for additional information regarding derivative financial instruments.
(d) Retained interest in securitized assets is reported in *Other assets* on our consolidated balance sheet.

	December 31, 2009						
	Level 1	Level 2	Level 3	Total			
Automotive Sector							
Assets							
Cash equivalents – financial instruments (a)							
U.S. government	\$ 30		\$ •	\$ 30			
Government-sponsored enterprises	•	949	•	949			
Government – non-U.S.	•	238	•	238			
Corporate debt		2,557	<u> </u>	2,557			
Total cash equivalents – financial instruments	30	3,744	•	3,774			
Marketable securities(b)							
U.S. government	9,130	•	•	9,130			
Government-sponsored enterprises	•	2,408	•	2,408			
Corporate debt	•	108	8	116			
Mortgage-backed and other asset-backed	•	191	17	208			
Equity	477	•	•	477			
Government – non-U.S.	•	1,283	•	1,283			
Other liquid investments (c)		901	<u></u> :	901			
Total marketable securities	9,607	4,891	25	14,523			
Derivative financial instruments							
Foreign exchange contracts	•	59	•	59			
Commodity contracts	•	8	7	15			
Other - warrants			2	2			
Total derivative financial instruments	•	67	9	76			
Total assets at fair value	\$ 9,637	\$ 8,702	\$ 34	\$18,373			
Liabilities							
Derivative financial instruments							
Foreign exchange contracts	\$ •	\$ 85	\$ •	\$ 85			
Commodity contracts	•	54	•	54			
Total derivative financial instruments	-	139	•	139			
Total liabilities at fair value	\$ •	\$ 139	\$ •	\$ 139			

[&]quot;Cash equivalents – financial instruments" in this table excludes time deposits, certificates of deposit, money market accounts, and other cash equivalents reported at par value on our balance sheet totaling \$2.1 billion as of December 31, 2009 for the Automotive sector, which approximates fair value. In addition to these cash equivalents, we also had cash on hand totaling \$3.9 billion as of December 31, 2009.

Marketable securities excludes an investment in Ford Credit debt securities held by the Automotive sector with a carrying value of \$646 million and an estimated fair value of \$656 million as of December 31, 2009; see Note 1 for additional detail.

Other liquid investments include certificates of deposit and time deposits with a maturity of more than 90 days at date of purchase.

	December 31, 2009				
	Level 1	Level 2	Level 3	Total	
Financial Services Sector					
Assets					
Cash equivalents – financial instruments (a)					
U.S. government	\$ 75	\$ •	\$ •	\$ 75	
Government-sponsored enterprises	•	400	•	400	
Government-non-U.S.	•	29	•	29	
Corporate debt.		75	:	75	
Total cash equivalents – financial instruments	75	504	-	579	
Marketable securities					
U.S. government	5,256	•	•	5,256	
Government-sponsored enterprises	•	1,098	•	1,098	
Corporate debt	•	159	4	163	
Mortgage-backed	•	237	•	237	
Government – non-U.S.	•	65	•	65	
Other liquid investments (b)		45	<u> </u>	45	
Total marketable securities	5,256	1,604	4	6,864	
Derivative financial instruments					
Interest rate contracts	•	1,234	420	1,654	
Foreign exchange contracts	•	22	•	22	
Cross currency interest rate swap contracts		203		203	
Total derivative financial instruments	•	1,459	420	1,879	
Retained interest in securitized assets		•	26	26	
Total assets at fair value	\$ 5,331	\$ 3,567	\$ 450	\$9,348	
Liabilities					
Derivative financial instruments					
Interest rate contracts	\$ •	\$ 409	\$ 437	\$ 846	
Foreign exchange contracts	•	46	•	46	
Cross-currency interest rate swap contracts	•	144	138	282	
Total derivative financial instruments		599	575	1,174	
Total liabilities at fair value	\$ •	\$ 599	\$ 575	\$1,174	

 ⁽a) "Cash equivalents – financial instruments" in this table excludes time deposits, certificates of deposit, money market accounts, and other cash equivalents reported at par value on our balance sheet totaling \$7.7 billion as of December 31, 2009 for the Financial Services sector, which approximates fair value. In addition to these cash equivalents, we also had cash on hand totaling \$2.8 billion as of December 31, 2009.
 (b) Other liquid investments include certificates of deposit and time deposits with a maturity of more than 90 days at date of purchase.

Reconciliation of Changes in Level 3 Balances

The following tables summarize the changes in Level 3 items measured at fair value on a recurring basis on our balance sheet for the periods ending March 31 (in millions):

	2010										
	Total Realized/ Net Fair Value at Unrealized Purchases/ Net December 31, Gains/ (Settlements) Into		l Purchases/ Net Transfers Fair (Settlements) Into/(Out of) Ma		March 31,	131, Instru					
Automotive Sector Marketable securities											
Corporate debt	\$	8	\$	•	\$	(8)	\$	\$	•	\$	•
Mortgage-backed and other asset-backed		17		(1)		(14)	•		2		•
Total marketable securities		25		(1)		(22)	-	Т	2		•
Derivative financial instruments, net		9		3		(1)	•		11		2
Total Level 3 fair value	\$	34	\$	2	\$	(23)	\$ •	\$	13	\$	2
Financial Services Sector											
Marketable securities - corporate debt	\$	4	\$	(4)	\$	•	\$ •	\$	•	\$	•
Derivative financial instruments, net		(155))	(63)		35	•		(183)		(34)
Retained interest in securitized assets		26		•		(13)			13		•
Total Level 3 fair value	\$	(125)	\$	(67)	\$	22	\$ •	\$	(170)	\$	(34)

(a) Includes option premiums (paid)/received on options traded during the quarter.

(b) For those assets and liabilities still held at reporting date.

	2009								
	Fair Va Decemb	er 31,	Total Realized/ Unrealized Gains/ (Losses)	1 1	Net Purchases/ Settlements) (a)	Net Transfers Into/(Out of) Level 3 (b)	Fair Value at March 31, 2009 (c)	Change In Unrealized Gains/ (Losses) on Instruments Still Held (d)	
Automotive Sector									
Marketable securities	ф	22	Φ (3	(10)	ф	Φ	Φ (2)	
Corporate debt	\$	22		2) \$	(13)		The second secon	\$ (2)	
Mortgage-backed and other asset-backed		127	(4	1)	(62)	(9)	52	(2)	
Equity		1		<u> </u>	<u> </u>	(1)	<u> </u>		
Total marketable securities		150	(5)	(75)	(10)	59	(4)	
Derivative financial instruments, net		(32)	(.	<u>s</u>) _	21		(14)		
Total Level 3 fair value	\$	118	\$ (9	9) \$	(54)	\$ (10)	\$ 45	\$ (6)	
Financial Services Sector									
Marketable securities - corporate debt	\$	5	\$ (4	1) \$		\$	\$ 1	\$ (4)	
Derivative financial instruments, net		(74)	(1:	5)	18	•	(71)	(1)	
Retained interest in securitized assets		92		2_	(7)	•	87	(2)	
Total Level 3 fair value	\$	23	\$ (1'	7) \$	11	\$ •	<u>\$ 17</u>	\$ (7)	

⁽a) Includes option premiums (paid)/received on options traded during the quarter.(b) "Transfers Into" represent the value at the end of the reporting period and "Transfers Out of" represent the value at the beginning of the reporting period.

(c) Refer to our 2009 Form 10-K Report for reconciliation of full year 2009.

(d) For those assets and liabilities still held at reporting date.

The following tables summarize the realized/unrealized gains/(losses) on Level 3 items by financial statement position for the period ending March 31 (in millions):

		2010							
	Autom Cost of	Inte Incon Other Oper	ating Se ome/ Other	rvices Compr Income/ Inc	Tota cher Realiz chensive Unreal come/ Gain (s) (a) (Losse	ed/ ized s/			
Automotive Sector									
Marketable securities	\$	• \$	(1) \$	• \$	• \$	(1)			
Derivative financial instrum	ents, net (b)	2	1	<u> </u>	<u> </u>	3			
Total Automotive sector		2	•	•	•	2			
Financial Services Sector									
Marketable securities		•	•	(4)	•	(4)			
Derivative financial instrum	ents, net (b)	•	•	(56)	(7)	(63)			
Retained interest in securitize	zed assets	•	•	(2)	2	<u> </u>			
Total Financial Services s	sector \$	• \$	• \$	(62) \$	(5) \$	(67)			

[&]quot;Other Comprehensive Income/(Loss)" on derivative financial instruments reflects foreign currency translation on non-U.S. dollar foreign affiliates.

See Note 14 for detail on financial statement presentation by hedge designation.

			2009		
	Automotive Cost of Sales	Automotive Interest Income and Other Non- Operating Income/ (Loss), Net	Financial Services Other Income/ (Loss), Net	Other Comprehensive Income/ (Loss) (a)	Total Realized/ Unrealized Gains/ (Losses)
Automotive Sector					
Marketable securities	\$ •	\$ (6)	•	\$	\$ (6)
Derivative financial instruments, net (b)	(4)	1			(3)
Total Automotive sector	(4)	(5)	•	•	(9)
Financial Services Sector					
Marketable securities	•	•	(4)	•	(4)
Derivative financial instruments, net (b)	•	•	(20)	5	(15)
Retained interest in securitized assets		•	4	(2)	2
Total Financial Services sector	\$ •	\$ •	\$ (20)	\$ 3	\$ (17)

[&]quot;Other Comprehensive Income/(Loss)" on derivative financial instruments reflects foreign currency translation on non-U.S. dollar foreign affiliates.
See Note 14 for detail on financial statement presentation by hedge designation.

Non-Recurring Fair Value Measurements

There were no non-recurring fair value measurements subsequent to initial recognition recorded during the first quarter of 2010.

Automotive Sector

During the first quarter of 2009, we recorded an other-than-temporary impairment of our investment in our U.S. consolidated dealerships of \$78 million in *Automotive cost of sales*. The fair value measurement used to determine the impairment was based on the market approach and reflected anticipated proceeds, expected to be *de minimis*. The fair value of our investment was classified in Level 2 of our fair value hierarchy. During the second quarter of 2009, we recorded an other-than-temporary impairment of our investment in the Bordeaux automatic transmission plant. The fair value measurement of \$241 million used to determine the impairment was based on the cost approach and considered the condition of the plant's fixed assets. The fair value of our investment was classified in Level 3 of our fair value hierarchy. See Note 12 for discussion of our held-for-sale impairment of Volvo.

Financial Services Sector

In March 2009, our Board of Directors approved potential sale of the Financial Services sector's investment in DFO Partnership. As a result, during the first quarter of 2009, we recorded an other-than-temporary impairment of the investment of \$141 million in *Financial Services equity in net income/(loss) of affiliated companies*. The investment in DFO Partnership was subsequently sold during the third quarter of 2009 and was no longer held on our balance sheet at December 31, 2009.

NOTE 3. CASH AND RESTRICTED CASH

Cash and cash equivalents that are restricted as to withdrawal or usage under the terms of certain contractual agreements are recorded as restricted in *Other assets* on our consolidated balance sheet.

Our Automotive sector restricted cash balances primarily include cash collateral required for letters of credit related to the sale of Hertz (see Note 16), cash collateral posted with counterparties related to derivatives (see Note 14) and cash collateral for bank guarantees. Additionally, restricted cash includes various escrow agreements related to insurance, customs, environmental matters and contractual obligations related to the sale or disposition of a business. Our Financial Services sector restricted cash balances primarily include cash collateral required to be held against loans with the European Investment Bank and cash held to meet certain local governmental and regulatory reserve requirements.

Restricted cash does not include required minimum balances, or cash securing debt issued through securitization transactions ("securitization cash"). See Note 9 for discussion of the minimum balance requirement related to the secured credit agreement that we initially entered into in December 2006, and securitization cash.

Restricted Cash

Restricted cash reflected on our balance sheet is as follows (in millions):

	M	arch 31, 2010	 December 31, 2009
Automotive sector	\$	736	\$ 777
Financial Services sector		408	 335
Total Company	\$	1,144	\$ 1,112
			_

NOTE 4. FINANCE RECEIVABLES - FINANCIAL SERVICES SECTOR

Retail finance receivables consist primarily of installment loans and direct financing lease contracts for new and used vehicles with retail customers, daily rental companies, government entities, and fleet customers. Wholesale finance receivables include dealer financing of new and used vehicles in inventory. Other finance receivables consist primarily of real estate, commercial and other collateralized loans, and accrued interest.

Net finance receivables were as follows (in millions):

		arch 31, 2010		ember 31, 2009
Retail (including direct financing leases)	\$	55,574	\$	58,229
Wholesale		21,456		22,370
Other finance receivables		3,638		3,611
Total finance receivables		80,668		84,210
Unearned interest supplements		(2,048)		(1,994)
Allowance for credit losses		(1,196)		(1,351)
Other		15		20
Finance receivables, net – sector balance sheet	\$	77,439	\$	80,885
Figure receivables not subject to friquely	¢	72,879	¢	76.001
Finance receivables, net, subject to fair value	D D	. ,		76,991
Fair value	3	74,251	3	76,066
Finance receivables, net – sector balance sheet	\$	77,439	\$	80,885
Reclassification of receivables purchased from Automotive sector to Other receivables, net		(3,602)		(3,889)
Finance receivables, net – consolidated balance sheet	\$	73,837	\$	76,996

NOTE 5. INVENTORIES

All inventories are stated at the lower of cost or market. Cost for a substantial portion of U.S. inventories is determined on a last-in, first-out ("LIFO") basis. LIFO was used for approximately 35% and 31% of inventories at March 31, 2010 and December 31, 2009, respectively. Cost of other inventories is determined on a first-in, first-out ("FIFO") basis.

Inventories are summarized as follows (in millions):

	rch 31, 2010	December 31, 2009
Raw materials, work-in-process and supplies	\$ 2,816	\$ 2,456
Finished products	 4,276	3,383
Total inventories under first-in, first-out method ("FIFO")	7,092	5,839
Less: Last-in, first-out method ("LIFO") adjustment	(800)	(798)
Total inventories	\$ 6,292	\$ 5,041

NOTE 6. VARIABLE INTEREST ENTITIES

A VIE is an entity that either (i) has insufficient equity to permit the entity to finance its activities without additional subordinated financial support or (ii) has equity investors who lack the characteristics of a controlling financial interest. A VIE is consolidated by its primary beneficiary. The primary beneficiary has both the power to direct the activities that most significantly impact the entity's economic performance and the obligation to absorb losses or the right to receive benefits from the entity that could potentially be significant to the VIE.

If we determine that we have operating power and the obligation to absorb losses or receive benefits, we consolidate the VIE as the primary beneficiary, and if not, we do not consolidate. Our involvement constitutes power that is most significant to the entity when we have unconstrained decision-making ability over key operational functions within the entity or we have the ability to exercise discretion in the servicing of financial assets related to securitization entities.

Assets recognized as a result of consolidating these VIEs do not represent additional assets that could be used to satisfy claims against our general assets. Conversely, liabilities recognized as a result of consolidating these VIEs do not represent additional claims on our general assets; rather, they represent claims against the specific assets of the consolidated VIEs.

Automotive Sector

VIEs of which we are the primary beneficiary:

At March 31, 2010, we have one VIE of which we are the primary beneficiary:

Tekfor Cologne GmbH ("Tekfor") is a 50/50 joint venture with Neumayer Tekfor GmbH. Tekfor produces transmission and chassis components for use in our vehicles. We provide financial support to Tekfor in the form of a revolving loan agreement. This loan is being used by Tekfor to refinance external debt

At December 31, 2009, in addition to Tekfor, we held interests in certain dealerships in North America. During the first quarter of 2009 we consolidated 64 dealerships as part of our Dealer Development Program. As we sold our ownership interest and liquidated certain other dealerships, that number was reduced during the year to 13 at December 31, 2009. In the first quarter of 2010, we acquired the remaining outside ownership interest and now consolidate the dealerships under the voting interest model.

The total consolidated VIE assets and liabilities reflected on our balance sheet are as follows (in millions):

Assets	arch 31, 2010	mber 31, 2009
Cash and cash equivalents	\$ 4	\$ 27
Other receivables, net	25	34
Inventories	22	106
Net property	31	154
Other assets	 •	 <u>1</u>
Total assets	\$ 82	\$ 322
Liabilities		
Payables	\$ 22	\$ 23
Accrued liabilities and deferred revenue	•	32
Debt	 •	 14
Total liabilities	\$ 22	\$ 69

The financial performance of Tekfor reflected on our statement of operations at March 31, 2010 includes sales of \$15 million and cost of sales, selling, administrative, and interest expense of \$18 million. The financial performance of Tekfor and the dealerships reflected on our statement of operations at March 31, 2009 include sales of \$497 million and cost of sales, selling, administrative, and interest expense of \$604 million.

VIEs of which we are not the primary beneficiary:

Joint Ventures that constitute significant VIEs of which we are not the primary beneficiary, and which are not consolidated as of March 31, 2010 and December 31, 2009, include:

- Getrag Ford Transmissions GmbH ("GFT") is a 50/50 joint venture with Getrag Deutsche Venture GmbH and Co. KG. Ford and its related parties purchase substantially all of the joint venture's output.
- Pininfarina Sverige, AB is a 40/60 joint venture between Volvo Cars and Pininfarina, S.p.A. Volvo purchases substantially all of the joint venture's output.
- Getrag All Wheel Drive AB is a 40/60 joint venture between Volvo Cars and Getrag Dana Holding GmbH. Volvo and its related parties purchase substantially all of the joint venture's output.

We purchase substantially all of the output of the joint ventures listed above. However, the associated operating agreements indicate that the power to direct economically significant activities of these joint ventures is either shared or rests with an unrelated party.

Additionally, the following entities (that are not joint ventures) are significant VIEs of which we are not the primary beneficiary, which are not consolidated as of March 31, 2010 and December 31, 2009:

- Hertz Vehicle Financing LLC was established in 2005, as part of the transaction to sell our interest in The Hertz Corporation ("Hertz"). We provided cash-collateralized letters of credit to support the payment obligations of Hertz Vehicle Financing LLC, a bankruptcy-remote special purpose entity which is thinly capitalized and wholly owned by Hertz. The arrangements outlined in the letters of credit do not grant Ford the power to direct activities that most significantly impact the entity's economic performance. The carrying value of our obligation related to these letters of credit, which will expire no later than December 21, 2011, was approximately \$8 million at March 31, 2010. For additional discussion of these letters of credit, see Note 16.
- Ford Motor Company Capital Trust II ("Trust II") was formed in 2002. We own 100% of Trust II's common stock, which is equal to 5% of Trust II's total equity. Operation of the trust is predetermined by Trust documents which cannot be modified.
- First Aquitaine Industries SAS ("First Aquitaine") operates a transmission plant in Bordeaux, France which manufactures automatic transmissions
 for Ford Explorer, Ranger and Mustang vehicles. During 2009 we transferred legal ownership of First Aquitaine to HZ Holding France and entered
 into a volume-dependent pricing agreement with the new owner to purchase transmissions through the end of the product cycle. After the transfer,
 Ford relinquished all decision-making rights within the entity.

Our maximum exposure to VIEs of which we are not the primary beneficiary at March 31, 2010 and December 31, 2009 are detailed as follows (in millions):

	March 31 2010	,	De	cember 31, 2009	Change in Maximum Exposure	
Investments	\$	440	\$	421	\$	19
Liabilities		(8)		(9)		1
Guarantees (off-balance sheet)		192		191		1
Total maximum exposure	\$	624	\$	603	\$	21

Financial Services Sector

VIEs of which the Financial Services sector is the primary beneficiary:

Our Financial Services sector (for these purposes, Ford Credit and Volvo Auto Bank) uses VIEs to issue asset-backed securities in transactions to public and private investors, bank conduits, and government-sponsored entities or others who obtain funding from government programs. The asset-backed securities are secured by finance receivables and interests in net investments in operating leases. Our Financial Services sector retains interests in its securitization transactions, including senior and subordinated securities issued by VIEs, rights to cash held for the benefit of the securitization investors such as cash reserves, and residual interests. Our Financial Services sector's involvement constitutes power that is most significant to the entity when we have unconstrained decision-making ability within the entity over key operational functions such as our ability to exercise discretion in the servicing of financial assets, to issue additional debt, to exercise a unilateral call option, to add assets to revolving structures, and to control the investment decisions. Therefore, the assets continue to be consolidated by our Financial Services sector.

The VIE transactions create and pass along risks to the variable interest holders, depending on the assets securing the debt and the specific terms of the transactions.

Our Financial Services sector aggregates and analyzes its transactions based on the risk profile of the product and the type of funding structure, including:

- Retail transactions consumer credit risk and prepayment risk.
- Wholesale transactions dealer credit risk.
- Net investments in operating lease transactions vehicle residual value risk, consumer credit risk, and prepayment risk.

As residual interest holder, our Financial Services sector is exposed to underlying residual and credit risk of the collateral, and may be exposed to interest rate risk. However, this risk is not incremental to the exposure our Financial Services sector has on the underlying assets. The amount of risk absorbed by our Financial Services sector's residual interests is generally represented by and limited to the amount of overcollaterization of its assets securing the debt and any cash reserves.

Our Financial Services sector has no obligation to repurchase or replace any securitized asset that subsequently becomes delinquent in payment or otherwise is in default. Securitization investors have no recourse to our Financial Services sector or its other assets for credit losses on the securitized assets, and have no right to require the Financial Services sector to repurchase the investments. Although not contractually required, Ford Credit regularly supports their wholesale securitization programs by repurchasing receivables of a dealer from the VIEs when the dealer's performance is at risk, which transfers the corresponding risk of loss from the VIE to Ford Credit. In order to continue to fund the wholesale receivables, Ford Credit also may contribute additional cash or wholesale receivables if the collateral falls below the required level. The cash contributions were \$20 million and \$0 at March 31, 2010 and December 31, 2009, respectively, and ranged from \$0 to \$375 million during the first quarter of 2010. In addition, while not contractually required, Ford Credit may purchase the commercial paper issued by Ford Credit's FCAR asset-backed commercial paper program.

From time to time, Ford Credit renegotiates the terms of its funding commitments and may reallocate the commitments globally. The Financial Services sector does not guarantee any asset-backed securities and generally has no obligation to provide liquidity or contribute cash or additional assets to the VIEs. However, in certain securitization transactions, Ford Credit has dynamic enhancements where it is required to support the performance of the securitization transactions by purchasing additional subordinated notes or increasing cash reserves.

VIEs that are exposed to interest rate or currency risk have reduced their exposure by entering into derivatives. In certain instances, Ford Credit has entered into offsetting derivative transactions with the VIE or the counterparty to protect the VIE from these risks that are not mitigated through derivative transactions between the VIE and its counterparty. See Note 14 for additional information regarding Ford Credit's derivatives.

The following table includes assets to be used to settle the debt and obligations of the Financial Services sector consolidated VIEs. The Financial Services sector may retain debt issued by the consolidated VIEs and this debt is excluded from the table below. The Financial Services sector holds the right to the excess cash flows from the assets that are not needed to pay the debt and obligations of the consolidated VIEs. The assets and debt reflected on our consolidated balance sheet are as follows (in billions):

Finance receivables	C	h and ash valents	Fin Recei I a Net In Ope	March 31, 2010 Finance Receivables, Net and Net Investment in Operating Leases			
Retail	\$	3.4	\$	38.6	\$	30.6	
Wholesale	<u> </u>	0.5	<u> </u>	15.8	<u> </u>	10.2	
Total finance receivables		3.9		54.4		40.8	
Net investment in operating leases		1.2		10.8		7.1	
Total*	\$	5.1	\$	65.2	\$	47.9	

^{*} Certain notes issued by the VIEs to affiliated companies served as collateral for accessing the European Central Bank ("ECB") open market operations program. This external funding of \$736 million at March 31, 2010 was not reflected as debt of the VIEs and is excluded from the table above, but was included in our consolidated debt. The finance receivables backing this external funding are included in the table above.

	Casi Casi Equiv	per 31, 2009 nance ivables, Net und vestment in rrating eases	Debt			
Finance receivables						
Retail	\$	3.1	\$	40.9	\$	31.2
Wholesale		0.5		16.5		8.4
Total finance receivables		3.6		57.4		39.6
Net investment in operating leases		1.3		10.2		6.6
Total*	\$	4.9	\$	67.6	\$	46.2

Certain notes issued by the VIEs to affiliated companies served as collateral for accessing the ECB open market operations program. This external funding of \$1.8 billion at December 31, 2009, was not reflected as debt of the VIEs and is excluded from the table above, but was included in our consolidated debt. The finance receivables backing this external funding are included in the table

Ford Credit's exposure based on the fair value of derivative instruments related to consolidated VIEs that support its securitization transactions is as follows (in millions):

		March 31, 2010				December 31, 2009			
		vative sset		rivative ability	Deriva Ass			rivative ability	
	A	SSCI		ability	7133	<u> </u>		ability	
Total derivative financial instruments*	\$	39	\$	504	\$	55	\$	528	

Ford Credit derivative assets and liabilities are included in *Other assets* and *Accrued liabilities and deferred revenue*, respectively, on our consolidated balance sheet.

The financial performance of the consolidated VIEs that support Ford Credit's securitization transactions reflected in our statement of operations is as follows (in millions):

		First Quart	er 2010			First Quarter 2009				
	Deri	vative	I	nterest		Derivative		Interest		
	Ex	pense	<u>F</u>	Expense	Expense		Expense			
VIE financial performance	\$	145	\$	332	\$	64	\$	478		

VIEs of which the Financial Services sector is not the primary beneficiary:

Ford Credit has an investment in a certain joint venture determined to be a VIE of which it is not the primary beneficiary. The joint venture provides consumer and dealer financing in its local markets and is financed by external debt and additional subordinated interest provided by the joint venture partner. The operating agreement indicates that the power to direct economically significant activities is shared with its joint venture partner, and the obligation to absorb losses or rights to receive benefits resides primarily with its joint venture partner. Ford Credit's investment in the joint venture is accounted for as an equity method investment and is included in *Equity in net assets of affiliated companies*. Ford Credit's maximum exposure to any potential losses associated with this VIE is limited to its equity investment, and amounted to \$64 million and \$67 million at March 31, 2010 and December 31, 2009, respectively.

NOTE 7. GOODWILL AND OTHER NET INTANGIBLE ASSETS

The components of goodwill and other net intangible assets are as follows (in millions):

			March 31, 201	10		December 31, 2009					
	Gross Carrying Amount		Carrying		Less: Accumulated Amortization		Net Carrying Amount	Gross Carrying Amount	Less: Accumulated Amortization	Net Carrying Amount	
Automotive Sector											
Ford Europe goodwill	\$	32	\$	\$	32	\$ 34	\$	\$ 34			
Manufacturing and production incentive rights	2	97	(241))	56	305	(228)	77			
License and advertising agreements		96	(34))	62	96	(32)	64			
Other intangible assets		74	(41)	_	33	74	(50)	24			
Total Automotive sector	4	99	(316))	183	509	(310)	199			
Financial Services Sector											
Ford Credit goodwill		9	•		9	9	•	9			
Other intangible assets		1	(1)	_	<u> </u>	1	(1)	•			
Total Financial Services sector		10	(1)		9	10	(1)	9			
Total Company	\$ 5	09	\$ (317)	\$	192	\$ 519	\$ (311)	\$ 208			

Changes in the goodwill balances are primarily attributable to the impact of foreign currency translation. We also have goodwill recorded within *Equity* in net assets of affiliated companies of \$34 million at March 31, 2010 and December 31, 2009.

NOTE 7. GOODWILL AND OTHER NET INTANGIBLE ASSETS (Continued)

Our intangible assets are comprised primarily of manufacturing and production incentive rights acquired in 2006 with a useful life of 4 years, license and advertising agreements with amortization periods of 5 years to 25 years, and other intangibles with various amortization periods (primarily patents, customer contracts, technology, and land rights).

Pre-tax amortization expense was as follows for the period ending March 31 (in millions):

	20	<u>10 </u>	2009		
Pre-tax amortization expense	\$	23	\$	18	

Amortization for current intangible assets is forecasted to be approximately \$80 million to \$90 million in 2010, \$10 million in 2011 through 2014, and \$7 million thereafter.

NOTE 8. RETIREMENT BENEFITS

We provide pension benefits and OPEB, such as health care and life insurance, to employees in many of our operations around the world.

Pension and OPEB expense is summarized as follows (in millions):

	First Quarter										
		Pension	ts								
	U.S. I	Plans	Non-U.S	S. Plans	OPEB						
	2010	2009	2010	2009	2010	2009					
Service cost	\$ 94	\$ 86	\$ 85	\$ 68	\$ 14	\$ 102					
Interest cost	631	674	325	292	84	223					
Expected return on assets	(791)	(822)	(347)	(303)	•	(33)					
Amortization of:											
Prior service costs/(credits)	93	94	19	19	(154)	(227)					
(Gains)/Losses and Other	2	2	65	37	24	21					
Separation programs	3	7	4	30	•	2					
(Gain)/Loss from curtailment						(2)					
Net expense/(income)	\$ 32	\$ 41	\$ 151	\$ 143	\$ (32)	\$ 86					

Plan Contributions

Our policy for funded plans is to contribute annually, at a minimum, amounts required by applicable laws and regulations. From time to time, we make contributions beyond those legally required.

Pension. In the first quarter of 2010, we contributed \$500 million to our worldwide pension plans (including Volvo), including \$100 million of benefit payments paid directly by the Company for unfunded plans. We expect to contribute from Automotive cash and cash equivalents an additional \$1 billion in 2010, including \$300 million of benefit payments paid directly by the Company for unfunded plans, for a total of \$1.5 billion this year (excluding any purchase price adjustment for pension plans at closing in connection with the sale of Volvo). Based on current assumptions and regulations, we do not expect to have a legal requirement to fund our major U.S. pension plans in 2010.

NOTE 9. DEBT AND COMMITMENTS

Our debt consists of short-term and long-term unsecured debt securities, convertible debt securities, and unsecured and secured borrowings from banks and other lenders. Debt issuances are placed directly by us or through securities dealers or underwriters and are held by institutional and retail investors. In addition, our Financial Services sector sponsors securitization programs that provide short-term and long-term asset-backed financing through institutional investors in the U.S. and international capital markets.

Debt outstanding is shown below (in millions):

	March 31, 2010		De	December 31, 2009	
Automotive Sector					
Debt payable within one year	ф	720	Ф	500	
Short-term (a) Long-term payable within one year	\$	739)	502	
Public unsecured debt securities		334		334	
Notes due to UAW Retiree Medical Benefits Trust (the "UAW VEBA Trust") unsecured portion (b)		859		859	
Secured term loan		77		77	
Secured revolving loan		3.000		•	
Other debt (a)		166		199	
Unamortized discount		(166)		(333)	
Total debt payable within one year		5.009		1.638	
Long-term debt payable after one year		3,007		1,030	
Public unsecured debt securities		5.260		5.260	
Convertible notes		3,454		3,454	
Subordinated convertible debentures		3,173		3,124	
Secured term loan		5,165		5,184	
Secured revolving loan		4,527		7,527	
Notes due to UAW VEBA Trust (b)					
Unsecured portion		6,720		6,720	
Secured portion		3,000		3,000	
U.S. Department of Energy loans		1,475		1,221	
Other debt		688		727	
Unamortized discount		(4,220)		(4,245)	
Total long-term debt payable after one year		29,242		31,972	
Total Automotive sector	<u>\$</u>	34,251	\$	33,610	
Fair value of debt	\$	33,305	\$	30,492	
Financial Services Sector					
Short-term debt					
Asset-backed commercial paper	\$	6,468	\$	6,369	
Other asset-backed short-term debt		2,603		4,482	
Ford Interest Advantage (c)		4,178		3,680	
Other short-term debt		879	_	1,088	
Total short-term debt		14,128		15,619	
Long-term debt					
Unsecured debt		0.005		7.220	
Notes payable within one year		8,825		7,338	
Notes payable after one year Asset-backed debt		29,799		33,888	
Notes payable within one year		19.652		18.962	
Notes payable after one year		24.102		23,163	
Unamortized discount		(478)		(530)	
Fair value adjustments (d)		258		231	
Total long-term debt	_	82,158		83,052	
	Φ.		Φ.		
Total Financial Services sector	\$	96,286	\$	98,671	
Fair value of debt	\$	98,580		100,231	
Total Automotive and Financial Services sectors	\$	130,537	\$	132,281	
Intersector elimination (e)		(432)		(646)	
Total Company	\$	130,105	\$	131,635	

⁽a) Includes \$519 million (short-term of \$514 million and long-term payable within one year of \$5 million) and \$177 million (short-term of \$174 million and long-term payable within one year of \$3 million) of debt with affiliated companies at March 31, 2010 and December 31, 2009, respectively.

Amortizing Guaranteed Secured Notes maturing June 30, 2022 owed to UAW VEBA Trust.

The Ford Interest Advantage program consists of Ford Credit's floating rate demand notes.

Adjustments related to designated fair value hedges of unsecured debt.

Debt related to Ford's acquisition of Ford Credit debt securities; see Note 1 for additional detail.

The fair value of debt presented above reflects interest accrued but not yet paid. Interest accrued on Automotive accrued liabilities and deferred revenue and was \$353 million and \$351 million at March 31, 2010 and December 31, 2009, respectively. Interest accrued on Financial Services sector debt is reported in *Financial Services other liabilities and deferred income* and was \$839 million and \$1.1 billion at March 31, 2010 and December 31, 2009, respectively. The change in the fair value of our debt in the first quarter of 2010 was primarily driven by improvements in the credit markets generally, and an improved market view of Ford specifically.

Maturities

Debt maturities at March 31, 2010 were as follows (in millions):

	20	10	2011	2012	2013	2014		Thereafter	Total Debt Maturities
Automotive Sector									
Public unsecured debt securities	\$	334	\$ •	\$ •	\$ •	\$	•	\$ 5,260	\$ 5,594
Unamortized discount (a)		•	•	•	•		•	(84)	(84)
Convertible notes		•	•	•	•		•	3,454	3,454
Unamortized discount (a)		•	•	•	•		•	(855)	(855)
Subordinated convertible debentures		•	•	•	•	13	38	2,985	3,173
Secured term loan		58	77	77	5,030		•	•	5,242
Secured revolving loan (b)		3,000	838	•	3,689		•	•	7,527
Notes due to UAW VEBA Trust (c)		859	859	1,238	1,238	1,2		5,147	10,579
Unamortized discount (a)		(166)	(617)	(593)	(531)	(4)	53)	(1,077)	(3,447)
U.S. Department of Energy loans		•	•	74	148		48	1,105	1,475
Short-term and other debt (d)		550	520	177	114		<u>23</u>	209	1,593
Total Automotive debt		4,635	1,677	973	9,688	1,13	34	16,144	34,251
Financial Services Sector									
Unsecured debt		9,495	11,969	7,283	4,775	3,69	99	6,460	43,681
Asset-backed debt	2	4,485	15,812	8,540	2,041	2	40	1,707	52,825
Unamortized discount (a)		(1)	(2)	(180)	(67)	(13	36)	(42)	(478)
Fair value adjustments (a) (e)		3	89	107	46		13		258
Total Financial Services debt	3	3,982	27,868	15,750	6,795	3,70	56	8,125	96,286
Intersector elimination (f)		(432)	•	•	•		•	·	(432)
Total Company	\$ 3	8,185	\$ 29,545	\$ 16,723	\$ 16,483	\$ 4,90	00	\$ 24,269	\$ 130,105

- Unamortized discount and fair value adjustments are presented based on maturity date of related debt. \$3 billion debt maturity in 2010 represents our election to make a prepayment in the second quarter of 2010 as committed on March 29,
- Amortizing Guaranteed Secured Notes maturing June 30, 2022 owed to UAW VEBA Trust. Primarily non-U.S. affiliate debt.

 Adjustments related to designated fair value hedges of unsecured debt.

- Debt related to Ford's acquisition of Ford Credit debt securities; see Note 1 for additional detail.

Automotive Sector

Public Unsecured Debt Securities

Our public unsecured debt securities outstanding were as follows (in millions):

	Aggregate Principal Amo Outstanding					
Title of Security	March 31, 2010	December 31, 2009				
9.50% Guaranteed Debentures due June 1, 2010	\$ 334	\$ 334				
6 1/2% Debentures due August 1, 2018	361	361				
8 7/8% Debentures due January 15, 2022	86	86				
6.55% Debentures due October 3, 2022 (a)	15	15				
7 1/8% Debentures due November 15, 2025	209	209				
7 1/2% Debentures due August 1, 2026	193	193				
6 5/8% Debentures due February 15, 2028	104	104				
6 5/8% Debentures due October 1, 2028 (b)	638	638				
6 3/8% Debentures due February 1, 2029 (b)	260	260				
5.95% Debentures due September 3, 2029 (a)	8	8				
6.15% Debentures due June 3, 2030 (a)	10	10				
7.45% GLOBLS due July 16, 2031 (b)	1,794	1,794				
8.900% Debentures due January 15, 2032	151	151				
9.95% Debentures due February 15, 2032	4	4				
5.75% Debentures due April 2, 2035 (a)	40	40				
7.50% Debentures due June 10, 2043 (c)	593	593				
7.75% Debentures due June 15, 2043	73	73				
7.40% Debentures due November 1, 2046	398	398				
9.980% Debentures due February 15, 2047	181	181				
7.70% Debentures due May 15, 2097	142	142				
Total public unsecured debt securities (d)	\$ 5,594	\$ 5,594				

- (a) Unregistered industrial revenue bonds.
- (b) Listed on the Luxembourg Exchange and on the Singapore Exchange.
- (c) Listed on the New York Stock Exchange.
- (d) Excludes 9 1/2% Debentures due September 15, 2011 and 9.215% Debentures due September 15, 2021 with outstanding balances at March 31, 2010 of \$167 million and \$180 million, respectively. These securities are on-lent to Ford Holdings to fund Financial Services activity and are reported as Financial Services debt.

2009 Debt Repurchases. In the first quarter of 2009, we repurchased through a private market transaction \$165 million principal amount of our outstanding public unsecured debt securities. As a result, we recorded a pre-tax gain of \$127 million (net of unamortized discounts, premiums and fees) in Automotive interest income and other non-operating income/(expense), net.

Convertible Notes

Convertible Notes due December 15, 2016

At March 31, 2010, we had outstanding \$2.875 billion of 4.25% Senior Convertible Notes due December 15, 2016 ("2016 Convertible Notes"), which includes \$375 million from the underwriters option to purchase additional convertible notes ("Over-Allotment Option"). The 2016 Convertible Notes pay interest semiannually at a rate of 4.25% per annum. The 2016 Convertible Notes are convertible into shares of Ford Common Stock, based on a conversion rate (subject to adjustment) of 107.5269 shares per \$1,000 principal amount of 2016 Convertible Notes (which is equal to a conversion price of \$9.30 per share, representing a 25% conversion premium based on the closing price of \$7.44 per share on November 3, 2009).

Upon conversion, we have the right to deliver, in lieu of shares of Ford Common Stock, cash or a combination of cash and Ford Common Stock. Holders may require us to purchase all or a portion of the 2016 Convertible Notes upon a change in control of the Company, or for shares of Ford Common Stock upon a designated event that is not a change in control, in each case for a price equal to 100% of the principal amount of the 2016 Convertible Notes being repurchased plus any accrued and unpaid interest to, but not including, the date of repurchase. We also may terminate the conversion rights at

any time on or after November 20, 2014 if the closing price of Ford Common Stock exceeds 130% of the then-applicable conversion price for 20 trading days during any consecutive 30-trading day period.

Liability and equity components of our 2016 Convertible Notes are summarized as follows (in millions):

	rch 31, 2010	December 31, 2009
Liability component		
Principal	\$ 2,875 \$	2,875
Unamortized discount	 (684)	(702)
Net carrying amount	\$ 2,191 \$	2,173
Equity component (recorded in Capital in excess of par value of stock)	\$ (702) \$	(702)

We recognized interest cost on our 2016 Convertible Notes as follows (in millions):

		First Quarter		
	2010		2009	
Contractual interest coupon	\$	30	\$	•
Amortization of discount		18		<u>.</u>
Total interest cost on 2016 Convertible Notes	\$	48	\$	_

The discount on the liability components of the 2016 Convertible Notes will amortize through November 16, 2016. The total effective rate on the liability component was 9.2% on the original offering and 8.6% on the Over-Allotment Option. If all \$2.9 billion of 2016 Convertible Notes were converted into shares as of March 31, 2010 at a share price of \$12.57, the share value would exceed the principal value of debt by about \$1 billion.

Convertible Notes due December 15, 2036

At March 31, 2010, we had outstanding \$579 million of 4.25% Senior Convertible Notes due December 15, 2036 ("2036 Convertible Notes"). The 2036 Convertible Notes pay interest semiannually at a rate of 4.25% per annum. The 2036 Convertible Notes are convertible into shares of Ford Common Stock, based on a conversion rate (subject to adjustment) of 108.6957 shares per \$1,000 principal amount of 2036 Convertible Notes (which is equal to a conversion price of \$9.20 per share, representing a 25% conversion premium based on the closing price of \$7.36 per share on December 6, 2006).

Upon conversion we have the right to deliver, in lieu of shares of Ford Common Stock, cash or a combination of cash and Ford Common Stock. Holders may require us to purchase all or a portion of the 2036 Convertible Notes for cash on December 20, 2016 and December 15, 2026 or upon a change in control of the Company, or for shares of Ford Common Stock upon a designated event that is not a change in control, in each case for a price equal to 100% of the principal amount of the 2036 Convertible Notes being repurchased plus any accrued and unpaid interest to, but not including, the date of repurchase. We may redeem for cash all or a portion of the 2036 Convertible Notes at our option at any time or from time to time on or after December 20, 2016 at a price equal to 100% of the principal amount of the 2036 Convertible Notes to be redeemed, plus accrued and unpaid interest to, but not including, the redemption date. We also may terminate the conversion rights at any time on or after December 20, 2013 if the closing price of Ford Common Stock exceeds 140% of the then-applicable conversion price for 20 trading days during any consecutive 30-trading day period.

Liability and equity components of our 2036 Convertible Notes are summarized as follows (in millions):

		March 31, 2010	December 31, 2009
Liability component			
Principal		\$ 579	\$ 579
Unamortized discount		(171)	(175)
Net carrying amount		\$ 408	\$ 404
Equity component (recorded in Capital in excess of par value of stock)		\$ (3,207)	\$ (3,207)
	21		

We recognized interest cost on our 2036 Convertible Notes as follows (in millions):

		First Quarter		
	2010		2009	
Contractual interest coupon	\$	6 \$	52	
Amortization of discount			34	
Total interest cost on 2036 Convertible Notes	\$	10 \$	86	

The discount on the liability component of the 2036 Convertible Notes will amortize through December 20, 2016, the first put date. The total effective rate on the liability component was 10.5%. If all \$579 million of 2036 Convertible Notes were converted into shares as of March 31, 2010 at a share price of \$12.57, the share value would exceed the principal value of debt by \$212 million.

Subordinated Convertible Debentures

At March 31, 2010, we had outstanding \$3 billion of 6.50% Junior Subordinated Convertible Debentures due 2032 ("Subordinated Convertible Debentures") and \$188 million of deferred interest, reported in *Automotive long-term debt*. The \$3 billion of Subordinated Convertible Debentures are due to Trust II, a subsidiary trust, and are the sole assets of Trust II. As of January 15, 2007, the Subordinated Convertible Debentures have become redeemable at our option.

At March 31, 2010, Trust II had outstanding 6.50% Cumulative Convertible Trust Preferred Securities with an aggregate liquidation preference of \$2.8 billion ("Trust Preferred Securities"). The Trust Preferred Securities are convertible into shares of Ford Common Stock, based on a conversion rate (subject to adjustment) of 2.8769 shares per \$50 principal amount of Trust Preferred Securities (which is equal to a conversion price of \$17.38 per share). We guarantee the payment of all distribution and other payments of the Trust Preferred Securities to the extent not paid by Trust II, but only if and to the extent we have made a payment of interest or principal on the Subordinated Convertible Debentures. As announced on March 27, 2009, we elected to defer future interest payments related to the Trust Preferred Securities for up to 5 years. Trust II is not consolidated by us as it is a VIE in which we do not have a significant variable interest and of which we are not the primary beneficiary.

During the first quarter of 2009, pursuant to a request for conversion, we issued an aggregate of 2,437,562 shares of Ford Common Stock, par value \$0.01 per share, in exchange for \$43 million principal amount of our Subordinated Convertible Debentures.

Secured Term Loan and Revolving Loan

Pursuant to our secured credit agreement (the "Credit Agreement"), at March 31, 2010, we had outstanding:

- \$838 million of revolving loans which bear interest of LIBOR plus a margin of 2.25%, maturing on December 15, 2011; \$6.7 billion of revolving loans which bear interest of LIBOR plus a margin of 3.25%, maturing on November 30, 2013, of which \$3 billion was repaid on April 6, 2010 and remains available for borrowing; and
- \$5.2 billion of a secured term loan maturing on December 15, 2013. The term loan principal amount amortizes at a rate of \$77 million (1% of original loan) per annum and bears interest at LIBOR plus a margin of 3.00%. We are required to use a portion of the cash proceeds from the sale of Volvo (discussed in Note 12) upon the closing thereof (expected in the third quarter of 2010) to partially prepay certain outstanding term loans under the Credit Agreement.

Under the Credit Agreement, we may designate certain of our domestic and foreign subsidiaries, including Ford Credit, as borrowers under the revolving facility. We and certain of our domestic subsidiaries that constitute a substantial portion of our domestic Automotive assets (excluding cash) are guarantors under the Credit Agreement, and future material domestic subsidiaries will become guarantors when formed or acquired.

Collateral. The borrowings of the Company, the subsidiary borrowers, and the guarantors under the Credit Agreement are secured by a substantial portion of our domestic Automotive assets (excluding cash). The collateral includes a majority of our principal domestic manufacturing facilities, excluding facilities to be closed, subject to limitations set forth in existing public indentures and other unsecured credit agreements; domestic accounts receivable; domestic inventory; up to \$4 billion of marketable securities or cash proceeds therefrom; 100% of the stock of our principal domestic subsidiaries,

including Ford Credit (but excluding the assets of Ford Credit); certain intercompany notes of Volvo Holding Company Inc., a holding company for Volvo, Ford Motor Company of Canada, Limited and Grupo Ford S. de R.L. de C.V. (a Mexican subsidiary); 66% to 100% of the stock of all major first tier foreign subsidiaries (including Volvo); and certain domestic intellectual property, including trademarks.

Covenants. The Credit Agreement requires ongoing compliance with a borrowing base covenant and contains other restrictive covenants, including a restriction on our ability to pay dividends. The Credit Agreement prohibits the payment of dividends (other than dividends payable solely in stock) on Ford Common and Class B Stock, subject to certain limited exceptions. In addition, the Credit Agreement contains a liquidity covenant requiring us to maintain a minimum of \$4 billion in the aggregate of domestic cash, cash equivalents, loaned and marketable securities and short-term VEBA assets and/or availability under the revolving credit facility.

With respect to the borrowing base covenant, we are required to limit the outstanding amount of debt under the Credit Agreement as well as certain permitted additional indebtedness secured by the collateral described above such that the total debt outstanding does not exceed the value of the collateral as calculated in accordance with the Credit Agreement.

Events of Default. In addition to customary payment, representation, bankruptcy and judgment defaults, the Credit Agreement contains cross-payment and cross-acceleration defaults with respect to other debt for borrowed money and a change in control default.

2010 Secured Revolver Actions. On March 29, 2010, we gave notice to our administrative agent under the Credit Agreement that we would make an optional prepayment of \$3 billion on April 6, 2010 of our revolving loans that are scheduled to mature on November 30, 2013. Such amount, which was repaid on April 6, 2010, will remain available for borrowing through November 2013 as the commitments of the revolving lenders were not reduced. As a result, at March 31, 2010, \$3 billion of revolving loans were reflected in the Automotive sector as Debt payable within one year.

2009 Secured Term Loan Actions. On March 27, 2009, Ford Credit purchased from term loan lenders under the Credit Agreement \$2.2 billion principal amount of the secured term loan for an aggregate cost of \$1.1 billion (including transaction costs). Consistent with previously-announced plans to return capital from Ford Credit to us, Ford Credit distributed the repurchased secured term loan to its immediate parent, Ford Holdings, whereupon the debt was forgiven. As a result of this transaction, we recorded a pre-tax gain of \$1.1 billion in the first quarter of 2009 in Automotive interest income and other non-operating income/(expense), net.

Notes Due to UAW VEBA Trust

At March 31, 2010 we had outstanding \$7.1 billion in amortizing notes due to the UAW VEBA Trust made up of a non-interest bearing Amortizing Guaranteed Secured Note maturing June 30, 2022 with a par value of \$6.7 billion ("Note A") and a non-interest bearing Amortizing Guaranteed Secured Note maturing June 30, 2022 with a par value of \$6.5 billion ("Note B"). For Note A, we had outstanding \$3.1 billion (\$4.7 billion par value net of \$1.6 billion unamortized discount) using an effective yield of 9.2%. For Note B, we had outstanding \$4 billion (\$5.9 billion par value net of \$1.9 billion unamortized discount) using an effective yield of 9.9%. The Notes allow for prepayments on the annual scheduled principal payment dates. The Notes are secured on a second lien basis, limited to the lesser of an aggregate \$3 billion or the outstanding principal amount of obligations thereunder, with collateral securing our obligations under the Credit Agreement.

Under Note B, we have the option, subject to certain conditions, of making each payment in cash, Ford Common Stock, or a combination of cash and Ford Common Stock. Any Ford Common Stock to be delivered in satisfaction of such payment obligation is to be valued based on its volume-weighted average price per share for the 30 trading-day period ending on the second business day prior to the relevant payment date.

DOE Loan

Pursuant to the Loan Arrangement and Reimbursement Agreement (the "Arrangement Agreement") with the DOE entered into on September 16, 2009, we had outstanding \$1.5 billion in loans as of March 31, 2010. Under the terms of the Arrangement Agreement, the DOE agreed to (i) arrange a 13-year multi-draw term loan facility (the "Facility") under the Advanced Technology Vehicles Manufacturing ("ATVM") Program in the aggregate principal amount of up to \$5.9 billion, (ii) designate us as a borrower under the ATVM Program and (iii) cause the Federal Financing Bank ("FFB")

to enter into the Note Purchase Agreement (the "Note Purchase Agreement") for the purchase of notes to be issued by us evidencing such loans under the Arrangement Agreement. Loans under the ATVM are made by and through the FFB, an instrumentality of the U.S. government that is under the general supervision of the U.S. Secretary of the Treasury.

The proceeds of advances under the Facility will be used to finance certain costs eligible for financing under the ATVM Program ("Eligible Project Costs") that are incurred through mid-2012 in the implementation of 13 advanced technology vehicle programs approved by the DOE (each, a "Project"). The Arrangement Agreement limits the amount of advances that may be used to fund Eligible Project Costs for each Project, and our ability to finance Eligible Project Costs with respect to a Project is conditioned on us meeting agreed timing milestones and fuel economy targets for that Project.

Maturity, Interest Rate and Amortization. Advances under the Facility may be requested through June 30, 2012, and the loans will mature on June 15, 2022 (the "Maturity Date"). Each advance bears interest at a blended rate based on the Treasury yield curve at the time such advance is borrowed, based on the principal amortization schedule for that advance, with interest payable quarterly in arrears. The principal amount of the loans is payable in equal quarterly installments commencing on September 15, 2012 and continuing through the Maturity Date. Per the Arrangement Agreement, we have the ability to voluntarily prepay all or a portion of any advance under the Facility at a prepayment price based on the Treasury yield curve at the time the prepayment is made. It is intended to replicate the price for such advance that would, if it were purchased by a third party and held to maturity, produce a yield to the third-party purchaser for the period from the date of purchase to the Maturity Date substantially equal to the interest rate that would be set on a loan from the Secretary of Treasury to the FFB to purchase an obligation having a payment schedule identical to the payment schedule of such advance for the period from the intended prepayment date to the Maturity Date.

Collateral. The \$5.9 billion commitment is comprised of two loans: (i) a \$1.5 billion note secured by a first priority lien on any assets purchased or developed with the proceeds of the loans, and (ii) a \$4.4 billion note secured by a junior lien on all of the collateral pledged under our Credit Agreement subordinated solely to (a) prior perfected security interests securing certain indebtedness, letters of credit, cash-management obligations and hedging obligations in an aggregate principal amount not to exceed \$19.1 billion as described in the First Amendment to the Arrangement Agreement and (b) certain other permitted liens described in the Arrangement Agreement.

Guarantees. Certain of our subsidiaries that, together with us, hold a substantial portion of the consolidated domestic Automotive assets (excluding cash) and that guarantee the Credit Agreement will guarantee our obligations under the Facility, and future material domestic subsidiaries will become guaranters when formed or acquired.

Affirmative Covenants. The Arrangement Agreement contains affirmative covenants substantially similar to those in the Credit Agreement (including similar baskets and exceptions), as well as certain other affirmative covenants required in connection with the ATVM Program, including compliance with ATVM Program requirements, introduction of advanced technology vehicles to meet or exceed projected overall annual fuel economy improvements and delivery of progress reports and independent auditor reports with respect to the Projects.

Negative Covenants. The Arrangement Agreement contains negative covenants substantially similar to those in the Credit Agreement. The Arrangement Agreement also contains a negative covenant substantially similar to the liquidity covenant in the Credit Agreement requiring that we not permit Available Liquidity (as defined in the Arrangement Agreement) to be less than \$4 billion.

Events of Default. In addition to customary payment, representation, bankruptcy and judgment defaults, the Arrangement Agreement contains cross-payment and cross-acceleration defaults with respect to other debt for borrowed money and a change in control default, as well as events of default specific to the facility.

Financial Services Sector

Unsecured Debt

Debt Repurchases. Through private market transactions, our Financial Services sector repurchased an aggregate of \$201 million and \$341 million principal amount of its outstanding notes in the first quarter of 2010 and 2009, respectively. As a result, our Financial Services sector recorded a pre-tax loss of \$7 million and a pre-tax gain of \$65 million (\$51 million related to Ford Holdings and \$14 million related to Ford Credit), net of unamortized premiums and discounts, in Financial Services other income/(loss), net in the first quarter of 2010 and 2009, respectively.

NOTE 9. DEBT AND COMMITMENTS (Continued)

Asset-Backed Debt

Our Financial Services sector transfers finance receivables and net investments in operating leases in structured transactions to fund operations and to maintain liquidity and the transactions are recorded as secured borrowings. The majority of our Financial Services sector's transactions are secured borrowings and the associated assets are not derecognized and continue to be reported on our financial statements.

The finance receivables and net investment in operating leases that have been included in structured transactions are only available for payment of the debt and other obligations issued or arising in the structured transactions. Cash and cash equivalent and marketable securities balances related to structured transactions are used only to support the structured transactions. The Financial Services sector holds the right to the excess cash flows not needed to pay the debt and other obligations issued or arising in each of the structured transactions. The debt has been issued either directly by us or by consolidated entities.

The following table shows the assets and liabilities related to our Financial Services sector's secured debt arrangements that are included in our financial statements (in billions):

VIEs (b)	Cash and Cash Equivalents and Marketable Securities (a)	March 31, 2010 Finance Receivables, Net and Net Investment in Operating Leases	Related Debt
Finance receivables	\$ 3.9	\$ 54.4	\$ 40.8
Net investment in operating leases	1.2	10.8	7.1
Total	5.1	65.2	47.9
Non-VIE			
Finance receivables (c)	0.5	5.1	4.8
Total securitization transactions			
Finance receivables	4.4	59.5	45.6
Net investment in operating leases	1.2	10.8	7.1
Total	\$ 5.6	\$ 70.3	\$ 52.7
		December 31, 2009	

	C	h and ash valents	Rece Net In Ope	nance vivables, Net and vestment in erating eases		elated Debt
VIEs (b)	ф	2.6	ф	57.4	¢.	20.6
Finance receivables	\$	3.6	\$	57.4	•	39.6
Net investment in operating leases		1.3		10.2		6.6
Total		4.9		67.6		46.2
Non-VIE						
Finance receivables (c)		0.3		6.1		6.7
Total securitization transactions						
Finance receivables		3.9		63.5		46.3
Net investment in operating leases		1.3		10.2		6.6
Total	\$	5.2	\$	73.7	\$	52.9

- (a) Includes marketable securities totaling \$256 million, which are pledged as collateral in a funding arrangement with the ECB.
- (b) Includes assets that can be used to settle obligations of the consolidated VIEs and the related debt of the VIEs. See Note 6 for additional information on Financial Services sector VIEs.
- (c) Certain debt issued by the VIEs to affiliated companies served as collateral for accessing the ECB open market operations program. This external funding of \$736 million and \$1.8 billion at March 31, 2010 and December 31, 2009, respectively was not reflected as a liability of the VIEs and is reflected as a non-VIE liability above. The finance receivables backing this external funding are reflected in VIE finance receivables.

Financial Services sector asset-backed debt also included \$94 million and \$97 million at March 31, 2010 and December 31, 2009 respectfully, that is secured by property.

NOTE 10. OTHER INCOME/(LOSS)

Automotive Sector

The following table summarizes amounts included in *Automotive interest income and other non-operating income/(expense)*, *net* for the periods ending March 31 (in millions):

	First (<u>Quarter</u>
	2010	2009
Interest income	\$ 47	\$ 62
Realized and unrealized gains/(losses) on cash equivalents and marketable securities	119	(6)
Gains/(Losses) on the sale of held-for-sale operations, equity and cost investments, and other dispositions	4	13
Gains/(Losses) on extinguishment of debt	•	1,279
Other *	19	4
Total	\$ 189	\$ 1,352

^{* 2009} includes \$9 million of expense in other income associated with the overall debt reduction actions discussed in Note 1.

Financial Services Sector

The following table summarizes the amounts included in Financial Services other income/(loss), net for the periods ending March 31 (in millions):

<u>FHS</u>	ու 	<u>ıarter</u>
$\overline{201}$.0	2009
Interest income (non-financing related) \$	16	\$ 44
Realized and unrealized gains/(losses) on cash equivalents and marketable securities	2	(13)
Gains/(Losses) on the sale of held-for-sale operations, equity and cost investments, and other dispositions	1	2
Gains/(Losses) on extinguishment of debt	(7)	65
Investment and other income related to sales of receivables	•	10
Insurance premiums earned, net	26	29
Other	88	(24)
Total <u>\$ 12</u>	26	\$ 113

NOTE 11. INCOME TAXES

Generally, for interim tax reporting we estimate one single tax rate for tax jurisdictions not subject to a valuation allowance, which is applied to the year-to-date ordinary income/(loss). We manage our operations by multi-jurisdictional business units, however, and thus are unable to reasonably compute one overall effective tax rate. Accordingly, our worldwide tax provision is calculated pursuant to U.S. GAAP, which provides that tax (or benefit) in each foreign jurisdiction not subject to valuation allowance be separately computed as ordinary income/(loss) occurs within the jurisdiction.

NOTE 12. HELD-FOR-SALE OPERATIONS, DISCONTINUED OPERATIONS, OTHER DISPOSITIONS, AND ACQUISITIONS

We classify disposal groups as held for sale when management, having the authority to approve the action, commits to a plan to sell the disposal group, the sale is probable within one year and the disposal group is available for immediate sale in its present condition. We also consider whether an active program to locate a buyer has been initiated, whether the disposal group is actively marketed for sale at a price that is reasonable in relation to its current fair value, and whether actions required to complete the plan indicate that it is unlikely that significant changes to the plan will be made or that the plan will be withdrawn. We classify disposal groups as discontinued operations when the criteria to be classified as held for sale have been met and we will not have any significant involvement with the disposal groups after the sale.

NOTE 12. HELD-FOR-SALE OPERATIONS, DISCONTINUED OPERATIONS, OTHER DISPOSITIONS, AND ACQUISITIONS (Continued)

Automotive Sector

Held-for-Sale Operations

Volvo. In the fourth quarter of 2008, we performed annual goodwill impairment testing for our Volvo reporting unit. We compared the carrying value of our Volvo reporting unit to its fair value, and concluded that the goodwill was not impaired. We performed this measurement relying primarily on the income approach, applying a discounted cash flow methodology. Our valuation was based on an in-use premise which considered a discount rate, after-tax return on sales rate, growth rate, and terminal value consistent with assumptions we believed principal market participants (i.e., other global automotive manufacturers) would use. This methodology produced appropriate valuations for entities we disposed of in recent years; in light of worsening economic conditions, however, we also considered other valuations, including a discounted cash flow analysis using more conservative assumptions than we initially used. This alternative analysis incorporated a significantly higher discount rate, offset partially by a higher growth rate; a much lower after-tax return on sales rate; and a lower terminal value. This alternative analysis reduced the valuation of our Volvo reporting unit by about 50%. Even this more conservative analysis, however, did not support an impairment of Volvo goodwill at year end.

As previously disclosed, in recent years we have undertaken efforts to divest non-core assets in order to allow us to focus exclusively on our global Ford brand. During the first quarter of 2009, based on our strategic review of Volvo and in light of our goal to focus on the global Ford brand, our Board of Directors committed to actively market Volvo for sale, notwithstanding the current distressed market for automotive-related assets. Accordingly, in the first quarter of 2009 we reported Volvo as held for sale.

Our commitment to actively market Volvo for sale also triggered a held-for-sale impairment test in the first quarter of 2009. We received information from our discussions with potential buyers that provided us a value for Volvo using a market approach, rather than an income approach. We concluded that the information we received from our discussions with potential buyers was more representative of the value of Volvo given the current market conditions, the characteristics of viable market participants, and our anticipation of a more immediate transaction for Volvo. These inputs resulted in a lower value for Volvo than the discounted cash flow method we had previously used.

In the first quarter of 2009, after considering deferred gains reported in *Accumulated other comprehensive income/(loss)*, we recognized a pre-tax impairment charge of \$650 million related to our total investment in Volvo. The impairment was recorded in *Automotive cost of sales*.

Had we not committed to actively market Volvo for sale, we would not have been afforded the benefit of the new information obtained in discussions with potential buyers. Rather, we would have continued to employ an in-use premise to test Volvo's goodwill and long-lived assets, using a discounted cash flow methodology with assumptions similar to those we used at year-end 2008.

On March 28, 2010, Ford entered into a definitive agreement to sell Volvo and related assets to Zhejiang Geely Holding Group for \$1.8 billion, subject to customary purchase price adjustments. The sale is expected to close in the third quarter of 2010.

NOTE 12. HELD-FOR-SALE OPERATIONS, DISCONTINUED OPERATIONS, OTHER DISPOSITIONS, AND ACQUISITIONS (Continued)

The assets and liabilities of Volvo classified as held for sale are as follows:

	March 31, 2010			ecember 31, 2009
Assets				
Receivables	\$	584	\$	420
Inventories		1,323		1,236
Net property		4,821		4,682
Goodwill		1,230		1,241
Other intangibles		228		204
Other assets		540		485
Impairment of carrying value		(650)		(650)
Total assets of the held-for-sale operations	\$	8,076	\$	7,618
Liabilities				
Payables	\$	2,092	\$	1,980
Pension liabilities		359		387
Warranty liabilities		258		358
Other liabilities		2,935		2,596
Total liabilities of the held-for-sale operations	\$	5,644	\$	5,321

Financial Services Sector

Held-for-Sale Operations

Primus Leasing Company Limited ("Primus Thailand"). In March 2009, Ford Credit completed the sale of Primus Thailand, its operation in Thailand that offered automotive retail and wholesale financing of Ford, Mazda and Volvo vehicles. As a result of the sale, Ford Credit received \$165 million in proceeds and recognized a de minimis pre-tax gain in Financial Services other income/(loss), net.

NOTE 13. AMOUNTS PER SHARE ATTRIBUTABLE TO FORD MOTOR COMPANY COMMON AND CLASS B STOCK

The calculation of diluted income/(loss) per share of Ford Common Stock and Class B Stock takes into account the effect of obligations, such as restricted stock unit awards, stock options and warrants, and convertible notes and securities, considered to be potentially dilutive.

Warrants

In conjunction with the transfer of assets to the UAW VEBA Trust on December 31, 2009, warrants to purchase 362,391,305 shares of Ford Common Stock at an exercise price of \$9.20 per share were issued. On April 6, 2010, the UAW VEBA Trust sold all such warrants to parties unrelated to us. In connection with the sale, the terms of the warrants were modified to provide for, among other things, net share settlement as the only permitted settlement method, thereby eliminating full physical settlement as an option, and elimination of certain of the transfer restrictions applicable to the underlying stock. The Company received no proceeds from the offering. All warrants are fully exercisable and expire January 1, 2013.

Option to Stock-Settle Debt

On December 31, 2009 we issued Note B to the UAW VEBA Trust (see our 2009 Form 10-K Report for additional discussion of Note B (referred to therein as "New Note B") and the Settlement Agreement pursuant to which it was issued). Under the terms of Note B, we have the exclusive right, subject to certain conditions, to make each principal payment due on any principal payment date (i.e., a fixed dollar amount) in cash, Ford Common Stock, or a combination of cash and Ford Common Stock. Under no circumstances does the UAW VEBA Trust have a right to require us to pay Note B by a transfer of Ford Common Stock.

If we elect to satisfy our scheduled payment by delivering shares of Ford Common Stock, the number of shares delivered would be determined by dividing the principal payment due by the volume-weighted average price (the "VWAP") per share of Ford Common Stock for the 30 trading days ending on the second business day immediately preceding the June 30 payment date each year. We may decide whether to settle our principal payment in shares at any time during the two-day period prior to the applicable payment date. As previously disclosed, we will use our discretion in determining which form of payment makes economic sense at the time of each required payment, balancing liquidity needs and preservation of shareholder value. We made our December 31, 2009 payment on Note B in cash. We have the option to prepay Note B in part or in full on each principal payment date, but only with cash.

NOTE 13. AMOUNTS PER SHARE ATTRIBUTABLE TO FORD MOTOR COMPANY COMMON AND CLASS B STOCK (Continued)

Our March 31, 2010 diluted earnings per share ("EPS") calculation includes approximately 464 million shares (using the VWAP of \$12.71 per share at March 29, 2010), assuming a hypothetical situation under which Note B could be satisfied in full with shares rather than cash for the fully accreted amount of the obligation, even though the terms of Note B permit only a prepayment in cash. This hypothetical option to settle Note B in full with stock reduced our diluted EPS in the first quarter by 3 cents per share. The number of shares used in this calculation may vary significantly based on the outstanding principal amount of Note B at each period-end, and based on the 30-day VWAP ending on the second business day prior to each period-end. If we decide to settle with shares each scheduled payment of Note B, the total number of shares issued may be materially different than the number used in our diluted EPS calculation.

Amounts Per Share

Basic and diluted income/(loss) per share were calculated using the following (in millions):

	First C	<u>Quarter</u>
	2010	2009
Basic and Diluted Income/(Loss) Attributable to Ford Motor Company		
Basic income/(loss) from continuing operations	\$2,085	\$(1,427)
Effect of dilutive 2016 Convertible Notes (a)	48	•
Effect of dilutive 2036 Convertible Notes (a)(b)	11	•
Effect of dilutive UAW VEBA Note B (a)	97	•
Effect of dilutive Trust Preferred Securities (a)(c)	51	
Diluted income/(loss) from continuing operations	\$2,292	\$(1,427)
Basic and Diluted Shares		
Average shares outstanding	3,366	2,398
Restricted and uncommitted-ESOP shares	(1)	(1)
Basic shares	3,365	2,397
Net dilutive options, warrants, and restricted and uncommitted-ESOP shares (d)	205	•
Dilutive 2016 Convertible Notes	309	•
Dilutive 2036 Convertible Notes (b)	63	•
Dilutive UAW VEBA Note B	464	•
Dilutive convertible Trust Preferred Securities (c)	163	
Diluted shares	4,569	2,397

- (a) As applicable, includes interest expense, amortization of discount, amortization of fees, and other changes in income or loss that would result from the assumed conversion.
- Not included in calculation of diluted EPS due to their antidilutive effect:
- (b) 531 million shares for the first quarter of 2009 and the related income effect for 2036 Convertible Notes.
- (c) 160 million shares for the first quarter of 2009 and the related income effect for Trust Preferred Securities.
- (d) 35 million contingently-issuable shares for first quarter of 2009.

NOTE 14. DERIVATIVE FINANCIAL INSTRUMENTS AND HEDGING ACTIVITIES

In the normal course of business, our operations are exposed to global market risks, including the effect of changes in foreign currency exchange rates, certain commodity prices, and interest rates. To manage these risks, we enter into various derivatives contracts. Foreign currency exchange contracts including forwards and options, are used to manage foreign exchange exposure. Commodity contracts including forwards and options are used to manage commodity price risk. Interest rate contracts including swaps, caps, and floors are used to manage the effects of interest rate fluctuations. Cross-currency interest rate swap contracts are used to manage foreign currency and interest rate exposures on foreign-denominated debt. Our derivatives are over-the-counter customized derivative transactions and are not exchange-traded. Management reviews our hedging program, derivative positions, and overall risk management strategy on a regular basis.

We have elected to apply hedge accounting to certain derivatives. Derivatives that receive designated hedge accounting treatment are documented and the relationships are evaluated for effectiveness at the time they are designated, as well as throughout the hedge period. Some derivatives do not qualify for hedge accounting; for others, we elect not to apply hedge accounting. Regardless of hedge accounting treatment, we only enter into transactions we believe will be highly effective at offsetting the underlying economic risk. Refer to our 2009 Form 10-K Report for a more detailed description of our derivative financial instruments and hedge accounting policies.

Income Effect of Derivative Instruments

The following tables summarize by hedge designation the pre-tax gains/(losses) recorded in Other comprehensive income/(loss) ("OCI"), reclassified from *Accumulated other comprehensive income/(loss)* ("AOCI") to income and recognized directly in income (in millions):

Automotive Sector	Recor	(Loss) ded in CI	Gain/Gain/Gain/Gain/Gain/Gain/Gain/Gain/	(Loss) ssified AOCI to	Rece	n/(Loss) ognized ncome
Designated Cash flow hedges:						
Foreign exchange contracts	\$	(5)	\$	(3)	\$	•
Derivatives not designated as hedging instruments: Foreign exchange contracts (a) Commodity contracts Other – warrants					\$	(155)
Total					¢	(157)
Total					Φ	(137)
Financial Services Sector Fair value hedges: Interest rate contracts Net interest settlements and accruals excluded from the assessment of hedge effectiveness					\$	53
Ineffectiveness (b)					Φ.	(2)
Total					\$	51
Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts (a) Cross currency interest rate swap contracts (a) Other – warrants					\$	29 (80) (8)
Total					\$	(59)

 ⁽a) Gains/(Losses) related to foreign currency derivatives were substantially offset by net revaluation impacts on foreign denominated assets and liabilities, which were recorded to the same statement of operations line item as the derivative gains/(losses).
 (b) Hedge ineffectiveness reflects change in fair value on derivatives of \$43 million gain and change in fair value on hedged debt of

⁽b) Hedge ineffectiveness reflects change in fair value on derivatives of \$43 million gain and change in fair value on hedged debt of \$45 million loss in first quarter 2010.

	First Quarter 2009						
Automotive Sector	Gain/(Loss) Recorded in OCI		Recla from	/(Loss) ssified AOCI to	Reco	n/(Loss) ognized ncome	
Designated Cash flow hedges:							
Foreign exchange contracts	\$	(55)	\$	28(a)	\$	(1)	
Commodity contracts	Φ.	<u>•</u>	Ф	4	Φ.	- (1)	
Total	\$	(55)	\$	32	\$	(1)	
Derivatives not designated as hedging instruments:					ф	7.5	
Foreign exchange contracts – operating exposures (b)					\$	75	
Foreign exchange contracts – investment portfolios (c) Commodity contracts						(1)	
Other – warrants						(30)	
Total					\$	44	
					_		
Financial Services Sector							
Fair value hedges:							
Interest rate contracts							
Net interest settlements and accruals excluded from the assessment of hedge effectiveness					\$	24	
Ineffectiveness (d)					Ф	(10)	
Total					\$	14	
					<u> </u>		
Derivatives not designated as hedging instruments: Interest rate contracts					\$	(95)	
Foreign exchange contracts (b)					Ψ	177	
Cross currency interest rate swap contracts (b)						73	
Other - warrants						(1)	
Total					\$	154	

- (a) Includes \$4 million gain reclassified from AOCI to income in first quarter 2009 attributable to transactions no longer probable to occur, primarily related to Volvo.
- (b) Gains/(Losses) related to foreign currency derivatives were partially offset by net revaluation impacts on foreign denominated assets and liabilities, which were recorded to the same statement of operations line item as the derivative gains/(losses).
- (c) Foreign exchange contracts investment portfolios on the balance sheet were \$1 million at March 31, 2009.
- (d) Hedge ineffectiveness reflects change in fair value on derivatives of \$1 million gain and change in fair value on hedged debt of \$11 million loss in first quarter 2009.

For our Automotive sector, we report in *Automotive cost of sales* on our consolidated statement of operations gains and losses on cash flow hedges and foreign exchange contracts on operating exposures and commodity contracts not designated as hedging instruments. Gains and losses on foreign exchange contracts on investment portfolios and other contracts not designated as hedging instruments are reported in *Automotive interest income and other non-operating income/(expense)*, net.

For our Financial Services sector, we report net interest settlements and accruals on fair value hedges (which are excluded from the assessment of hedge effectiveness) in *Interest expense* on our consolidated statement of operations. Foreign currency revaluation on accrued interest for fair value hedges was less than \$1 million in the first quarter of 2010 and 2009. Ineffectiveness on fair value hedges and gains and losses on interest rate contracts not designated as hedging instruments are reported in *Financial Services other income/(loss)*, net. Gains and losses on foreign exchange and cross currency interest rate swap contracts not designated as hedging instruments are reported in *Selling*, administrative, and other expenses.

Balance Sheet Effect of Derivative Instruments

The following tables summarize the estimated fair value of our derivative financial instruments (in millions):

		March 31, 2010				
	N	Fair	Value of ssets		Value of bilities	
Automotive Sector						
Cash flow hedges: Foreign exchange contracts	\$	90	\$	•	\$	8
Foreign exchange contracts	Þ	90	Ф	Ÿ	Ф	0
Derivatives not designated as hedging instruments:						
Foreign exchange contracts		4,327		59		120
Commodity contracts		657		12		29
Other – warrants		12		3		•
Total derivatives not designated as hedging instruments		4,996		74		149
Total Automotive sector derivative instruments	\$	5,086	\$	74	\$	157
Financial Services Sector						
Fair value hedges:						
Interest rate contracts	\$	6,239	\$	392	\$	•
Derivatives not designated as hedging instruments:						
Interest rate contracts		63,610		1,082		677
Foreign exchange contracts		4,107		6		85
Cross currency interest rate swap contracts		2,642		72		302
Total derivatives not designated as hedging instruments		70,359		1,160	_	1,064
Total Financial Services sector derivative instruments	\$	76,598	\$	1,552	\$	1,064
			Decembe	er 31, 2009		
		Fair Val				Value of bilities
	No	บนบทลเร	A	ssets	Lia	
Automotive Sector	N	otionais	A	ssets	<u>Lia</u>	bilities
Cash flow hedges:				ssets		
		118	A \$	ssets	Lia	5
Cash flow hedges: Foreign exchange contracts				ssets		
Cash flow hedges:				• 59		
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments:		118		•		5
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts		118 4,255		59 15 2		5 80 54
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts		118 4,255 980		59 15		5 80 54
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants		118 4,255 980 12		59 15 2		5 80 54
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments		4,255 980 12 5,247		59 15 2 76		80 54 •
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector		4,255 980 12 5,247		59 15 2 76		80 54 •
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments		4,255 980 12 5,247		59 15 2 76		80 54 •
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts	\$ <u>\$</u>	4,255 980 12 5,247 5,365	\$ 	59 15 2 76	\$	5 80 54 • 134 139
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges:	\$ <u>\$</u>	118 4,255 980 12 5,247 5,365 6,309	\$ 	59 15 2 76 76 385	\$	5 80 54 • 134 139
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts	\$ <u>\$</u>	118 4,255 980 12 5,247 5,365 6,309 68,527 4,386	\$ 	59 15 2 76 76 385	\$	5 80 54 • 134 139
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts Cross currency interest rate swap contracts	\$ <u>\$</u>	118 4,255 980 12 5,247 5,365 6,309 68,527 4,386 3,873	\$ 	59 15 2 76 76 385	\$	80 54 • 134 139 • 846 46 282
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts	\$ <u>\$</u>	118 4,255 980 12 5,247 5,365 6,309 68,527 4,386	\$ 	59 15 2 76 76 385	\$	5 80 54 • 134 139
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts Cross currency interest rate swap contracts	\$ <u>\$</u>	118 4,255 980 12 5,247 5,365 6,309 68,527 4,386 3,873	\$ 	59 15 2 76 76 385	\$	80 54 • 134 139 • 846 46 282
Cash flow hedges: Foreign exchange contracts Derivatives not designated as hedging instruments: Foreign exchange contracts Commodity contracts Other – warrants Total derivatives not designated as hedging instruments Total Automotive sector derivative instruments Financial Services Sector Fair value hedges: Interest rate contracts Derivatives not designated as hedging instruments: Interest rate contracts Foreign exchange contracts Cross currency interest rate swap contracts Total derivatives not designated as hedging instruments	\$ <u>\$</u>	4,255 980 12 5,247 5,365 6,309 68,527 4,386 3,873 76,786	\$ 	59 15 2 76 76 76 385 1,269 22 203 1,494	\$	80 54 • 134 139 • 846 46 282 1,174

In our consolidated balance sheet, we report derivative assets in *Other assets*, and derivative liabilities in *Payables* and *Accrued liabilities and deferred revenue* for Automotive and Financial Services sectors, respectively. To ensure consistency in our treatment of derivative and non-derivative exposures with regard to our master agreements, we do not net our derivative position by counterparty for purposes of balance sheet presentation and disclosure.

The notional amounts of the derivative financial instruments do not necessarily represent amounts exchanged by the parties and, therefore, are not a direct measure of our exposure to the financial risks described above. The amounts exchanged are calculated by reference to the notional amounts and by other terms of the derivatives, such as interest rates, foreign currency exchange rates or commodity volumes and prices.

Counterparty Risk and Collateral

Use of derivatives exposes us to the risk that a counterparty may default on a derivative contract. We establish exposure limits for each counterparty to minimize this risk and provide counterparty diversification. Substantially all of our derivative exposures are with counterparties that have long-term credit ratings of single-A or better. The aggregate fair value of derivative instruments in asset positions on March 31, 2010 was about \$2 billion, representing the maximum loss that we would recognize at that date if all counterparties failed to perform as contracted. We enter into master agreements with counterparties that generally allow for netting of certain exposures; therefore, the actual loss we would recognize if all counterparties failed to perform as contracted would be significantly lower.

We include an adjustment for non-performance risk in the fair value of derivative instruments. At March 31, 2010, our adjustment for non-performance risk relative to a measure based on an unadjusted inter-bank deposit rate (e.g., LIBOR) reduced derivative assets by \$0 and \$14 million for Automotive and Financial Services sectors, respectively, and reduced derivative liabilities by \$0 and \$11 million for Automotive and Financial Services sectors, respectively. See Note 2 for more detail on valuation methodologies.

We post cash collateral with certain counterparties based on our net position with regard to foreign currency and commodity derivative contracts. As of March 31, 2010, we posted \$109 million in cash collateral related to derivative instruments, which is included in restricted cash and reported in *Other assets* on our consolidated balance sheet.

NOTE 15. SEGMENT INFORMATION

(In millions)			Au	tomotive	Sector			
EVECTO ON A DIVER 2010	Ford North America	Ford South America	Ford Europe	Ford Asia Pacific <u>Africa</u>	Volvo	Jaguar Land Rover	Other	Total
FIRST QUARTER 2010 Sales/Revenues								
External customer	\$ 14,132	\$ 2,014	\$ 7,647	\$1,578	\$3,523	\$ •	\$ •	\$28,894
Intersegment	201		167	•	11	•	•	379
Income/(Loss)								
Income/(Loss) before income taxes	1,223	203	75	22	188	•	(391)	
Total assets at March 31 (a)								81,956
FIRST QUARTER 2009								
Sales/Revenues								
External customer	\$ 10,018	\$ 1,404	\$ 5,769	\$1,165	\$2,624	\$ •	\$ •	\$20,980
Intersegment	146	•	171	•	12	•	•	329
Income/(Loss)								
Income/(Loss) before income taxes	(802)) 63	(590)	(104)	(915)	(2)	840	(1,510)
Total assets at March 31								74,017

	Financial Services Sector							Total Company		
FIRST QUARTER 2010 Sales/Revenues	Fo	ord Credit		Other Financial Services	_	Elims	Total	_	Elims (b)	Total
External customer	\$	2,594	\$	78	\$	• \$	2,672	\$	•	\$ 31,566
Intersegment	·	130		3		•	133		(512)	•
Income/(Loss)										
Income/(Loss) before income taxes		828		(13)		•	815		•	2,135
Total assets at March 31 (a)		115,555		8,558		(7,052)	117,061		(3,532)	195,485
FIRST QUARTER 2009										
Sales/Revenues										
External customer	\$	3,330	\$	80	\$	• \$	3,410	\$	•	\$ 24,390
Intersegment		105		4		•	109		(438)	•
Income/(Loss)										
Income/(Loss) before income taxes		(36)		(116)		•	(152)		•	(1,662)
Total assets at March 31		132,307		10,820		(9,072)	134,055		(3,237)	204,835

⁽a) As reported on our sector balance sheet.

NOTE 16. COMMITMENTS

Guarantees

At March 31, 2010 and December 31, 2009, the following guarantees and indemnifications were issued and outstanding:

Guarantees related to affiliates and third parties. We guarantee debt and lease obligations of certain joint ventures, as well as certain financial obligations of outside third parties including suppliers to support our business and economic growth. Expiration dates vary through 2017, and guarantees will terminate on payment and/or cancellation of the obligation. A payment by us would be triggered by failure of the guaranteed party to fulfill its obligation covered by the guarantee. In some circumstances, we are entitled to recover from the third party amounts paid by us under the guarantee. However, our ability to enforce these rights is sometimes stayed until the guaranteed party is paid in full, and may be limited in the event of insolvency of the third party or other circumstances. Maximum potential payments under guarantees total \$247 million and \$219 million at March 31, 2010 and December 31, 2009, respectively. The carrying value of our recorded liabilities related to guarantees was \$24 million and \$30 million at March 31, 2010 and December 31, 2009, respectively. Our performance risk under these guarantees is reviewed regularly, and has resulted in no changes to our initial valuations.

⁽b) Includes intersector transactions occurring in the ordinary course of business.

NOTE 16. COMMITMENTS (Continued)

In December 2005, we completed the sale of Hertz. As part of this transaction, we provided cash-collateralized letters of credit in an aggregate amount of \$200 million to support the asset-backed portion of the buyer's financing for the transaction. Our commitment to provide the letters of credit expires no later than December 21, 2011 and supports the payment obligations of Hertz Vehicle Finance LLC under one or more series of asset-backed notes. The letters of credit can be drawn upon on any date funds allocated to pay interest on the asset-backed notes are insufficient to pay scheduled interest payments, principal amounts due on the legal final maturity date, or when the balance of assets supporting the asset-backed notes is less than the outstanding balance of the asset-backed notes. The carrying value of our deferred gain related to the letters of credit was \$8 million and \$9 million at March 31, 2010 and December 31, 2009, respectively. We believe future performance under these letters of credit is remote.

Indemnifications. In the ordinary course of business, we execute contracts involving indemnifications standard in the industry and indemnifications specific to a transaction, such as the sale of a business. These indemnifications might include claims relating to any of the following: environmental, tax, and shareholder matters; intellectual property rights; power generation contracts; governmental regulations and employment-related matters; dealers, supplier, and other commercial contractual relationships; and financial matters, such as securitizations. Performance under these indemnities generally would be triggered by a breach of terms of the contract or by a third-party claim.

As part of the sale of Jaguar Land Rover, we provided the buyer a customary set of indemnifications, some of which were subject to an aggregate limit of \$805 million (which expired June 1, 2009) and some of which (e.g., warranties related to our capacity and authority to enter into the transaction, our ownership of the companies sold and the shares of those companies free from encumbrances and certain tax covenants) are unlimited in amount. Outstanding claims of \$139 million relating to the \$805 million indemnification as well as indemnifications relating to certain warranties described in the preceding sentence continue. We believe that the probability of payment under these claims and indemnifications is remote.

We also are party to numerous indemnifications which do not limit potential payment; therefore, we are unable to estimate a maximum amount of potential future payments that could result from claims made under these indemnities.

Warranty

Included in the warranty cost accruals are costs for basic warranty coverages on vehicles sold. Additional service actions, such as product recalls and other customer service actions, are not included in the warranty reconciliation below, but are also accrued for at the time of sale. Estimates for warranty costs are made based primarily on historical warranty claim experience. Product warranty accruals accounted for in *Accrued liabilities and deferred revenue* were as follows (in millions):

	First Q	uarter
	2010	2009
Beginning balance	\$3,147	\$3,239
Payments made during the period	(582)	(634)
Changes in accrual related to warranties issued during the period	441	308
Changes in accrual related to pre-existing warranties	95	371
Foreign currency translation and other	(35)	(45)
Ending balance	\$3,066	\$3,239
		4.5

NOTE 17. EQUITY/(DEFICIT) ATTRIBUTABLE TO FORD MOTOR COMPANY AND NONCONTROLLING INTERESTS

 $Components\ of\ equity/(deficit)\ attributable\ to\ Ford\ Motor\ Company\ and\ its\ noncontrolling\ interests\ are\ as\ follows\ (in\ millions):$

		First Quarter 2010		First Quarter 2009				
	Equity/(Deficit) Attributable to Ford Motor Company	Equity/(Deficit) Attributable to Noncontrolling Interests	Total Equity/ (Deficit)	Equity/(Deficit) Attributable to Ford Motor Company	Equity/(Deficit) Attributable to Noncontrolling Interests	Total Equity/ (Deficit)		
Beginning balance, January 1	\$ (7,820)	\$ 38	\$(7,782)	\$ (15,721)	\$ 350	\$ (15,371)		
Total comprehensive income/(loss)								
Net income/(loss)	2,085	•	2,085	(1,427)	(8)	(1,435)		
Other comprehensive income/(loss):								
Foreign currency translation	(489)	•	(489)	(447)	•	(447)		
Net gain/(loss) on derivative instruments	(1)	•	(1)	(87)	•	(87)		
Employee benefit-related	157	•	157	(5)	•	(5)		
Net holding gain/(loss)	(2	•	(2)	(1)		(1)		
Total other comprehensive income/(loss)	(335	•	(335)	(540)		(540)		
Total comprehensive income/(loss)	1,750	•	1,750	(1,967)	(8)	(1,975)		
Other changes in equity:								
Capital in excess of par value of stock for debt								
conversion, employee benefit plans, and other	596	•	596	110	•	110		
Adoption of the accounting standard on								
VIE consolidation	•	•	•	•	(269)	(269)		
Dividends	•	•	•	•	•	•		
Other	(1)	•	(1)	1	<u>(5)</u>	(4)		
Ending balance, March 31	\$ (5,475)	\$ 38	\$(5,437)	\$ (17,577)	\$ 68	\$ (17,509)		

Report of Independent Registered Public Accounting Firm

To Board of Directors and Stockholders Ford Motor Company:

We have reviewed the accompanying consolidated balance sheet of Ford Motor Company and its subsidiaries as of March 31, 2010, and the related consolidated statements of operations and comprehensive income for the three-month periods ended March 31, 2010 and 2009 and the condensed consolidated statement of cash flows for the three-month periods ended March 31, 2010 and 2009. These interim financial statements are the responsibility of the Company's management.

The accompanying sector balance sheets and the related sector statements of operations and of cash flows are presented for purposes of additional analysis and are not a required part of the basic financial statements. Such information has been subjected to the review procedures applied in the review of the basic financial statements.

We conducted our review in accordance with the standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with the standards of the Public Company Accounting Oversight Board (United States), the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our review, we are not aware of any material modifications that should be made to the accompanying consolidated interim financial statements for them to be in conformity with accounting principles generally accepted in the United States of America.

We previously audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheet as of December 31, 2009 and the related consolidated statements of operations, of stockholders' equity, and of cash flows for the year then ended (not presented herein), and in our report dated February 25, 2010, we expressed an unqualified opinion (with an explanatory paragraph relating to changes in the method of accounting for noncontrolling interests and convertible debt instruments that may be settled in cash upon conversion (including partial cash settlement) in 2009 and the method of accounting for uncertain tax positions in 2007) on those consolidated financial statements. As discussed in Note 1 to the accompanying consolidated financial statements, the Company adopted the Financial Accounting Standards Board's new standard on accounting for variable interest entity consolidation on January 1, 2010. The accompanying December 31, 2009 consolidated balance sheet reflects this change.

/s/ PricewaterhouseCoopers LLP

PricewaterhouseCoopers LLP

Detroit, Michigan May 7, 2010

FIRST QUARTER RESULTS OF OPERATIONS

Our worldwide net income attributable to Ford Motor Company was \$2.1 billion or \$0.50 per share of Common and Class B Stock in the first quarter of 2010, an improvement of \$3.5 billion from a net loss attributable to Ford Motor Company of \$1.4 billion or \$0.60 per share of Common and Class B Stock in the first quarter of 2009.

Results by sector are shown below (in millions):

	First Quarter				
	2010	2009*	2010 Over/ (Under) 2009		
Income/(Loss) before income taxes					
Automotive sector	\$1,320	\$(1,510)	\$ 2,830		
Financial Services sector	815	(152)	967		
Total Company	2,135	(1,662)	3,797		
Provision for/(Benefit from) income taxes	50	(227)	277		
Income/(Loss) from continuing operations	2,085	(1,435)	3,520		
Income/(Loss) from discontinued operations		•	•		
Net income/(loss)	2,085	(1,435)	3,520		
Less: Income/(Loss) attributable to noncontrolling interests	·	(8)	8		
Net income/(loss) attributable to Ford Motor Company	\$2,085	\$(1,427)	\$ 3,512		

^{*} Adjusted for the effect of the new accounting standard on VIE consolidation; see Note 1 of the Notes to the Financial Statements for additional detail.

Income/(Loss) before income taxes includes certain items ("special items") that we have grouped into "Personnel and Dealer-Related Items" and "Other Items" to provide useful information to investors about the nature of the special items. The first category includes items related to our efforts to match production capacity and cost structure to market demand and changing model mix and therefore helps investors track amounts related to those activities. The second category includes items that we do not generally consider to be indicative of our ongoing operating activities, and therefore allows investors analyzing our pre-tax results to identify certain infrequent significant items that they may wish to exclude when considering the trend of ongoing operating results.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Continued)

The following table details special items in each category by segment or business unit (in millions):

	Firs	t Quarter -	 Incom 	ne/(Loss)
Personnel and Dealer-Related Items:	2	2010		2009
Automotive Sector				
Ford North America				
Personnel-reduction programs	\$	(53)	\$	(170)
U.S. dealer actions		(15)		(81)
Retiree health care and related charges		20		(178)
Job Security Benefits/Transition Assistance Plan ("JSB/TAP")		18		292
Total Ford North America		(30)		(137)
Ford Europe				
Personnel-reduction programs		(32)		(5)
Ford Asia Pacific Africa				
Personnel-reduction programs		<u>(1</u>)		<u>(7</u>)
Total Personnel and Dealer-Related Items - Automotive sector		(63)		(149)
Other Items:				
Automotive Sector				
Other Automotive				
Gain on debt reduction actions and related costs		•		1,270
Volvo				
Held-for-sale impairment and related items *		188		(666)
Jaguar Land Rover				
Other		<u> </u>		(2)
Total Other Items – Automotive sector		188		602
Financial Services Sector				
DFO Partnership impairment		•		(141)
Gain on purchase of Ford Holdings debt securities		<u> </u>		51
Total Other Items – Financial Services sector		-		(90)
Total	\$	125	\$	363

^{*} First quarter 2010 includes operating profit of \$49 million.

Included in *Provision for/(Benefit from) income taxes* are tax benefits of \$199 million and \$3 million for the first quarter of 2010 and 2009, respectively, that we consider to be special items. These primarily consist of the tax effects of the pre-tax special items listed above, and release of our valuation allowance related to our Canadian financial services operations.

Discussion of Automotive and Financial Services sector results of operations below is on a pre-tax basis, and includes special items unless otherwise specifically noted. Discussion of overall Automotive cost changes, including structural cost changes (e.g., manufacturing and engineering, pension/OPEB, overhead, etc.), is primarily at prior-year exchange and excludes special items and discontinued operations. In addition, costs that vary directly with production volume, such as material, freight, and warranty costs, are measured at prior-year volume and mix.

AUTOMOTIVE SECTOR

Results of Operations

Details by segment or business unit of *Income/(Loss) before income taxes* are shown below for the first quarter of 2010 and 2009 (in millions), with Volvo and Jaguar Land Rover separated out from "ongoing" subtotals:

	First Quarter						
	2010	2009	2010 Over/(Under) 2009				
Ford North America	\$1,223	\$ (802)	\$ 2,025				
Ford South America	203	63	140				
Ford Europe	75	(590)	665				
Ford Asia Pacific Africa	22	(104)	126				
Total ongoing Automotive operations	1,523	(1,433)	2,956				
Other Automotive *	(391)	840	(1,231)				
Total ongoing Automotive	1,132	(593)	1,725				
Volvo	188	(915)	1,103				
Jaguar Land Rover		(2)	2				
Total Automotive sector	\$1,320	\$(1,510)	\$ 2,830				

^{*} For first quarter 2010, the loss of \$391 million is more than explained by net interest expense of \$492 million, which was comprised of about \$550 million of interest expense, offset partially by interest income. In addition, there was \$101 million of favorable fair market value adjustment, associated primarily with our investment in Mazda.

Details by segment of Automotive revenues ("sales") and wholesale unit volumes for the first quarter of 2010 and 2009 are shown below:

	First Quarter							
			es (a) illions)				esales (b) ousands)	
	2010	2009	2010 Over/(Ur 2009	ider)	2010	2009	2010 Over/(Uno 2009	der)
Ford North America	\$14.1	\$10.0	\$ 4.1	41%	547	350	197	56%
Ford South America	2.0	1.4	0.6	43	101	93	8	9
Ford Europe (c)	7.7	5.8	1.9	33	416	343	73	21
Ford Asia Pacific Africa (d)	1.6	1.2	0.4	35	189	131	58	44
Volvo	3.5	2.6	0.9	34	92	69	23	33
Total Automotive sector	\$28.9	\$21.0	\$ 7.9	38	1,345	986	359	36

(a) 2010 over/(under) 2009 sales percentages are computed using unrounded sales numbers.

(c) Included in wholesale unit volumes are Ford brand vehicles sold in Turkey by our unconsolidated affiliate, Ford Otosan, totaling about 10,000 and 6,000 units in the first quarters of 2010 and 2009, respectively. "Sales" above do not include revenue from those units.

⁽b) Wholesale unit volumes include all Ford brand units (whether produced by Ford or by an unconsolidated affiliate), units manufactured by Ford that are sold to other manufacturers, units distributed for other manufacturers, and local-brand vehicles produced by our Chinese unconsolidated affiliate Jiangling Motors Corporation ("JMC"). Revenue from certain vehicles in wholesale unit volumes (specifically, Ford brand vehicles produced and distributed by our unconsolidated affiliates, as well as JMC brand vehicles) are not included in our revenue. Vehicles sold to daily rental car companies that are subject to a guaranteed repurchase option, as well as other sales of finished vehicles for which the recognition of revenue is deferred (e.g., consignments), also are included in wholesale unit volumes.

⁽d) Included in wholesale unit volumes of Ford Asia Pacific Africa are Ford brand and JMC brand vehicles sold in China by our unconsolidated affiliates totaling about 118,000 and 68,000 units in the first quarters of 2010 and 2009, respectively. "Sales" above does not include revenue from these units.

Details of Automotive sector market share for selected markets for the first quarter of 2010 and 2009, along with the level of dealer stocks as of March 31, 2010 and 2009, are shown below:

				D	ealer-Owned	
		Market Share			(in thousa	nds)
Market	2010	2009	2010 Over/(Under) 2009	2010	2009	2010 Over/(Under) 2009
United States (b)	16.5%	13.9%	2.6 pts.	405	410	(5)
South America (b) (c)	10.7	10.9	(0.2)	28	30	(2)
Europe (b) (d)	9.4	9.4	•	187	225	(38)
Asia Pacific Africa (e) (f)	2.0	2.1	(0.1)	59	54	5
Volvo – United States/Europe (d)	0.5/1.4	0.6/1.3	(0.1)/0.1	11/29	15/33	(4)/(4)

- (a) Dealer-owned stocks represent our estimate of vehicles shipped to our customers (dealers) and not yet sold by the dealers to their retail customers.
- (b) Includes only Ford and, in certain markets (primarily United States), Lincoln and Mercury brands.
- (c) South America market share is based, in part, on estimated vehicle registrations for our six major markets (Argentina, Brazil, Chile, Colombia, Ecuador and Venezuela).
- (d) Europe market share is based, in part, on estimated vehicle registrations for the 19 European markets we track (described in "Item 1. Business" of our 2009 Form 10-K Report).
- (e) Asia Pacific Africa market share is based, in part, on estimated vehicle sales for our 12 major markets (Australia, China, Japan, India, Indonesia, Malaysia, New Zealand, Philippines, South Africa, Taiwan, Thailand and Vietnam); market share includes Ford brand and JMC brand vehicles sold in China by our unconsolidated affiliates.
- (f) Dealer-owned stocks for Asia Pacific Africa include primarily Ford brand vehicles and JMC brand vehicles sold in China by our unconsolidated affiliates, as well as a small number of units distributed for other manufacturers.

Overall Automotive Sector

The improvement in results is more than explained by favorable volume and mix (\$1.9 billion), favorable net pricing (\$1.1 billion), and the non-recurrence of the Volvo held-for-sale impairment (\$650 million), offset partially by the non-recurrence of a prior year gain on debt reduction actions (\$1.3 billion).

The increase in revenues is more than explained by favorable volume, favorable changes in currency exchange, and favorable net pricing.

Ford North America Segment. The improvement in results is more than explained by favorable volume and mix, favorable net pricing, lower retiree health care and related charges and lower personnel reduction actions, offset partially by lower adjustments to reserves for JSB/TAP and unfavorable cost changes. The unfavorable cost changes are more than explained by the non-recurrence of a favorable prior year warranty reserve adjustment and higher net product costs, offset partially by lower structural costs (primarily manufacturing and engineering costs).

Ford South America Segment. The increase in earnings is more than explained by favorable changes in currency exchange rates and favorable net pricing offset partially by unfavorable cost changes. The unfavorable cost changes reflect higher net product costs, and higher structural costs (including higher manufacturing and engineering costs and higher overhead costs).

Ford Europe Segment. The improvement in results primarily reflects favorable volume and mix (more than explained by the non-recurrence of prior-year stock reductions and higher industry volumes), favorable cost changes and higher parts profits. The favorable cost changes reflect lower product costs and lower warranty costs.

Ford Asia Pacific Africa Segment. The improvement in results is more than explained by higher profits of unconsolidated China affiliates, favorable net pricing, and favorable volume and mix (primarily as a result of high industry volumes).

Other Automotive. The decline in results is more than explained by the non-recurrence of a prior year gain on debt reduction actions.

Volvo Segment. The improvement in results primarily reflects the non-recurrence of the held-for-sale impairment, favorable volume and mix, held-for-sale cessation of depreciation, and favorable net pricing, offset partially by unfavorable cost changes primarily to support increased production.

FINANCIAL SERVICES SECTOR

Results of Operations

Details of the Financial Services sector Revenues and Income/(Loss) before income taxes for the first quarter of 2010 and 2009 are shown below:

				First Qua	rter			
			venues pillions)	Income		Before In millions		Taxes
	2010	2009	2010 Over/(Under) 2009	2010	2	2009	Over	2010 /(Under) 2009
Ford Credit Other Financial Services	\$ 2.6 0.1	\$ 3.3 0.1	\$ (0.7) 0.0	\$ 828 (13)	\$	(36) (116)	\$	864 103
Total	\$ 2.7	\$ 3.4	\$ (0.7)		\$	(152)	\$	967

Ford Credit

The improvement in pre-tax results was more than explained by lower residual losses on returned vehicles and lower depreciation expense for leased vehicles due to higher auction values (about \$440 million); a lower provision for credit losses (about \$440 million); non-recurrence of net losses related to market valuation adjustments to derivatives, shown as unallocated risk management in the table below (\$54 million); higher financing margin primarily attributable to lower borrowing costs (about \$50 million); and lower operating costs (about \$30 million). These factors were offset partially by lower volume, primarily related to lower average receivables (about \$130 million).

Results of Ford Credit's operations and unallocated risk management for the first quarter of 2010 and 2009 are shown below:

	First Quarter					
Income/(Loss) before income taxes	2010	2009	2010 Over/(Under) 2009			
North America operations	\$702	\$ (45)	\$ 747			
International operations	96	33	63			
Unallocated risk management*	30	(24)	54			
Income/(Loss) before income taxes	828	(36)	864			
Provision for/(Benefit from) income taxes	_300	_(23)	323			
Net income/(loss)	\$528	\$(13)	\$ 541			

^{*} Consists of gains and losses related to market valuation adjustments to derivatives primarily related to movements in interest rates.

The improvement in pre-tax results for Ford Credit's North America operations primarily reflects lower residual losses on returned vehicles and lower depreciation expense for leased vehicles due to higher auction values, a lower provision for credit losses, higher financing margin, and lower operating costs. These factors were offset partially by lower volume. The increase in pre-tax earnings for Ford Credit's International operations primarily reflects a lower provision for credit losses, and lower losses on residual based products, offset partially by lower volume. The change in unallocated risk management reflects the non-recurrence of net losses related to market valuation adjustments to derivatives, primarily related to movements in interest rates.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations (Continued)

Ford Credit's net finance receivables and net investment in operating leases are shown below (in billions):

	March 31, 		ber 31, 009	O	2010 ver/(Under) 2009
Receivables - On-Balance Sheet			_		_
Finance receivables					
Retail installment	\$ 53.8	\$	56.3	\$	(2.5)
Wholesale	21.5		22.4		(0.9)
Other	2.5		2.4		0.1
Unearned interest supplements	(2.0)		(1.9)		(0.1)
Allowance for credit losses	 (1.2)		(1.3)		0.1
Finance receivables, net	74.6		77.9		(3.3)
Net investment in operating leases	 13.3		14.6		(1.3)
Total receivables – on-balance sheet (a)(b)	\$ 87.9	\$	92.5	\$	(4.6)
Memo:					
Total receivables – managed (c)	\$ 89.9	\$	94.5	\$	(4.6)
Total receivables – serviced (d)	90.0		94.6		(4.6)

2010

- (a) At March 31, 2010 and December 31, 2009, includes finance receivables before allowance for credit losses of \$59.8 billion and \$64.4 billion, respectively, that have been sold for legal purposes in securitization transactions, but continue to be included in Ford Credit's financial statements. In addition, at March 31, 2010 and December 31, 2009, includes net investment in operating leases before allowance for credit losses of \$10.9 billion and \$10.4 billion, respectively, that have been included in securitization transactions, but continue to be included in Ford Credit's financial statements. These underlying securitized assets are available only for payment of the debt and other obligations issued or arising in the securitization transactions; they are not available to pay Ford Credit's other obligations or the claims of its other creditors. Ford Credit holds the right to the excess cash flows not needed to pay the debt and other obligations issued or arising in each of these securitization transactions. See Note 9 of the Notes to the Financial Statements for more information regarding securitization transactions.
- (b) Includes allowance for credit losses of \$1.4 billion and \$1.5 billion at March 31, 2010 and December 31, 2009, respectively.
- (c) Includes on-balance sheet receivables, excluding unearned interest supplements related to finance receivables of about \$2 billion and \$1.9 billion at March 31, 2010 and December 31, 2009, respectively; and includes off-balance sheet retail receivables of about \$100 million at December 31, 2009.
- (d) Includes managed receivables and receivables sold in whole-loan sale transactions where Ford Credit retains no interest, but which it continues to service of about \$100 million at March 31, 2010 and December 31, 2009.

The decrease in receivables from year-end 2009 primarily reflects the transition of Jaguar, Land Rover, Mazda, and Volvo financing to other finance providers and lower industry and financing volumes in 2009 and 2010 compared with prior years. At March 31, 2010, the Jaguar, Land Rover, and Mazda financing portfolio represented about 7% of Ford Credit's managed receivables and the Volvo financing portfolio represented about 4% of Ford Credit's managed receivables. These percentages will decline over time.

The following table shows Ford Credit's worldwide charge-offs (credit losses net of recoveries) for the various categories of financing during the periods indicated. The loss-to-receivables ratios, which equal charge-offs on an annualized basis divided by the average amount of receivables outstanding for the period, excluding the allowance for credit losses and unearned interest supplements related to finance receivables, are shown below.

	First Quarter						
		010	2	009	0	2010 over/(Und 2009	er)
On-Balance Sheet							
Charge-offs (in millions)	\$	133	\$	332	\$	(199)	
Loss-to-receivables ratio		0.58%		1.21%		(0.63)	pts.
Memo:							
Charge-offs – managed (in millions)	\$	133	\$	335	\$	(202)	
Loss-to-receivables ratio – managed		0.58%		1.22%		(0.64)	pts.

The decrease in charge-offs from a year ago primarily reflects lower losses in the United States and Europe. Charge-offs in the United States decreased due to lower repossessions, lower severity, and lower wholesale and dealer loan net losses.

Shown below is Ford Credit's allowance for credit losses and its allowance for credit losses as a percentage of end-of-period receivables (finance receivables, excluding unearned interest supplements, and net investment in operating leases, excluding the allowance for credit losses) for its on-balance sheet portfolio:

	arch 31, 2010	De	cember 31, 2009	_	2010 Over/(Unde 2009	er)
Allowance for credit losses (in millions)	\$ 1,356	\$	1,549	\$	(193)	
Allowance as a percentage of end-of-period receivables	1.49%		1.61%		(0.12)	pts.

The allowance for credit losses is estimated using a combination of models and management judgment, and is based on such factors as portfolio quality, historical loss performance, and receivable levels. The decrease in the allowance for credit losses is consistent with the decrease in charge-offs and included about \$95 million for management's judgment regarding higher retail installment and lease repossession assumptions compared with historical trends used in Ford Credit's models. At December 31, 2009, Ford Credit's allowance for credit losses included about \$215 million for management's judgment regarding higher retail installment and lease repossession assumptions and higher wholesale and dealer loan default assumptions.

In purchasing retail finance and lease contracts, Ford Credit uses a proprietary scoring system that classifies contracts using several factors, such as credit bureau information, credit bureau scores (e.g., FICO score), customer characteristics, and contract characteristics. In addition to Ford Credit's proprietary scoring system, it considers other factors, such as employment history, financial stability, and capacity to pay. As of March 31, 2010, about 5% of the outstanding U.S. retail finance and lease contracts in Ford Credit's serviced portfolio were classified as high risk at contract inception, about the same as year-end 2009.

Residual Risk

Ford Credit is exposed to residual risk on operating leases and similar balloon payment products where the customer may return the financed vehicle to Ford Credit. Residual risk is the possibility that the amount Ford Credit obtains from returned vehicles will be less than its estimate of the expected residual value for the vehicle. Ford Credit estimates the expected residual value by evaluating recent auction values, return volumes for its leased vehicles, industry-wide used vehicle prices, marketing incentive plans, and vehicle quality data.

North America Retail Operating Lease Experience. Ford Credit uses various statistics to monitor its residual risk:

- Placement volume measures the number of leases Ford Credit purchases in a given period;
- Termination volume measures the number of vehicles for which the lease has ended in the given period; and
- Return volume reflects the number of vehicles returned to Ford Credit by customers at lease-end.

The following table shows operating lease placement, termination, and return volumes for Ford Credit's North America operations, which accounted for 98% of its total investment in operating leases at March 31, 2010 (in thousands, except for percentages):

	First Q	<u>uarter</u>
	2010	2009
Placements	29	20
Terminations	97	84
Returns	72	75
Memo:		
Return rates	74%	89%

The increase in placement volumes primarily reflects higher industry volumes, higher market share, and changes in Ford's marketing programs. The increase in termination volumes primarily reflects higher placement volumes in 2007 and the first half of 2008. The decrease in return volumes primarily reflects lower return rates, consistent with improved auction values relative to Ford Credit's expectations of lease-end values at the time of contract purchase.

U.S. Ford, Lincoln, and Mercury Brand Retail Operating Lease Experience

The following table shows return volumes for Ford Credit's Ford, Lincoln, and Mercury brand U.S. operating lease portfolio. Also included are auction values at constant first quarter 2010 vehicle mix for lease terms comprising 57% of Ford Credit's active Ford, Lincoln, and Mercury brand U.S. operating lease portfolio (in thousands, except for percentages):

	First Quarter		
	2010	2009	
Returns			
24-Month term	16	16	
36-Month term	19	22	
39-Month term	9	6	
Total returns	44	44	
Memo:			
Return rates	71%	89%	
Auction Values at Constant First Quarter 2010 Vehicle Mix			
24-Month term	\$18,940	\$17,020	
36-Month term	15,285	13,115	

Ford, Lincoln, and Mercury brand U.S. return volumes in first quarter 2010 were equal to the same period a year ago, primarily reflecting higher terminations, offset by a lower return rate (down 18 percentage points) consistent with improved auction values relative to Ford Credit's expectations of lease-end values at the time of contract purchase. The increase in auction values at constant first quarter 2010 mix primarily reflected the overall auction value improvement in the used vehicle market. Auction values, at constant first quarter 2010 mix, improved compared with fourth quarter 2009 for vehicles under 24-month and 36-month leases by \$380 per unit and \$490 per unit, respectively. The year-over-year improvement in auction values experienced in the first quarter of 2010 is not expected to continue to the same extent through the end of the year.

LIQUIDITY AND CAPITAL RESOURCES

Automotive Sector

Our industry has been heavily impacted by the global economic crisis, which included a sudden and substantial decline in global industry sales volume. The dramatic decline in industry sales volume, combined with tight credit markets, other economic factors, and the costs associated with transforming our business, put significant pressure on our Automotive liquidity. While the economic environment is improving, recoveries in key markets are modest due to weak labor markets and tight credit. We believe that our continued focus on delivering on our plan is the right strategy to achieve our objectives. Our Automotive liquidity strategy includes ensuring that we have sufficient funding available with a high degree of certainty throughout the business cycle; our long-term goal is to improve our core Automotive operations so that we have a high degree of certainty about our capability to generate cash from our operations.

Gross Cash. Automotive gross cash includes cash and cash equivalents and net marketable securities. In 2008, we reclassified out of our Automotive gross cash calculation the Temporary Asset Account ("TAA") securities. Gross cash is detailed below as of the dates shown (in billions):

	M	arch 31, 2010		December 31, 2009	March 31, 2009		December 31, 2008
Cash and cash equivalents	\$	12.8	\$	9.7	\$ 7.7	\$	6.1
Marketable securities (a)		12.5		15.2	13.5		9.3
Total cash and marketable securities		25.3		24.9	21.2		15.4
UAW-Ford TAA/Other (b)		•	_	•	(0.3)	_	(2.3)
Gross cash	\$	25.3	\$	24.9	\$ 20.9	\$	13.1

⁽a) Included at March 31, 2010 and December 31, 2009 are Ford Credit debt securities that we purchased, which are reflected in the table at a carrying value of \$432 million and \$646 million, respectively; the estimated fair value is \$439 million and \$656 million, respectively. Also included are Mazda marketable securities with a fair value of \$551 million and \$447 million at March 31, 2010 and December 31, 2009, respectively.

⁽b) Amount transferred to UAW-Ford TAA that, due to consolidation, was shown in Cash and cash equivalents and Marketable securities.

In managing our business, we classify changes in Automotive gross cash into two categories: operating-related and other (which includes the impact of certain special items, contributions to funded pension plans, the net effect of the change in the TAA and VEBA on gross cash, certain tax-related transactions, acquisitions and divestitures, capital transactions with the Financial Services sector, dividends paid to shareholders, and other – primarily financing-related). Our key liquidity metrics are operating-related cash flow, which best represents the ability of our Automotive operations to generate cash, and Automotive gross cash. We believe the cash flow analysis reflected in the table below is useful to investors because it includes in operating-related cash flow elements that we consider to be related to our Automotive operating activities (e.g., capital spending) and excludes cash flow elements that we do not consider to be related to the ability of our operations to generate cash. This differs from a cash flow statement presented in accordance with U.S. GAAP and differs from *Cash flows from operating activities of continuing operations*, the most directly comparable U.S. GAAP financial measure.

Changes in Automotive gross cash are summarized below (in billions):

	Fi	irst Qu	arter
	20:	10 (a)	2009
Gross cash at end of period	\$	25.3	\$ 20.9
Gross cash at beginning of period		24.9	13.1
Total change in gross cash	\$	0.4	\$ 7.8
Operating-related cash flows			
Automotive income/(loss) before income taxes (excluding special items)	\$	1.2	\$ (2.0)
Capital expenditures		(0.9)	(1.1)
Depreciation and special tools amortization		0.9	1.0
Changes in receivables, inventories and trade payables		(0.4)	0.8
Other (b)		(0.6)	(1.7)
Subtotal		0.2	(3.0)
Up-front subvention payments to Ford Credit		(0.3)	(0.5)
Total operating-related cash flows		(0.1)	(3.5)
Other changes in gross cash			
Cash impact of personnel-reduction programs and Job Security Benefits accrual		(0.1)	(0.3)
Contributions to funded pension plans		(0.3)	(0.4)
Net effect of TAA/VEBA on gross cash (c)		•	2.0
Net receipts from Financial Services sector		0.5	0.3
Net proceeds from/(payments on) Automotive sector debt		0.5	9.9
Equity issuances, net (primarily related to the equity distribution agreement described below)	1	0.5	•
Other (d)		(0.6)	(0.2)
Total change in gross cash	\$	0.4	\$ 7.8

- (a) Except as noted, see note (d) below, 2010 data exclude Volvo.
- (b) First quarter 2010 Other Operating-related cash flows were primarily driven by timing differences of payables and receivables between the Automotive and Financial Services sectors in the normal course of business.
 (c) As previously disclosed in "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" in our
- (c) As previously disclosed in "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" in our 2008 Form 10-K Report, in January 2009 we liquidated the assets in the TAA established pursuant to the Retiree Health Care Settlement Agreement, and replaced the assets with a promissory note owing by Ford to Ford-UAW Holdings LLC, allowing us access to the TAA assets as another available source of liquidity for use in our operations during 2009.
- (d) Primarily relates to net cash flows from Volvo, investments to support unconsolidated subsidiaries and the impact of exchange on our non-US dollar cash balances.

Included within operating-related cash flows are cash flows related to changes in our working capital balances (i.e., trade receivables, trade payables, and inventories), and receivables and payables between the Automotive and Financial Services sectors associated primarily with the Automotive sector's vehicle wholesales. These cash flows generally are subject to seasonal timing differences. For example, we typically experience cash flow timing differences associated with inventories due to our annual December shutdown period, when inventories usually are at the lowest level of the year. This drawdown of inventories creates significant cash inflows during the fourth quarter. As production resumes in January, we replenish our inventory stocks and generally experience resulting cash outflows, reflected in "Changes in receivables, inventories and trade payables" in the table above. Additionally, as a result of our December shutdown, Automotive receivables from the Financial Services sector, reflected in "Other" in Operating-related cash flows above, largely are collected by the end of the fourth quarter with a normal increase in the receivables balance in January as production resumes.

Shown in the table below is a reconciliation between financial statement Cash flows from operating activities of continuing operations and operating-related cash flows (calculated as shown in the table above) for the first quarter of 2010 and 2009 (in billions):

	Fi	rst Qu	arter
	201	0 (a)	2009
Cash flows from operating activities of continuing operations	\$	0.6	\$ (2.4)
Items included in operating-related cash flows			
Capital expenditures		(0.9)	(1.1)
Net transactions between Automotive and Financial Services sectors (b)		(0.7)	(0.6)
Net cash flows from non-designated derivatives		•	0.2
Items not included in operating-related cash flows			
Cash impact of JSB/TAP accrual and personnel-reduction programs		0.1	0.3
Contributions to funded pension plans		0.3	0.4
Tax refunds, tax payments, and tax receipts from affiliates		•	(0.3)
Other (c)		0.5	
Operating-related cash flows	\$	(0.1)	\$ (3.5)

(a) Except as noted, see note (c) below, 2010 data exclude Volvo.

(b) Primarily payables and receivables between the Automotive and Financial Services sectors in the normal course of business.

(c) 2010 includes Volvo cash flows.

Equity and Equity-Linked Issuances. On December 4, 2009, we entered into an equity distribution agreement with certain broker-dealers pursuant to which we would offer and sell shares of Ford Common Stock from time to time for an aggregate offering price of up to \$1 billion. Sales of Ford Common Stock under this agreement are expected to be made over a several-month period by means of ordinary brokers' transactions on the New York Stock Exchange at market prices or as otherwise agreed. During the first quarter of 2010, we issued 32.1 million shares of Common Stock for an aggregate price of \$373 million resulting in net proceeds of \$370 million, which will be used for general corporate purposes. Since inception through March 31, 2010, under this agreement we issued 41.9 million shares of Common Stock for an aggregate price of \$470 million, resulting in net proceeds of \$466 million. There were no issuances under this program in April 2010.

Secured Credit Agreement. At March 31, 2010, the revolving credit facility of our secured Credit Agreement totaled \$8.1 billion, of which (i) \$8 billion was utilized (including \$438 million to support letters of credit), (ii) \$7.2 billion matures on November 30, 2013 and (iii) \$886 million matures on December 15, 2011. Also at March 31, 2010, term loans outstanding under the Credit Agreement totaled \$5.2 billion.

On April 6, 2010, Ford paid down \$3 billion of the drawn amount of the revolving credit facility. This payment will reduce Automotive gross cash and debt reflected on our second quarter 2010 balance sheet by \$3 billion. The action does not affect Automotive liquidity, as the repaid amounts remain available for borrowing through November 2013.

Other Automotive Credit Facilities. At March 31, 2010, we had \$613 million of other contractually-committed Automotive credit facilities with financial institutions, including \$25 million of worldwide Automotive unsecured credit facilities and \$588 million of local credit facilities to foreign Automotive affiliates. Of the \$613 million of contractually-committed credit facilities, \$143 million has been utilized. Of the \$470 million available for use, \$47 million expire in 2010, \$59 million expire in 2013, and \$364 million expire in 2014.

On April 9, 2010, we terminated the \$25 million of worldwide Automotive unsecured credit facilities.

Net Cash/(Debt). Our Automotive sector net debt calculation is detailed below (in billions):

	March 31, 2010		December 3 2009	31,
Gross cash	\$	25.3	\$	24.9
Less:				
Long-term debt		29.3		32.0
Debt payable within one year		5.0		1.6
Total debt		34.3		33.6
Net cash/(debt)	\$	(9.0)	\$	(8.7)

As discussed above, included within our debt payable within one year is our April 6, 2010 \$3 billion repayment of our revolving credit facility that matures in 2013.

See Note 9 of the Notes to the Financial Statements for our debt maturity table as of March 31, 2010 and additional debt disclosures.

Liquidity Sufficiency. One of the four key priorities of our business plan is to finance our plan and improve our balance sheet. The actions described above are consistent with this priority. Based on our planning assumptions, we believe that we have sufficient liquidity and capital resources to continue to transform our business, invest in new products that customers want and value, pay our debts and obligations as and when they come due, and provide a cushion within the uncertain global economic environment. We will continue to look for opportunities to improve our balance sheet, primarily by working to improve our underlying business to generate positive Automotive operating-related cash flow.

Financial Services Sector

Ford Credit

Debt. At March 31, 2010, Ford Credit's debt was \$94.2 billion compared with \$96.3 billion at year-end 2009. At March 31, 2010, unsecured long-term debt (including notes payable within one year) was down about \$2 billion from year-end 2009, primarily reflecting about \$3 billion of debt maturities, repurchases, and calls, offset partially by about \$1 billion of new unsecured long-term debt issuance. Remaining unsecured long-term debt maturities were as follows: 2010 • \$4 billion; 2011 • \$11 billion; 2012 • \$7 billion; and the remainder thereafter. At March 31, 2010, asset-backed long-term debt (including notes payable within one year) was up about \$2 billion from year-end 2009, reflecting asset-backed long-term debt issuance in excess of amortization.

Funding Strategy. Ford Credit's funding strategy is to maintain sufficient liquidity to meet short-term funding obligations by having a substantial cash balance and committed funding capacity. Despite recent upgrades to its credit ratings, Ford Credit remains rated below investment grade. As a result, Ford Credit continues to rely heavily on securitization funding as this market continues to be more cost effective than unsecured funding and allows Ford Credit access to a different investor base. Ford Credit plans to meet a significant portion of its 2010 funding requirements through securitization transactions, and access the unsecured debt market on occasion. In addition, Ford Credit has various alternative business arrangements for select products and markets that reduce its funding requirements while allowing Ford Credit to support us (e.g., Ford Credit's partnering in Brazil for retail financing and FCE Bank ple's ("FCE") partnering with various institutions in Europe for full service leasing and retail and wholesale financing). Ford Credit is continuing to explore and execute such alternative business arrangements.

Much of the negative impact Ford Credit experienced from the volatility and disruptions in the asset-backed securitization market since August 2007 has subsided. Ford Credit's access to the public and private funding markets, credit spreads on most transactions, and derivative capacity have improved. Additionally, Ford Credit has been able to complete unsecured debt issuances and securitization transactions with longer maturities. During 2010, Ford Credit has completed about \$11 billion of funding, including: two unsecured debt issuances, one of which was for a term of ten years; about \$800 million of wholesale and retail subordinated notes in the United States, Canada and Germany; two wholesale securitization transactions eligible under the U.S. Federal Reserve's Term Asset-Backed Securities Loan Facility ("TALF") with terms of three and five years, respectively; a private non-TALF wholesale securitization transaction that also had a term of five years; and a non-TALF lease securitization transaction.

Compared to year-end 2009, Ford Credit reduced its committed capacity by \$1.5 billion as a result of its lower receivables balance. First quarter 2010 renewals (about 82% of the \$2 billion that came due) and added capacity were more than offset by voluntary terminations of certain excess capacity. About 66% of Ford Credit's \$32.3 billion of committed capacity as of March 31, 2010 is up for renewal during the remainder of 2010 and 14% is up for renewal in the second quarter of 2010. Ford Credit's renewal strategy is to optimize capacity and maintain sufficient liquidity to protect its global funding needs. Most of Ford Credit's asset-backed committed facilities enable Ford Credit to obtain term funding up to the time that the facilities expire. Any outstanding debt at the maturity of the facilities remains outstanding and is repaid as underlying assets liquidate. Ford Credit's ability to obtain funding under its committed asset-backed liquidity programs is subject to having a sufficient amount of assets eligible for these programs, and for certain programs, having the ability to obtain derivatives to manage the interest rate risk.

Ford Credit's funding plan is subject to risks and uncertainties, many of which are beyond its control, including disruption in the capital markets for the types of asset-backed securities used in Ford Credit's asset-backed funding. Potential impacts of industry events on Ford Credit's ability to access debt and derivatives markets, or renew its committed liquidity programs in sufficient amounts and at competitive rates, represents another risk to Ford Credit's funding plan. As a result, Ford Credit may need to further reduce the amount of finance receivables and operating leases it purchases or originates, thereby reducing its ongoing profits and adversely affecting its ability to support the sale of Ford vehicles.

Government-Sponsored Securitization Funding Programs. Government-sponsored securitization funding programs helped stabilize the asset-backed securitization market. In the first quarter of 2010, most of Ford Credit's funding was completed in public and private markets without utilizing government-sponsored securitization funding programs.

Commercial Paper Funding Facility ("CPFF"): The CPFF became operational in October 2008 and purchased unsecured and asset-backed commercial paper from U.S. issuers. In 2008, Ford Credit registered to sell up to \$16 billion from its asset-backed commercial paper program ("FCAR") to the CPFF. FCAR issued a total of \$9 billion of commercial paper to the CPFF in 2009, all of which had matured by September 30, 2009. The CPFF ceased purchasing commercial paper on February 1, 2010.

Term Asset-Backed Securities Loan Facility: TALF began in March 2009 to make credit available by restoring liquidity in the asset-backed securitization market. Under TALF, the Federal Reserve Bank of New York ("FRBNY") made loans to holders of TALF-eligible asset-backed securities. The loans were equal to the market value of the asset-backed securities less a discount. Interest rates on most TALF loans were 100 basis points over the respective term benchmark rate, and discounts varied according to the assets supporting the asset-backed securities. The TALF program revived the asset-backed securitization market by attracting new investors who purchased asset-backed securities, receiving higher spreads on these securities than the spreads they pay on their loans from FRBNY. As investor demand increased due to the liquidity provided by TALF, spreads generally narrowed on Ford Credit's issuances and the percentage of non-TALF investors increased. As the spread on certain asset-backed securities fell below the 100 basis point spread on TALF loans, Ford Credit's TALF-eligible asset-backed securities were purchased almost exclusively by non-TALF investors.

The balance of Ford Credit's TALF-eligible asset-backed securities outstanding at year-end 2009 was \$8.1 billion. In the first quarter of 2010, Ford Credit issued \$2.3 billion of TALF-eligible wholesale asset-backed securities and amortization totaled about \$700 million. The balance of TALF-eligible asset-backed securities outstanding at March 31, 2010 was \$9.7 billion. TALF expired in March 2010. The following table summarizes Ford Credit's \$12.6 billion of TALF-eligible issuances since program inception, including the weighted average spread of the triple-A rated notes over the relevant benchmark rates for securitization transactions:

Date	Issuer	Issuance Size (in billions)	Weighted Average Spread (basis points)
Retail Installment			
March 2009	Ford Credit Auto Owner Trust 2009 – A	\$ 3.0	295
June 2009	Ford Credit Auto Owner Trust 2009 – B	1.9	9 161
July 2009	Ford Credit Auto Owner Trust 2009 – C	1.0	165
September 2009	Ford Credit Auto Owner Trust 2009 – D	2.	1 83
Wholesale			
October 2009	Ford Credit Master Owner Trust 2009 – 2	1.:	5 155
January 2010	Ford Credit Master Owner Trust 2010 – 1	1.3	3 165
March 2010	Ford Credit Master Owner Trust 2010 – 3	1.0	170
Retail Lease			
June 2009	Ford Credit Auto Lease Trust 2009 – A	0.3	3 211

European Central Bank ("ECB") Open Market Operations: FCE is eligible to access liquidity through the ECB's open market operations program. This program allows eligible counterparties to use eligible assets (including asset-backed securities) as collateral for short-term liquidity. The present term limitation is three months; however, in the past the term has been as long as one year. The funding efficiency of liquidity provided under this program is typically lower than if the asset-backed securities were placed in the public or private markets, because the ECB applies its own market valuation to the collateral and a discount to the original face value of the asset-backed securities. The market valuation and discount vary by the term of the asset, asset type, and jurisdiction of the asset. During 2009, FCE used the ECB's open market operations program to provide additional liquidity at a time when access to the asset-backed securitization market was limited and costs for such funding were significantly higher than in the past. FCE had about \$800 million and \$1.8 billion of funding from the ECB relating to asset-backed securities and other marketable securities at March 31, 2010 and December 31, 2009, respectively. FCE has continued to sell notes previously posted as collateral for ECB funding in the

secondary markets. These secondary sales have improved FCE's funding efficiency, demonstrated its market access, and reduced its reliance on government-sponsored programs. Ford Credit expects FCE's utilization of the ECB open market operations program to decline.

Funding Plan. The following table illustrates Ford Credit's public and private term funding issuances in 2009 and through April 30, 2010, and its planned issuances for 2010 (in billions):

	2010				
	Full Year Forecast			Through April 30	Actual 2009
Public Term Funding					
Unsecured	\$	3 – 6	\$	2	\$ 5
Securitization transactions (a)		9 - 14		6	15
Total public term funding	\$	12 – 17	\$	8	\$ 20
Private Term Funding (b)	\$	7 – 11	\$	3	\$ 11

- (a) Includes public securitization transactions and securitization transactions issued under Rule 144A of the Securities Act of 1933.
- (b) Includes private term debt, securitization transactions, and other term funding; excludes sales to Ford Credit's FCAR program.

Through April 30, 2010, Ford Credit completed about \$8 billion of public term funding transactions, including about \$2 billion of retail asset-backed securitization transactions in the United States, Canada, and Europe, about \$2 billion of wholesale asset-backed securitization transactions in the United States, about \$2 billion of lease asset-backed securitization transactions in the United States.

Through April 30, 2010, Ford Credit completed about \$3 billion of private term funding transactions (excluding its on-balance sheet asset-backed commercial paper program), primarily reflecting retail, lease, and wholesale asset-backed securitization transactions in the United States and Canada.

Liquidity. The following table illustrates the various sources of Ford Credit's liquidity as of the dates shown (in billions):

	March 31, 2010			ember 31, 2009
Cash, cash equivalents, and marketable securities (a)	\$	20.7	\$	17.3
Committed liquidity programs		22.0 (b	2)	23.2
Asset-backed commercial paper		9.1	")	9.3
Credit facilities		1.2		1.3
Committed capacity		32.3		33.8
Committed capacity and cash		53.0		51.1
Less: Capacity in excess of eligible receivables		(9.0)		(6.5)
Less: Cash, cash equivalents, and marketable securities to support on-balance sheet securitization transactions		(5.6)		(5.2)
Liquidity		38.4		39.4
Less: Utilization		(17.3)		(18.3)
Liquidity available for use	\$	21.1	\$	21.1

- (a) Excludes marketable securities related to insurance activities.
- (b) Capacity as of April 1, 2010.

At March 31, 2010, committed capacity and cash shown above totaled \$53 billion, of which \$38.4 billion could be utilized (after adjusting for capacity in excess of eligible receivables of \$9 billion and cash required to support on-balance sheet securitization transactions of \$5.6 billion). At March 31, 2010, \$17.3 billion was utilized, leaving \$21.1 billion available for use (including \$15.1 billion of cash, cash equivalents, and marketable securities, but excluding marketable securities related to insurance activities and cash, cash equivalents, and marketable securities to support on-balance sheet securitization transactions).

At March 31, 2010, Ford Credit's liquidity available for use was about the same as year-end 2009. Liquidity available for use was 23% of managed receivables, compared with 22% at year-end 2009. In addition to the \$21.1 billion of liquidity available for use, \$9 billion of capacity in excess of eligible receivables provides Ford Credit with an alternative for funding future purchases or originations and gives Ford Credit flexibility to shift capacity to alternate markets and asset classes.

Cash, Cash Equivalents, and Marketable Securities. At March 31, 2010, Ford Credit's cash, cash equivalents, and marketable securities (excluding marketable securities related to insurance activities) totaled \$20.7 billion, compared with \$17.3 billion at year-end 2009. In the normal course of Ford Credit's funding activities, Ford Credit may generate more proceeds than are required for its immediate funding needs. These excess amounts are maintained primarily as highly liquid investments, which provide liquidity for Ford Credit's short-term funding needs and give it flexibility in the use of its other funding programs. Cash, cash equivalents, and marketable securities include amounts to be used only to support Ford Credit's on-balance sheet securitization transactions of \$5.6 billion at March 31, 2010 and \$5.2 billion at December 31, 2009.

Committed Liquidity Programs. Ford Credit and its subsidiaries, including FCE, have entered into agreements with a number of bank-sponsored asset-backed commercial paper conduits ("conduits") and other financial institutions whereby such parties are contractually committed, at Ford Credit's option, to purchase from Ford Credit eligible retail or wholesale assets or to purchase or make advances under asset-backed securities backed by retail, lease or wholesale assets for proceeds of up to \$22 billion at April 1, 2010 (\$9.9 billion retail, \$8.5 billion wholesale and \$3.6 billion supported by various retail, lease or wholesale assets) of which about \$7 billion are commitments to FCE. These committed liquidity programs have varying maturity dates, with \$18.2 billion having maturities within the next twelve months (of which \$6.2 billion relates to FCE commitments), and the remaining balance having maturities between September 2011 and March 2013. Ford Credit plans to achieve capacity renewals consistent with its lower receivables balance and to optimize capacity and maintain sufficient liquidity to protect its global funding needs. Ford Credit's ability to obtain funding under these programs is subject to having a sufficient amount of assets eligible for these programs as well as its ability to obtain interest rate hedging arrangements for certain securitization transactions. Ford Credit's capacity in excess of eligible receivables would protect it against the risk of lower than planned renewal rates. At March 31, 2010, \$10.2 billion of these commitments were in use. These programs are free of material adverse change clauses, restrictive financial covenants (for example, debt-to-equity limitations and minimum net worth requirements), and credit rating triggers that could limit Ford Credit's ability to obtain funding. However, the unused portion of these commitments may be terminated if the performance of the underlying assets deteriorates beyond specified levels. Based on Ford Credit's experience and knowledge as se

Asset-Backed Commercial Paper. At March 31, 2010, Ford Credit had \$9.1 billion of contractually-committed liquidity facilities provided by banks to support its FCAR program. Of the \$9.1 billion of contractually-committed liquidity facilities, \$4.4 billion expire in 2010 and \$4.7 billion expire in 2012. Utilization of these facilities is subject to conditions specific to the FCAR program and Ford Credit's having a sufficient amount of eligible assets for securitization. The FCAR program must be supported by liquidity facilities equal to at least 100% of its outstanding balance. At March 31, 2010, \$9.1 billion of FCAR's bank liquidity facilities were available to support FCAR's asset-backed commercial paper, subordinated debt, or FCAR's purchase of Ford Credit's asset-backed securities. At March 31, 2010, the outstanding commercial paper balance for the FCAR program was \$6.5 billion.

Credit Facilities. At March 31, 2010, Ford Credit and its majority-owned subsidiaries, including FCE, had \$1.2 billion of contractually-committed unsecured credit facilities with financial institutions, of which \$625 million were available for use. Of the credit facilities available for use, \$308 million expire in 2010, \$260 million expire in 2011, and \$57 million expire in 2012. Of the \$1.2 billion of contractually-committed credit facilities, almost all are FCE worldwide credit facilities. The FCE worldwide credit facilities may be used, at FCE's option, and by any of FCE's direct or indirect, majority-owned subsidiaries. FCE will guarantee any such borrowings. All of the worldwide credit facilities are free of material adverse change clauses, restrictive financial covenants, and credit rating triggers that could limit Ford Credit's ability to obtain funding.

Leverage. Ford Credit uses leverage, or the debt-to-equity ratio, to make various business decisions, including evaluating and establishing pricing for retail, wholesale, and lease financing, and assessing its capital structure. Ford Credit refers to its shareholder's interest as equity. Ford Credit calculates leverage on a financial statement basis and on a managed basis.

The following table illustrates the calculation of Ford Credit's financial statement leverage (in billions, except for ratios):

	March 31, 2010		December 31, 2009	
Total debt	\$	94.2	\$	96.3
Equity		10.8		11.0
Financial statement leverage (to 1)		8.7		8.8

The following table illustrates the calculation of Ford Credit's managed leverage (in billions, except for ratios):

	arch 31, 2010	D	ecember 31, 2009
Total debt	\$ 94.2	\$	96.3
Securitized off-balance sheet receivables outstanding	0.0		0.1
Retained interest in securitized off-balance sheet receivables	0.0		0.0
Adjustments for cash, cash equivalents, and marketable securities (a)	(20.7)		(17.3)
Adjustments for derivative accounting (b)	 (0.2)		(0.2)
Total adjusted debt	\$ 73.3	\$	78.9
Equity	\$ 10.8	\$	11.0
Adjustments for derivative accounting (b)	 (0.1)		(0.2)
Total adjusted equity	\$ 10.7	\$	10.8
Managed leverage (to 1)	6.9		7.3

⁽a) Excludes marketable securities related to insurance activities.

Ford Credit plans its managed leverage by considering prevailing market conditions and the risk characteristics of its business. At March 31, 2010, Ford Credit's managed leverage was 6.9 to 1, compared with 7.3 to 1 at December 31, 2009. Ford Credit's managed leverage is significantly below the threshold of 11.5 to 1 set forth in the Amended and Restated Support Agreement with us. In the first quarter of 2010, Ford Credit paid a cash distribution of \$500 million to its parent.

On-Balance Sheet Arrangements

The majority of Ford Credit's securitization transactions are included in its financial statements. Ford Credit expects its future securitization transactions to be on-balance sheet. Ford Credit believes on-balance sheet arrangements are more transparent to its investors. Securitized assets are only available to repay the related asset-backed debt and to pay other securitization investors and other participants. These underlying securitized assets are available only for payment of the debt and other obligations issued or arising in the securitization transactions; they are not available to pay Ford Credit's other obligations or the claims of Ford Credit's other creditors. Ford Credit holds the right to the excess cash flows not needed to pay the debt and other obligations issued or arising in each of these securitization transactions. This debt is not Ford Credit's legal obligation or the legal obligation of its other subsidiaries. Assets and associated liabilities related to Ford Credit's on-balance sheet securitization transactions are as follows (in billions):

	2010	_ L	2009
Total outstanding principal amount of finance receivables and net investment in operating leases included in on-balance sheet			_
securitizations (a)	\$ 70.7	\$	74.8
Cash, cash equivalents, and marketable securities to be used only to support the on-balance sheet securitizations	5.6	(b)	5.2
Debt payable only out of collections on the underlying securitized assets and related enhancements	52.7		52.9

Manah 21

See Note 9 of the Notes to the Financial Statements for more information regarding the accounting treatment for securitization transactions. Ford Credit expects its future securitization transactions to be on-balance sheet.

Off-Balance Sheet Arrangements

Ford Credit has not entered into any off-balance sheet arrangements (off-balance sheet securitization transactions and whole-loan sale transactions, excluding sales of businesses and liquidating portfolios) since the first quarter of 2007, which is consistent with Ford Credit's plan to execute on-balance sheet securitization transactions.

Total Company

Equity/(Deficit). At March 31, 2010, Total equity/(deficit) attributable to Ford Motor Company was negative \$5.5 billion, an improvement of \$2.3 billion compared with December 31, 2009. The improvement is more than explained by favorable changes in Retained earnings (primarily related to first quarter 2010 net income attributable to Ford) and favorable changes in Capital in excess of par value of stock (primarily equity issuances).

⁽b) Primarily related to market valuation adjustments to derivatives due to movements in interest rates. Adjustments to debt are related to designated fair value hedges and adjustments to equity are related to retained earnings.

⁽a) Before allowance for credit losses of about \$800 million and \$1.1 billion at March 31, 2010 and December 31, 2009, respectively. Excludes about \$400 million and \$0 in finance receivables related to Volvo Auto Bank at March 31, 2010 and December 31, 2009, respectively.

⁽b) Included are marketable securities totaling \$256 million, which are pledged as collateral in a funding arrangement with the ECB.

Credit Ratings. Our short- and long-term debt is rated by four credit rating agencies designated NRSROs by the SEC:

- DBRS Limited ("DBRS");
- Fitch, Inc. ("Fitch");
- Moody's Investors Service, Inc. ("Moody's"); and
- Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc. ("S&P").

In several markets, locally recognized rating agencies also rate us. A credit rating reflects an assessment by the rating agency of the credit risk associated with a corporate entity or particular securities issued by that entity. Their ratings of us are based on information provided by us and other sources. Credit ratings are not recommendations to buy, sell or hold securities and are subject to revision or withdrawal at any time by the assigning rating agency. Each rating agency may have different criteria for evaluating company risk and, therefore, ratings should be evaluated independently for each rating agency. Lower credit ratings generally result in higher borrowing costs and reduced access to capital markets.

The following rating actions have been taken by these NRSROs since January 1, 2010:

Ford

On January 11, 2010, Fitch upgraded Ford's corporate rating to B- from CCC and the senior secured rating to BB- from B+; the outlook remains positive.

On March 17, Moody's upgraded Ford's Corporate Family Rating (CFR) to B2 from B3, our secured credit facility to Ba2 from Ba3 and our senior unsecured rating to B3 from Caa1. The ratings are under review for further possible upgrade.

On March 30, DBRS upgraded Ford's issuer rating to B from B (low), our senior secured credit facilities to BB (low) from B (high) and long-term debt to CCC (high) from CCC. The ratings trend is positive.

On April 28, S&P revised its outlook on Ford to positive from stable and affirmed its ratings.

On April 29, Fitch upgraded Ford's issuer default rating to B from B-, our senior secured rating to BB from BB- and our senior unsecured rating to CCC from CC. The outlook remains positive.

Ford Credit

On January 11, 2010, Fitch upgraded Ford Credit's corporate rating to B- from CCC, the senior unsecured rating to B+ from B, and the short-term rating to B from C; the outlook remains positive.

On March 4, Moody's upgraded Ford Credit's senior unsecured rating to B2 from B3 with a stable outlook.

On March 17, Moody's upgraded Ford Credit's senior unsecured rating to B1 from B2. The ratings are under review for further possible upgrade. Moody's also initiated a CFR for Ford Credit of B1.

On March 30, DBRS upgraded Ford Credit's issuer and long-term debt ratings to B (high) from B. It also upgraded Ford Credit's short-term rating to R-4 from R-5. The long-term ratings trend is positive while the short-term ratings trend is stable.

On April 28, S&P revised its outlook on Ford Credit to positive from stable and affirmed its ratings.

On April 29, Fitch upgraded Ford Credit's issue default rating to B from B- and affirmed its short-term rating at B. The outlook remains positive.

The following chart summarizes certain of the credit ratings and the outlook presently assigned to Ford and Ford Credit by these four NRSROs:

			N	RSRO RATINO	SS		
		For	d			Ford Credit	
	Issuer Default/ Corporate/ Issuer Rating	Long-Term Senior Unsecured	Senior Secured	Outlook / Trend	Long-Term Senior Unsecured	Short-Term Unsecured	Outlook / Trend
DBRS	В	CCC (high)	BB (low)	Positive	B (high)	R-4	Positive**
Fitch	В	CCC	BB	Positive	BB-	В	Positive
Moody's	B2	В3	Ba2	Review	B1	NP	Review
S&P	B-	CCC	B-	Positive	B-*	NR	Positive

^{*} S&P assigns FCE a long-term senior unsecured rating of B, maintaining a one notch differential versus Ford Credit.

^{**} DBRS has assigned a stable trend to Ford Credit's short-term ratings.

OUTLOOK

We are encouraged by our momentum, and we continue to believe that our plan – to aggressively restructure our business to operate profitably, accelerate development of new products customers want and value, finance our plan and improve our balance sheet, and work together effectively as one team to leverage our global resources – provides the right strategy to achieve our objectives. For additional discussion of the economic environment and discussion and assessment of the risks and opportunities to our planning assumptions, see "Item 1A. Risk Factors," "Overview," "Outlook," and "Critical Accounting Estimates" in our Form 2009 10-K Report, and updates thereto in this Report.

Our projected vehicle production is as follows:

		Second Quarter 2010 (in thousands)	
	Vehicle Unit Production*	Over/(Under) Second Quarter 2009	
Ford North America	625	174	
Ford South America	135	25	
Ford Europe	448	50	
Ford Asia Pacific Africa	213	73	

Includes production of Ford and JMC brand vehicles to be sold by our unconsolidated affiliates.

Overall, the year-over-year increase in second quarter 2010 production reflects continuing strong customer demand for our products, maintenance of competitive stock levels, and the non-recurrence of prior-year stock reductions. Our projection for second quarter production for Ford North America is 30,000 units higher than our previous guidance given in March 2010, reflecting strengthening demand in the United States and Canada.

Our planning assumptions for 2010 include the following:

Industry Volume (a)	Full-Year Plan	Full-Year Outlook	Memo: First Quarter 2010
(million units)			
-United States	11.5 - 12.5	On Track	11.3
-Europe (b)	13.5 - 14.5	14.0 - 15.0	16.0
Operational Metrics			
Compared with prior year:			
–Quality	Improve	On Track	Improved
-Automotive Structural Costs (c)	Somewhat Higher	On Track	\$100 Million Lower
–U.S. Market Share (Ford, Lincoln, and Mercury)	Equal / Improve	On Track	16.5%
–U.S. Share of Retail Market (d)	Equal / Improve	On Track	14.1%
-Europe Market Share (b)	Equal	On Track	9.4%
Absolute amount:			
-Automotive Operating-Related Cash Flow (e)	Positive	On Track	\$(100) Million
-Capital Spending	\$4.5 Billion to \$5 Billion	On Track	\$900 Million

Although excess industry capacity continues to persist in key markets and the recovery is moderated in some regions by continued labor market weakness and tight consumer credit conditions, we do expect global industry sales volume to increase by 5% to 10% this year over last year. As shown above, we have increased our outlook for the 19 markets we track in Europe, reflecting strong first quarter industry sales volume tempered by uncertainty about the extent of payback from government vehicle purchase incentives which began last year and are now ending.

Looking ahead, we anticipate that global central banks are likely to reduce stimulus by retiring special lending programs and begin modest policy interest rate increases, although we anticipate interest rates likely will remain relatively low in 2010 and supportive of economic recovery.

Seasonally adjusted annual rate; includes medium and heavy vehicles. For the 19 markets we track in Europe as defined in "Item 1. Business" of our 2009 Form 10-K Report. (b)

Structural cost changes are measured primarily at prior-year exchange, and exclude special items and discontinued operations. (c)

⁽d) Estimate.

See "Liquidity and Capital Resources" discussion above for reconciliation to U.S. GAAP.

We experienced significant positive net pricing in 2009; we expect year-over-year improvement in 2010, but this will be a smaller magnitude than in 2009.

As reported, we have achieved significant structural cost reductions over the past four years; as a result, we expect full-year Automotive structural costs to be somewhat higher as we increase production to meet demand and increase investment in new products.

We expect net product costs to increase in 2010, as commodity cost increases have resumed in conjunction with the emergence of economic recovery and are expected to continue throughout the year. We expect the net impact of other product-related cost changes to be flat. We also will continue to be affected by currency volatility.

We remain on track to achieve positive Automotive operating-related cash flow for full-year 2010. Our Automotive operating-related cash outflow of \$100 million in the first quarter is consistent with our plan, and primarily reflects seasonal timing differences associated with inventories and receivables due to our annual December shutdown period. See the discussion of changes in Automotive gross cash in "Liquidity and Capital Resources" above for further information about these timing differences.

We expect that the rate of capital spending for the remainder of 2010 will be higher than the first quarter 2010 rate to support our product programs.

As previously reported, at the end of March 2010 we entered into a definitive agreement (attached hereto as Exhibit 10.1) to sell Volvo and related assets to Zhejiang Geely Holding Group for \$1.8 billion, subject to purchase price adjustments to reflect Volvo's pension liabilities, debt, cash and working capital at closing, the net effect of which could decrease significantly the cash proceeds we receive. We expect the sale to close in the third quarter, and as such we have not included Volvo in our 2010 planning assumptions.

Ford Credit now expects full-year pre-tax profits to be about the same as 2009, primarily reflecting recent improvements in used-vehicle auction values and credit loss performance, which are expected to offset the effects of lower average receivables and the non-recurrence of certain favorable factors experienced during 2009. At year-end 2010, Ford Credit anticipates managed receivables to be in the range of \$80 billion to \$90 billion. Ford Credit also expects to pay distributions in 2010 of about \$2 billion (up from \$1.5 billion projected previously), including the first quarter cash distribution of \$500 million. Ford Credit will continue to assess its ability to make future distributions based on its available liquidity and managed leverage objectives.

Based on our improving business performance, the gradually strengthening economy, and our updated planning assumptions, we now expect to deliver solid profits with positive Automotive operating-related cash flow for full-year 2010, and continued improvement in 2011.

Risk Factors

Statements included herein may constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements are based on expectations, forecasts, and assumptions by our management and involve a number of risks, uncertainties, and other factors that could cause actual results to differ materially from those stated, including, without limitation:

- Further declines in industry sales volume, particularly in the United States or Europe, due to financial crisis, deepening recession, geo-political events, or other factors:
- Decline in market share:
- Lower-than-anticipated market acceptance of new or existing products;
- An increase in or acceleration of market shift beyond our current planning assumptions from sales of trucks, medium- and large-sized utilities, or other more profitable vehicles, particularly in the United States;
- A return to elevated gasoline prices, as well as the potential for volatile prices or reduced availability;
- Continued or increased price competition resulting from industry overcapacity, currency fluctuations, or other factors;
- Adverse effects from the bankruptcy, insolvency, or government-funded restructuring of, change in ownership or control of, or alliances entered into by a major competitor:
- A prolonged disruption of the debt and securitization markets;
- Fluctuations in foreign currency exchange rates, commodity prices, and interest rates;
- Economic distress of suppliers that may require us to provide substantial financial support or take other measures to ensure supplies of components or materials and could increase our costs, affect our liquidity, or cause production disruptions;
- Single-source supply of components or materials;
- Labor or other constraints on our ability to restructure our business;
- Work stoppages at Ford or supplier facilities or other interruptions of production;
- Substantial pension and postretirement health care and life insurance liabilities impairing our liquidity or financial condition;
- Worse-than-assumed economic and demographic experience for our postretirement benefit plans (e.g., discount rates or investment returns); Restriction on use of tax attributes from tax law "ownership change;"
- The discovery of defects in vehicles resulting in delays in new model launches, recall campaigns, or increased warranty costs;
- Increased safety, emissions, fuel economy, or other regulation resulting in higher costs, cash expenditures, and/or sales restrictions;
- Unusual or significant litigation or governmental investigations arising out of alleged defects in our products, perceived environmental impacts, or
- A change in our requirements for parts or materials where we have long-term supply arrangements that commit us to purchase minimum or fixed quantities of certain parts or materials, or to pay a minimum amount to the seller ("take-or-pay" contracts);
- Adverse effects on our results from a decrease in or cessation of government incentives related to capital investments;
- Adverse effects on our operations resulting from certain geo-political or other events;
- Substantial levels of Automotive indebtedness adversely affecting our financial condition or preventing us from fulfilling our debt obligations (which may grow because we are able to incur substantially more debt, including additional secured debt);
- Failure of financial institutions to fulfill commitments under committed credit facilities;
- Inability of Ford Credit to obtain competitive funding;
- Inability of Ford Credit to access debt, securitization, or derivative markets around the world at competitive rates or in sufficient amounts due to credit rating downgrades, market volatility, market disruption, or other factors;
- Higher-than-expected credit losses;
- Increased competition from banks or other financial institutions seeking to increase their share of financing Ford vehicles;
- Collection and servicing problems related to finance receivables and net investment in operating leases;
- Lower-than-anticipated residual values or higher-than-expected return volumes for leased vehicles;
- New or increased credit, consumer, or data protection or other laws or regulations resulting in higher costs and/or additional financing restrictions; and
- Inability to implement our One Ford plan.

We cannot be certain that any expectation, forecast, or assumption made in preparing forward-looking statements will prove accurate, or that any projection will be realized. It is to be expected that there may be differences between projected and actual results. Our forward-looking statements speak only as of the date of their initial issuance, and we do not undertake any obligation to update or revise publicly any forward-looking statement, whether as a result of new information, future events or otherwise. For additional discussion of these risks, see "Item 1A. Risk Factors" in our 2009 Form 10-K Report.

CRITICAL ACCOUNTING ESTIMATES

Based on events occurring subsequent to December 31, 2009, we are updating certain of the Critical Accounting Estimates disclosed in our 2009 Form 10-K Report.

Valuation of Deferred Tax Assets

Nature of Estimates Required. Deferred tax assets and liabilities are recognized based on the future tax consequences attributable to temporary differences that exist between the financial statement carrying value of assets and liabilities and their respective tax bases, and operating loss and tax credit carryforwards on a taxing jurisdiction basis. We measure deferred tax assets and liabilities using enacted tax rates that will apply in the years in which we expect the temporary differences to be recovered or paid.

U.S. GAAP standards of accounting for income taxes require a reduction of the carrying amounts of deferred tax assets by recording a valuation allowance if, based on the available evidence, it is more likely than not (defined as a likelihood of more than 50%) such assets will not be realized. The valuation of deferred tax assets requires judgment in assessing the likely future tax consequences of events that have been recognized in our financial statements or tax returns and future profitability. Our accounting for deferred tax consequences represents our best estimate of those future events. Changes in our current estimates, due to unanticipated events or otherwise, could have a material impact on our financial condition and results of operations.

Assumptions and Approach Used. In assessing the need for a valuation allowance, we consider both positive and negative evidence related to the likelihood of realization of the deferred tax assets. If, based on the weight of available evidence, it is more likely than not the deferred tax assets will not be realized, we record a valuation allowance. The weight given to the positive and negative evidence is commensurate with the extent to which the evidence may be objectively verified. As such, it is generally difficult for positive evidence regarding projected future taxable income exclusive of reversing taxable temporary differences to outweigh objective negative evidence of recent financial reporting losses. U.S. GAAP states that a cumulative loss in recent years is a significant piece of negative evidence that is difficult to overcome in determining that a valuation allowance is not needed against deferred tax assets.

This assessment, which is completed on a taxing jurisdiction basis, takes into account a number of types of evidence, including the following:

- Nature, frequency, and severity of current and cumulative financial reporting losses. A pattern of objectively measured recent financial reporting losses is heavily weighted as a source of negative evidence. In certain circumstances, historical information may not be as relevant due to changed circumstances;
- Sources of future taxable income. Future reversals of existing temporary differences are heavily-weighted sources of objectively verifiable positive evidence. Projections of future taxable income exclusive of reversing temporary differences are a source of positive evidence only when the projections are combined with a history of recent profits and can be reasonably estimated. Otherwise, these projections are considered inherently subjective and generally will not be sufficient to overcome negative evidence that includes relevant cumulative losses in recent years, particularly if the projected future taxable income is dependent on an anticipated turnaround to profitability that has not yet been achieved. In such cases, we generally give these projections of future taxable income no weight for the purposes of our valuation allowance assessment pursuant to U.S. GAAP; and
- Tax planning strategies. If necessary and available, tax planning strategies would be implemented to accelerate taxable amounts to utilize expiring carryforwards. These strategies would be a source of additional positive evidence and, depending on their nature, could be heavily weighted.

See Note 23 of the Notes to the Financial Statements in our 2009 Form 10-K Report for more information regarding deferred tax assets.

Sensitivity Analysis. In 2006, our net deferred tax position in the United States changed from a net deferred tax liability position to a net deferred tax asset position. In our assessment of the need for a valuation allowance, we heavily weighted the negative evidence of cumulative financial reporting losses in recent periods and the positive evidence of future reversals of existing temporary differences. Although a sizable portion of our North American losses in recent years were the result of charges incurred for restructuring actions, impairments, and other special items, even without these charges we still would have incurred significant operating losses. Accordingly, we considered our pattern of recent losses to be relevant to our analysis. Considering this pattern of recent relevant losses and the uncertainties associated with projected future taxable income exclusive of reversing temporary differences, we gave no weight to projections showing future U.S. taxable income for purposes of assessing the need for a valuation allowance. As a result of our assessment, we concluded that the net deferred tax assets of our U.S. entities required a full valuation allowance. We also recorded a full valuation allowance on the net deferred tax assets of certain foreign entities, such as Germany, Canada, and Spain, as the realization of these foreign deferred tax assets are reliant upon U.S.-source taxable income

At December 31, 2006, we reported a \$7.3 billion valuation allowance against our deferred tax assets (including \$2.7 billion resulting from the adoption of the revised standard on accounting for defined benefit pension and other postretirement benefit plans). During 2007, we recorded an additional valuation allowance of \$700 million. Losses during 2008, primarily in the United States, increased the valuation allowance by \$9.3 billion to a balance of \$17.3 billion at December 31, 2008. The valuation allowance increased by \$100 million in 2009, which reflects a \$1.1 billion increase related to charges to other comprehensive income, partially offset by a \$1 billion decrease as a result of operating profits. At March 31, 2010, the valuation allowance decreased to \$16.4 billion reflecting a \$200 million decrease due to operating income in the first three months; the remaining decrease primarily relates to the pending sale of our Volvo operations.

A sustained period of profitability in our North America operations is required before we would change our judgment regarding the need for a full valuation allowance against our net deferred tax assets. Accordingly, although we were profitable during the first quarter of 2010, we continue to record a full valuation allowance against the net deferred tax assets in the United States and foreign entities discussed above. Although the weight of negative evidence related to cumulative losses is decreasing as we deliver on our One Ford plan, we believe that this objectively-measured negative evidence outweighs the subjectively-determined positive evidence and, as such, we currently do not anticipate a change in judgment regarding the need for a full valuation allowance in 2010. The consumption of tax attributes to offset expected operating profits during 2010, however, would reduce the overall level of deferred tax assets subject to valuation allowance.

At March 31, 2010 and December 31, 2009 our net deferred tax assets, net of the valuation allowances of \$16.4 billion and \$17.4 billion, respectively, were \$1.3 billion and \$1.1 billion, respectively. Unlike our U.S. operations where, considering the pattern of relevant losses and the uncertainties associated with projected future taxable income exclusive of reversing temporary differences, we gave no weight to projections showing future taxable income, these net deferred tax assets relate to certain operations outside North America where we generally have had a long history of profitability and believe it is more likely than not that the net deferred tax assets will be realized through future taxable earnings. Accordingly, we have not established a valuation allowance on our remaining net deferred tax assets. Most notably, at March 31, 2010 and December 31, 2009, we recognized a net deferred tax asset of \$1.3 billion and \$1.5 billion, respectively, in our U.K. Automotive operations, primarily based upon the tax return consolidation of our Automotive operations with our U.K. FCE operation. Our U.K. FCE operation has a long history of profitability, and we believe it will provide a source of future taxable income that can be reasonably estimated. If in the future FCE profits in the United Kingdom decline, additional valuation allowances may be required. We will continue to assess the need for a valuation allowance in the future.

ACCOUNTING STANDARDS ISSUED BUT NOT YET ADOPTED

There are no applicable accounting standards issued but not yet adopted to report this period.

OTHER FINANCIAL INFORMATION

The interim financial information included in this Quarterly Report on Form 10-Q for the periods ended March 31, 2010 and 2009 has not been audited by PricewaterhouseCoopers LLP ("PricewaterhouseCoopers"). In reviewing such information, PricewaterhouseCoopers has applied limited procedures in accordance with professional standards for reviews of interim financial information. Readers should restrict reliance on PricewaterhouseCoopers' reports on such information accordingly. PricewaterhouseCoopers is not subject to the liability provisions of Section 11 of the Securities Act of 1933 for its reports on interim financial information, because such reports do not constitute "reports" or "parts" of registration statements prepared or certified by PricewaterhouseCoopers within the meaning of Sections 7 and 11 of the Securities Act of 1933.

ITEM 3. Quantitative and Qualitative Disclosures About Market Risk.

Automotive Sector

Foreign Currency Risk. The net fair value of foreign exchange forward and option contracts (including adjustments for credit risk) as of March 31, 2010 was a liability of approximately \$69 million compared to a net fair value liability of \$26 million as of December 31, 2009. The potential decrease in fair value of foreign exchange forward and option contracts (excluding adjustments for credit risk), assuming a 10% adverse change in the underlying foreign currency exchange rates versus the U.S. dollar, would be approximately \$614 million at March 31, 2010 and was \$622 million as of December 31, 2009. If adjustments for credit risk were to be included, the decrease would be smaller.

Commodity Price Risk. The net fair value of commodity forward and option contracts (including adjustments for credit risk) as of March 31, 2010 was a liability of \$17 million, compared to a liability of \$39 million as of December 31, 2009. The potential decrease in fair value of commodity forward and option contracts (excluding adjustments for credit risk), assuming a 10% decrease in the underlying commodity prices, would be approximately \$16 million at March 31, 2010, compared with a decrease of \$20 million at December 31, 2009. If adjustments for credit risk were to be included, the decrease would be smaller.

Financial Services Sector

Interest Rate Risk. To provide a quantitative measure of the sensitivity of Ford Credit's pre-tax cash flow to changes in interest rates, Ford Credit uses interest rate scenarios that assume a hypothetical, instantaneous increase or decrease in interest rates of 100 basis points (or 1%) across all maturities, as well as a base case that assumes that interest rates remain constant at existing levels. These interest rate scenarios are purely hypothetical and do not represent Ford Credit's view of future interest rate movements. The differences in pre-tax cash flow between these scenarios and the base case over a twelve-month period represent an estimate of the sensitivity of Ford Credit's pre-tax cash flow. Under this model, Ford Credit estimates that at March 31, 2010, all else constant, such an increase in interest rates would increase pre-tax cash flow by \$23 million over the next twelve months, compared with \$27 million at December 31, 2009. The sensitivity analysis presented above assumes a one-percentage point interest rate change to the yield curve that is both instantaneous and parallel. In reality, interest rate changes are rarely instantaneous or parallel and rates could move more or less than the one percentage point assumed in Ford Credit's analysis. As a result, the actual impact to pre-tax cash flow could be higher or lower than the results detailed above.

ITEM 4. Controls and Procedures.

Evaluation of Disclosure Controls and Procedures. Alan Mulally, our Chief Executive Officer ("CEO"), and Lewis Booth, our Chief Financial Officer ("CFO"), have performed an evaluation of the Company's disclosure controls and procedures, as that term is defined in Rule 13a-15(e) of the Securities Exchange Act of 1934, as amended ("Exchange Act"), as of March 31, 2010, and each has concluded that such disclosure controls and procedures are effective to ensure that information required to be disclosed in our periodic reports filed under the Exchange Act is recorded, processed, summarized and reported within the time periods specified by SEC rules and forms, and that such information is accumulated and communicated to the CEO and CFO to allow timely decisions regarding required disclosures.

Changes in Internal Control over Financial Reporting. On January 1, 2010, we adopted the new accounting standard for consolidation of VIEs. Pursuant to the new standard, we deconsolidated the majority of our Automotive sector VIEs, and account for our interest in these unconsolidated affiliates using the equity investment method.

PART II. OTHER INFORMATION

ITEM 1. Legal Proceedings.

Environmental Matters

Dearborn Research and Engineering Center (previously reported in "Item 3. Legal Proceedings" on page 35 of our 2009 Form 10-K Report). In April 2010, we reached a settlement with the Michigan Department of Natural Resources and Environment ("MDNRE," formerly the Michigan Department of Environmental Quality) regarding alleged violations at our Dearborn Research and Engineering Center. The settlement, which will be finalized after a public comment period, includes a penalty of \$80,640.

Other Matters

SEC Pension and Post-Employment Benefit Accounting Inquiry. On October 14, 2004, the SEC's Division of Enforcement notified us that it was conducting an inquiry into the methodology used to account for pensions and other post-employment benefits. We were one of a number of companies to receive a request for information as part of this inquiry, and completed submission of all information requested as of April 2007. Since then, no further request had been made by the SEC, and on March 12, 2010 we received confirmation that this matter has been concluded.

ITEM 2. Unregistered Sales of Equity Securities and Use of Proceeds.

During the first quarter of 2010, we purchased shares of Ford Common Stock as follows:

Period	Total Number of Shares Purchased*	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly- Announced Plans or Programs	Maximum Number (or Approximate Dollar Value) of Shares that May Yet Be Purchased Under the Plans or Programs
Jan. 1, 2010 through Jan. 31, 2010	•	\$ •	•	**
Feb. 1, 2010 through Feb. 28, 2010	50,695	11.12	•	**
March 1, 2010 through March 31, 2010	19,919	12.57	<u> </u>	**
Total/Average	70,614	11.53	•	**

^{*} We presently have no publicly-announced repurchase program in place. Shares were acquired from our employees or directors in accordance with our various compensation plans as a result of share withholdings to pay: (i) income tax related to the lapse of restrictions on restricted stock or the issuance of unrestricted stock; and (ii) the exercise price and related income taxes with respect to certain exercises of stock options.

** No publicly-announced repurchase program in place.

ITEM 5. Other Information.

Governmental Standards

Fuel Quality and Content

European Requirements. Legislators in the European Union ("EU") continue to discuss biomass sustainability criteria, as the use of automotive fuel derived from biomass increases, primarily driven by the EU renewable energy directive calling for 10% renewable energy content for transportation purposes (including renewable electricity for electric vehicles, as well as biomass-based fuel). Fuel content requirements already have been amended to allow "B5" diesel (including up to 5% biomass-based fuel) and "E5" gasoline (including up to 5% ethanol); it is expected that the renewable energy directive will lead to approval of "E10" gasoline (including up to 10% ethanol), and "B7" diesel (including up to 7% biomass-based fuel), and further measures may be necessary to reach the 10% target. At the same time, the automotive industry and oil industry continue to work to identify a set of potential fuel scenarios that might satisfy the target, and assess the most likely routes to success.

Motor Vehicle Safety

U.S. Requirements. In the wake of recent Toyota product recalls, the U.S. Senate and House have introduced new automotive safety-related bills. The legislative proposals include additional vehicle standards related to brake override technology, pedal placement, electric systems, event data recorders and keyless ignition systems. The proposed legislation also contains a presumption favoring public disclosure of early warning data under the Transportation Recall Enhancement, Accountability, and Documentation ("TREAD") Act, new "Vehicle Safety User Fees", and increased civil penalties for manufacturers. Ford, independently and through the Alliance of Automobile Manufacturers, is working with legislators as the legislation is being drafted.

Motor Vehicle Fuel Economy

- *U.S. Requirements Federal Standards.* The EPA issued a determination in December 2009 under the Clean Air Act that greenhouse gases ("GHGs") endangered the public health and welfare, and thereafter finalized GHG regulations for mobile and stationary sources. In early 2010, several industry groups filed a petition for review of the endangerment finding, which now have been consolidated into a single case in the U.S. Court of Appeals for the District of Columbia Circuit. Because the mobile source GHG rules and the PSD Tailoring Rule (discussed in "Governmental Standards" in our 2009 Form 10-K Report) are predicated on the EPA's endangerment finding, the outcome of this litigation could affect implementation of these rules.
- *U.S. Requirements "One National Standard" for Model Years 2012-2016.* Since the May 2009 announcement of an agreement in principle among the automobile industry, the federal government, and the state of California concerning motor vehicle GHG emissions and fuel economy regulations, various steps have been taken to implement the agreement in principle, including the following actions in the first quarter of 2010:
 - The parties have filed joint motions to dismiss the federal litigation in California, Vermont, and Rhode Island regarding federal preemption of state motor vehicle GHG standards.
 - The California Air Resources Board ("CARB") adopted the modifications to its regulations required to implement the agreement in principle.
 - The EPA and National Highway Traffic Safety Administration promulgated a final rule setting forth harmonized GHG and fuel economy standards for the 2012-2016 model years.

The federal GHG and fuel economy standards are very challenging, requiring new light-duty vehicles to achieve an industry average fuel economy of approximately 35.5 miles per gallon by the 2016 model year. We believe that we will be able to comply with the California GHG standards for the 2009-2011 period, and the harmonized federal corporate average fleet economy ("CAFE")/GHG standards for the 2012-2016 period, as a result of aggressive actions to improve fuel economy built into our cycle plan.

Also in April 2010, CARB announced plans to abandon the "cool glazing" standards that it previously had considered.

Other National Requirements. In March 2010, the Canadian province of Quebec provided clarification regarding its GHG regulations, indicating it would adopt an approach similar to the new model in the U.S. "One National Standard." Also, the Canadian federal government has published for public comment a draft rule for car and light truck GHG emissions; a final rule is expected later this year.

ITEM 6. Exhibits.

Please see exhibit index below.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

FORD MOTOR COMPANY

(Registrant)

May 7, 2010 Date:

By: /s/ Bob Shanks
Bob Shanks
Vice President and Controller
(Chief Accounting Officer)

EXHIBIT INDEX

Designation	Description	Method of Filing
Exhibit 10.1	Stock Purchase Agreement by and among Ford Motor Company, Volvo Personvagnar Holding AB, Mintime North America, LLC and Geely Sweden AB for the Sale and Purchase of Volvo Car Corporation and Volvo Cars of North America, LLC	Filed with this Report.
Exhibit 10.2	Executive Compensation Recoupment Policy	Filed with this Report.*
Exhibit 10.3	Amendment dated May 5, 2010 to the Consulting Agreement between Ford Motor Company and Edsel B. Ford $\rm II$	Filed with this Report.*
Exhibit 12	Ford Motor Company and Subsidiaries Calculation of Ratio of Earnings to Combined Fixed Charges	Filed with this Report.
Exhibit 15	Letter of PricewaterhouseCoopers LLP dated May 7, 2010 relating to financial information	Filed with this Report.
Exhibit 31.1	Rule 15d-14(a) Certification of CEO	Filed with this Report.
Exhibit 31.2	Rule 15d-14(a) Certification of CFO	Filed with this Report.
Exhibit 32.1	Section 1350 Certification of CEO	Furnished with this Report.
Exhibit 32.2	Section 1350 Certification of CFO	Furnished with this Report.
Exhibit 101.INS	XBRL Instance Document	Furnished with this Report.**
Exhibit 101.SCF	H XBRL Taxonomy Extension Schema Document	Furnished with this Report.**
Exhibit 101.CAI	XBRL Taxonomy Extension Calculation Linkbase Document	Furnished with this Report.**
Exhibit 101.LAI	3 XBRL Taxonomy Extension Label Linkbase Document	Furnished with this Report.**
Exhibit 101.PRE	E XBRL Taxonomy Extension Presentation Linkbase Document	Furnished with this Report.**
Exhibit 101.DEF	XBRL Taxonomy Extension Definition Linkbase Document	Furnished with this Report.**

Management contract or compensatory plan or arrangement. Submitted electronically with this Report.

STOCK PURCHASE AGREEMENT
BY AND AMONG
FORD MOTOR COMPANY,
VOLVO PERSONVAGNAR HOLDING AB,
MINTIME NORTH AMERICA, LLC
AND
GEELY SWEDEN AB
FOR THE SALE AND PURCHASE OF
VOLVO CAR CORPORATION
AND
VOLVO CARS OF NORTH AMERICA, LLC

March 28, 2010

TABLE OF CONTENTS

			Page
1	DEF	FINITIONS	1
2	SAI	E AND PURCHASE	1
•	2 . 1	Sale and Purchase	1
	2 . 2	Delivery of Pre-Closing Estimates	2
	2 . 3	Payment of the Initial Purchase Price	3
	2	Discharge of Intra-Group Balances	3
	2	Closing Statement Adjustments to the Purchase Price	4
	5 2	Pension Adjustment to the Purchase Price	5
	6 2 <u>.</u>	Closing	5
	7 2	Goodwill Deposit	6
3	8 REF	PRESENTATIONS AND WARRANTIES OF PARENT AND SWEDEN SELLER	6
•	3	Organization and Standing; Target Companies; Company Subsidiaries	6
	3	Authority and Capacity; No Conflicts; Binding Obligation; No Liquidation	7
	3 . 3	Governmental Approvals	8
	3	Formation Documents and Bylaws; Books and Records	9
	3	Capitalization	9
	5	Ownership and Transfer of the Sweden Company Shares and the U.S. Company Interest	10
		Directors and Officers	10
	7 3	Financial Statements and Liabilities	10
	8 3 9	Title to and Sufficiency of Assets	11
	3 . 1	Conduct of Business; Absence of Certain Changes or Events	12
	0	Material Contracts	12
	1 1 3	Taxes	15
	1 2	I UACO	13
	3	Real Property	18
	3	Intellectual Property	20
	1 4		20
	-		

1	Insurance	25
5 3	Litigation	25
1 6 3	Labor and Employment	26
1 7		
3 1	Employee Benefits	26
8 3	Data Privacy	28
1 9 3	Material Environmental Permits	28
2 0		
3 2	Environmental Site Assessments	28
3	Contaminated Land	28
2 2 3	Material Expenditure	29
2 3 3		
3 2 4 3	Product Recalls	29
3 . 2	Product Liability	29
5 3	Warranties	29
2 6 3	Compliance with Law; Permits	29
2 7	Compinance with Law, Fermits	29
3	Information Technology and Functionality	30
2 8 3	Suppliers	31
2 9 3		
3 3 0 3	JLR Agreements and AMLL Agreements	32
	NSC Reorganization	32
3 1 3	Exclusivity of Representations and Warranties	32
3 2		

EXECUTION VERSION

4	REF	PRESENTATIONS AND WARRANTIES OF U.S. BUYER AND SWEDEN BUYER	32
•	4	Organization and Standing; Capitalization	32
	1 4	Authority and Capacity; No Conflicts; Binding Obligation; No Liquidation	33
	2		
	4	Governmental Approvals	34
	3	Financial Capacity	34
	4		
	4	Investment Intent and Experience	35
	5	U.S. Dealers	35
5	6	CLOSING COVENANTS	36
5	5	E-CLOSING COVENANTS Governmental Filings	36
	1	Governmental Pinings	30
	5	Consultation Obligations	38
	2 5	Conduct of Business	39
	3		
	5	Access to Company Information	43
	4 5	No Control of the Business	43
	5		
	5	Notification of Certain Matters; Supplemental Disclosure	44
	5	Contract Consents	44
	7	Termination of ID Intercompany Associated Termination of Douges of Attorney	15
	5 8	Termination of IP Intercompany Agreements; Termination of Powers of Attorney	45
	5	Information Barriers Protocol	45
	9 5	Benefit Plans in Canada	45
	1	Beliefit I fails in California	73
	0 5	Benefit Plans in the U.S.	46
	1		
	5	Defined Benefit Plan in the Netherlands	46
	1		
	5	Replacement of Insurance Policies	46
	1		
	5	IT Transition Plan	47
	1 4		
	5	Replacement Vehicle Finance Arrangements	47
	1 5		
	5	Leakage	47
	1 6		
	5	Non-solicitation	47
	1 7		

	5 1	Financing	48
	8 5	Retiree Medical Plans	49
	1 9		
	5 . 2	UK Pension Matters	49
	0 5	CFMA Payments	50
	2		30
	5 . 2 2	Allocation of Special GHG Credits	51
	5 . 2 3	Patent License Agreement	51
	5	Working Capital Forecast	52
6	4 AD	DITIONAL AGREEMENTS	52
	6	U.S. Tax Treatment of the Sale and Purchase of the U.S. Company Interest	52
	1		
	6	Tax Liability	52
	2 6	Transfer Taxes	55
	3 6	Tax Returns	55
	4 6	Conduct of Tax Proceedings	60
	5 6	Group Relief and Tax Losses	63
	6	Tax Cooperation	64
	7 6	Group Arrangements	65
	8 6	BL3 Property	65
	9 6	Intercompany Accounts and Agreements	66
	1 0	Shared Contracts	66
	6 1	Shared Confidences	00
	6	PZEV Credits	67
	1		

EXECUTION VERSION

6 1	CAFE Filing	68
3 6	GHG Allowances	68
1 4		
6	Information Barriers Protocol	69
1		
5 6	No Relevant Transfer	69
1 6		
6	Employment	69
1		
7 6	Maintenance of Employee Benefits Plan	70
1 8		
6	Equity Plan Awards	71
1		
9	Defined Contribution Plans Transfers	71
2 0		
6	Defined Benefit Plan in the U.S.	71
2		
1 6	Defined Benefit Plan in the Netherlands	72
2 2 6		
6	Defined Benefit Pension Plan in Japan	72
2 3		
3 6	Consent to Transfer; Consultation Obligations	73
2 4		
6	Non-Solicitation of Certain Employees	74
2 5		
6	Retained Company Name Changes	74
2		
6	Insurance Claims	74
6	insurance Craims	/4
2 7		
6	Directors and Officers Discharge of Liability; Indemnification	75
2 8		
8 6	Records and Assistance	75
	Records and Assistance	75
2 9		
6	Releases	76
3 0		
0 6	Wrong Pockets	77
		•
3 1		
6	Replacement Vehicle Finance Arrangements	78
3 2		

	6	Further Actions	78
	3 3		
	6 3 4	Confidentiality	79
	6	News Releases; Public Announcements	79
	3 5		
	6	Assignment of PLA	80
	3 6 6	Maintenance of Back-to-Back Financing Commitment	80
	. 3	Maintenance of Back-to-Back I mainting Communicati	00
7	7	NDITIONS PRECEDENT TO OBLIGATIONS OF PARENT AND SWEDEN SELLER	81
•	7	Representations and Warranties	81
	1 7	Governmental Approvals; Buyer Governmental Approvals	81
	. 2		01
	7	Performance of Obligations	81
	7	No Law	81
	4 7	Buyers Officer's Certificate	81
	5		01
	7	Documents prior to or at Closing	82
8	6 CO	NDITIONS PRECEDENT TO OBLIGATIONS OF U.S. BUYER AND SWEDEN BUYER	82
•	8	Representations and Warranties	82
	1 8	Governmental Approvals; Buyer Governmental Approvals	82
	2		
	8	Performance of Obligations	82
	8	No Company Material Adverse Effect	83
	4 8	No Law	83
	5		02
	8 6	Parent Officer's Certificate	83
	8	Documents prior to or at Closing	83
	7 8	U.S. EPA GHG Credits	83
9	8	OSING	84
,	9	Deliveries by Parent and Sweden Seller	84
	1		
	9	Deliveries by U.S. Buyer and Sweden Buyer	85
1	2 SUI	RVIVAL OF CLAIMS; INDEMNIFICATION; REMEDIES	86
	1	Survival of Claims	86
	0		
	10.2	2 Indemnification by Parent	87

EXECUTION VERSION

100)	Limitations on Indemnification by Parent	88
3 1 0		Indemnification by U.S. Buyer, Sweden Buyer	89
4 1 0)	Limitations on Indemnification by U.S. Buyer, Sweden Buyer	90
5 1 0)	Effect of Investigation; Waiver; No Target Company or Company Subsidiary Liability	90
1 0)	Third Party Claims	91
7 1 0)	Environmental Claims	93
8 1 0)	Adjustment of Losses	96
9 1 0		Payments	97
0 1 0		Treatment of Payments	97
1 1 1 0		Mitigation; Exclusive Remedy	97
2 T		MINATION	98
1 1		Termination	98
1 1 1		Effect of Termination	99
2 l N 2	ЛIS	CELLANEOUS	100
1 2		Notices	100
1 1 2		Amendment	101
1 2		Waivers	102
3 1 2 4		Successors and Assigns	102
1 2		Payments	103
51 2		Deductions from Payments; Gross-up	103

1 2	Currency Indemnity	104
7 1 2	Entire Agreement	105
8 1 2	Schedules; Exhibits; Appendices Preamble; Recitals	105
9 1 2	Headings; Interpretation; Governing Language	105
0 1 2	Counterparts; Effectiveness	106
1 1 2	Severability	106
2 1 2	Specific Performance; Injunctive Relief	107
3 1 2	Expenses	107
1 2 1 5	No Brokers	108
1 2 1	No Third Party Beneficiaries	108
6 1 2	Representation by Counsel; Mutual Drafting	108
7 1 2	Governing Law	108
8 1 2	Dispute Resolution	108
9 1 2 2 0	Waiver of Sovereign Immunity	110
0 1 2 2 1	Waiver of Jury Trial	110

LIST OF APPENDICES

Appendix 1	Definitions
Appendix 1	Deminions

LIST OF SCHEDULES

		Schedule A-1	Post-Closing Adjustments
Schedule A-2	Pension Adjustment Calcu		
		Schedule B	Disclosures
Schedule C	Governmental Approvals		
		Schedule D	Knowledge of Parent
Schedule E	Permitted Encumbrances		
		Schedule F	NSC Reorganization
Schedule G	Knowledge of Buyer		
0.1.1.1.1	T.C D D	Schedule H	Operation of Business
Schedule I	Information Barriers Proto		1' 601 1'11 11 15''' 11 11
0 1 1 1 17	Schedule J	Owne	ership of Shanghai Holdco and Beijing Holdco
Schedule K	Surviving Contracts	0 1 1 1 7	DZEM C. 1'
C 1 1 1 1 1 1	W. I. C. L. IE W.	Schedule L	PZEV Credits
Schedule M	Working Capital Facility		A
C -1 11- O	Brokers	Schedule N	Assurances
Schedule O		Schedule P	Janes Defined Denefit Dlan
Schedule Q			Japan Defined Benefit Plan
Schedule Q	Statutory Audit Companie	Schedule R	Replacement Policies
Schedule S	Buyer Governmental App		Replacement Folicies
Schedule 5		chedule T	Target Companies IT System
Schedule U	Excepted Markets	chedule 1	Target Companies II System
Benedule 6	Excepted Warkets	Schedule V	Excess GHG Allowances
Schedule W	Parent IT System	Belledule v	Excess GIIG / mowances
	Turent II System	Schedule X	Excluded Contracts
Schedule Y	Company Registered Intel		
		nedule Z	Resigning Directors and Officers
Schedule AA	Company Financial Stater	nents	ž ž
		nedule BB	Transaction Financial Statements
Schedule CC	Reconciliation of Account	ts	
	S	Schedule DD	Interim Financial Statements
Schedule EE	Collective Consultation R	equirements	
	Schedu	ıle FF	Non-Controlled IP Knowledge of Parent
Schedule GG	Cycle Plan		
		Schedule HH	Business Plan

Stock Purchase Agreement

v

LIST OF EXHIBITS

Binding Acquisition Financing Commitment Letters Exhibit I Form of LLC Interest Assignment and Assumption Agreement Exhibit II Form of Assignment and Assumption Agreement Exhibit III Exhibit IV Confidentiality Agreement Request for DB Asset Split Exhibit V Form of Information Technology Transitional Services Agreement Exhibit VI Exhibit VII Form of Dispute Resolution Agreement Exhibit VIII Form of Intellectual Property License Agreement Form of Intellectual Property Assignment Agreement Exhibit IX Form of Design, Development and Research Cooperation Agreement Exhibit X Exhibit XI Form of Engines and Stampings Supply Agreement (Parent to Sweden Company) Exhibit XII Form of Engines and Stampings Supply Agreement (Sweden Company to Parent) Exhibit XIII Form of Conrods and Brake Discs Supply Agreement (Floby) Form of DW 10b/c Supply Agreement Exhibit XV Exhibit XIV Form of Camshaft Supply Agreement Form of Tooling Agreement Exhibit XVI Exhibit XVII Forms of Secondment Agreements Form of Transitional Services Agreement Exhibit XVIII Exhibit XIX Form of NSC Transitional Services Agreement Form of Vehicle Finance Separation Agreement Exhibit XX Exhibit XXI Form of VAB Transitional Services Agreement Exhibit XXII **BL3 Form Leases** Exhibit XXIII Form of Ford Netherlands Pension Fund Transfer Agreement Exhibit XXIV Form of Letter Agreement regarding Allocation of GHG Credits Exhibit XXV Form of PLA Assignment and Assumption Agreement Exhibit XXVI Form of German IP Pledge Agreement

Exhibit XXVII

Exhibit XXVIII

Stock Purchase Agreement

Back-to-Back Financing Commitment

Seller Financing Agreement

STOCK PURCHASE AGREEMENT

This STOCK PURCHASE AGREEMENT (the "Stock Purchase Agreement") is entered into as of March 28, 2010 by and among Ford Motor Company, a corporation organized under the Law of the State of Delaware ("Parent"), Volvo Personvagnar Holding AB, a corporation organized under the Law of Sweden ("Sweden Seller"), Mintime North America, LLC, a limited liability company organized under the Law of the State of Delaware ("U.S. Buyer"), and Geely Sweden AB, a corporation organized under the Law of Sweden ("Sweden Buyer," together with U.S. Buyer "Buyers" and collectively with Parent and Sweden Seller, the "Parties").

Recitals of the Parties

WHEREAS, Parent owns all of the issued and outstanding interests in U.S. Company (the "U.S. Company Interest") and Sweden Seller owns all of the issued and outstanding shares of capital stock of the Sweden Company (the "Sweden Company Shares");

WHEREAS, Parent desires to sell the U.S. Company Interest to U.S. Buyer, and Sweden Seller desires to sell the Sweden Company Shares to Sweden Buyer, and U.S. Buyer desires to acquire from Parent the U.S. Company Interest, and Sweden Buyer desires to acquire from Sweden Seller the Sweden Company Shares, at the price and upon the terms and subject to the conditions set forth in this Stock Purchase Agreement; and

WHEREAS, Parent desires to transfer, or to grant license of, certain Intellectual Property Rights (or to procure that certain of its Affiliates do so) to Sweden Company immediately following the acquisition by Buyers of the Sweden Company Shares and the U.S. Company Interest upon the terms and subject to the conditions set forth in the Intellectual Property Agreements;

NOW, THEREFORE, in consideration of the representations, warranties, covenants and other agreements set forth in this Stock Purchase Agreement, and upon the terms and subject to the conditions set forth in this Stock Purchase Agreement, the Parties agree as follows:

1. DEFINITIONS

For purposes of this Stock Purchase Agreement, capitalized terms used but not otherwise defined in this Stock Purchase Agreement shall have the meanings set forth in Appendix 1, except as otherwise expressly provided.

2. SALE AND PURCHASE

2.1 Sale and Purchase

(a) Upon the terms and subject to the conditions of this Stock Purchase Agreement, on the Closing Date:

- (i) Parent shall sell, transfer and deliver the U.S. Company Interest to U.S. Buyer free and clear of Encumbrances;
- (ii) Sweden Seller shall sell, transfer and deliver the Sweden Company Shares to Sweden Buyer free and clear of Encumbrances;
- (iii) U.S. Buyer shall purchase the U.S. Company Interest from Parent;
- (iv) Sweden Buyer shall purchase the Sweden Company Shares from Sweden Seller; and, immediately thereafter,
- (v) Parent shall deliver the Intellectual Property Agreements duly executed by Parent, each of the other Ford Entities and Sweden

Company,

in exchange for payment of the Purchase Price to Parent (acting on behalf of itself, Sweden Seller and Ford Global) by Sweden Buyer (acting on behalf of (A) itself, with respect to the Sweden Company Shares, (B) U.S. Buyer, with respect to the U.S. Company Interest and (C) Sweden Company, with respect to the Intellectual Property Rights transferred or licensed under the Intellectual Property Agreements).

(b) The "**Purchase Price**" shall be equal to the Initial Purchase Price, as adjusted by: (i) the Intra-Group Balances Price Adjustment pursuant to Section 2.5(b)(i); (ii) the Net External Indebtedness Adjustment pursuant to Section 2.5(b)(ii); (iii) the Net Working Capital Adjustment pursuant to Section 2.5(b)(iii) and (iv) the Total Pension Adjustment pursuant to Section 2.6 and any special payment pursuant to paragraph 10 of Schedule A-2.

2.2 Delivery of Pre-Closing Estimates

- (a) Parent shall:
- (i) deliver to Buyers not less than six Business Days before Closing (or the anticipated Closing, as applicable) its good faith calculations of the Estimated Net External Indebtedness, Estimated Intra-Group Payables, Estimated Intra-Group Receivables, Estimated Net Working Capital and Estimated Pension Adjustment;
- (ii) consider in good faith any objections prepared by or on behalf of Buyers and delivered not less than four Business Days before Closing (or the anticipated Closing, as applicable), relating to any calculation provided pursuant to Section 2.2(a)(i) and, to the extent agreed by Parent in its sole discretion (acting in good faith), amend such calculations in light of any such objections; and
- (iii) deliver to Buyers not less than three Business Days before Closing any updated and finalized calculations of the Estimated Net External Indebtedness, Estimated Intra-Group Payables, Estimated Intra-Group Receivables, Estimated Net Working Capital and Estimated Pension Adjustment to the extent amended by Parent pursuant to Section 2.2(a)(ii).

- (b) The information to be provided to Buyers as contemplated by Section 2.2(a) with respect to the aggregate of such Estimated Intra-Group Receivables and Estimated Intra-Group Payables shall include the amount estimated as payable to be paid pursuant to Section 2.4(a) and Section 2.4(b), together with a summary schedule of the previous calendar month actual intra-group balances identifying payables and receivables by business unit on which the estimates are based, and the adjustment for activity made to the previous calendar month actual intra-group balances in calculating the Estimated Intra-Group Balances.
- (c) In the event that following the delivery of the estimates described in this Section 2.2 the Closing does not occur on the first Business Day of the following calendar month, such estimates shall be deemed null and void for all purposes of this Stock Purchase Agreement.

2.3 Payment of the Initial Purchase Price

On the Closing Date, Sweden Buyer shall pay, or cause the payment of, the Initial Purchase Price less the Goodwill Deposit and the Seller Financing Amount (the "Net Initial Purchase Price") by wire transfer in immediately available funds into a bank account of Parent (which account shall have been designated by Parent no later than two Business Days prior to the Closing Date). Such amount shall be paid in accordance with Section 12.5. The part of the Initial Purchase Price as is constituted by the Goodwill Deposit will be satisfied by release of the Goodwill Deposit to Parent by the Escrow Agent in accordance with Section 2.8(b) and the terms of the Escrow Agreement. The part of the Initial Purchase Price as is constituted by the Seller Financing Amount will be satisfied by execution, delivery and performance of the Seller Financing Agreement.

2.4 Discharge of Intra-Group Balances

- (a) On the Closing Date:
- (i) Sweden Buyer (as agent for the relevant Target Company or Company Subsidiary owing the relevant Intra-Group Payables) shall pay an amount equal to the aggregate of the Estimated Intra-Group Payables (if any) to Parent (for itself or, as applicable, as agent for and on behalf of the relevant member of Parent Group to which the relevant Intra-Group Payables are owed); and
- (ii) Parent (for itself or, as applicable, as agent for and on behalf of the relevant member of Parent Group owing the relevant Intra-Group Receivables) shall pay an amount equal to the aggregate of the Estimated Intra-Group Receivables (if any) to Sweden Buyer (as agent for the relevant Target Company or Company Subsidiary to which the relevant Intra-Group Receivables are owed).
- (b) Five Business Days following issuance of the Final Closing Statement in accordance with Schedule A-1, the following payments shall be made in respect of any Intra-Group Payables and Intra-Group Receivables:
- (i) if the actual amount of the Intra-Group Payables is greater than the Estimated Intra-Group Payables or the amount of the Intra-Group Receivables is less than the Estimated Intra-Group Receivables, then Sweden Buyer (as agent for the relevant Target Company or Company Subsidiary) shall pay to Parent (for itself or, as applicable, as agent for and on behalf of the relevant member of Parent Group) an amount equal to the difference (less any amount already settled); and

(ii) if the actual amount of the Intra-Group Receivables is greater than the Estimated Intra-Group Receivables or the amount of t
Intra-Group Payables is less than the Estimated Intra-Group Payables, then Parent (for itself or, as applicable, as agent for and on behalf of the relevant
member of Parent Group) shall pay to Sweden Buyer (as agent for the relevant Target Company or Company Subsidiary) an amount equal to the differen
(less any amount already settled), and

the Intra-Group Payables and Intra-Group Receivables shall be treated as fully paid and discharged.

- (c) Any payment required to be made pursuant to Section 2.4(b) shall be paid by Parent or Sweden Buyer (as the case may be) together with (and in substitution for any interest accruing in accordance with the applicable terms and conditions of any Intra-Group Payable or Intra-Group Receivable) an amount equal to interest at the Applicable Rate, for the period from (but excluding) the Closing Date to (and including) the date of such payment, calculated on a daily basis.
- (d) Notwithstanding anything to the contrary in Section 12.6, any sums which Parent or Sweden Buyer is obliged to pay pursuant to Section 2.4(a) shall be aggregated and set-off against each other, and the sums which each is respectively obliged to pay one another pursuant to Section 2.4(b) shall be aggregated and set-off against each other. Whichever of Sweden Buyer or Parent then remains with a payment obligation under Section 2.4(a), and separately under Section 2.4(b), shall, in each case, make the applicable payment only of the net amount payable with respect to each of Section 2.4(a) and Section 2.4(b). Where net amounts are paid or received, the book entries made in the accounts of the relevant debtor and creditor shall reflect the gross payments being effected to reflect the discharge of Intra-Group Payables and Intra-Group Receivables.
- (e) If and to the extent that any of the mechanisms in this Stock Purchase Agreement for the settlement or discharge of any sums which Parent or Sweden Buyer is obliged to pay pursuant to this Section 2.4 would cause any person to breach exchange control or equivalent restrictions in any jurisdiction, or to violate any other Law, the Parties will co-operate in good faith to agree and implement an alternate mechanism which achieves the objectives of this Stock Purchase Agreement and avoids such breach or violation.

2.5 Closing Statement Adjustments to the Purchase Price

- (a) Following Closing, Parent and Sweden Buyer shall give effect to Schedule A-1.
- (b) Five Business Days following the issuance of the Final Closing Statement in accordance with Schedule A-1, adjustments to the Initial Purchase Price shall be made by Parent (for itself and on behalf of Sweden Seller) or, as applicable, by Sweden Buyer (for itself and on behalf of U.S. Buyer) as follows:
- (i) Sweden Buyer shall pay to Parent the amount equal to the aggregate amount by which the Intra-Group Balances is greater than the Estimated Intra-Group Balances, or, as applicable, Parent shall pay to Sweden Buyer the amount equal to the aggregate amount by which the Estimated Intra-Group Balances is greater than the Intra-Group Balances;
- (ii) Parent shall repay to Sweden Buyer an amount equal to the amount by which the Net External Indebtedness is greater than the Estimated Net External Indebtedness or, as applicable,

Sweden Buyer shall pay to Parent an amount equal to the amount by which the Estimated Net External Indebtedness is greater than the Net External Indebtedness; and

(iii) Sweden Buyer shall pay to Parent the amount equal to the amount by which the Net Working Capital is greater than the Estimated Net Working Capital or, as applicable, Parent shall pay to Sweden Buyer the amount equal to the amount by which the Estimated Net Working Capital is greater than the Net Working Capital,

provided that, for the avoidance of doubt and in order to avoid duplication, no item shall be included in the calculation of more than one of (w) the Net External Indebtedness Adjustment, (x) the Net Working Capital Adjustment, (y) the Intra-Group Balances Price Adjustment or (z) the Total Pension Adjustment.

- (c) Any payment required to be made pursuant to Section 2.5(b) shall be paid by Parent or Sweden Buyer (as the case may be) together with an amount equal to interest on such payment at the Applicable Rate, for the period from (but excluding) the Closing Date, to (and including) the date of payment, calculated on a daily basis.
- (d) Notwithstanding anything to the contrary in Section 12.6, any sums which Parent and Sweden Buyer are obliged to pay to one another pursuant to this Section 2.5 shall be aggregated and set-off against each other and whichever of Parent or Sweden Buyer is left with any payment obligation shall make an applicable payment.

2.6 Pension Adjustment to the Purchase Price

- (a) Following Closing, Parent and Buyers shall give effect to Schedule A-2 to determine the Total Pension Adjustment and to give effect to any special payment under paragraph 10 of Schedule A-2.
- (b) Within five Business Days after the determination of the Total Pension Adjustment in accordance with Schedule A-2 (i) Parent (for itself and on behalf of Sweden Seller) shall repay to Sweden Buyer (for itself and on behalf of U.S. Buyer) the amount equal to the amount by which the Total Pension Adjustment is greater than the Estimated Pension Adjustment or (ii) Sweden Buyer (for itself and on behalf of U.S. Buyer) shall pay to Parent (for itself and on behalf of Sweden Seller) an amount equal to the amount by which the Estimated Pension Adjustment is greater than the Total Pension Adjustment.
- (c) Any payment required to be made pursuant to Section 2.6(b) shall be paid by Parent or Sweden Buyer (as the case may be) together with an amount equal to interest on such payment at the Applicable Rate, for the period from (but excluding) the Closing Date to (and including) the date of payment, calculated on a daily basis.

2.7 Closing

(a) Closing shall occur at 9:00 a.m. London time on the first date that is (i) the first Business Day of a calendar month and (ii) on or after the third Business Day following the satisfaction or, to the extent permitted, written waiver, of each of the conditions set forth in Article 7 and Article 8 (other than those conditions that by their nature will be satisfied at Closing, but subject to the satisfaction or waiver of such conditions), or on such other date and time as the Parties may agree in writing (the "Closing Date").

(b) Closing will occur at the offices of Hogan & Hartson in London, or at such other place as the Parties may agree in writing prior to the Closing Date.

2.8 Goodwill Deposit

- (a) Upon the date of this Stock Purchase Agreement, Sweden Buyer shall provide irrevocable instructions to deposit on March 29, 2010 with HSBC Holdings plc, London, (the "Escrow Agent") the amount of \$50,000,000 (the "Goodwill Deposit"). The maintenance in escrow and release of the Goodwill Deposit shall be governed by the terms of the Escrow Agreement, in accordance with Section 2.8(b), Section 11.2(b) or Section 11.2(c), as applicable.
- (b) Upon Closing, Parent shall, and Sweden Buyer shall cause Beijing Holdco on behalf of Sweden Buyer to, promptly execute and deliver to Escrow Agent, in accordance with the terms of the Escrow Agreement, a joint instruction to immediately release (i) the Goodwill Deposit to Parent in part satisfaction of the Initial Purchase Price by wire transfer of immediately available funds, and (ii) any interest accrued on the Goodwill Deposit to Sweden Buyer by wire transfer of immediately available funds.

3. REPRESENTATIONS AND WARRANTIES OF PARENT AND SWEDEN SELLER

Except as disclosed to Buyers in Schedule B as delivered on the date of this Stock Purchase Agreement (without regard to any subsequent amendment) (it being understood that, in each case, (a) a disclosure against a specific representation or warranty shall be deemed a disclosure against any other representation or warranty in this Article 3 to which such disclosure may relate, if the applicability of a disclosure to such other representation or warranty is reasonably apparent, and (b) no reference to, or disclosure of, an item on Schedule B shall be construed as an admission or indication that such item or other matter is material or that such item or other matter is required to be referred to or disclosed on Schedule B), Parent and Sweden Seller represent and warrant to U.S. Buyer and Sweden Buyer, as of the date of this Stock Purchase Agreement and as of Closing or at such time as is otherwise specified in any Section of this Article 3, as follows:

3.1 Organization and Standing; Target Companies; Company Subsidiaries

- (a) Each of Parent, Sweden Seller, U.S. Company and Sweden Company is duly organized, validly existing and in good standing under the Law of its respective jurisdiction of incorporation or formation, as applicable. Each Target Company has the requisite corporate power and authority to own or lease and to operate its assets and to carry on its respective business as currently conducted.
- (b) Each Target Company is duly qualified to conduct business as a foreign company and is in good standing in each of the jurisdictions where the nature of the business conducted by such Target Company or the character of the assets owned, leased or otherwise held by such Target Company makes such qualification necessary, except to the extent the absence of such qualification or good standing would not, individually or in the aggregate, reasonably be expected to be material to the Business.
- (c) Schedule B Part 3.1(c) sets forth a true, complete and correct list of the Company Subsidiaries and sets forth with respect to each Company Subsidiary the jurisdiction of formation.

- (d) The Target Companies and the Company Subsidiaries do not own, directly or indirectly, any capital stock, membership interest, partnership interest, joint venture interest or other equity interest in any other Person, except for (i) passive holdings of less than 5% of the share capital in listed or non-listed entities, (ii) the Company Subsidiaries, and (iii) as set forth on Schedule B Part 3.1(d).
- (e) Except as set forth on Schedule B Part 3.1(e), each Company Subsidiary is duly organized, validly existing and in good standing under the Law of its respective jurisdiction of incorporation or formation, as applicable, and has the requisite corporate power and authority to own or lease and to operate its assets, and to carry on its respective business as currently conducted.
- (f) Except as set forth on Schedule B Part 3.1(e), each Company Subsidiary is duly qualified to conduct business as a foreign company and is in good standing in each of the jurisdictions where the nature of the business or the character of the assets owned, leased or otherwise held by such Company Subsidiary makes such qualification necessary, except to the extent the absence of such qualification or good standing would not, individually or in the aggregate, reasonably be expected to be material to the Business.

3.2 Authority and Capacity; No Conflicts; Binding Obligation; No Liquidation

- (a) Each member of Parent Group has the requisite corporate power and authority to enter into the Transaction Documents to which it is a party and to consummate the transactions contemplated by the Transaction Documents.
- (b) Subject to satisfaction of the Governmental Approvals and the Buyer Governmental Approvals, the execution, delivery and performance by each member of Parent Group of the Transaction Documents to which it is party, and the consummation by each member of Parent Group and Target Companies and Company Subsidiaries, as applicable, of the transactions contemplated by such Transaction Documents, have been duly authorized by all requisite action of the relevant member of Parent Group, Target Company and Company Subsidiary, with such authorization in full force and effect, and do not:
- (i) conflict with or violate (A) any material Law in any material respect or (B) a provision of the Formation Document or the Bylaws of any member of Parent Group, a Target Company or Company Subsidiary; or
- (ii) conflict with, result in a breach of, constitute a default under, give rise to a right of prepayment, termination, cancellation or acceleration of an obligation or loss of a material benefit or to increased, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of an Encumbrance upon assets under, any Material Contract, except to the extent that such conflict, breach, default, prepayment, termination, cancellation, acceleration of obligation, loss of material benefit, increase, acceleration or guaranty of rights or entitlements or creation of Encumbrance would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impede the ability of a member of Parent Group to consummate the transactions contemplated by the Transaction Documents.
- (c) This Stock Purchase Agreement has been duly executed and delivered by Parent and Sweden Seller and, at Closing, each member of Parent Group will have duly executed and delivered each Separation Agreement to which it is a party. Assuming the due authorization, execution and delivery by

- U.S. Buyer and Sweden Buyer of this Stock Purchase Agreement, this Stock Purchase Agreement constitutes, and the Separation Agreements when executed will constitute, legal, valid and binding obligations of each such member of Parent Group, enforceable against each such member of Parent Group, in each case in accordance with their respective terms, except as such enforceability may be subject to the effects of Law.
- (d) Except as set forth on Schedule B Part 3.2(d), since December 31, 2008, with respect to each of Parent, Sweden Seller, each Target Company and each Company Subsidiary:
- (i) to the Knowledge of Parent, no order, whether executive, legislative or judicial, judgment, ruling, award, decree, directive, writ, injunction, investigation or proceedings by a Governmental Entity (an "Order") has been entered, no resolution has been adopted, and no petition has been presented to a court of competent jurisdiction for the purpose of its winding up or liquidation;
- (ii) no provisional liquidator, manager, administrator, conservator or administrative receiver, or equivalent in its jurisdiction of incorporation or formation, has been appointed, including with respect to all or substantially all of its assets, and, to the Knowledge of Parent, no petition or application for such appointment has been filed or presented to a court of competent jurisdiction;
 - (iii) no assignment of all or substantially all of its assets has been made in favor of its creditors; and
- (iv) to the Knowledge of Parent, no proceeding under bankruptcy, insolvency, reorganization, moratorium or other similar Law has been initiated.

3.3 Governmental Approvals

- (a) Other than the approvals, consents, authorizations, filings, waiting periods terminations or expirations and notices set forth on Schedule C (the "Governmental Approvals"), no approval, consent, authorization or Permit from, or filing or notice to, a Governmental Entity, and no expiration or termination of a mandatory waiting period by a Governmental Entity, is required in connection with the execution and delivery by Parent and Sweden Seller of this Stock Purchase Agreement or the execution by the relevant members of Parent Group, the Target Companies and Company Subsidiaries of the Separation Agreements to which each is a party or the consummation by the relevant members of Parent Group, the Target Companies and Company Subsidiaries of the transactions contemplated by the Transaction Documents, except for such approvals, consents, authorizations, Permits, filings, notices, or expirations or terminations (i) the failure of which to obtain, make, submit or occur would not be material, or (ii) as are identified by U.S. Buyer and Sweden Buyer on Schedule S (the "Buyer Governmental Approvals").
- (b) To the Knowledge of Parent, there is no fact or circumstance related to any member of Parent Group, the Target Companies or Company Subsidiaries that would reasonably be expected to delay the receipt or satisfaction of the Governmental Approvals or the expiration or termination of any waiting periods required under any of the Governmental Approvals.

3.4 Formation Documents and Bylaws; Books and Records

- (a) Except as set forth on Schedule B Part 3.4(a), Parent and Sweden Seller have made available to Buyers copies of: (a) the Formation Document of each Target Company and each Company Subsidiary, as in effect on the date of this Stock Purchase Agreement; and (b) to the extent applicable, the Bylaws of each Target Company and each Company Subsidiary, as in effect on the date of this Stock Purchase Agreement.
- (b) The stock certificate and the shareholders' register of Sweden Company, the board and shareholders' meeting minutes and written board and shareholders' resolutions of Sweden Company for the period since January 1, 2006, and the board meeting minutes and written board and members' resolutions of U.S. Company for the period since January 1, 2004 (copies of the material terms which have been made available to Buyers) are true, complete and correct in all material respects.

3.5 Capitalization

- (a) The share capital of Sweden Company is registered to be at least SEK 500,000,000 divided into 500,000 shares of capital stock and at most SEK 2,000,000,000 divided into 2,000,000 shares of capital stock. The Sweden Company Shares consist of 723,530 shares of capital stock with quota value of SEK 1,000 per share all of which are duly authorized, have been validly issued, are fully paid and are owned beneficially and of record by Sweden Seller.
- (b) The capital stock of U.S. Company consists of one membership interest, which is duly authorized, has been validly issued, is fully paid and non-assessable, and is owned beneficially and of record by Parent.
- (c) All of the Sweden Company Shares were, and the U.S. Company Interest was, issued in compliance with Law. Sweden Company Shares are and were, and the U.S. Company Interest is and was, not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or similar right, Order, the Formation Document of the relevant Target Company or any Agreement to which Parent, Sweden Seller or a Target Company is party or is otherwise bound.
- (d) Schedule B Part 3.1(c) sets forth a true, complete and correct list with respect to each Company Subsidiary of the authorized and outstanding shares of capital stock, membership interest or other interest that confers the right to receive a share of the profits and losses, or distribution of assets, of the issuing entity of such Company Subsidiary and the owner(s) of record of such interests of the Company Subsidiaries. All of the issued and outstanding shares of capital stock, membership interests, joint venture interests and other equity interests of each Company Subsidiary directly or indirectly owned beneficially and of record by a Target Company, are owned free and clear of Encumbrances, and all such interests have been duly authorized and validly issued and are fully paid and, as applicable, non-assessable.
- (e) No shares of capital stock, membership interests, joint venture interests or other equity interests of a Target Company or Company Subsidiary have been reserved. There are no outstanding securities convertible into, or exchangeable for, such interests of a Target Company or Company Subsidiary, and no outstanding options, warrants, purchase rights, subscription rights or similar equity instruments or Agreements to purchase or to subscribe for such interests or would otherwise require a Target Company or Company Subsidiary to issue, sell or otherwise cause such interests to become outstanding. No Target Company or Company Subsidiary has outstanding or authorized any stock

appreciation, phantom stock, profit participation or similar rights. No Target Company or Company Subsidiary has outstanding any bonds, debentures, notes or other obligations the holders of which have the right to vote (or are convertible into, or exercisable or exchange for, securities or interests having the same right to vote) on any matter. No Target Company or Company Subsidiary has entered into a legally binding obligation to sell or issue convertible securities, options, rights, warrants, purchase rights, subscription rights or similar equity instruments, except as contemplated by this Stock Purchase Agreement. There are no Agreements affecting or relating to the voting, issuance, purchase, redemption, repurchase or transfer of capital stock, membership interests, joint venture interests or other equity interests of a Target Company or Company Subsidiary.

3.6 Ownership and Transfer of the Sweden Company Shares and the U.S. Company Interest

Except as set forth on Schedule B Part 3.6, Parent has good, valid and marketable title, free and clear of Encumbrances, to the U.S. Company Interest, with full right and lawful authority to sell, transfer, assign and deliver the U.S. Company Interest to U.S. Buyer pursuant to this Stock Purchase Agreement. Sweden Seller has good, valid and marketable title, free and clear of Encumbrances, to the Sweden Company Shares, with full right and lawful authority to sell, transfer, assign and deliver the Sweden Company Shares to Sweden Buyer pursuant to this Stock Purchase Agreement. Upon delivery of and payment for the U.S. Company Interest and the Sweden Company Shares at Closing pursuant to this Stock Purchase Agreement, U.S. Buyer and Sweden Buyer will acquire good, valid and marketable title to the U.S. Company Interest and the Sweden Company Shares respectively, free and clear of Encumbrances, other than Encumbrances arising solely as a result of actions by U.S. Buyer or Sweden Buyer or as a result of a breach of any of Buyers' representations and warranties set forth in Article 4.

3.7 Directors and Officers

A true, complete and correct list of the current directors and officers, or as applicable, senior management, of each Target Company and each Company Subsidiary is set forth on Schedule B Part 3.7, including each such person's name and position(s).

3.8 Financial Statements and Liabilities

- (a) Set forth on Schedule AA are complete and correct copies of: (i) the audited consolidated balance sheet of Target Companies and Company Subsidiaries as of December 31, 2007 and December 31, 2008 (the balance sheet as of December 31, 2008 only, the "Company Balance Sheet"), (ii) the audited consolidated income statements of Target Companies and Company Subsidiaries for each of the three years in the period ended December 31, 2008 and (iii) the audited consolidated cash flow statements of Target Companies and Company Subsidiaries for each of the three years in the period ended December 31, 2008, together with all notes thereto, and an opinion or report of the independent auditors to the Volvo Automotive Combined Operating Group, certifying such financial statements (collectively, the "Company Financial Statements" and to the extent such financial statements are as of December 31, 2008 or for the one year ended December 31, 2008, the "2008 Financial Statements").
- (b) The Target Companies' and Company Subsidiaries' system of internal controls over the financial reporting for the Business is sufficient, in all material respects, to provide reasonable assurance that transactions are recorded as necessary to permit preparation of the Company Financial Statements in conformity with U.S. GAAP.

- (c) The Company Financial Statements (i) have been prepared in accordance with the books and records of each Target Company and each Company Subsidiary and (ii) present fairly, in all material respects, the financial position on a consolidated basis as of December 31, 2007 and December 31, 2008 and the results of operations and cash flows of Target Companies and Company Subsidiaries on a consolidated basis for each of the three years in the period ended December 31, 2008 in conformity with U.S. GAAP.
- (d) Schedule CC sets forth, in all material respects, a list of specified adjustments made to the 2008 Financial Statements in the preparation of the Transaction Financial Statements.
- (e) The Interim Financial Statements have been prepared on the same basis and using the accounting principles, policies, procedures and practices consistently applied, including in relation to accounting discretion and judgment, as used in the Transaction Financial Statements.
- (f) Each of the audited statutory accounts for the Statutory Audit Companies for the period ended December 31, 2008 (copies of which have been made available to Buyers) (i) has been prepared in all material respects in accordance with applicable Local GAAP and Law on a basis consistent with the prior period and (ii) presents fairly, in all material respects, the financial position of the relevant entity as of December 31, 2008 and of its profits and losses for the period ended December 31, 2008.
- (g) The Target Companies and Company Subsidiaries do not have any Liabilities of a nature required to be reflected on a balance sheet prepared in accordance with U.S. GAAP, except (i) as disclosed, reflected or reserved against in the Company Balance Sheet and (ii) for Liabilities and obligations that are not individually or in the aggregate material to the Business and were incurred in the Ordinary Course of Business since December 31, 2008
- (h) The Target Companies and Company Subsidiaries do not have, and for the 18 months following the Closing Date will not have, any Liabilities of a nature required to be reflected on a balance sheet prepared in accordance with U.S. GAAP (including accruals) relating to any business conducted by a Target Company or Company Subsidiary which business has been sold or discontinued in the five-year period prior to the date of this Stock Purchase Agreement (a "**Discontinued Business Liability**"), except as disclosed, reflected or reserved against in the Company Balance Sheet. For the purposes of this Section 3.8(h), whether a Discontinued Business Liability is of a nature required to be reflected on a balance sheet prepared in accordance with U.S. GAAP shall be determined with reference to the financial position of the Business immediately prior to Closing.

3.9 Title to and Sufficiency of Assets

- (a) Excluding for the purposes of this Section 3.9(a) all Owned Real Property, Leased Real Property, and Real Property Leases and Intellectual Property Rights: (i) a Target Company or a Company Subsidiary has good and valid title to, or a valid leasehold interest in, the Assets free and clear of all Encumbrances except for Permitted Encumbrances; (ii) all leases with respect to leasehold interests are in full force and effect and constitute valid and binding obligations of each other party thereto and none of the Target Companies, the Company Subsidiaries or any other party thereto is in breach of any of the terms of such lease; and (iii) all leased personal property used by, or in connection with, the Business is in the condition required of such property by the terms of the applicable lease Agreement.
- (b) To the Knowledge of Parent, the Material Real Property Improvements and all items of tangible personal property which are owned, leased or used by the Target Companies and the Company

Subsidiaries and are material to the Business are structurally sound, free from material defect (subject to normal wear and tear given the use and age of such assets), have been maintained in all material respects in accordance with generally accepted industry practice and are usable in the Ordinary Course of Business.

(c) Assuming receipt of all of the Governmental Approvals and Buyer Governmental Approvals, the rights, properties and assets (including Intellectual Property Rights, real property, IT Applications and IT Infrastructure and Permits) of the Target Companies and Company Subsidiaries and the facilities, services and Intellectual Property Rights, IT Applications and IT Infrastructure to which the Target Companies and Company Subsidiaries have a contractual right, including pursuant to the Transaction Documents, together include all rights, properties, assets, facilities and services that are necessary for Target Companies and Company Subsidiaries to carry on the Business immediately after Closing in all material respects in the manner in which it is carried on as of the date of this Stock Purchase Agreement.

3.10 Conduct of Business; Absence of Certain Changes or Events

Except as contemplated by this Stock Purchase Agreement, as required by Law or as set forth on Schedule B Part 3.10, since the date of the Interim Financial Statements through the date of this Stock Purchase Agreement:

- (a) each Target Company and each Company Subsidiary has conducted its respective business only in the Ordinary Course of Business;
- (b) there has not been any Company Material Adverse Effect that is continuing; and
- (c) no action has been taken by Parent, Sweden Seller, any Target Company or any Company Subsidiary that would, if taken after the date of this Stock Purchase Agreement, constitute a breach of Section 5.3.

3.11 Material Contracts

- (a) The term "Material Contracts" shall mean each of the following Agreements, excluding the IP Contracts, to which a Target Company or Company Subsidiary is a party (except as otherwise provided in this Section 3.11):
 - Purchasing Agreements;
- (ii) Agreements (other than Intercompany Agreements) under which such Target Company or Company Subsidiary has directly or indirectly, other than in the Ordinary Course of Business (A) borrowed, or agreed to borrow, money from, or issued, or agreed to issue, any note, bond, debenture or similar instrument or otherwise incurred any indebtedness to, any Person (other than a Target Company or Company Subsidiary), with an aggregate liability in excess of \$3,000,000, or (B) made, or agreed to make, any advance, loan, extension of credit or capital contribution to, or other investment of debt or equity in, any Person (other than a Target Company or Company Subsidiary) for any Agreement under which a Target Company or Company Subsidiary has an aggregate liability in excess of \$3,000,000;

- (iii) Agreements for any joint venture, partnership or similar arrangement or an Agreement relating to holding, voting or transferring any securities or other interest by a Target Company or Company Subsidiary (including any stockholders' or members' agreement);
- (iv) Agreements relating to an acquisition to be made by such Target Company or Company Subsidiary of an operating business or the capital stock, membership interest, joint venture interest or other equity interest of another Person, in each case for consideration in excess of \$3,000,000;
- (v) Agreements relating to the sale or purchase of fixed assets or real estate having a value individually in excess of \$5,000,000, under which such sale or purchase has not been consummated;
- (vi) employment Agreements that have an aggregate annual future liability in excess of \$1,000,000, except any Agreement relating to the employment of any Designated Employee identified in schedule A to the Secondment Agreements;
 - (vii) Local Collective Agreements;
- (viii) covenants not to compete or other covenants restricting the development, manufacture, marketing or distribution of the products and services of a Target Company or Company Subsidiary which restrictions adversely and materially affect the Business, or any Agreement containing non-solicitation provisions or otherwise limiting or prohibiting a Target Company or Company Subsidiary from conducting the Business;
- (ix) Agreements with any current or former officer, director (or nominee for director) or employee (other than indemnification, employment-related Agreements);
 - (x) Surviving Contracts which are not Transaction Documents;
- (xi) leases, subleases or similar Agreements with any Person (other than a Target Company or Company Subsidiary) under which such Target Company or Company Subsidiary is a lessor or sublessor of, or makes available for use to any Person, any Owned Real Property, Leased Local Real Property or Leased Foreign Real Property or any portion of any premises otherwise occupied by such Target Company or Company Subsidiary, in each case that are material to the operation of the Business;
- (xii) Agreements under which (A) any Person (other than a Target Company or Company Subsidiary) has, directly or indirectly, guaranteed indebtedness, liabilities or obligations of a Target Company or Company Subsidiary or (B) any Target Company or Company Subsidiary has, directly or indirectly, guaranteed indebtedness, liabilities or obligations of any Person (other than a Target Company or Company Subsidiary), for each Agreement under which the aggregate liability is in excess of \$3,000,000;
- (xiii) Agreements granting an Encumbrance (other than a Permitted Encumbrance) upon any Asset the aggregate value of which exceeds \$5,000,000:

- (xiv) without duplication of any other subsection of this Section 3.11(a), Agreements providing for indemnification by any Target Company or Company Subsidiary to any Person except for any Agreement related to, or arising out of, a commercial arrangement consistent with past practice and entered into in the Ordinary Course of Business, in each case that are material to the operation of the Business;
- (xv) confidentiality or non-disclosure Agreements (other than confidentiality Agreements or provisions included in standard employment, contractor or consultancy Agreements or standard terms and conditions, inclusive of standard purchase order terms and conditions, of a Target Company or Company Subsidiary), in each case that are material to the operation of the Business;
- (xvi) without duplication of Section 3.11(a)(v), Agreements for the sale of any asset of a Target Company or Company Subsidiary (other than inventory sales in the Ordinary Course of Business) or the grant of any preferential rights to purchase any such asset (including any discounted purchase price) or requiring the consent of any Person to transfer thereof, in any such case involving assets the aggregate value of which exceeds \$5,000,000;
- (xvii) hedging or factoring Agreements to which Sweden Company is a party related to currency exchange, interest rates, commodity prices or similar Agreement;
- (xviii) without duplication of the provisions in Section 3.11(a)(i) to Section 3.11(a)(xvii), Agreements involving expected aggregate annual payments by or to any Target Company or Company Subsidiary of more than \$3,000,000, except for any Agreement related to, or arising out of, a commercial arrangement consistent with past practice and entered into in the Ordinary Course of Business; and
- (xix) Agreements other than as set forth above to which any Target Company or Company Subsidiary is a party pursuant to which a Target Company or Company Subsidiary receives a benefit or has an obligation to perform that is (A) material to the Business, including any such Agreements between a Target Company or Company Subsidiary and CFMA, or (B) if breached, terminated or not renewed would reasonably be expected to materially impair the operation of the Business.
- (b) Except as set forth on Schedule B Part 3.11(b), each of the Production Agreements is governed by the Production Purchasing Global Terms and Conditions.
- (c) (1) Schedule B Part 3.11(c)(i) sets forth a true, complete and correct list of the Shared Contracts which are Material Contracts to which a member of Parent Group is a party and pursuant to which a Target Company or Company Subsidiary is a party or receives a benefit or has an obligation to perform;
- (ii) Schedule B Part 3.11(c)(ii) sets forth a true, complete and correct list of the Shared Contracts which are Material Contracts to which a Target Company or Company Subsidiary is a party and pursuant to which a member of Parent Group is a party or receives a benefit or has an obligation to perform.

All Material Contracts are valid, binding and in full force and effect and are enforceable by the applicable Target Company or Company Subsidiary in accordance with their respective terms, except as such enforceability may be subject to the effects of Law. The applicable Target Company or Company Subsidiary that is a party to each Material Contract has performed in all material respects all obligations required to be performed by it to date under such Material Contract and neither it nor, to the Knowledge of Parent, any other party to any Material Contract (with or without the lapse of time or the giving of notice, or both) is in default or breach in any material respect of any Material Contract. Except as set forth on Schedule B Part 3.11(d), to the Knowledge of Parent, since December 31, 2008, no Target Company or Company Subsidiary has received written notice of termination or non-renewal, or intention to terminate or to not renew from any other party to a Material Contract. The material terms of the Material Contracts as in effect on the date of this Stock Purchase Agreement, together with all amendments thereto, have been made available to Buyers.

3.12 Taxes

Except as set forth on Schedule B Part 3.12:

- (a) All Tax Returns required to have been filed by, or with respect to, a Target Company or Company Subsidiary have been timely filed, taking into account any extension of time to file that has been granted or obtained, and all such Tax Returns have been duly, completely and accurately prepared in all material respects. All Taxes shown to be due and payable on such Tax Returns have been paid. All material records required to be maintained for Tax purposes by or with respect to a Target Company or Company Subsidiary have been maintained.
 - (b) Each Target Company and each Company Subsidiary has paid all material Taxes due and payable by each of them.
- (c) No Tax on any Target Company or Company Subsidiary arose in respect of, by reference to or in consequence of the Reorganization that has not been paid prior to the date of this Stock Purchase Agreement.
- (d) Each Target Company and each Company Subsidiary has withheld and has timely paid to the appropriate Governmental Entity all material Taxes required to have been withheld.
- (e) No material deficiency or material Claim for Tax has been asserted or assessed by a Governmental Entity against or with respect to a Target Company or Company Subsidiary that has not been satisfied by payment, settled (such settlement having been satisfied by payment) or withdrawn.
 - (f) There are no Tax liens on the assets of a Target Company or Company Subsidiary, except for Permitted Encumbrances.
- (g) To the Knowledge of Parent, there are no outstanding waivers or Agreements extending the statute of limitations for any period with respect to Taxes to which a Target Company or Company Subsidiary may be subject.
- (h) To the Knowledge of Parent, no Target Company or Company Subsidiary is the beneficiary of an extension of time within which to file a Tax Return.

- (i) The accruals for deferred Taxes and reserves for Taxes are accurately reflected on the Company Financial Statements in accordance with U.S. GAAP as of December 31, 2008.
- (j) Except with respect to the Reorganization, since December 31, 2008, no Target Company or Company Subsidiary has been involved in any transaction which has given or would reasonably be expected to give rise to a liability to Tax on any Target Company or Company Subsidiary (or would have given or would reasonably be expected to give rise to such a liability but for the availability of any Relief) other than Tax arising from transactions entered into by it in the Ordinary Course of Business.
- (k) No Target Company or Company Subsidiary has a material legal or contractual obligation that will be binding on such Target Company or Company Subsidiary following Closing to pay the Taxes of any other Person, other than a Target Company or a Company Subsidiary.
- (l) No Target Company or Company Subsidiary has received a Tax ruling or entered into a Tax agreement with a Governmental Entity which ruling or agreement is, by its terms, applicable to Tax periods following Closing and binding upon such Target Company or such Company Subsidiary on the Closing Date.
- (m) To the Knowledge of Parent, no Target Company or Company Subsidiary is, or has, since January 1, 2005, been subject to an examination or audit by any Tax Authority (including any formal enquiry under a statutory procedure comparable to that which is conducted in the United Kingdom by Her Majesty's Revenue and Customs), other administrative proceeding, dispute or litigation with respect to the determination of any Tax liability. To the Knowledge of Parent, no Tax Authority has provided notice that it will institute an examination or audit of a Target Company or Company Subsidiary that has not yet commenced.
- (n) No transaction in respect of which any consent, ruling, confirmation or clearance (for purposes of this Section 3.12(n), each a "ruling") was required or sought from any Tax Authority has been entered into or carried out by any Target Company or Company Subsidiary without such ruling having first been properly obtained. All information supplied to any Tax Authority in connection with any such ruling fully and accurately disclosed all facts and circumstances material to the giving of such ruling. Any transaction for which such ruling was obtained has been carried out in accordance with the terms of such ruling and the application on which the ruling was based and at a time when such ruling was valid and effective. No facts or circumstances have arisen since any such ruling was obtained which would cause the ruling to become invalid or ineffective.
- (o) No Tax Authority has operated or agreed to operate any special arrangement (being an arrangement which is not based on relevant legislation or any published practice) in relation to any Target Company's or Company Subsidiary's affairs which arrangement has or would be reasonably expected to have a material impact on the tax liabilities of, or tax practices or policies of, a Target Company or Company Subsidiary.
- (p) Each Target Company or Company Subsidiary is and has at all times been resident for Tax purposes in the country pursuant to the Laws of which (or the Laws of a political subdivision of which) it has been formed, and is not and has not at any time been treated as resident in any other country for any Tax purpose (including any double taxation arrangement). No Target Company or Company Subsidiary is subject to Tax in any country other than the country pursuant to the Laws of which (or the Laws of a political subdivision of which) it has been formed by virtue of having a permanent establishment or other place of business in that country.

- (q) No Target Company or Company Subsidiary is liable for any Tax as the agent of any other Person or business or constitutes a permanent establishment of any other Person, business or enterprise for any Tax purpose.
 - (r) Each Target Company or Company Subsidiary:
- (i) to the extent required by VAT legislation, is registered for the purposes of VAT in its jurisdiction of incorporation and in any other jurisdiction and has been so registered at all times that it has been required to be registered by VAT legislation, and such registration is not subject to any conditions imposed by or agreed with the relevant Tax Authority;
 - (ii) has complied in all material respects with the terms of VAT legislation;
- (iii) is not, and has not been treated as, a member of a group that includes a Person that is neither a Target Company nor a Company Subsidiary for the purposes of VAT legislation, and has not applied for such treatment;
 - (iv) has not been required to provide any security in respect of any amount of VAT; and
- (v) has maintained and obtained at all times and in all material respects complete, correct and up-to-date records, invoices and other documents (as the case may be) requisite for the purposes of VAT legislation and has preserved such records, invoices and other documents in such form and for such periods as are required by VAT legislation.
 - (s) Each Target Company or Company Subsidiary:
- (i) has complied in all material respects with the terms of Customs Duties legislation, including any applicable registration procedures;
- (ii) has maintained and obtained at all times and in all material respects complete, correct and up-to-date records, invoices and other documents (as the case may be) requisite for the purposes of Customs Duties legislation and has preserved such records, invoices and other documents in such form and for such period as are required by Customs Duties legislation;
- (iii) has not been required to provide any security in respect of any amount of Customs Duties, beyond customary postings of a surety bond or similar security with respect to customary import activities as may be required by a Governmental Entity with jurisdiction over importation activities.
- (t) All material VAT and Customs Duties payable by any Target Company or Company Subsidiary upon the supply, acquisition, use or importation of any assets imported or owned by such Target Company or Company Subsidiary, or any services supplied to or by it, has been paid in full.
- (u) In respect of all documents which establish or are necessary to establish the title of any Target Company or Company Subsidiary to any asset, or by virtue of which any Target Company or Company Subsidiary has any right, all applicable stamp duties or registration charges or similar duties or charges have been duly paid.

- (v) Since the finalization of the 2008 Financial Statements, no intentional action has been taken by Parent or Parent Group for the principal purpose of reducing the net operating loss carryforwards for Swedish Tax purposes of a Target Company or Company Subsidiary; provided, however, that notwithstanding the foregoing, none of the following shall be a breach of this Section 3.12(v): (A) tax-deductible group contributions by Parent Group to Sweden Company in an aggregate amount of no more than SEK 150,000,000 subsequent to the finalization of the 2008 Financial Statements, (B) actions taken by any Target Company or Company Subsidiary in the Ordinary Course of Business (provided that any tax-deductible group contributions entered into between Parent Group and any Target Company or Company Subsidiary shall not be considered to be in the Ordinary Course of Business for the purposes of this Section 3.12(v)), (C) actions taken by Parent or Parent Group in connection with the NSC Reorganization or (D) actions taken as contemplated by the Transaction Documents.
 - (w) U.S. Company is disregarded as an entity separate from Parent for U.S. federal income tax purposes.

3.13 Real Property

- (a) Schedule B Part 3.13(a) sets forth a true, complete and correct list of the Owned Real Property.
- (b) Schedule B Part 3.13(b) sets forth a true, complete and correct list of the Local Real Property Leases, the properties to which the Local Real Property Leases relate (the portion of such properties demised under the Local Real Property Leases, the "Leased Local Real Property"), the Foreign Real Property Leases, and the properties to which the Foreign Real Property Leases relate (the portion of such properties demised under the Foreign Real Property Leases, the "Leased Foreign Real Property").
- (c) Except as set forth on Schedule B Part 3.13(c), (i) each Target Company or Company Subsidiary has good and marketable fee simple title, registration of ownership in the applicable land register (or has equivalent applicable evidence of ownership) to the Owned Real Property and owns the Material Real Property Improvements located thereon, free and clear of Encumbrances, except for Permitted Encumbrances; (ii) no Person other than a Target Company or Company Subsidiary has a right, other than pursuant to a Permitted Encumbrance, to use, occupy or lease all or any portion of the Owned Real Property, except as would not materially impair the ability of the applicable Target Company or Company Subsidiary to use such Owned Real Property in the Business immediately after Closing; (iii) to the Knowledge of Parent, none of the Material Real Property Improvements encroaches in any material respect on any easement, right of way or other encumbrance which burdens any portion of the Owned Real Property in any way which would materially impair the current use of the relevant Owned Real Property; (iv) to the Knowledge of Parent, no structures, facilities or other improvements on any parcel adjacent to the Owned Real Property encroach in any material respect onto the Owned Real Property in any way which would materially impair the current use of the relevant Owned Real Property; (v) to the Knowledge of Parent, Parent or Sweden Seller has delivered to Buyers true, complete and correct copies of all of the deeds and other instruments in possession of Parent, Sweden Seller, a Target Company or Company Subsidiary relating to the Owned Real Property, and copies of all title insurance policies, opinions, abstracts and surveys in possession of Parent, Sweden Seller, a Target Company or Company Subsidiary; and (vi) there are no outstanding options or rights of first refusal to purchase any Owned Real Property or any portion thereof or interest therein.

- Except as set forth on Schedule B Part 3.13(d), true, complete and correct copies of the Real Property Leases have been made available to Buyers. Except as set forth on Schedule B Part 3.13(d), and otherwise in accordance with the terms and conditions of the applicable Real Property Leases: (i) a Target Company or Company Subsidiary has a valid lease or leasehold interest, or an equivalent applicable interest, to the Leased Real Property, free and clear of Encumbrances, except for Permitted Encumbrances; (ii) to the Knowledge of Parent, no structures, facilities or other improvements on any parcel adjacent to the Leased Real Property encroach in any material respect onto the Leased Real Property in any way which would materially impair the current use by the relevant Target Company or Company Subsidiary of the relevant Leased Real Property; (iii) to the Knowledge of Parent, Target Company or Company Subsidiary has registration of ownership in the applicable land register (or has equivalent applicable evidence of ownership) in relation to the Leased Real Property located in Belgium; and (iv) no Person other than a Target Company or Company Subsidiary has a right, other than pursuant to a Permitted Encumbrance, to use, occupy or lease all or any portion of the Leased Real Property, except as would not materially impair the ability of the applicable Target Company or Company Subsidiary to use such Leased Real Property in the operation of the Business.
- (e) Other than pursuant to a Permitted Encumbrance, to the Knowledge of Parent, no Target Company or Company Subsidiary has served or received written notice of any subsisting default or breach in any material respect of any Real Property Lease, except as would not materially impair the current use and/or occupation of the relevant Leased Real Property. Other than pursuant to a Permitted Encumbrance, no Target Company or Company Subsidiary has assigned its interest under a Real Property Lease, or subleased all or a portion of the space demised thereby, to a third party.
- (f) There is no pending or, to the Knowledge of Parent, threatened in writing, condemnation of any portion of the Owned Real Property or the Leased Real Property by a Governmental Entity, except as would not materially impair the ability to use and/or occupy such Owned Real Property or Leased Real Property in the operation of the Business. Except as set forth on Schedule B Part 3.13(f), there is no pending or, to the Knowledge of Parent, threatened in writing fire, health, safety, building, zoning or other land use regulatory proceeding relating to any portion of the Owned Real Property or Leased Real Property by a Governmental Entity, except as would not materially impair the ability to use and/or own such Owned Real Property or Leased Real Property in the operation of the Business. No Governmental Entity having the power of eminent domain over the Owned Real Property and Leased Real Property has commenced, or to the Knowledge of Parent, intends to exercise the power of eminent domain or similar power with respect to all or any part of the Owned Real Property or Leased Real Property.
- (g) Except as set forth on Schedule B Part 3.13(g), the Owned Real Property and Leased Real Property have adequate rights of way and access to public ways and public utilities and such other access rights as are necessary for the operation of the Business at such Owned Real Property and Leased Real Property.
- (h) All water, sewer, gas, electric, drainage and other utilities required by the use and operation of the Owned Real Property and the Leased Real Property in the conduct of the businesses of the Target Companies and the Company Subsidiaries are installed, are fully operable and are adequate to service the Owned Real Property and the Leased Real Property and, to the Knowledge of Parent, there are no disputes, actions or violations that would reasonably be expected to result in termination of such services, in each case except as would not materially impair the ability to use and/or own such Owned Real Property or Leased Real Property in the conduct of the Business. To the Knowledge of Parent, there are no disputes, actions or violations that would reasonably be expected to result in termination or

material reduction in the current access from the Owned Real Property or Leased Real Property to existing roads or to utility services presently serving the Owned Real Property and Leased Real Property, except as would not materially impair the ability to use and/or own such Owned Real Property or Leased Real Property in the conduct of the Business.

3.14 Intellectual Property

- (a) Schedule Y sets forth a true, complete and correct list of all material Company Registered Intellectual Property and the registered owner(s) of each such Intellectual Property Right.
- (b) A member of Parent Group owns all rights and interests in, and has title to, the Parent Assigned Intellectual Property and the Parent Licensed Intellectual Property, except (i) as set forth in Schedule B Part 3.14(g) and Schedule B Part 3.14(v), and (ii) to the extent any Parent Assigned Intellectual Property or Parent Licensed Intellectual Property is jointly owned by a third party. Schedule B Part 3.14(b) lists joint owners of the material Parent Assigned Intellectual Property.
- (c) As of the Closing Date, the Ford Entities are entitled to assign, or procure the assignment by a member of Parent Group of, the Parent Assigned Intellectual Property in accordance with the Intellectual Property Assignment Agreement and to grant the rights to use the Parent Licensed Intellectual Property in accordance with the Intellectual Property License Agreement.
- (d) A Target Company or Company Subsidiary owns all of the rights and interests in, and has title to, the Company Intellectual Property (excluding the Non-Controlled IP), and a Non-Controlled Company owns all of the rights and interest in and has title to the Non-Controlled IP, in each case except (i) as set forth in Schedule B Part 3.14(g) and Schedule B Part 3.14(v), (ii) with respect to any Non-Controlled IP (other than the Non-Controlled Volvo Brand IP), where the failure to have such rights and interest would not reasonably be expected to have a material adverse effect on the Business; and (iii) to the extent any Company Intellectual Property is jointly owned by a third party. Schedule Y lists joint owners of the material Company Registered Intellectual Property.
- (e) The Intellectual Property Rights to be licensed to Buyer Group under the Intellectual Property License Agreement constitute all of the Intellectual Property Rights that are owned by members of Parent Group or, with respect to the Gemini Engine IP, the PSA Group, with respect to the technologies specified in the Documented Plans.
- (f) As of the Closing Date, Target Company, Company Subsidiary, a member of Parent Group, or a Non-Controlled Company has paid all application, filing, registration, maintenance and renewal fees due in respect of: (i) the Company Registered Intellectual Property and the Non-Controlled Volvo Brand IP; (ii) to the Non-Controlled IP Knowledge of Parent, the registered Non-Controlled IP (other than the Non-Controlled Volvo Brand IP); (iii) the registered Parent Assigned Intellectual Property; and (iv) the registered Parent Licensed Intellectual Property, except in each case, to the extent that such Target Company, Company Subsidiary, member of Parent Group or a Non-Controlled Company has determined not to prosecute, maintain or renew, any Intellectual Property Rights which are not material to the Business.
- (g) Except as set forth on Schedule B Part 3.14(g), to the Knowledge of Parent and, with respect to the Non-Controlled IP, to the Non-Controlled IP Knowledge of Parent, there is no Claim (including opposition, cancellation and reexamination proceedings) filed with a Governmental Entity and pending against a Target Company, Company Subsidiary, Non-Controlled Company or member of Parent

Group challenging the registrability, validity or enforceability of: (i) any Company Registered Intellectual Property or registered Non-Controlled IP; (ii) any registered Parent Assigned Intellectual Property; or (iii) any registered Parent Licensed Intellectual Property. Neither the Target Companies, the Company Subsidiaries, any member of Parent Group, nor any Non-Controlled Company has received a written notice since the date that is 24 months prior to the date of this Stock Purchase Agreement that remains unresolved challenging the registrability, validity or enforceability of: (i) any Company Registered Intellectual Property; (ii) any registered Non-Controlled Volvo Brand IP, or to the Non-Controlled IP Knowledge of Parent, any other registered Non-Controlled IP; (iii) any registered Parent Assigned Intellectual Property; or (iv) any registered Parent Licensed Intellectual Property that is material to the Business, in each case except as set forth on Schedule B Part 3.14(g).

- (h) Schedule B Part 3.14(h) sets forth a list of all Material Third Party Licenses In.
- (i) Schedule B Part 3.14(i) sets forth a list of all Material Third Party Licenses Out.
- (j) The Material Third Party Licenses In and the Material Third Party Licenses Out are valid, binding and in full force and effect in accordance with their terms. No member of Parent Group, no Target Company and no Company Subsidiary that is a party to a Material Third Party License In or a Material Third Party License Out nor, to the Knowledge of Parent, any other party to a Material Third Party License In or Material Third Party License Out, is in default or breach in any material respect of any such Material Third Party License In or Material Third Party License Out.
- (k) No member of Parent Group, no Target Company and no Company Subsidiary has received written notice of any pending Claims filed with a Governmental Entity in connection with, and neither the Target Companies, the Company Subsidiaries nor any member of Parent Group has given, nor to the Knowledge of Parent received, written notice since the date that is 24 months prior to the date of this Stock Purchase Agreement of an alleged breach that remains unresolved of any Material Third Party License In or Material Third Party License Out, except as set forth on Schedule B Part 3.14(k).
- (I) No member of Parent Group, no Target Company and no Company Subsidiary that is a party to a Material Third Party License In or Material Third Party License Out has given to any other party to such agreement, or has received, written notice of termination or non-renewal, or intention to terminate or not renew any Material Third Party License In or Material Third Party License Out. Except as set forth on Schedule B Part 3.14(1), the Target Companies and Company Subsidiaries will not lose any of their respective rights under any Material Third Party License In to use any Intellectual Property Rights of third parties (i) that are used, by such Target Companies or Company Subsidiaries, or (ii) with respect to the technologies specified in the Documented Plans, in each case as a result of the execution of this Stock Purchase Agreement or the Separation Agreements or the consummation of the transactions contemplated by the Transaction Documents.
- (m) Sweden Company is not in default or breach of the Volvo Trademark Holding Shareholders' Agreement in any material respect. To the Knowledge of Parent, no other party to such Agreement is in default or breach of it in any material respect.
- (n) No member of Parent Group, no Target Company and no Company Subsidiary has received notice of any pending Claims filed with a Governmental Entity in connection with any alleged breach of the Volvo Trademark Holding Shareholders' Agreement. To the Knowledge of Parent, no party to the Volvo Trademark Holding Shareholders' Agreement has given written notice since the date that is 24 months prior to the date of this Stock Purchase Agreement of any alleged breach of such Agreement that remains unresolved.

- (o) To the Knowledge of Parent, no party to the AB Volvo License Agreement is in default or breach of that Agreement in any material respect. Neither Target Company, Company Subsidiary nor any member of Parent Group has given notice in writing to AB Volvo alleging that AB Volvo is in breach of the Volvo Trademark Holding Shareholders' Agreement or the AB Volvo License Agreement.
- (p) Schedule B Part 3.14(p)(1) sets forth a true and accurate list of mailing and merchandising catalogues used by AB Volvo and Parent Group as of March 31, 1999 (copies of which have been made available to Sweden Buyer) that, together with the merchandise products identified in Schedule B Part 3.14(p)(2), identify substantially all of the merchandise products on and in relation to which Sweden Company is licensed under the Volvo Trademark Holding License Agreement to use the Volvo Cars Licensed Trademarks (as defined in the Volvo Trademark Holding License Agreement).
- All Agreements settling Claims or proceedings, covenants not to sue and release agreements (i) to which a Target Company or Company Subsidiary is a party or (ii) to which a member of Parent Group is party and pursuant to which Target Company or a Company Subsidiary receives a license or other benefit, in each case with respect to Intellectual Property Rights used in the Business or with respect to the technologies specified in the Documented Plans ("IP Settlement Agreements") are valid, binding and in full force and effect and are enforceable by the applicable Target Company or Company Subsidiary on their respective terms, except as such enforceability may be subject to the effects of Law. No member of Parent Group, Target Company or Company Subsidiary, nor to the Knowledge of Parent, any other party to any IP Settlement Agreement, is in default or breach in any material respect of such IP Settlement Agreement. No member of Parent Group, no Target Company and no Company Subsidiary that is a party to any such IP Settlement Agreement has received written notice of termination or non-renewal, or intention to terminate or not renew, from any other party to such IP Settlement Agreement. To the Knowledge of Parent, there are no grounds on which any Target Company or Company Subsidiary will lose the benefit of any IP Settlement Agreement as a result of the execution of this Stock Purchase Agreement or the Separation Agreement or the consummation of the transactions contemplated by the Transaction Documents, except as set forth in Schedule B Part 3.14(q).
 - (r) No Compulsory Licenses have been granted and remain in effect:
 - (i) for any Company Intellectual Property (excluding the Non-Controlled IP);
 - (ii) to the Non-Controlled IP Knowledge of Parent, for any Non-Controlled IP;
 - (iii) for any Parent Assigned Intellectual Property; or
- (iv) for any Parent Licensed Intellectual Property, to the extent such Compulsory Licenses would reasonably be expected to prevent, restrict or otherwise adversely affect the rights of Buyer Group to disclose, use, exploit or sublicense the Parent Licensed Intellectual Property in accordance with the terms of the Intellectual Property License Agreement.
- (s) Except for non-exclusive licenses granted in the Ordinary Course of Business, the (i) Company Intellectual Property (excluding the Non-Controlled IP) and the Parent Assigned Intellectual Property, and (ii) the Non-Controlled Volvo Brand IP, and to the Non-Controlled IP Knowledge of Parent, any other Non-Controlled IP, in each case, are free and clear of Encumbrances.

- (t) Except for non-exclusive licenses granted in the Ordinary Course of Business, the Parent Licensed Intellectual Property is free and clear of Encumbrances that would reasonably be expected to prevent, restrict or otherwise adversely affect the rights of Buyer Group to disclose, use, exploit or sublicense the Parent Licensed Intellectual Property in accordance with the terms of the Intellectual Property License Agreement.
- (u) The Intellectual Property Rights (i) owned by or licensed to the Target Companies and the Company Subsidiaries, (ii) the Parent Assigned Intellectual Property and (iii) the licenses of Intellectual Property Rights to be granted to Buyer Group under the Intellectual Property License Agreement, together comprise all the Intellectual Property Rights existing as of the Closing Date required for the Target Companies and the Company Subsidiaries to:
- (i) fulfill their obligations under, and receive the benefit of, all the Separation Agreements and, if applicable, the Gemini Agreements (as defined in the Intellectual Property License Agreement); and
- (ii) fulfill their obligations under the JLR Agreements and the AMLL Agreements (each as defined in the Intellectual Property License Agreement) and all other material agreements with third parties.
- (v) Except as set forth on Schedule B Part 3.14(v): (i) to the Knowledge of Parent, none of the operations of the Target Companies nor the Company Subsidiaries as currently conducted infringe the Intellectual Property Rights of a third party; (ii) no member of Parent Group, no Target Company and no Company Subsidiary has received written notice of any pending Claim that has been filed with a Governmental Entity against a Target Company, Company Subsidiary or member of Parent Group asserting that the operations of the Target Companies or the Company Subsidiaries as currently conducted infringe the Intellectual Property Rights of a third party; and (iii) neither a Target Company, Company Subsidiary nor a member of Parent Group has received written notice from a third party since the date that is 24 months prior to the date of this Stock Purchase Agreement that remains unresolved alleging that the operations of the Target Companies or the Company Subsidiaries infringe the Intellectual Property Rights of a third party or otherwise disputing the rights of the Target Companies or the Company Subsidiaries to use any Intellectual Property Rights owned or used by them.
- (w) Except as set forth on Schedule B Part 3.14(w): (i) to the Knowledge of Parent (except with respect to the Non-Controlled IP, to the Non-Controlled IP Knowledge of Parent), no third party is infringing any Company Intellectual Property, any Parent Assigned Intellectual Property, or any Parent Licensed Intellectual Property that is material to the Business; (ii) there are no pending Claims filed with a Governmental Entity (A) by a Target Company, Company Subsidiary or member of Parent Group asserting infringement by a third party of any Company Intellectual Property (excluding the Non-Controlled IP), any Parent Assigned Intellectual Property, or any Parent Licensed Intellectual Property that is material to the Business; or (B) by a Non-Controlled Company asserting infringement by a third party of any Non-Controlled IP; and (iii) none of a Target Company, Company Subsidiary nor a member of Parent Group, nor, to the Non-Controlled IP Knowledge of Parent, a Non-Controlled Company, has sent written notice to a third party since the date that is 24 months prior to the date of this Stock Purchase Agreement that remains unresolved alleging that the third party is infringing any Company Intellectual Property, any Parent Assigned Intellectual Property, or any Parent Licensed Intellectual Property that is material to the Business.

- (x) To the Knowledge of Parent, transactions relating to Company Registered Intellectual Property and registered Parent Assigned Intellectual Property that are material to the Business have been registered with appropriate Governmental Entities within applicable time limits as required by Law.
- (y) The Target Companies and Company Subsidiaries have implemented and maintained industry-standard confidentiality policies and procedures to protect against unauthorized disclosure of material confidential information. The Target Companies, Company Subsidiaries and members of Parent Group have used commercially reasonable efforts but no less than a reasonable degree of care to maintain all confidential information of or used by the Target Companies and Company Subsidiaries as confidential and to limit disclosure thereof, except disclosure to third parties subject to written confidentiality obligations. To the Knowledge of Parent, there have been no disclosures of material confidential information of or used by the Target Companies and Company Subsidiaries other than in the Ordinary Course of Business and subject to written confidentiality obligations.
- (z) Except as set forth on Schedule B Part 3.14(z), there are no written Agreements or, to the Knowledge of Parent, oral agreements that prevent, restrict or otherwise adversely affect (i) the disclosure, use, exploitation, sublicensing or assignment of the Company Intellectual Property or Parent Assigned Intellectual Property by Buyer Group, or (ii) the disclosure, use, exploitation or sublicensing of the Parent Licensed Intellectual Property by Buyer Group in accordance with the terms of the Intellectual Property License Agreement.
- (aa) The Target Companies and the Company Subsidiaries own or have a license to use in the Business (on terms that will allow (i) the Business to be conducted in the manner in which it is conducted as at the date of this Stock Purchase Agreement and (ii) the Documented Plans to be carried out) any Intellectual Property Rights created, developed or invented by each current or former Employee and individual contractor of a Target Company, Company Subsidiary, which Intellectual Property Rights are used in the Business or with respect to technologies specified in the Documented Plans.
- (bb) Except as set forth on Schedule B Part 3.14(bb), to the Knowledge of Parent, there are no pending Claims from any current or former Employees of a Target Company or Company Subsidiary in any jurisdiction for compensation or remuneration for patentable inventions invented or copyright works or design rights created, as used by any Target Company or Company Subsidiary.
- (cc) Records for all Company Registered Intellectual Property (except for registered Non-Controlled IP) and Parent Assigned Intellectual Property have been maintained in the Ordinary Course of Business and are within the possession or control of, or are accessible to, a Target Company or Company Subsidiary. To the Non-Controlled IP Knowledge of Parent, records for all registered Non-Controlled IP have been maintained in the Ordinary Course of Business and are within the possession or control of, or are accessible to, a Target Company, Company Subsidiary or a Non-Controlled Company. Copies of all Material Third Party Licenses In and Material Third Party Licenses Out have been maintained in the Ordinary Course of Business. Such copies are within the possession or control of, a Target Company Subsidiary.
- (dd) To the Knowledge of Parent, there are no pending Claims filed with a Governmental Entity against a Target Company, Company Subsidiary or member of Parent Group (in relation to the operation of the Business) asserting a violation of moral rights or rights of a similar nature, and neither the Target Companies, the Company Subsidiaries nor any member of Parent Group has received written notice of any such Claim that remains unresolved since the date that is 24 months prior to the date of this Stock Purchase Agreement which would reasonably be expected to adversely affect: (i) the use by the Target Companies and Company Subsidiaries of the Company Intellectual Property or Parent Assigned

Intellectual Property; or (ii) the use in accordance with the terms of the Intellectual Property License Agreement of the Parent Licensed Intellectual Property.

- (ee) Target Group members and their authorized sublicensees can use the Ford Broad License Vehicle IP (as such term is defined in the Intellectual Property License Agreement) and use and sell Products (as such term is defined in the Intellectual Property License Agreement) based on, incorporating or using the Ford Broad License Vehicle IP without using the Ford Business Materials (as such term is defined in the Intellectual Property License Agreement).
- (ff) None of any of Target Company's or Company Subsidiary's current Associated Companies (as such term is defined in the Intellectual Property License Agreement) uses the Ford Limited License Vehicle IP (as such term is defined in the Intellectual Property License Agreement) or the Ford Limited License IT, except as set forth in Schedule B Part 3.14(ff).

3.15 Insurance

- (a) The Replacement Policies are (or, if not yet in effect, shall be from the applicable effective date) maintained in such amounts, with such deductibles and against such risks and losses as are (i) reasonable for the Business consistent with the operation of the Business in the 12 months prior to the date of this Stock Purchase Agreement as if the Business had been operated independent of Parent Group and (ii) required by Law and Agreements to which a Target Company or Company Subsidiary is a party.
- (b) Schedule B Part 3.15(b) sets forth a true, complete and correct list of all pending claims and claims history under the Ford Excess (Umbrella) Liability Insurance Program for the Target Companies and Company Subsidiaries (i) in relation to product liability claims, since the date that is 24 months prior to the date of this Stock Purchase Agreement, and (ii) in relation to all other claims, since January 1, 2006.
- (c) To the Knowledge of Parent, all Insurance Policies are in full force and effect and are enforceable in accordance with their terms. All premiums due and payable thereon have been paid. Except in connection with the arrangement of the Replacement Policies, no notice of cancellation or termination has been received with respect to any Insurance Policy as such policy applies to a Target Company or Company Subsidiary.
- (d) To the Knowledge of Parent, (i) no Target Company or Company Subsidiary has done or omitted to do anything that has rendered, or would reasonably be expected to render, an Insurance Policy void or voidable and (ii) all events that have occurred and are known by Parent that would reasonably be expected to lead to a material future claim by a Target Company or Company Subsidiary on an Insurance Policy have been duly notified.

3.16 Litigation

(a) Except as set forth on Schedule B Part 3.16(a), there is no Claim (i) related to product liability or (ii) in any individual case or instance seeking damages in excess of \$500,000, or, to the Knowledge of Parent, that has been threatened in writing since the date that is 24 months prior to the date of this Stock Purchase Agreement, which Claim has not been withdrawn, terminated, settled or otherwise closed, against a Target Company or Company Subsidiary or any of its material assets. Except as identified on Schedule B Part 3.16(a), none of the Claims required to be listed in Schedule B Part 3.16(a),

if adversely determined, would, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

- (b) There is no pending Claim against any current or, to Knowledge of Parent, former director or officer of a Target Company or Company Subsidiary for actions taken in the role of director or officer and with respect to which the Target Company or Company Subsidiary has or would reasonably be expected to have an indemnification obligation.
- (c) Except as set forth on Schedule B Part 3.16(c), there is no unsatisfied material Order or material penalty against or affecting the Target Companies or Company Subsidiaries or any of their properties or assets, and there is no Order to which the Target Companies or Company Subsidiaries or any of their respective properties or assets are subject, or under which the Target Companies or Company Subsidiaries are in default.

3.17 Labor and Employment

- (a) To the Knowledge of Parent, there are no pending or threatened Industrial Actions relating to Employees.
- (b) Except as set forth on Schedule B Part 3.17(b), (i) to the Knowledge of Parent, there are no locations at which the Employees are represented by labor unions or work councils at the local level (any such represented employees, "Represented Employees") and (ii) there are no material local collective Agreements applicable to Represented Employees (any such local collective Agreements, "Local Collective Agreements").
- (c) To the Knowledge of Parent, each Target Company and each Company Subsidiary is in material compliance with all Local Collective Agreements.
- (d) Parent has made available in the Legal Data Room a list of the Employees (on an anonymous basis), setting forth salary, age, length of service and notice period required for termination of service (each as of the dates specified on such lists), which list, to the Knowledge of Parent, is true, complete and correct in all material respects.

3.18 Employee Benefits

- (a) Schedule B Part 3.18(a) sets forth (i) to the Knowledge of Parent, a list of all material Plans required by Law and (ii) a list of all other material Plans (including any employment Agreement with respect to any Employee whose salary grade is at or above that of a Leadership Level 3 Employee, except any Designated Employee as identified in schedule A to the Ford Secondment Agreement) and any arrangement for the provision of material Retirement Benefits in which any Target Company or Company Subsidiary has ever participated and under which the Target Company or Company Subsidiary has any Liability after Closing.
- (b) Except as set forth on Schedule B Part 3.18(b), all Plans are in material compliance with Law and have been operated in all material respects in accordance with their terms and, to the Knowledge of Parent, there are no circumstances which would reasonably be expected to result in material non-compliance with any Law.

- (c) Except (i) as set forth on Schedule B Part 3.18(c), (ii) for Plans required by Law and (iii) for Plans that are publicly available, Parent and Sweden Seller have made available to Buyers copies, or written descriptions where no written Plan exists, of all material governing documentation of all material Plans, including any amendments to such Plans, and, to the Knowledge of Parent, have made available to Buyers the most recent actuarial valuation, and all such documentation is true, complete and correct in all material respects.
- (d) All contributions required to be made by Law or pursuant to the terms of a Plan have been timely made and the funded status of the Plans (assets and obligations) have been appropriately valued on the 2008 Financial Statements as of December 31, 2008 based on U.S. GAAP. Except as set forth on Schedule B Part 3.18(d), to the Knowledge of Parent, and except as required by Law or a Local Collective Agreement, there is no proposal to materially increase contribution rates to material Plans in any jurisdiction.
- (e) Other than routine claims for benefits, there are no material Claims (or Claims which when aggregated with other Claims dealing with the same Plan would be material) pending or, to the Knowledge of Parent, threatened in writing since the date that is 12 months prior to the date of this Stock Purchase Agreement by or on behalf of any of the Plans, by or in respect of a current or former Employee or agent of a current or former Employee involving such Plan or the assets of such Plan with an aggregate value of more than \$500,000, and there are no circumstances which would reasonably be expected to give rise to any such Claim.
- (f) To the Knowledge of Parent, and except as required by Law or a Local Collective Agreement, no proposal has been announced and no written promise, assurance or guarantee has been given to or in respect of any Employee or former Employee to continue or make any discretionary increases to any defined benefit Retirement Benefits (whether provided for under the Plans or otherwise).
- (g) To the Knowledge of Parent, each Plan that is operated in the European Union satisfies the requirements to provide equal benefits and equal access for men and women to the extent required by article 141 of the Treaty Establishing the European Community (as amended by the Treaty of Amsterdam, dated October 2, 1997), including any additional requirements of local Law relating to such requirements in each country within the European Union
- (h) Except as set forth on Schedule B Part 3.18(h), no debt under sections 75 or 75A of the UK Pensions Act 1995 has become due to any of the Plans and is unpaid and, to Parent's Knowledge, no such debt has become due from any Target Company or Company Subsidiary to any other retirement benefit arrangement and is unpaid and, as of the date of this Stock Purchase Agreement, there are no circumstances which would reasonably be expected to give rise to such debt becoming due on, before or with effect from the Closing Date and remaining unpaid either from any Target Company or Company Subsidiary. To Parent's Knowledge, no financial support direction or contribution notice under sections 38 to 51 of the UK Pensions Act has been issued to any Target Company or Company Subsidiary (and, to the Knowledge of Parent, there is no investigation pending as of the date of this Stock Purchase Agreement by the UK Pensions Regulator that would reasonably be expected to lead to such a financial support direction or contribution notice being issued) in relation to any of the Plans and any other retirement benefit arrangement in relation to which Target Company or Company Subsidiary is an associate or connected person of an employer (for the purposes of the UK Pensions Act). To the Knowledge of Parent, no Target Company or Company Subsidiary has, since April 27, 2004, been a party to or given knowing assistance to an act or deliberate failure to act, that would reasonably be expected to enable the UK Pensions Regulator to issue a contribution notice under section 38 of the UK Pensions Act.

3.19 Data Privacy

- (a) Each Target Company and each Company Subsidiary has complied in all material respects with Parent's global policies on personal data privacy, except as otherwise required by Law.
- (b) Except as set forth on Schedule B Part 3.19, since the date that is 24 months prior to the date of this Stock Purchase Agreement, no Target Company or Company Subsidiary has received, or is continuing to administer (i) a written enforcement notice of non-compliance with the EU Data Protection Directive from a Governmental Entity or regulatory authority; (ii) to the Knowledge of Parent, a Claim for compensation by an individual whose personal data is processed by a Target Company or Company Subsidiary for breach of the EU Data Protection Directive; or (iii) to the Knowledge of Parent, any other notice alleging that a Target Company or Company Subsidiary has not complied with the EU Data Protection Directive.

3.20 Material Environmental Permits

- (a) Schedule B Part 3.20 sets forth a true, complete and correct list of the Material Environmental Permits. The Material Environmental Permits have been duly issued to, and maintained and complied with in all material respects by, each Target Company and Company Subsidiary.
- (b) To the Knowledge of Parent, no action has been taken by a Governmental Entity to revoke, withdraw, terminate, cancel, modify, suspend or to cause failure to renew a Material Environmental Permit, and no circumstances exist, including the execution of this Stock Purchase Agreement or the Separation Agreements or the consummation of the transactions contemplated hereby and thereby, which would reasonably be expected to give rise to such action, except where the consequence of such action would not, individually or in the aggregate, reasonably be expected to materially impair the operation of the Business.

3.21 Environmental Site Assessments

Each of Parent, Sweden Seller each Target Company and each Company Subsidiary has provided to the Environmental Consultants all material information and data requested by the Environmental Consultants for the purposes of carrying out their site investigations and audits with respect to producing the Phase I Environmental Site Assessments, and there are no material omissions from such information or data provided.

3.22 Contaminated Land

Except as set forth on Schedule B Part 3.22, there is no contamination of soil or groundwater by Hazardous Materials at, on, under, within or migrating from a facility or site owned, used or occupied by a Target Company or Company Subsidiary as of the date of this Stock Purchase Agreement (or to the Knowledge of Parent, owned, used or occupied by a Target Company or Company Subsidiary during the five-year period prior to the date of this Stock Purchase Agreement), which contamination requires investigation, remediation, or other response or action pursuant to Environmental Law, except for such remediation or response as would not, individually or in the aggregate, be likely to result in expenditure in excess of \$500,000.

3.23 Material Expenditure

Except as set forth on Schedule B Part 3.23, to the Knowledge of Parent, no material expenditure is required in the 36 months following the date of this Stock Purchase Agreement, in order to upgrade, modify, replace or improve any Site as required under any Environmental Law or Material Environmental Permit.

3.24 Product Recalls

As of the date of this Stock Purchase Agreement, Schedule B Part 3.24 sets forth a true, complete and correct list of mandatory or voluntary Product Recalls by a Target Company, Company Subsidiary or member of Parent Group with respect to Vehicles supplied within the seven years prior to the date of this Stock Purchase Agreement.

3.25 Product Liability

- (a) Except as set forth on Schedule B Part 3.25, to the Knowledge of Parent, no Target Company or Company Subsidiary has, and, with respect to the Business, none of Parent, Sweden Seller or their respective Affiliates has, received any written notice since the date that is two years prior to the date of this Stock Purchase Agreement relating to any Claim involving use of any of the products (or any part or component) designed, manufactured, serviced or sold, or services performed, by a Target Company or Company Subsidiary, including for negligence, strict liability, design or manufacturing defect, conspiracy, failure to warn, or breach of express or implied warranties of merchantability or fitness for any purpose or use, or from any alleged breach of implied warranties or representations, or any alleged non-compliance with any applicable Law pertaining to product liability matters.
- (b) All Vehicles being sold by a Target Company, Company Subsidiary or member of Parent Group in the European Union, and to the Knowledge of Parent, in any other territory, comply in all material respects with the relevant Product Regulatory Requirements as in effect on the date of this Stock Purchase Agreement for the relevant country in which such Vehicles are sold.

3.26 Warranties

Schedule B Part 3.26 sets forth a true, complete and correct list of the forms of written manufacturer (a) warranties, (b) extended warranties, (c) extended service plans, (d) indemnification obligations, (e) guarantees and (f) warranty policies, in each case issued by Sweden Company on behalf of the Target Group for the 36 months prior to the date of this Stock Purchase Agreement in respect of any products and services of each Target Company and Company Subsidiary, which are in effect as of December 11, 2009.

3.27 Compliance with Law; Permits

- (a) Except as set forth on Schedule B Part 3.27(a), to the Knowledge of Parent, the Business currently is, and since the date that is 36 months prior to the date of this Stock Purchase Agreement has been, in all material respects operated in compliance with Law.
- (b) Except as set forth on Schedule B Part 3.27(b), since the date that is 36 months prior to the date of this Stock Purchase Agreement, no Target Company or Company Subsidiary has received

written notice from a Governmental Entity of an actual or alleged material violation of, or failure to comply with, Law.

(c) All material Permits (other than Material Environmental Permits) issued or granted to a Target Company or Company Subsidiary ("Material Non-Environmental Permits") are validly held by a Target Company or Company Subsidiary and are in full force and effect, and the applicable Target Company or Company Subsidiary has complied in all material respects with all terms and conditions of such Material Non-Environmental Permits. Since the date that is 12 months prior to the date of this Stock Purchase Agreement, none of Parent, Sweden Seller, the Target Companies or Company Subsidiaries has received notice of any Claim relating to revocation, withdrawal, termination, cancellation, suspension or modification of any such Permit. Except as set forth on Schedule B Part 3.27(c), no Material Non-Environmental Permits will be subject to revocation, withdrawal, termination, cancellation, suspension, modification, or non-renewal as a result of the execution and delivery of this Stock Purchase Agreement or the Separation Agreements or the consummation of the transactions contemplated by the Transaction Documents.

3.28 Information Technology and Functionality

- (a) All IT Applications and IT Infrastructure used by a Target Company or a Company Subsidiary as of the date of this Stock Purchase Agreement are either (i) owned by a Target Company or a Company Subsidiary, or (ii) licensed or leased to a Target Company or a Company Subsidiary under a written Third Party IT Agreement, or (iii) to the Knowledge of Parent, form part of the Parent IT System. Copies of all material Third Party IT Agreements entered into by a Target Company as of the date of this Stock Purchase Agreement and on terms other than Sweden Company's or Parent's standard terms and conditions or supplements have been made available to Buyers.
- (b) All IT Applications and IT Infrastructure used by a Target Company or a Company Subsidiary as of the Closing Date will be either (i) owned by a Target Company or a Company Subsidiary, or (ii) licensed or leased to a Target Company or a Company Subsidiary under a written Third Party IT Agreement, or (iii) form part of the Parent IT System (taking into account any updates on or after the Closing Date as contemplated by section 16 of the IT Agreement).
- (c) To the Knowledge of Parent, the list of IT Applications and IT Infrastructure identified at Schedule W comprises all of the IT Applications and IT Infrastructure (i) owned by a member of Parent Group or provided to a member of Parent Group or to a Target Company or a Company Subsidiary pursuant to a third party agreement to which a member of Parent Group is a party, and (ii) used by a Target Company or Company Subsidiary (or any of their respective suppliers, dealers or customers in or in relation to one or more Vehicles), in each case as of the date of this Stock Purchase Agreement.
- (d) The list of IT Applications and IT Infrastructure identified at Schedule W comprises all of the IT Applications and IT Infrastructure (i) owned by a member of Parent Group or provided to a member of Parent Group or to a Target Company or a Company Subsidiary pursuant to a third party agreement to which a member of Parent Group is a party, and (ii) used by a Target Company or Company Subsidiary (or any of their respective suppliers, dealers or customers in or in relation to one or more Vehicles), in each case as of the Closing Date, except in either case, as such list may be updated in accordance with section 16 of the IT Agreement.

- (e) The IT Applications and IT Infrastructure of the Target Companies IT System that are owned by a Target Company or Company Subsidiary are free and clear from all Encumbrances, except for IT Permitted Encumbrances.
- (f) To the Knowledge of Parent, no member of Parent Group, no Target Company or Company Subsidiary that is a party to a Third Party IT Agreement, nor, to the Knowledge of Parent, any counterparty to it, is in default or breach of such Agreement in any material respect, or is in default or breach of such Agreement in any respect which gives rise to an entitlement to another party to such Agreement to terminate or suspend such Agreement.
- (g) To the Knowledge of Parent, no member of Parent Group, Target Company or Company Subsidiary that is party to a Third Party IT Agreement has received written notice of termination or non-renewal, or intention to terminate or not renew, from any counterparty to such Agreement, except as would not reasonably be expected to materially impair the operation of the Target Companies IT System.
- (h) Except as set forth in Schedule B Part 3.28(h), since the date that is 12 months prior to the date of this Stock Purchase Agreement, the IT Systems have not:
- (i) been the cause of a material disruption to the production of Vehicles by a Target Company or Company Subsidiary or otherwise materially impaired the operation of the Business;
- (ii) been infected by a software virus, bug, logic bomb, Trojan horse or any other extraneous program, script or executable code that distorts their proper functioning, permits unauthorized access or disables them without the consent of the user thereof, except as has not materially impaired the operation of the Business; or
- (iii) to the Knowledge of Parent, been accessed by an unauthorized Person, except to the extent such access has been detected and controlled in the Ordinary Course of Business through security practices established to control unauthorized access of such IT Systems.
- (i) Parent, Sweden Seller, the Target Companies and Company Subsidiaries maintain in respect of their respective businesses and consistent with Parent's "Information Technology Policy Manual", "Global Information Standard I", and "Global Information Standard II" corporate policies, data backups, information technology disaster recovery arrangements and hardware and software support and maintenance to minimize the risk of occurrence of any disruption, infection or access referred to in Section 3.28(h), and copies of all such policies have been made available to Buyers.

3.29 Suppliers

Schedule B Part 3.29 sets forth a true, complete and correct list of the top 20 (as measured by annual spend by the Target Companies and Company Subsidiaries) production or non-production suppliers of goods and services to the Target Companies and Company Subsidiaries as of December 31, 2009 and for the year ended December 31, 2008. Except as set forth on Schedule B Part 3.29, since the date of the Interim Financial Statements, there has not been (a) any material adverse change in the business relationship of any Target Company with any supplier of merchandise named in Schedule B Part 3.29 or (b) any adverse change in any material term (including credit terms, other than temporary changes that are no longer in place) of the supply agreements or related Agreements with any such supplier.

3.30 JLR Agreements and AMLL Agreements

- (a) Schedule B Part 3.30(a) sets forth a true, complete and correct list of the JLR Agreements.
- (b) Schedule B Part 3.30(b) sets forth a true, complete and correct list of the AMLL Agreements.

3.31 NSC Reorganization

The NSC Reorganization has been completed by Parent prior to the date of this Stock Purchase Agreement.

3.32 Exclusivity of Representations and Warranties

Except for the representations and warranties contained in the Separation Agreements or in this Article 3 (as modified by Schedule B delivered on the date of this Stock Purchase Agreement), none of Parent, Sweden Seller or any other Person makes any express or implied representation or warranty with respect to Parent, Sweden Seller, a Target Company, a Company Subsidiary or any of their respective Affiliates, the Business or Assets, or the transactions contemplated by this Stock Purchase Agreement or the Separation Agreements, and Parent and Sweden Seller disclaim all other representations or warranties, whether made by Parent, Sweden Seller or any of their respective Affiliates, directors, employees, agents or representatives, whether included in the Due Diligence Materials or otherwise made orally or in writing to any member of Buyer Group or Geely Group or their respective representatives. Notwithstanding anything to the contrary in this Section 3.32, neither the scope of the representations and warranties set forth in the Separation Agreements or in this Stock Purchase Agreement, nor the absence of any representation or warranty from this Stock Purchase Agreement or any Separation Agreement, shall (or shall be deemed to) limit, modify or otherwise affect, any Claim based on actual fraud or willful misconduct.

4. REPRESENTATIONS AND WARRANTIES OF U.S. BUYER AND SWEDEN BUYER

U.S. Buyer and Sweden Buyer represent and warrant to Parent and Sweden Seller, as of the date of this Stock Purchase Agreement and as of Closing or at such time as is otherwise specified in any Section of this Article 4, as follows:

4.1 Organization and Standing; Capitalization

- (a) Each of U.S. Buyer and Sweden Buyer is duly organized, validly existing and in good standing under the Law of its respective jurisdiction of incorporation or formation, as applicable, and has the requisite corporate power and authority to own or lease and to operate its assets and to carry on its respective business as currently conducted.
- (b) Beijing Holdco Controls Sweden Buyer and indirectly owns a majority of the issued and outstanding shares of capital stock of Sweden Buyer and Sweden Buyer owns the sole membership interest of U.S. Buyer.

- (c) Schedule J sets forth a true, complete and correct list, with respect to each of Shanghai Holdco and Beijing Holdco, of the authorized and outstanding shares of capital stock, membership interest or other interest that confers the right to receive a share of the profits and losses, or distribution of assets, of the issuing entity of each company and the owner(s) of record of such interests of each of Shanghai Holdco and Beijing Holdco.
- (d) Zhejiang Geely Holding Group Co., Ltd. Controls each of Beijing Holdco, Shanghai Holdco, Sweden Buyer and U.S. Buyer. Except for between members of Geely Group, there are no Agreements with respect to Control of Beijing Holdco, Shanghai Holdco, Sweden Buyer, U.S. Buyer, Sweden Company, U.S. Company or the Project Companies, or otherwise granting operational control, including with respect to the current or future Intellectual Property Rights of Sweden Company, over any such Person.

4.2 Authority and Capacity; No Conflicts; Binding Obligation; No Liquidation

- (a) Each member of Buyer Group has the requisite corporate power and authority to enter into the Transaction Documents to which it is party and to consummate the transactions contemplated by the Transaction Documents.
- (b) Subject to satisfaction of the Governmental Approvals and the Buyer Governmental Approvals, the execution, delivery and performance by each member of Buyer Group of this Stock Purchase Agreement, and the consummation of the transactions contemplated by this Stock Purchase Agreement, have been duly authorized by all requisite actions of U.S. Buyer and Sweden Buyer, with such authorization in full force and effect, and do not:
- (i) conflict with or violate (A) any material Law in any material respect or (B) a provision of the Formation Document or the Bylaws of U.S. Buyer or Sweden Buyer; or
- (ii) conflict with, result in a breach of, constitute a default under, give rise to a right of prepayment, termination, cancellation or acceleration of an obligation or loss of material benefit or to increased, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of an Encumbrance upon assets under, any Agreement to which a member of Buyer Group is a party or by which their respective properties or assets are bound, except to the extent that such conflict, breach, default, prepayments, termination, cancellation, acceleration of obligation, loss of material benefit, increase or acceleration or guaranty of rights or entitlements or creation of Encumbrance would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impede the ability of U.S. Buyer or Sweden Buyer to consummate the transactions contemplated by the Transaction Documents.
- (c) This Stock Purchase Agreement has been duly executed and delivered by U.S. Buyer and Sweden Buyer. Assuming the due authorization, execution and delivery by Parent and Sweden Seller of this Stock Purchase Agreement, this Stock Purchase Agreement constitutes a legal, valid and binding obligations of U.S. Buyer and Sweden Buyer, enforceable against U.S. Buyer and Sweden Buyer in accordance with its terms, except as such enforceability may be subject to the effects of Law.
 - (d) With respect to each of U.S. Buyer and Sweden Buyer:

- (i) to the Knowledge of Buyer, no Order has been entered, no resolution has been adopted, and no petition has been presented to a court of competent jurisdiction for the purpose of its winding up or liquidation;
- (ii) no provisional liquidator, manager, administrator, conservator or administrative receiver, or equivalent in its jurisdiction of incorporation or formation, has been appointed, including with respect to all or substantially all of its assets, and, to the Knowledge of Buyer, no petition or application for such appointment has been filed or presented to a court of competent jurisdiction;
 - (iii) no assignment of all or substantially all of its assets has been made in favor of its creditors; and
- (iv) to the Knowledge of Buyer, no proceeding under bankruptcy, insolvency, reorganization, moratorium or other similar Law has been initiated.

4.3 Governmental Approvals

- (a) Buyers and their Affiliates have made or obtained, as applicable, all approvals, consents, authorizations, filings, waiting periods terminations or expirations and notices to or from a Governmental Entity that are required by Law for Buyers to execute, deliver and perform the Escrow Agreement.
- (b) Other than the Governmental Approvals and the Buyer Governmental Approvals, no approval, consent, authorization or Permit from, or filing or notice to, a Governmental Entity, and no expiration or termination of a mandatory waiting period by a Governmental Entity, is required in connection with the execution and delivery by U.S. Buyer and Sweden Buyer of this Stock Purchase Agreement or the execution by the relevant members of Buyer Group of the Separation Agreements to which each is a party or the consummation by the relevant members of Buyer Group of the transactions contemplated by the Transaction Documents, except for such approvals, consents, authorizations, Permits, filings, notices, or expirations or terminations the failure of which to obtain, make, submit or occur would not be material.
- (c) To the Knowledge of Buyer, there is no fact or circumstance related to any member of Buyer Group that would reasonably be expected to delay the receipt or satisfaction of the Governmental Approvals or the Buyer Governmental Approvals or the expiration or termination of any waiting periods required under any of the Governmental Approvals or the Buyer Governmental Approvals.

4.4 Financial Capacity

(a) At or prior to the date of this Stock Purchase Agreement, Buyers have delivered a copy of the executed commitment letter and framework agreements, attached to this Stock Purchase Agreement as Exhibit I (the "Binding Acquisition Financing Commitment Letters"), from Daqing State-owned Assets Operating Co., Ltd. and Shanghai Jiding Industrial Area Development (Group) Co., Ltd. to provide financing in an aggregate amount of RMB 4,000,000,000 to Buyers to fund the Purchase Price (the "Acquisition Financing"). The Acquisition Financing, when funded in accordance with the Binding Acquisition Financing Commitment Letters, will, together with cash on hand held by Buyers, provide financing sufficient to pay the Purchase Price as well as all other amounts required to be paid by Buyers under this Stock Purchase Agreement. The Binding Acquisition Financing Commitment Letters are valid

and in full force and effect and have not been modified or rescinded, in whole or in part, and no event has occurred which, with or without notice, lapse of time or both, would constitute a default or breach on the part of Buyers under any term or condition of the Binding Acquisition Financing Commitment Letters. Except as described in, or provided for in, the Binding Acquisition Financing Commitment Letters, there are no (i) conditions precedent to the obligations of Daqing State-owned Assets Operating Co., Ltd and Shanghai Jiading Industrial Development (Group) Co., Ltd. to fund the Acquisition Financing or (ii) causes of termination of the Binding Acquisition Financing Commitment Letters, the Acquisition Financing or any material commitment thereunder. All approvals, consents, authorizations or Permits from, or filings or notices to, a Governmental Entity which are required for Daqing State-owned Assets Operating Co. and Shanghai Jiading Industrial Area Development (Group) Co., Ltd. to provide the Acquisition Financing, including approvals from the relevant State-owned assets supervisory and regulatory authority, have been obtained. To the Knowledge of Buyer, there is no fact or circumstance that would reasonably be expected to prevent or delay the satisfaction of any term or condition of the Binding Acquisition Financing Commitment Letters or that would permit Daqing State-owned Assets Operating Co., Ltd or Shanghai Jiading Industrial Development (Group) Co., Ltd. to reduce the total amount of the Acquisition Financing or impose any additional conditions precedent to the availability of the Acquisition Financing.

(b) Notwithstanding the provisions of Section 4.4(a) and Section 4.4(b), the obligations of U.S. Buyer and Sweden Buyer under this Stock Purchase Agreement, including the obligation to proceed with Closing subject only to satisfaction, or waiver, of the conditions precedent set forth in Article 8 of this Stock Purchase Agreement, are not subject to the availability to U.S. Buyer or Sweden Buyer on the Closing Date of sufficient financing to fund payment of the Purchase Price.

4.5 Investment Intent and Experience

U.S. Buyer is acquiring the U.S. Company Interest, and Sweden Buyer is acquiring the Sweden Company Shares, for its own account, solely for the purpose of investment and not with a view to, or for sale in connection with, distribution thereof in violation of U.S. federal or state securities Laws, or the securities Laws of other jurisdictions. U.S. Buyer has not agreed to, and is under no obligation to, transfer the U.S. Company Interest, or to grant any rights in the U.S. Company Interest to any other Person, and Sweden Buyer has not agreed to, and is under no obligation to, transfer the Sweden Company Shares to, or to grant any rights in the Sweden Company Shares to, any other Person except in each case in compliance with U.S. federal or state securities Laws, or the securities Laws of other jurisdictions. Buyers (a) have knowledge and experience in investments of the type contemplated by this Stock Purchase Agreement, (b) are capable of evaluating the merits and risks of this investment and (c) can bear the economic risks of this investment and can afford the complete loss of such investment. Buyers understand that the exemptions from registration under the U.S. federal and state securities of blue sky Law relied upon by Parent in connection with the sale of the U.S. Company Interest pursuant to this Stock Purchase Agreement are based in part on the representations and warranties provided in this Section 4.5.

4.6 U.S. Dealers

U.S. Buyer and Sweden Buyer acknowledge that the U.S. Dealer Agreements are subject to certain U.S. dealer franchise protection statutes that limit the ability of U.S. Company to terminate or modify the U.S. Dealer Agreements, and that the change in ownership of U.S. Company contemplated by this Stock Purchase Agreement does not by itself provide grounds or occasion to terminate or modify the U.S. Dealer Agreements.

5. PRE-CLOSING COVENANTS

5.1 Governmental Filings

- (a) As soon as practicable after the execution and delivery by the Parties of this Stock Purchase Agreement (but in no event later than 20 Business Days from the date of this Stock Purchase Agreement (the "Filing Deadline")):
- (i) each of Parent, Sweden Seller, and Buyers shall separately make, or shall cause their respective Affiliates to make, the filings required by each of them relating to the Governmental Approval set forth on Schedule C Part 1;
- (ii) Buyers shall provide all notices and make all filings relating to the Governmental Approvals set forth on Schedule C Part 2A and Schedule C Part 3 and shall make, or shall cause their respective Affiliates to make, the filings relating to the Buyer Governmental Approvals set forth on Schedule S Part 1, in each case to the extent not already made prior to the date of this Stock Purchase Agreement, and provided, however, that to the extent Buyers, after exercising commercially reasonable efforts, are unable to collect by the Filing Deadline the requisite information and documentation necessary for submission of such filings from Persons outside of Buyers' control, Buyers shall make any such filings as soon as practicable and in any event will use all commercially reasonable efforts to make such filings no later than 30 Business Days from the date of this Stock Purchase Agreement (the "Extended Deadline"); and
- (iii) Buyers shall collect all information necessary to determine, in good faith consultation with the competition regulatory counsel to Parent, whether any approval, consent, authorization of Permit from, or filing or notice to, or expiration or termination of a mandatory waiting period by, a Governmental Entity in addition to the Governmental Approvals or the Buyer Governmental Approvals are required under applicable Law (which shall include for the purposes of this clause any antitrust or merger control law) (any such approvals, consents, authorizations, Permits, filings, notices, or expirations or terminations, the "Additional Governmental Approvals"), and provide written notice of the Additional Governmental Approvals to Parent and Sweden Seller.
- (b) No later than three Business Days following receipt of the Buyer Governmental Approvals set forth on Schedule S Part 1, Buyers shall make or cause to be made all filings relating to the Buyer Governmental Approvals set forth on Schedule S Part 2.
- (c) Buyers shall make the filings relating to the Governmental Approval set forth on Schedule C Part 2B1a no later than five Business Days after the European Commission takes jurisdiction under Article 4(5) of the EC Merger Regulation, or, in the event that the European Commission does not accept such filings within five Business Days after taking jurisdiction under Article 4(5) of the EC Merger Regulation, no later than one Business Day after the European Commission's agreement to accept the filings relating to the Governmental Approval set forth on Schedule C Part 2B1a ("European Commission Filing Deadline"), provided, however, that to the extent Buyers, after exercising commercially reasonable efforts, are unable to collect by the European Commission Filing Deadline the requisite information and documentation necessary for submission of such filings from Persons outside of Buyers' control, Buyers shall make the filings relating to the Governmental Approval set forth on Schedule C Part 2B1a as soon as practicable and in any event will use all commercially reasonable efforts to make such filings no later than the Extended Deadline. If the European Commission has not taken

jurisdiction under Article 4(5) of the EC Merger Regulation, Buyers shall make the filings relating to the Governmental Approvals set forth on Schedule C Part 2B1b no later than 20 Business Days from the date of Buyer being informed by the European Commission that it has not taken jurisdiction under Article 4(5) of the EC Merger Regulation ("Member State Filing Deadline"), provided, however, that to the extent Buyers, after exercising commercially reasonable efforts, are unable to collect by the Member State Filing Deadline the requisite information and documentation necessary for submission of such filings from Persons outside of Buyers' control, Buyers shall make the filings relating to the Governmental Approval set forth on Schedule C Part 2B1b as soon as practicable and in any event will use all commercially reasonable efforts to make such filings no later than the Extended Deadline.

- (d) No later than 30 Business Days from the date of this Stock Purchase Agreement, Buyers shall make the filings, if any relating to the Additional Governmental Approvals ("Additional Governmental Approvals Filing Deadline") provided, however, that to the extent Buyers, after exercising commercially reasonable efforts, are unable to collect by the Additional Governmental Approvals Filing Deadline the requisite information and documentation necessary for submission of such filings from Persons outside of Buyers' control, Buyers shall make such filings as soon as practicable and in any event will use all commercially reasonable efforts to make such filings no later than the Extended Deadline.
- (e) Subject to restrictions imposed by Law, each of Parent and Sweden Seller on the one hand, and Buyers on the other hand shall, and shall cause their respective affiliates to:
- (i) promptly supply the other with any information that may be reasonably required in order to make any notices or filings to be made pursuant to Section 5.1(a) or Section 5.1(b);
- (ii) promptly notify the other of the occurrence of any event that is required to be set forth in an amendment or supplement to any notices or filings made pursuant to Section 5.1(a) or Section 5.1(b) and cooperate with each other in filing promptly with the relevant Governmental Entity such amendment or supplement;
- (iii) promptly notify the other upon the receipt of any relevant comments, questions or requests, including requests for the production of any documents or information or the amendment or supplement of any filing, from a Governmental Entity relating to the transactions contemplated by this Stock Purchase Agreement;
- (iv) use commercially reasonable efforts to consult in advance and cooperate with each other in connection with any analyses, appearances, presentations, memoranda, briefs, arguments and proposals to be made or submitted by or on behalf of any Party to a Governmental Entity relating to the transactions contemplated by this Stock Purchase Agreement; and
- (v) except to the extent such disclosure is prohibited by Law, provide, upon the request of the other Party, copies of all correspondence between itself or its Affiliates and any Governmental Entity relating to the transactions contemplated by this Stock Purchase Agreement, it being understood that the Parties may designate any competitively sensitive materials provided to the other under this Section 5.1(c)(v) as "Outside Counsel Only" with such materials and the information contained therein to be provided only to the outside legal counsel of the receiving Party and not to be disclosed by

such outside legal counsel to employees, officers or directors of the receiving Party without the prior written consent of the Party providing such materials.

- (f) Subject to Section 5.1(e), each Party shall be responsible for its own costs and expenses, including legal fees and disbursements, incurred by it in respect of making its own filings that are required and as set forth in this Section 5.1, provided that where the Parties agree to appoint joint counsel in respect of any filing that is required pursuant to this Section 5.1, at Closing, Parent shall reimburse Buyers for 50% of all legal fees and disbursements paid by Buyers to such local counsel, and Buyers shall reimburse Parent for 50% of all legal fees and disbursements paid by Parent to such local counsel, in each case to the extent not already reimbursed prior to Closing.
- (g) Buyers shall, or shall cause their respective Affiliates to, promptly pay all filing fees required in connection with any Governmental Approval and any Buyer Governmental Approval, provided that Parent shall, at Closing, reimburse Buyers for 50% of all such fees paid by Buyers to the extent not already reimbursed prior to Closing.
- (h) From the date of this Stock Purchase Agreement until Closing, Buyers shall not, and shall use commercially reasonable efforts to cause each member of Geely Group not to, acquire or agree to acquire, by merging with or into or consolidating with, or by purchasing a substantial portion of the assets of or equity in, or by any other manner, any assets or Person, if the execution and delivery of a definitive Agreement relating to, or the consummation of, such acquisition would reasonably be expected to: (i) impose any delay in obtaining, or increase the risk of not obtaining, consents of a Governmental Entity necessary to consummate the transactions contemplated by this Stock Purchase Agreement or the expiration or termination of any applicable waiting period; (ii) increase the risk of a Governmental Entity seeking or entering an Order prohibiting the consummation of the transactions contemplated by this Stock Purchase Agreement; (iii) increase the risk of not being able to remove any such Order on appeal or otherwise; or (iv) otherwise prevent or delay the consummation of the transactions contemplated by this Stock Purchase Agreement.
- (i) Notwithstanding any provision to the contrary in this Stock Purchase Agreement, Buyers and members of Buyer Group shall not be required hereunder to (i) enter into any settlement, undertaking, consent decree, stipulation or other Agreement (collectively, "Undertakings") with any Governmental Entity in connection with this Stock Purchase Agreement, any Transaction Document or any other agreements contemplated hereby or (ii) divest or otherwise hold separate (including by establishing a trust or otherwise), or take any other action (or otherwise agree to do any of the foregoing) with respect to the subsidiaries, businesses, assets or properties of the Target Companies, Buyers or members of Buyer Group (collectively, "Divestitures"), except to the extent such Undertakings or Divestitures are not material to the combined operations of the Buyer Group and Target Companies and the Company Subsidiaries.

5.2 Consultation Obligations

- (a) From the date of this Stock Purchase Agreement until Closing, Parent shall, and shall cause each Target Company and each Company Subsidiary to (i) comply with their respective Collective Consultation Requirements in Sweden and (ii) use commercially reasonable efforts to comply with their respective Collective Consultation Requirements as set forth in Schedule EE.
- (b) Subject to restrictions imposed by Law, Buyers shall use commercially reasonable efforts to cooperate with Parent to comply with Parent's obligations in Section 5.2(a).

5.3 Conduct of Business

Except as expressly contemplated by this Stock Purchase Agreement, as set forth on Schedule H, as otherwise directed or consented to in writing by U.S. Buyer or Sweden Buyer, which consent may not be unreasonably withheld, conditioned or delayed (it being understood that U.S. Buyer's or Sweden Buyer's failure to respond to a properly delivered request for such consent within five Business Days shall be deemed consent for the purposes of this Section 5.3 to all matters to which such request relates), as required pursuant to a legally binding obligation in existence on the date of this Stock Purchase Agreement or as required by Law, from the date of this Stock Purchase Agreement until Closing:

- (a) Subject to Section 5.3(b), Parent shall conduct and operate the Business, and shall cause Sweden Seller, each Target Company and each Company Subsidiary to cause the Business to be conducted and operated, only in the Ordinary Course of Business; and
 - (b) Parent shall (with respect to the Business), and shall cause Sweden Seller, each Target Company and each Company Subsidiary to:
 - (i) use commercially reasonable efforts to:
- (A) keep available the services of the Employees and preserve the relationships of the Target Companies and Company Subsidiaries with customers, suppliers, licensees, and distributors;
- (B) maintain the Assets in the same state of repair, order and condition as the Assets are in on the date of this Stock Purchase Agreement (except for normal wear and tear); and
 - (C) maintain in full force and effect all material Permits;
 - (ii) maintain its books and records, including any Company Records, in accordance with past practice;
- (iii) not take any action that would reasonably be expected to, result in any of the conditions set forth in Article 7 or Article 8 not being satisfied;
- (iv) not amend or otherwise modify in any respect, or propose an amendment or modification in any respect to, its Formation Document or its Bylaws;
- (v) not (A) issue, sell, pledge, grant, transfer or otherwise dispose of (or authorize the issuance, sale, pledge, grant, transfer or other disposition) of any Equity Interests, (B) create, permit, allow or suffer to exist any Encumbrance in respect of any Equity Interests, or (C) redeem, purchase or otherwise acquire any Equity Interest, directly or indirectly, or effect a split, reclassification or other change in, or of, any Equity Interest;
- (vi) not declare, set aside, make or pay any dividend or other distribution, payable in stock, property or otherwise, with respect to any equity or debt securities or interests;

- (vii) not make a loan or advance to an Employee or a member of its board of directors, or equivalent;
- (viii) not grant an increase in the compensation or benefits, or otherwise increase the compensation or benefits payable, or to become payable to a member of the board of directors of Sweden Company or a member of the Sweden Company Management Team, by more than 2% per annum except such increases as may generally apply to salaried employees of Parent or grant any rights to retention, severance or termination pay to, or, except in the Ordinary Course of Business, enter into any new (or amend any existing) employment, retention, severance or other Agreement with, any such Person;
- (ix) not enter into, materially amend or exercise a discretion (in each case which would reasonably be expected to have a material impact) under any existing Plan (or any plan, arrangement or other Agreement that would reasonably be expected to be a Plan if so adopted or amended) for an Employee or a member of their respective boards of directors, or equivalent, but excluding amendments or modifications as may generally apply to salaried employees of Parent or amending or modifying the period (from that currently provided for) of exercisability of options granted under any Plan or authorizing cash payments in exchange for any options to acquire securities or other interests in Parent;
- (x) not hire or terminate the employment of any Employee whose salary grade is at or above that of a Leadership Level 3 Employee, except (A) as required to fill any positions that are Leadership Level 3 or higher and become vacant after the date of this Stock Purchase Agreement, or (B) any termination for cause;
- (xi) except as required by Law or by the provisions of a Local Collective Agreement, not grant an increase in the aggregate compensation or benefits payable by the Target Companies or Company Subsidiaries by more than 1.5% in the aggregate calculated by reference to the aggregate wages, salaries, other remunerations and social security expenses (including pensions) payable for employees of Sweden Company and its subsidiaries, as reported in accordance with Swedish GAAP in the Ford VHC AB Annual Report for 2008 for Ford VHC AB and its subsidiaries on the date of the Company Balance Sheet;
- (xii) not enter into any commitment to make future capital expenditures that are in the aggregate greater than \$10,000,000, or that in any individual case is greater than \$2,000,000, except in accordance with the forecasts and plans, copies of which have been made available to Buyers;
 - (xiii) not assume, guarantee or endorse the obligations of, any Person in an individual amount in excess of \$3,000,000;
- (xiv) not incur a Liability that individually calls for payment by a Target Company or Company Subsidiary of an amount in excess of \$3,000,000 except for (A) contributions to Plans made in the Ordinary Course of Business or (B) subject to Section 5.3(xx), entering into Agreements relating to, or arising out of, a commercial arrangement consistent with past practice and entered into in the Ordinary Course of Business;

EXECUTION VERSION

- (xv) except for inventory sold or leased, not sell, transfer, lease, pledge or license on an exclusive basis to a third party, or place an Encumbrance (other than a Permitted Encumbrance) upon an Asset of a Target Company or Company Subsidiary with an original purchase price or cost basis exceeding \$5,000,000;
- (xvi) not acquire (including by merger or consolidation or acquisition of stock or assets) any interest in any corporation, partnership, other business organization or Person or any division thereof;
- (xvii) not make, revoke or change any Tax election, adopt or change any Tax accounting method, practice or period, grant or request a waiver or extension of any limitation on the period for audit and examination or assessment and collection of Tax, file any amended Tax Return or settle or compromise any contested Tax liability;
- (xviii) not change its methods of accounting or accounting practices, policies, principles or procedures, including revenue recognition procedures except as required by U.S. GAAP, Swedish GAAP or Law (which required changes will be provided to U.S. Buyer and Sweden Buyer in writing prior to any such change being made);
- (xix) not settle a Claim, in any individual case or instance in excess of \$1,000,000, against a Target Company or Company Subsidiary in connection with the Business or the assets of such Target Company or Company Subsidiary;
- (xx) not amend or modify in any material respect terminate, cancel or request any material modification or material amendment to any Material Contract, or enter into any Agreement that would, if entered into as of the date of this Stock Purchase Agreement, be a Material Contract;
- (xxi) with respect to the Owned Real Property or Leased Real Property, not enter into, modify or amend in any material respect, terminate, or renew any material lease of, or reciprocal easement Agreement, operating Agreement or other material Agreement relating to, any such real property;
- (xxii) except as permitted under Section 5.3(b)(xxvi), not sell or transfer or otherwise dispose of title to or ownership of any Company Intellectual Property, Parent Assigned Intellectual Property or Parent Licensed Intellectual Property;
- (xxiii) not grant any licenses of, or other rights to use, other than non-exclusive licenses or rights to use granted in the Ordinary Course of Business:
- (A) any Company Intellectual Property, Parent Assigned Intellectual Property or Sweden Company's rights under the Volvo Trademark Holding License Agreement; or
- (B) any Parent Licensed Intellectual Property that would prevent, restrict or otherwise adversely affect Buyer Group's ability to disclose, use, exploit or sublicense that Parent Licensed Intellectual Property in accordance with the Intellectual Property License Agreement;

- (xxiv) not place any new Encumbrances, other than non-exclusive licenses or rights to use granted in the Ordinary Course of Business, over the:
- (A) Company Intellectual Property, Parent Assigned Intellectual Property or Sweden Company's rights under the Volvo Trademark Holding License Agreement; or
- (B) Parent Licensed Intellectual Property, other than Encumbrances that will not prevent, restrict or otherwise adversely affect Sweden Company's ability to disclose, use, exploit or sublicense such Parent Licensed Intellectual Property in accordance with the Intellectual Property License Agreement;
- (xxv) not amend or modify in a manner that would materially adversely affect the rights that a Target Company or Company Subsidiary, as applicable, has to use Intellectual Property Rights under, or terminate, cancel or request any such modification or amendment to or in respect of:
 - (A) any Material Third Party License In or Material Third Party License Out;
 - (B) any IP Settlement Agreement;
 - (C) other than in the Ordinary Course of Business, any Third Party IT Agreement; or
 - (D) the Volvo Trademark Holding Shareholders' Agreement or AB Volvo License Agreement;
- (xxvi) pay all application, filing, registration, maintenance and renewal fees due in respect of, and take all other actions required to maintain, any Company Registered Intellectual Property, the registered Parent Assigned Intellectual Property or the registered Parent Licensed Intellectual Property, except to the extent that a Target Company, Company Subsidiary or member of Parent Group determines not to prosecute, maintain or renew, any Intellectual Property Rights which are not material to the Business;
 - (xxvii) not make any admission or take any action in respect of, or settle, any material Claim that relates to any of the following:
 - (A) Company Intellectual Property or Parent Assigned Intellectual Property; or
- (B) Parent Licensed Intellectual Property, except to the extent that such admission, action or settlement would not prevent, restrict or adversely affect Buyer Group's ability to use such Parent Licensed Intellectual Property in accordance with the Intellectual Property License Agreement;
- (xxviii) use commercially reasonable efforts to procure that Volvo Trademark Holding AB reasonably undertakes and/or continues ongoing administrative actions, including countering oppositions against its own applications and lodging oppositions in respect of third party applications, which Volvo Trademark Holding AB determines are reasonably required to safeguard Volvo Trademark

Holding AB's (and its licensees') use of the VOLVO mark (and transliterations of such mark) in all relevant territories and classes;

- (xxix) not vote, or otherwise exercise any rights, in favor of causing any Non-Controlled Company to act contrary to the covenants in Section 5.3(b)(xxii) to Section 5.3(b)(xxvii) in respect of any Non-Controlled IP; and
 - (xxx) not authorize or enter into any Agreement or otherwise make any legally enforceable commitment to do any of the foregoing.

For the avoidance of doubt, without prejudice to Section 5.3(b)(xxix), any actions of the type set forth in this Section 5.3 taken by a Non-Controlled Company with respect to Non-Controlled IP shall not be deemed actions taken by Parent, Sweden Seller, a Target Company or any Company Subsidiary.

5.4 Access to Company Information

From the date of this Stock Purchase Agreement until Closing, except as otherwise prohibited by Law or the terms of an Agreement entered into prior to the date of this Stock Purchase Agreement, or as would be reasonably expected to violate the attorney-client privilege of Parent, Sweden Seller, Target Companies or Company Subsidiaries with their respective legal counsel, Parent and Sweden Seller shall provide, and Parent shall cause the Target Companies and Company Subsidiaries to provide, to Buyers or representatives of Buyers reasonable access upon reasonable prior written notice, during normal business hours, to the offices, books, records, working papers, Tax returns, members of the boards of directors, Employees, consultants and contractors of each Target Company and each Company Subsidiary, and shall furnish representatives of Buyers with such financial and operating data and other information with respect to the Business and Assets, including the reports of the IT Steering Committee (which will include any updates to the IT Transition Plan), as Buyers may reasonably request, so long as such access shall not result in an unreasonable interference with the operation of the Business. For the avoidance of doubt, this Section 5.4 shall not apply to any information, data or materials related to the design, development or manufacture of any vehicles or components, any vehicle technology or cycle plans other than the Documented Plans.

5.5 No Control of the Business

- (a) U.S. Buyer and Sweden Buyer acknowledge and agree that nothing contained in this Stock Purchase Agreement shall give U.S. Buyer or Sweden Buyer, directly or indirectly, the right to control or direct the operations of a Target Company or Company Subsidiary prior to Closing.
- (b) Prior to Closing, Parent shall cause the Target Companies to exercise, consistent with the terms and conditions of this Stock Purchase Agreement, complete control and supervision over their respective operations, and the operations of their respective Subsidiaries as exercised on the date of this Stock Purchase Agreement. Notwithstanding any provision to the contrary in this Stock Purchase Agreement, no consent of U.S. Buyer or Sweden Buyer shall be required with respect to any matter set forth in Section 5.3 or elsewhere in this Stock Purchase Agreement to the extent requiring such consent would violate Law.

5.6 Notification of Certain Matters; Supplemental Disclosure

- (a) Parent and Sweden Seller shall have the continuing obligation, from the date of this Stock Purchase Agreement until Closing, reasonably promptly to notify Buyers in writing with respect to any matter arising subsequent to the date of this Stock Purchase Agreement that would reasonably be expected to prevent the satisfaction of the conditions precedent in Section 8.1, provided that (i) no such notification nor updates and revisions to Schedule B shall have any effect for the purpose of determining the satisfaction of the conditions set forth in Article 8 and (ii) no such notification nor revision to Schedule B shall have effect with respect to any entitlement to indemnification as set forth in Section 10.2.
- (b) Parent shall reasonably promptly provide to Buyers any information Buyers may reasonably request with respect to any matter of which Buyers are notified pursuant to Section 5.6(a).

5.7 Contract Consents

- (a) From the date of this Stock Purchase Agreement until Closing, Parent shall use commercially reasonable efforts at Parent's cost to procure third party consents required:
- (i) with respect to Third Party IT Agreements, to enable Parent to perform the Parent IT Transition Services and each Target Company to perform the Target Companies IT Transition Services; and
- (ii) with respect to any other Material Contracts and IP Contracts (other than the Third Party IT Agreements which are the subject of Section 5.7(a)(i)), in each case to which a Target Company or Company Subsidiary is a party, to allow the Target Companies and Company Subsidiaries to continue to benefit from such Material Contracts and IP Contracts on the same terms following Closing.
- (b) If and to the extent that any Intellectual Property Rights (x) used by a Target Company or Company Subsidiary or (y) with respect to the technologies specified in the Documented Plans, are licensed under a Consent Contract and cannot be used by the applicable Target Company or Company Subsidiary after Closing without obtaining a consent, approval, waiver or the like from a third party, from the date of this Stock Purchase Agreement until Closing, Parent shall, or shall cause the relevant member of Parent Group to, use commercially reasonable efforts at its own cost to, either:
 - (i) obtain such third party consent, approval, waiver or the like; or
- (ii) to the extent permitted under the relevant Consent Contract, grant a sublicense to the applicable Target Company or Company Subsidiary to use, in substantially the same manner as used or permitted to be used prior to Closing, the Intellectual Property Rights licensed under such Consent Contract.
- (c) Parent shall not, and shall cause the relevant member of Parent Group not to, terminate or amend any Consent Contract in any way that might prevent, restrict or otherwise adversely affect the use of any Intellectual Property Rights by the applicable Target Company or Company Subsidiary, without the prior written consent of such Target Company or Company Subsidiary.

- (d) Buyers shall use commercially reasonable efforts, at Parent's request, to cooperate with Parent in connection with Parent's obligations in Section 5.7(a) and Section 5.7(b).
- (e) For the avoidance of doubt, nothing in this Stock Purchase Agreement shall require Parent or Buyers directly or indirectly to pay any amounts to third parties in order to procure the consents described in this Section 5.7.

5.8 Termination of IP Intercompany Agreements; Termination of Powers of Attorney

- (a) Prior to the Closing Date, Parent shall, or shall procure that the relevant members of Parent Group shall use commercially reasonable efforts to (a) execute a deed of termination with effect as of Closing for each Intercompany Agreement under which Intellectual Property Rights are licensed (excluding any Surviving Contracts, Shared Contracts or Excluded Contracts) and (b) deliver to Buyers copies of all such deeds of termination. If, notwithstanding compliance with this Section 5.8(a), a deed of termination is not executed for any such Intercompany Agreement prior to Closing, Parent shall, or shall cause the relevant member of Parent Group to, continue to use commercially reasonable efforts after Closing to (a) execute promptly a deed of termination in respect of such Intercompany Agreements, to take effect retroactively as of Closing, and (b) deliver to Buyers copies of such deeds of termination.
- (b) Prior to the Closing Date, to the extent Buyers request in writing that Parent and Sweden Seller terminate any specific powers of attorney with effect as of Closing, Parent and Sweden Seller shall, and shall cause Target Companies and Company Subsidiaries to, use commercially reasonable efforts to terminate any such powers of attorney and deliver copies of any such terminations to Buyers.

5.9 Information Barriers Protocol

Prior to the Closing Date, Parent shall, and shall cause the Target Companies and Company Subsidiaries to provide written notification to each of the Relevant Employees of Parent Group and Target Companies and Company Subsidiaries, which notification shall advise of the existence and purpose of the Information Barriers Protocol set forth on Schedule I.

5.10 Benefit Plans in Canada

- (a) At or prior to Closing, Parent shall, and shall cause Ford Motor Company of Canada, Limited, to procure that benefit accruals applicable to each Employee participating in Ford Motor Company of Canada, Limited Pension Plan #3 ("Pension Plan #3") or the Ford Motor Company of Canada Excess Plan shall cease and each such Employee shall be fully vested in the Employee's accrued benefit under such Plan.
- (b) At or prior to Closing, Parent shall, and shall cause the Company Subsidiary in Canada to, use commercially reasonable efforts to (i) adopt and establish two new defined contribution plans, and, where required by Law, related trusts, (the "Replacement Canada Defined Contribution Plans") with terms and conditions that are comparable to the terms and conditions of the Ford Motor Company of Canada, Limited Pension Plan #6 ("Pension Plan #6") and the Ford Savings and Stock Investment Program (the "SSIP"), respectively; (ii) make the applications and filings to obtain any approvals from a Governmental Entity that are required by Law for each such Replacement Canada Defined Contribution Plan to operate as a retirement plan in Canada; and (iii) procure that, at or prior to Closing, any Employee who is participating in Pension Plan #3, Pension Plan #6 or the SSIP no longer participates as an active

participant accruing benefits in such plans and commences participation in the Replacement Canada Defined Contribution Plans.

5.11 Benefit Plans in the U.S.

- (a) At or prior to Closing, Parent shall, and shall cause the relevant member of Parent Group to, procure that benefit accruals applicable to each Employee participating in the Ford General Retirement Plan or Ford UAW Retirement Plan (collectively, the "U.S. Employee Pensions Plans," and such Employees, the "U.S. Employee Pension Participants") shall cease and that each such U.S. Employee Pension Participant who is not a U.S. Hourly Employee Pension Participant shall be fully vested in the Employee's accrued benefit under the Plan.
- (b) At or prior to Closing, Parent shall use commercially reasonable efforts to amend the Pension Plan of Volvo Cars of North America (the "Volvo U.S. Defined Benefit Plan") to provide for the participation of the U.S. Employee Pension Participants to whom a Local Collective Agreement applies (the "U.S. Hourly Employee Pension Participants") pursuant to such Local Collective Agreements.
- (c) At or prior to Closing, Parent shall, and shall cause the Company Subsidiary in the United States to, use commercially reasonable efforts to (i) adopt and establish a new defined contribution plan, and, where required by Law, a related trust (the "Replacement U.S. Defined Contribution Plan"), with terms and conditions that are comparable to the terms and conditions of the Ford Retirement Plan (the "FRP"); (ii) make the applications and filings to obtain any approvals from a Governmental Entity that are required by Law for such Replacement U.S. Defined Contribution Plan to operate as a retirement plan in the United States; and (iii) procure that, at or prior to Closing, any Employee who is participating in the FRP no longer participates as an active participant accruing benefits in such plan and commences participation in the Replacement U.S. Defined Contribution Plan.
- (d) At or prior to Closing, Parent shall use commercially reasonable efforts to procure that any U.S. Employee Pension Participant who is not a U.S. Hourly Employee Pension Participant shall commence participation in the Replacement U.S. Defined Contribution Plan.

5.12 Defined Benefit Plan in the Netherlands

Parent shall, and shall cause the applicable Company Subsidiary in the Netherlands to, use commercially reasonable efforts to:

- (a) prior to Closing, adopt and establish a new funding vehicle (the "Volvo Netherlands Funding Vehicle") to finance the future benefit accruals of Employees resident in the Netherlands who participate in the Volvo Netherlands Pension Plan (the "Volvo Netherlands Defined Benefit Plan"); and
- (b) prior to Closing, make the applications and filings to obtain any approvals from a Governmental Entity that are required by Law for the Volvo Netherlands Funding Vehicle to operate in the Netherlands.

5.13 Replacement of Insurance Policies

(a) Parent shall arrange for the separate insurance policies in respect of the Target Companies and the Company Subsidiaries and their respective businesses, assets, directors and

employees set forth on Schedule R (the "Replacement Policies") to take effect on the dates specified in Schedule R.

(b) For the avoidance of doubt, nothing contained in this Stock Purchase Agreement shall extinguish, limit or otherwise affect or be deemed to affect any obligation on the part of any insurer under any policy of insurance that provides cover to any member of Parent Group on or before the Closing Date.

5.14 IT Transition Plan

From the date of this Stock Purchase Agreement until Closing, Parent, Target Companies and Company Subsidiaries shall use commercially reasonable efforts to comply in all material respects with their respective obligations to implement section 5.1.2, section 5.1.3, section 5.1.4 and section 12.10 of the Form of Information Technology Transitional Services Agreement attached as Exhibit VI (giving effect to all definitions used in such sections and the interpretations set forth in section 1.2.1 thereof), as if such Agreement were in full force and effect between Parent and Sweden Company for the purpose of this Section 5.14.

5.15 Replacement Vehicle Finance Arrangements

From the date of this Stock Purchase Agreement until Closing, Parent shall cause Sweden Company to use commercially reasonable efforts to implement replacement Vehicle Finance Arrangements in all Markets (the "Replacement Vehicle Finance Arrangements").

5.16 Leakage

From the Measuring Time until Closing, Parent and Sweden Seller shall not permit, and Parent shall procure that the Target Companies and Company Subsidiaries and each of their respective directors, officers, employees and Affiliates shall not permit, there to be (a) any transaction with (whether an acquisition, disposal, transaction at undervalue or overvalue, or otherwise), payment or gift to, guarantee of any obligation of, any member of Parent Group or related party by a Target Company or Company Subsidiary, (b) payment (or declaration) of any dividends, incurrence of management fees, charges or liabilities (including transaction bonuses and transaction related costs), or granting of indemnities or guarantees or redemption, repurchase or reductions in share capital, by a Target Company or Company Subsidiary in favor of any member of Parent Group or related party or (c) any Agreement or arrangement to do any of the foregoing and any Tax, or social security, relating to any of the foregoing, provided that Target Companies and Company Subsidiaries shall be permitted to engage in transactions and arrangements for goods and services consistent with past practice. For the purposes of this Section 5.16, "related party" shall mean any Person with respect to whom any member of Parent Group is (x) entitled to exercise or control the exercise of 10% or more of the total combined voting power of all classes of shares entitled to vote or (y) the owner of an interest of 10% or more in the capital or profits of such Person.

5.17 Non-solicitation

Until this Stock Purchase Agreement shall have been terminated pursuant to its terms, Parent and Sweden Seller will not, and will cause the Target Companies and Company Subsidiaries and its and their respective directors, officers, employees, Affiliates and other agents and representatives (including any

investment banking, legal or accounting firm retained by any of them and any individual member or employee of the foregoing) (each, an "Agent") not to, directly or indirectly: (a) initiate, knowingly solicit or seek any inquiries or the making or implementation of any proposal or offer (including any proposal or offer to its stockholders or members or any of them) with respect to a merger, acquisition, consolidation, recapitalization, liquidation, dissolution, equity investment or similar transaction involving the Target Companies or Company Subsidiaries, or any purchase of all or any substantial portion of the assets that would be material to the operation of the Business as conducted as of the date of this Stock Purchase Agreement, or of the securities or interests of the Target Companies or Company Subsidiaries, regardless of the form of transaction (a "Proposal") (b) engage in any negotiations concerning, or provide any confidential or non-public information or data to, or have any substantive discussions with, any Person relating to a Proposal, (c) otherwise cooperate in any effort or attempt to make, implement or accept a Proposal, or (d) enter into an Agreement with any Person, other than Buyers or their Affiliates, relating to a Proposal. Parent shall not directly or indirectly release any Persons other than U.S. Buyer or Sweden Buyer from any confidentiality or standstill provisions or Agreements relating to any transaction of the kind contemplated by this Section 5.17.

5.18 Financing

From the date of this Stock Purchase Agreement until Closing:

- (a) Buyers shall (i) maintain in effect the Binding Acquisition Financing Commitment Letters, (ii) use commercially reasonable efforts to negotiate definitive Agreements with respect thereto on terms and conditions contemplated by the Binding Acquisition Financing Commitment Letters and execute and deliver to Parent a copy thereof promptly following such execution, (iii) use commercially reasonable efforts to satisfy on a timely basis all conditions applicable to Buyers in the Binding Acquisition Financing Commitment Letters that are within the control of a member of Buyer Group and comply with all obligations of a member of Buyer Group thereunder and (iv) enforce the rights of each member of Buyer Group under the Binding Acquisition Financing Commitment Letters in the event of a breach by the financing sources, including seeking specific performance of the parties thereunder.
- (b) In the event that all conditions to the Binding Acquisition Financing Commitment Letters have been satisfied, or upon funding will be satisfied, Buyers shall use their commercially reasonable efforts to cause the Persons providing such Acquisition Financing to fund on the Closing Date the Acquisition Financing (together with funds from other funding sources) necessary to consummate the Closing, and to fund, when applicable, any portion of the Acquisition Financing required to satisfy any obligations of Buyers in connection with this Stock Purchase Agreement, in each case, including by Buyers taking enforcement action (including seeking specific performance) to cause such Persons to fund such Acquisition Financing.
- (c) If any portion of the Acquisition Financing becomes unavailable, or Buyers become aware of any event or circumstance that makes any portion of the Acquisition Financing unavailable on the terms and conditions contemplated in the Binding Acquisition Financing Commitment Letters, Buyers shall use their commercially reasonable efforts to arrange and obtain alternative financing from alternative financing sources in an equivalent amount upon conditions no less favorable than those in the Binding Acquisition Financing Commitment Letters as promptly as practicable following the occurrence of such event. Buyers shall give Parent prompt notice of any material breach by any party to the Binding Acquisition Financing Commitment Letters or of any such letter not likely to be satisfied, in each case, of which Buyers become aware, or any termination of the Binding Acquisition Financing Commitment Letters.

- Buyers shall use commercially reasonable efforts to secure a working capital facility (other than the Entrustment Loan Agreement) from a bank prior to Closing for the operation of Sweden Company which (i) is in an amount equal to or greater than the amount set forth in Schedule M Part 1, (ii) is on terms and conditions that, when taken together, are customary for such working capital facilities and consistent with the reasonable requirements of the Business and (iii) permits all available funds to be disbursed and used outside the PRC (the "Working Capital Facility") provided, however, that entry into any Working Capital Facility shall be at the absolute discretion of Buyers.
- (e) Buyers shall keep Parent reasonably informed on a reasonably current basis of the status of its efforts to arrange the Acquisition Financing and the Working Capital Facility.
- (f) Parent shall cause the Target Companies to provide reasonable cooperation and assistance to Buyers in connection with the arrangement of the Acquisition Financing and the Working Capital Facility, so long as such cooperation and assistance would not result in an unreasonable interference with the operation of the Business, provided that nothing in this Section 5.18 shall require Parent to prepare or provide any financial statements or financial or operating reports not otherwise included in the Due Diligence Materials or prepared in the Ordinary Course of Business.

5.19 Retiree Medical Plans

Parent shall, and shall cause U.S. Company to, use commercially reasonable efforts to:

- (a) prior to Closing, adopt and establish a new retiree medical program for Employees resident in the United States (other than for those Represented Employees employed in Suwannee, Georgia and Ontario, California); and
- (b) on or prior to Closing, provide that Employees resident in the United States will no longer be eligible to receive benefits pursuant to the retiree medical program under the Ford Motor Company Health Care Plan for Salaried Employees (the "Ford Retiree Medical Program"), except for such Employees who are fully eligible to commence receiving benefits pursuant to such plan as of Closing, which Employees shall be eligible to commence receiving benefits pursuant to such plan (on the same basis as other employees of Parent retiring at the same time with similar eligibility characteristics) effective as of their retirement from U.S. Company.

For the avoidance of doubt, Buyers are not assuming any Liability under the Ford Retiree Medical Program and, except to the extent required under the terms of a Local Collective Agreement for Represented Employees, no accrued service under the Ford Retiree Medical Program will be credited to the new retiree medical program established pursuant to Section 5.19(a).

5.20 UK Pension Matters

(a) From the date of this Stock Purchase Agreement until Closing, Parent may at its option make an application (and may at any time withdraw or abandon such application) for clearance statements under sections 42 and 46 of the UK Pensions Act (the "Clearance Statements") in relation to the Volvo Car UK Pension Scheme and the transactions under or referred to in the Transaction Documents and the Buyers' associated financing of these transactions. If Parent elects not to make such an application or withdraws or abandons such application, Buyers may, at Buyers' option, make or continue with such an application in favor of Buyers and such other Buyer Affiliates as shall be specified by Buyers, on the terms set out in this Section 5.20. The application for the Clearance Statements shall be:

- (i) in a form agreed between Parent and the Buyers;
- (ii) where the application is made by Parent, in favor of Parent, Target Companies and Company Subsidiaries and such of Parent's Affiliates and their respective directors and officers as Parent shall determine to be appropriate and, at Buyers' option (but in respect only of such of the Clearance Statements as are issued under section 42 of the UK Pensions Act, if the UK Pensions Regulator so determines), in favor of Buyers and such other Buyer Affiliates as shall be specified by Buyers; and
- (iii) on terms that include a contribution to the Volvo Car UK Pension Scheme from the Company Subsidiary in the United Kingdom at or prior to Closing of an amount determined by Parent in its sole discretion.
 - (b) Parent and the Buyers agree that each of them (to the extent permitted by Law):
 - (i) shall cooperate in the preparation and submission to the UK Pensions Regulator of the application for the Clearance Statements;
- (ii) shall promptly provide the other with information relevant to the application for the Clearance Statements which information shall be complete and accurate in all material respects and shall respond reasonably and promptly to any questions arising;
- (iii) shall not make representations about the other to the trustees of the Volvo Car UK Pension Scheme or to the UK Pensions Regulator without either the prior approval of the other or the other being present at any telephone call or meeting at which such representation is made;
- (iv) hereby consents to disclosure of the information provided under Section 5.20(b)(ii) above to the UK Pensions Regulator and to the trustees of the Volvo Car UK Pension Scheme (subject to such confidentiality undertakings from the trustees as Parent and the Buyers shall reasonably require); and
- (v) shall be given the opportunity on reasonable notice to attend any meeting or telephone call with the trustees of the Volvo Car UK Pension Scheme or the UK Pensions Regulator in connection with the application for the Clearance Statements.

5.21 CFMA Payments

Parent shall cause Sweden Company to cause CFMA, with effect from and after Closing, to make the following technology licensing payments to Sweden Company for the Volvo branded products manufactured by CFMA: (a) the "Licensing Fee" payable to Ford Global following the Closing Date pursuant to article 4.2.1 of the S40 TLC, and (b) the "Recurring Royalty" payable to Ford Global following the Closing Date pursuant to article 4.2.1 of the S80 TLC. The Parties acknowledge and agree that all other terms and conditions of the S40 TLC and S80 TLC shall remain unchanged and in full force and effect.

5.22 Allocation of Special GHG Credits

- (a) At or prior to Closing, Parent shall allocate to Sweden Company pursuant to the form of letter agreement between Parent and Sweden Company attached hereto as Exhibit XXIV the following GHG Credits (the "Special GHG Credits"):
 - (i) all 2009 MY Volvo Credits and 2010 MY Volvo Credits, and
- (ii) an amount of Ford Credits equal to 200,000 tons of GHG Credits, reduced by the aggregate amount of 2009 MY Volvo Credits and 2010 MY Volvo Credits allocated by Parent to Sweden Company pursuant to clause (i).
- (b) Buyers and Parent agree to take any and all actions reasonably necessary under the EPA GHG Emissions Regulations to give effect to the allocation of Special GHG Credits under this Section 5.22.

5.23 Patent License Agreement

- (a) Prior to Closing, Parent may enter into a Patent License Agreement with Ford Global (the "PLA") pursuant to which Ford Global grants a license to Ford and its current Affiliates with respect to the Volvo Assigned Patents (excluding the SPA Platform Patents and the VED and VEP Patents) provided that such license must be on the following terms (the "Required Terms"): equivalent terms to those terms in the Intellectual Property License Agreement that will apply to the Volvo Limited License Vehicle IP upon the Intellectual Property License Agreement becoming effective at Closing, except that:
- (i) the license under the PLA in respect of the Volvo Hybrid Patents and the P2 Platform Patents shall be on the same terms as the license granted in respect of the Volvo Limited License Separation IP that constitutes an Improvement to, or of, the Volvo Hybrid Patents, the P2 Platform Existing Patents or the P2 Platform Patents under the Intellectual Property License Agreement; and
 - (ii) the PLA shall not include:
- (A) an indemnity from Ford Global to Ford on equivalent terms to that given by Ford under sections 16.1 or 16.2 of the Intellectual Property License Agreement;
- (B) terms equivalent to those in section 17.2 of the Intellectual Property License Agreement, provided that this shall not limit or amend the terms of section 17.2 of the Intellectual Property License Agreement in any way;
- (C) terms of the Intellectual Property License Agreement that are binding on Ford and its current Affiliates under the Intellectual Property License Agreement with respect to the Volvo Assigned Patents licensed under the PLA; and
- (D) a right for Ford or any of its Affiliates to encumber by way of charge, pledge, or assignment of any of, its rights and benefits relating to the Volvo Assigned Patents licensed under the PLA; provided that this shall not limit any rights of Ford or any of its Affiliates under section 24.2.3 of the Intellectual Property License Agreement.

(b) Capitalised terms used in this Section 5.23 that are not defined in this Stock Purchase Agreement shall have the meanings given to them in the Intellectual Property License Agreement.

5.24 Working Capital Forecast

Parent shall deliver to Buyers not less than 15 Business Days before Closing (or the anticipated Closing, as applicable) Parent's good faith forecast of the total cash flow (including working capital funding) requirements of the Business as at Closing and for 60 calendar days following Closing (the "Working Capital Forecast").

6. ADDITIONAL AGREEMENTS

6.1 U.S. Tax Treatment of the Sale and Purchase of the U.S. Company Interest

Each of Parent, U.S. Buyer and Sweden Buyer acknowledges and agrees that the purchase and sale of the U.S. Company Interest pursuant to this Agreement shall be treated as a purchase and sale of the assets of U.S. Company for U.S. federal income Tax purposes.

6.2 Tax Liability

- (a) Subject to the provisions of Section 6.3 and any applicable limitations set forth in Article 10 (but without regard to any disclosure in Schedule B) and with effect from Closing, Parent covenants to pay U.S. Buyer and Sweden Buyer (as applicable) an amount equal to:
- (i) any Actual Tax Liability of a Target Company or Company Subsidiary for (A) a taxable year or taxable period that ends on or before the Closing Date (a "**Pre-Closing Period**") and (B) with respect to a taxable period that includes but does not end on the Closing Date (a "**Straddle Period**"), for the portion of such Straddle Period ending on and including the Closing Date;
- (ii) any Actual Tax Liability which is properly attributable to a member of Parent Tax Group (including Tax for which a member of Parent Tax Group is primarily liable but has failed to discharge, and Tax for which a member of Buyer Group or Geely Group or a Target Company or Company Subsidiary is jointly liable but where the amount is properly attributable to a member of Parent Tax Group);
 - (iii) any liability to:
 - (A) make to any member of Parent Tax Group any payment for Group Relief; or
- (B) repay to any member of Parent Tax Group the whole or any part of any payment for Group Relief previously received,

in each case pursuant to any agreement or arrangement entered into by any Target Company or Company Subsidiary on or before Closing;

(iv) any payment by a Target Company or Company Subsidiary (for purposes of this Section 6.2(a)(iv), the "payor") to a Person other than a Target Company, Company Subsidiary or a member of Buyer Group or Geely Group (for purposes of this Section 6.2(a)(iv), the "payee") pursuant to any breach of representation or warranty in respect of or relating to Tax or pursuant to a covenant to pay or to indemnify the payee for Taxes imposed upon or economically borne by the payee with respect to any period ended on or before Closing, and any costs reasonably and properly suffered or incurred by any Target Company or Company Subsidiary as a result of or in connection with any such claim, provided that such payment is required pursuant to an Agreement entered into prior to Closing pursuant to which Agreement the payor specifically agreed to indemnify or pay the payee for such Taxes or gave such representation or warranty in respect of or relating to Tax in connection with the sale and purchase of a business or part of a business or a business entity:

(v) any Deemed Tax Liability,

provided, however, that Parent shall not be liable to make a payment (x) in respect of any Tax or Tax Liability to the extent that provision or reserve in respect of such Tax or Tax Liability has been made in the Final Closing Statement and such Tax or Tax Liability is reflected in and reduces Net Working Capital or is reflected in and increases Net External Indebtedness, or (y) in respect of any Tax or Tax Liability described in Section 6.2(a)(i) (and, for the avoidance of doubt, not described in Section 6.2(a)(ii)) to the extent that such Tax or Tax Liability is imposed on a Target Company or Company Subsidiary, or for which a Target Company or Company Subsidiary may otherwise be liable, as a result of transactions occurring on the Closing Date subsequent to Closing (Taxes described in clauses (x) or (y) of this proviso, collectively, the "Excluded Taxes"). Insofar as provided in Section 6.2(c), Parent or Sweden Seller (as applicable) shall be entitled to any refund of Taxes allocable to any Pre-Closing Period and, with respect to a Straddle Period, the portion of such Straddle Period ending on and including the Closing Date, where, or to the extent that, such refund is not a Buyer's Relief, and does not arise from the use of a Buyer's Relief.

- (b) With effect from Closing, Buyer Group shall be liable for and pay and, to the extent that Buyer Group fails to pay and any member of Parent Group is required to pay, U.S. Buyer and Sweden Buyer shall pay Sweden Seller and Parent (as applicable), subject to the provisions of Section 6.3 and any applicable limitations set forth in Article 10, an amount equal to (i) all Taxes imposed on a Target Company or Company Subsidiary for any taxable year or taxable period that begins after the Closing Date and, with respect to a Straddle Period, for the portion of such Straddle Period beginning after the Closing Date (other than Taxes for which Parent is liable pursuant to the provisions of Section 6.2(a)), and (ii) Excluded Taxes.
- (c) If, following Closing, a member of Buyer Group receives a refund of, or credit for Taxes of a Target Company or Company Subsidiary allocable to (i) any Pre-Closing Period or (ii) with respect to a Straddle Period, the portion of such Straddle Period ending on and including the Closing Date (not being a refund or credit which is a Buyer's Relief, or a refund or credit which arises from the use of a Buyer's Relief), then Buyers shall pay to Parent an amount equal to such refund or credit; provided that this Section 6.2(c) shall not apply where the Tax refund or credit obtained arises in respect of an overpayment of Tax prior to Closing where such overpayment would not have been an overpayment but for any deduction, loss or credit arising in and allocable to a Tax period or portion of a Straddle Period (determined in accordance with the provisions of Section 6.2(d)) following Closing. Subject to Section 6.2(e), Buyers shall be entitled to any refund of, or credit for, Taxes allocable to a taxable year or period that begins after the Closing Date and, with respect to a Straddle Period, the portion of such Straddle Period beginning after the Closing Date, and if any such amount is received by any member of

Parent Group then Parent shall pay such amount to U.S. Buyer or Sweden Buyer on behalf of the relevant Target Company or Company Subsidiary. Any amount of refund or credit payable by Buyers or Parent pursuant to this Sections 6.2(c) shall, in the case of a refund, be paid promptly following receipt, and in the case of a credit, shall be paid promptly following the date on which such credit results in a decrease of Liability for Tax, and shall be paid net of any Tax suffered on such refund or credit.

- (d) For purposes of this Section 6.2, Taxes for a Straddle Period will be allocated between (x) the portion of the Straddle Period that begins at the beginning of the Straddle Period and ends at the end of the Closing Date, and (y) the remaining portion of the Straddle Period in the following manner:
- (i) Any Tax other than real and personal property Taxes described in clause (ii) of this Section 6.2(d) will be allocated based on a "closing of the books" as of the end of the Closing Date.
- (ii) Real and personal property Taxes with respect to property held by a Target Company or Company Subsidiary at Closing will be prorated based on the ratio of the number of days in the portion of the Straddle Period ending on the Closing Date to the number of days in the remaining portion of the Straddle Period; provided, however, that if the transactions contemplated by this Stock Purchase Agreement result in the reassessment of the value of property owned by a Target Company or Company Subsidiary to a higher amount for property Tax purposes, or the imposition of property Taxes at a rate that is different than the rate that would have been imposed if such transactions had not occurred, then the calculation of the amount of such property Taxes allocable to the portion of the Straddle Period ending on the Closing Date shall be based on the assessed value and Tax rate that would have applied had such transactions not occurred.
- (e) If any payment is made by Parent pursuant to Section 6.2(a) in respect of a Tax Liability or other matter, and there is a decrease in the Tax Liability of a member of Buyer Group or a successor thereof for a taxable year or taxable period beginning after the Closing Date or for the portion of any Straddle Period beginning after the Closing Date which decrease would not have arisen but for the Tax Liability or other matter in respect of which the payment by Parent was made or the circumstances giving rise thereto, U.S. Buyer or Sweden Buyer shall refund to Parent an amount equal to the amount of such decrease (less any Tax incurred in respect of such decrease), up to a maximum amount equal to the payment made under Section 6.2(a) in respect of the relevant Tax Liability (less any Tax suffered on that payment, except to the extent that the payment made by U.S. Buyer or Sweden Buyer to Parent pursuant to this Section 6.2(e) (the "Tax Refund Payment") results in a decrease in liability to Tax of a member of Buyer Group which would not have arisen but for the making of the Tax Refund Payment. For the avoidance of doubt, where the Tax Refund Payment results in the cancellation or reduction of the Tax which would otherwise be suffered on the payment made by Parent in respect of the relevant Tax Liability, such cancellation or reduction shall be a decrease in liability to Tax of a member of Buyer Group for the purposes of the previous sentence), provided that no amount shall be refunded where the decrease is in an amount of Tax which would otherwise have given rise to a claim under Section 6.2(a). The provisions of this Section 6.2(e) shall apply, *mutatis mutandis*, where a payment is made by U.S. Buyer or Sweden Buyer pursuant to Section 6.2(b), and there is a resulting decrease in the Tax liability of a member of Parent Group, as if the references to "Parent" and "Sweden Seller" were references to "U.S. Buyer" and "Sweden Buyer" and read with any other necessary modifications.
- (f) If Parent is obligated to make a payment to U.S. Buyer or Sweden Buyer under Section 6.2(a)(i) or Section 6.2(a)(ii) with respect to an Actual Tax Liability of a Target Company or a Company Subsidiary and if Parent does not make such payment by the close of a due date that is the later of (i) the date such Actual Tax Liability was paid to the applicable Tax Authority and (ii) the tenth

Business Day following the date U.S. Buyer or Sweden Buyer, as the case may be, delivers a written demand for such payment to Parent, then the payment which Parent is obligated to make shall be increased by an amount equal to the Default Interest accruing on such payment from the close of such due date to the date Parent makes such payment (including such interest). If Parent is obligated to make a payment to U.S. Buyer or Sweden Buyer under Section 6.2(a) (iv) with respect to any payment made or costs suffered or incurred by a Target Company or Company Subsidiary to which Section 6.2(a)(iv) applies and if Parent does not make such payment by the close of a due date that is the later of (x) the date the Target Company or Company Subsidiary involved made such payment or suffered or incurred such costs and (y) the tenth Business Day following the date U.S. Buyer or Sweden Buyer, as the case may be, delivers a written demand for such payment to Parent, then the payment which Parent is obligated to make shall be increased by an amount equal to the Default Interest accruing on such payment from the close of such due date to the date Parent makes such payment (including such interest). If U.S. Buyer or Sweden Buyer is obligated to make a payment to Parent or Sweden Seller under Section 6.2(b) with respect to an Actual Tax Liability of a Target Company or a Company Subsidiary and if U.S. Buyer or Sweden Buyer, as the case may be, does not make such payment by the close of a due date that is the later of (A) the date such Actual Tax Liability was paid to the applicable Tax Authority and (B) the tenth Business Day following the date Parent or Sweden Seller, as the case may be, delivers a written demand for such payment to U.S. Buyer or Sweden Buyer, as the case may be, then the payment which U.S. Buyer or Sweden Buyer is obligated to make shall be increased by an amount equal to the Default Interest accruing on such payment from the close of such due date to the date U.S. Buyer or Sweden Buyer makes such payment (including such interest). For the avoidance of doubt and to avoid a duplicative accrual of interest, no interest shall accrue pursuant to the foregoing provisions of this Section 6.2(f) with respect to any day for which the Actual Tax Liability itself has increased due to the accrual of interest payable to the Tax Authority for such day or with respect to any day for which interest has accrued pursuant to the provisions of Section 6.4(c)(v). Notwithstanding any provision of this Stock Purchase Agreement to the contrary, the provisions of Section 12.6(d) shall not apply to the payment of interest pursuant to the provisions of this Section 6.2(f).

6.3 Transfer Taxes

Notwithstanding any provision to the contrary in this Stock Purchase Agreement, any real property transfer, stamp Tax, stock transfer Tax, or other similar transfer Tax (together "**Transfer Taxes**") imposed on the transactions contemplated by this Stock Purchase Agreement shall be borne equally by (i) U.S. Buyer and Sweden Buyer, on the one hand, and (ii) Parent and Sweden Seller, on the other hand, except to the extent that such Transfer Taxes are imposed by a Tax Authority in Sweden and Sweden Buyer is legally required to account for such Transfer Taxes, in which case such Transfer Taxes shall be borne by Sweden Buyer.

6.4 Tax Returns

(a) Subject to and in accordance with the provisions of this Section 6.4, Parent or its duly authorized agents shall prepare, file or cause to be filed when due, taking into account all extensions properly obtained, all Tax Returns that are required to be filed on or before the Closing Date by or with respect to each Target Company and each Company Subsidiary, and Parent shall timely remit or cause to be timely remitted Taxes shown due and payable on such Tax Returns. Except as otherwise required by Law or caused by a change in Law, all such Tax Returns, to the extent relating to a Target Company or Company Subsidiary, shall be prepared and filed in a manner consistent with past practice (and without unreasonable delay), and no position will be taken, election made or method of accounting adopted that is inconsistent with positions taken, elections made or methods of accounting used in preparing and filing

similar Tax Returns in prior periods. With respect to Tax returns that are required to be filed after the Closing Date, Parent shall also, at its own cost, prepare, file or cause to be filed when due, taking into account all extensions properly obtained, (i) all U.S. federal, state and local income Tax Returns that report income and deductions of U.S. Company for Pre-Closing Periods and that are prepared on the basis that U.S. Company is a disregarded entity for income Tax purposes, and (ii) all U.S. federal, state and local income Tax Returns for Straddle Periods that report income and deductions of U.S. Company with respect to any period of time occurring on or before the Closing Date and that are prepared on the basis that U.S. Company is a disregarded entity for income Tax purposes, and (in each case) Parent shall timely remit or cause to be timely remitted Taxes shown due and payable on such Tax Returns. The cost to any Target Company or Company Subsidiary in preparing or filing such Tax Returns shall be provided for, reserved or accrued in the Final Closing Statement and shall be reflected in and reduce Net Working Capital or reflected in and increase Net External Indebtedness.

(b)

- (i) Subject to and in accordance with the provisions of this Section 6.4 and except as otherwise provided in Section 6.4(a), U.S. Buyer or Sweden Buyer or their duly authorized agents shall prepare, file or cause to be filed when due, taking into account all extensions properly obtained, all Tax Returns with respect to Pre-Closing Periods and all Tax Returns with respect to Straddle Periods, in each case that are required to be filed after the Closing Date by or with respect to a Target Company or Company Subsidiary, and U.S. Buyer or Sweden Buyer, as the case may be, shall timely remit to cause to be timely remitted Taxes shown due and payable on such Tax Returns. Except as otherwise required by Law or caused by a change in Law, all such Tax Returns shall be prepared and filed in a manner consistent with past practice, and no position will be taken, election made or method of accounting adopted that is inconsistent with positions taken, elections made or methods of accounting used in preparing and filing similar Tax Returns in prior periods.
- (ii) Parent shall be liable for and shall pay to U.S. Buyer or Sweden Buyer (as applicable) an amount equal to (A) all external costs and expenses reasonably incurred by U.S. Buyer or Sweden Buyer (or the relevant Target Company or Company Subsidiary) in complying with its obligations in respect of Tax Returns for Pre-Closing Periods pursuant to Section 6.4(b)(i) and (B) such proportion of any external costs and expenses reasonably incurred by U.S. Buyer or Sweden Buyer (or the relevant Target Company or Company Subsidiary) in complying with its obligations in respect of Tax Returns for Straddle Periods pursuant to Section 6.4(b)(i) as corresponds to the proportion of the Straddle Period which falls on or prior to the Closing Date, provided (in each case) that the appointment of external agents to assist with those obligations is consistent with the past practice of the Target Company or Company Subsidiary concerned. If, and to the extent that, any member of Parent Group, other than the Target Company or Company Subsidiary concerned, was primarily responsible for or materially assisted in the preparation and filing of the Tax Returns for the relevant Target Company or Company Subsidiary) in appointing external agents to undertake those responsibilities or to give such assistance after Closing in relation to Tax Returns for which U.S. Buyer or Sweden Buyer is responsible pursuant to Section 6.4(b)(i) shall be costs and expenses for which Parent is liable to the extent provided in clause (A) or (B) (as the case may be) of the preceding sentence despite not being consistent with past practice. Parent shall pay U.S. Buyer or Sweden Buyer any amounts for which Parent is liable pursuant to this Section 6.4(b)(ii) by the later of (a) the date when U.S. Buyer or Sweden Buyer or the Target Company or Company Subsidiary concerned becomes liable to pay or incur such costs; and (b) the tenth Business Day following the date U.S. Buyer or Sweden Buyer, as the case may be, delivers a written demand for such payment to Parent.

- (c) With respect to Tax Returns prepared by Parent pursuant to subsection (a), on the one hand, or prepared by U.S. Buyer or Sweden Buyer pursuant to subsection (b), on the other hand:
- (i) The Party responsible for preparing a Tax Return (for purposes of this Section 6.4, the "**preparing party**") shall provide to the other Party (for purposes of this Section 6.4, the "**other party**") a draft of such Tax Return at least 30 days before such Tax Return is required to be filed, and, in the case of a Tax Return for a Straddle Period, the preparing party shall at the same time provide to the other party a written calculation of the allocation pursuant to Section 6.2(d) of the Tax liability with respect to such Tax Return; provided, however, that notwithstanding any provision of this Stock Purchase Agreement to the contrary, Parent shall have no obligation to provide U.S. Buyer or any Affiliate of U.S. Buyer with any U.S. federal, state or local income Tax Return (including any schedules, attachments or other information relating to such Tax Return) that is prepared on the basis that U.S. Company is a disregarded entity.
- (ii) To the extent permitted by Law, the preparing party shall promptly provide the other party (or shall procure that the other party is promptly provided) with all information reasonably necessary to enable the other party to review each draft Tax Return and allocation provided by the preparing party to the other party pursuant to Section 6.4(c)(i), provided that in the case of a Tax Return for a Straddle Period, U.S. Buyer or Sweden Buyer may redact any such information to the extent reasonably necessary to preserve the confidentiality of commercially sensitive confidential information.
- (iii) The other party shall notify the preparing party of proposed revisions to any draft Tax Return or allocation provided by the preparing party to the other party pursuant to Section 6.4(c)(i) within 15 days of the receipt by the other party from the preparing party of such draft Tax Return and, in the case of a Tax Return for a Straddle Period, such allocation. Such proposed revisions shall be addressed in the following manner:

(A)

a. In the case of a Tax Return for a Pre-Closing Period as to which U.S. Buyer or Sweden Buyer is the preparing party and Parent is the other party, the preparing party shall incorporate the revisions proposed by the other party into such Tax Return and shall timely file such Tax Return as so revised with the Taxing Authority, provided that (x) the preparing party shall not be obligated to incorporate a specific revision to the extent that incorporation of such specific revision would cause an Inconsistency (as defined in Section 6.4(c)(iii)(D) below), and (y) the other party shall consider in good faith any objections the preparing party makes to any other specific revision but shall not be required to accept such objections (and, if it does not accept such objections, then, subject to the other provisions of this Section 6.4(c)(iii)(A), the revision to the Tax Return shall be made notwithstanding such objections). If the preparing party believes that the incorporation of a specific revision would cause an Inconsistency, it shall (by no later than the start of the fifth day prior to the due date of such Tax Return and with written notice to the other party) present the question to an internationally recognized firm of accountants appointed by agreement between Parent or Sweden Seller (as applicable) and U.S. Buyer or Sweden Buyer (as applicable), or, in default of agreement, by the President from time to time of the Institute of Chartered Accountants in England and Wales (the "Independent Tax Accountant") for its prompt determination.

b. Where the preparing party believes that the incorporation of a specific revision would cause such Tax Return to be false, misleading or inaccurate in any respect then

except to the extent that, prior to the due date of such Tax Return, the Independent Tax Accountant determines that such specific revision would not cause such Tax Return to be false, misleading or inaccurate in any respect, the Tax Return shall be filed without such specific revision proposed by the other Party. In any other case, except to the extent that, prior to the due date of such Tax Return, the Independent Tax Accountant determines that a specific revision would cause such Tax Return to be contrary to a specific requirement of Law or of the last sentence of Section 6.4(b)(i) the Tax Return shall be filed with the revisions proposed by the other Party. If, subsequent to the Tax Return having being filed, the Independent Tax Accountant makes a determination which has been taken in the Tax Return which has been filed, an amended Tax Return shall be filed to reflect the determination of the Independent Tax Accountant.

(B) In the case of a Tax Return for a Pre-Closing Period as to which Parent is the preparing party and U.S. Buyer or Sweden Buyer is the other party, the preparing party shall consider in good faith any revisions proposed by the other party but shall not be obligated to accept such revisions, provided that the preparing party shall be required to incorporate a specific revision to the extent that the incorporation of such specific revision is required in order to avoid an Inconsistency, it shall (by no later than the start of the fifth day prior to the due date of such Tax Return and with written notice to the preparing party) present the question to the Independent Tax Accountant for its prompt determination. Except to the extent that, prior to the due date of such Tax Return, shall be filed without the revisions proposed by the other party. If, subsequent to the Tax Return having being filed, the Independent Tax Accountant makes a determination which is contrary to the position which has been taken in the Tax Return which has been filed, an amended Tax Return shall be filed to reflect the determination of the Independent Tax Accountant.

(C)

a. In the case of a Tax Return for a Straddle Period as to which U.S. Buyer or Sweden Buyer is the preparing party and Parent is the other party, the preparing party shall incorporate the reasonable revisions proposed by the other party into such Tax Return to the extent that they are materially relevant to the liability of the other party under this Agreement and are not materially relevant to Tax which would economically be borne by the preparing party. The preparing party shall consider in good faith any other revisions proposed by the other party which are both materially relevant to the liability of the other party under this Agreement and materially relevant to Tax which would economically be borne by the preparing party (giving due weight to each of these considerations bearing in mind the relative amounts of Tax at stake) but shall not be obligated to accept such revisions, except to the extent that the incorporation of such specific revision is required in order to avoid an Inconsistency. If the other party believes that the incorporation of such revision is required in order to avoid an Inconsistency. If the other party believes that the incorporation of such revision is required in order to avoid an Inconsistency it shall (by no later than the start of the fifth day prior to the due date of such Tax Return and with written notice to the preparing party) present the question to the Independent Tax Accountant for its prompt determination.

b. Where the greater part of the Straddle Period falls on or before the Closing Date, except to the extent that, prior to the due date of such Tax Return, the Independent Tax Accountant determines that a specific revision is not required in order to avoid an Inconsistency, the Tax Return shall be filed with the revisions proposed by the other party, provided that if the preparing party (acting reasonably and in good faith) believes that a specific revision would cause such Tax Return to be false, misleading, incomplete or inaccurate in any respect then the Tax Return shall be filed without such specific revision. If, subsequent to the Tax Return having being filed, the Independent Tax Accountant

makes a determination which is contrary to the position which has been taken in the Tax Return which has been filed, an amended Tax Return shall be filed to reflect the determination of the Independent Tax Accountant.

- c. Where the greater part of the Straddle Period falls after the Closing Date, except to the extent that, prior to the due date of such Tax Return, the Independent Tax Accountant determines that a specific revision is required in order to avoid an Inconsistency, the Tax Return shall be filed without the revisions proposed by the other party. If, subsequent to the Tax Return having being filed, the Independent Tax Accountant makes a determination which is contrary to the position which has been taken in the Tax Return which has been filed, an amended Tax Return shall be filed to reflect the determination of the Independent Tax Accountant.
- (D) For the purposes of this Section 6.4(c)(iii), "**Inconsistency**" means that a Tax Return is (i) contrary to a specific requirement of Law, (ii) materially inconsistent with the requirements of the second sentence of Section 6.4(a) or the last sentence of Section 6.4(b)(i) (as applicable), or (iii) false, misleading or inaccurate in any respect.
- (iv) Subject to the proviso in Section 6.4(c)(i), by no later than the date each Tax Return is filed, the preparing party shall provide to the other party a complete copy of such Tax Return as filed.
- (v) By no later than the third Business Day prior to the date each Tax Return is to be filed or each Tax payment is otherwise due, the other party shall pay to the preparing party the portion (if any) of the Actual Tax Liability due with respect to such Tax Return or Tax payment for which the other party is responsible pursuant to Section 6.2. In the event that the preparing party and the other party disagree as to the proper amount of such payment, the other party shall pay to the preparing party the amount that the other party believes to be proper, and the disagreement as to the amount claimed by the preparing party shall be submitted to Independent Tax Accountant for its expeditious determination. Upon determination by Independent Tax Accountant of the additional amount (if any) of Actual Tax Liability properly payable by the other party, the other party shall pay to the preparing party such additional amount together with Default Interest with respect to the period from the date such additional amount should have been paid pursuant to the first sentence of this Section 6.4(c)(v) to the date such additional amount is actually paid. For the avoidance of doubt, the provisions of this Section 6.4(c)(v) are intended to facilitate the timely payment by the Parties to the applicable Tax Authorities of the Actual Tax Liabilities as such payments are due; the determination of the ultimate responsibility of the respective Parties for Tax is controlled by Section 6.2. Accordingly, the calculation of Independent Tax Accountant, shall not be dispositive of, or in any way prejudice, the determination of any Party's ultimate liability pursuant to Section 6.2.
 - (d) Subject to the proviso in Section 6.4(c)(i), the preparing party shall procure that:
- (i) the other party is kept fully informed of the progress of all matters relating to the Tax Returns for Pre-Closing Periods and Straddle Periods;
- (ii) the other party, promptly at the end of each calendar month, receives copies of all written correspondence with any Tax Authority insofar as it is relevant to Pre-Closing Periods or Straddle Periods;

- (iii) no Tax Return is submitted to any Tax Authority which is not, so far as preparing party is aware, true, complete and correct in all respects, and not misleading; and
- (iv) no material correspondence is submitted to, or any material agreement reached with, any Tax Authority without the prior written approval of the other party (which approval will not be withheld, conditioned or delayed unreasonably).

For the purposes of Section 6.4(d)(iv) above, "material correspondence" and "material agreements" are Tax Returns and other correspondence and Agreements which the preparing party considers, or ought reasonably to consider, would be of material importance to the other party as regards either the future relationship with the Tax Authority concerned, the indemnification obligations of the other party pursuant to this Stock Purchase Agreement, or the liability to Tax of any Target Company or Company Subsidiary or any other member of Parent Group, Buyer Group or Geely Group.

- (e) None of U.S. Buyer, Sweden Buyer or any member of Buyer Group or Geely Group shall, or shall cause or permit a Target Company or Company Subsidiary to, amend, refile or otherwise modify, or grant an extension of any statute of limitation with respect to, a Tax Return relating in whole or in part to a Target Company or Company Subsidiary with respect to any Pre-Closing Period or Straddle Period other than (A) where such amendment, refiling, modification or extension would not increase the liability of Parent or Sweden Seller under this Stock Purchase Agreement; or (B) with the prior written consent of Parent, which consent shall not to be unreasonably withheld, conditioned or delayed.
- (f) In the event that Parent at any time in its sole discretion determines that Sweden Company may be considered a shell company (Sw. skalbolag) under the Swedish Income Tax Act (Sw. Inkomstskattelagen (1999:1229)), Sweden Company shall timely file a cash company Tax Return (Sw. skalbolagsdeklaration) with the relevant Tax Authority with respect to the sale and transfer of the Sweden Company Shares by Sweden Seller pursuant to this Stock Purchase Agreement. In the event that such cash company Tax Return is to be filed on or before the Closing Date, Parent shall procure that Sweden Company timely file such Tax Return, and, in the event that such cash company Tax Return is to be filed after the Closing Date, Sweden Buyer shall procure that Sweden Company timely file such Tax Return, provided that (a) Parent gives Sweden Buyer notice of its determination that such cash company Tax Return is required to be filed, on a timely basis and in any event at least 25 Business Days before such cash company Tax Return is required to be filed; (b) so far as permitted by law, such cash company Tax Return shall be filed within 30 days of the Parties entering into this Stock Purchase Agreement; and (c) Parent Company shall be liable for and shall pay to U.S. Buyer or Sweden Buyer (as applicable) an amount equal to all external costs and expenses reasonably incurred by Sweden Buyer (or Sweden Company) in complying with its obligations pursuant to this Section 6.4(f), on the date which is the later of (a) the date when Sweden Buyer or Sweden Company becomes liable to pay or incur such costs; and (b) the tenth Business Day following the date Sweden Buyer delivers a written demand for such payment to Parent.

6.5 Conduct of Tax Proceedings

(a) Upon receipt by a member of Buyer Group of written notice of a Tax audit, examination or assessment, or related administrative or court proceeding, either pending or threatened in writing (a "**Tax Proceeding**"), which would reasonably be expected to give rise to a Tax Claim against Parent, U.S. Buyer or Sweden Buyer shall promptly (and in any event within 15 days of the receipt by Buyer Group of such written notice) notify Parent and shall provide Parent with copies of all notices or other

correspondence or documents from the relevant Tax Authority received by a member of Buyer Group with respect to each such Tax Proceeding.

- (b) In respect of a Tax Proceeding for a Pre-Closing Period or in respect of any Tax Proceeding for a Straddle Period which is not relevant to Tax which would be economically borne by U.S. Buyer or Sweden Buyer, the Parties agree as follows: (i) Parent shall have the right to represent, at its own expense, each Target Company's and each Company Subsidiary's interests in any Tax Proceeding which would reasonably be expected to give rise to a Tax Claim against Parent and, at Parent's expense, to employ counsel of its choice that is reasonably acceptable to Buyers. Within 30 days of the receipt by Parent of the written notice of such a Tax Proceeding, Parent shall notify U.S. Buyer or Sweden Buyer, as the case may be, as to whether it elects to represent the Target Company or Company Subsidiary to which such Tax Proceeding relates; and (ii) if, in connection with any such Tax Proceeding, Parent does not elect to represent such Target Company or Company Subsidiary, U.S. Buyer or Sweden Buyer shall continue to keep Parent fully informed of the status and conduct of such Tax Proceeding (including through compliance with the provisions of Section 6.5(a)), shall consult with Parent with respect to the conduct of such Tax Proceeding, and, upon the request of Parent (and subject to Section 6.5(d)), shall permit and enable Parent to participate in such Tax Proceeding at Parent's expense. U.S. Buyer and Sweden Buyer shall not, and shall procure that no member of Buyer Group shall, settle any claim for Taxes for which Parent may be liable pursuant to Section 6.2, without the prior written consent of Parent, which consent shall not to be unreasonably withheld, conditioned or delayed.
 - (c) Parent shall be liable for, and shall pay to U.S. Buyer or Sweden Buyer (as applicable) an amount equal to:
- (i) any external costs reasonably incurred by U.S. Buyer or Sweden Buyer or the relevant Target Company or Company Subsidiary in connection with any action taken by U.S. Buyer or Sweden Buyer or the relevant Target Company or Company Subsidiary in respect of any such Tax Proceeding to which Section 6.5(b)(ii) applies, including any external costs reasonably incurred by U.S. Buyer or Sweden Buyer or the relevant Target Company or Company Subsidiary in satisfying or settling (pursuant to the requirements of Section 6.5(b)(ii)) the Tax Liability which is the subject of such Tax Proceeding; and
- (ii) any external costs reasonably incurred by U.S. Buyer or Sweden Buyer or the relevant Target Company or Company Subsidiary which are directly attributable to any action taken by U.S. Buyer or Sweden Buyer or the relevant Target Company or Company Subsidiary in obtaining any refund or credit for Taxes to which Section 6.2(c) applies.

Parent shall pay U.S. Buyer or Sweden Buyer any amounts for which Parent is liable pursuant to this Section 6.5(c) by the later of (x) the date when U.S. Buyer or Sweden Buyer or the Target Company or Company Subsidiary concerned becomes liable to pay or incur such costs; and (y) the tenth Business Day following the date U.S. Buyer or Sweden Buyer, as the case may be, delivers a written demand for such payment to Parent.

(d) In the event that a payment of Tax is required by Law as a precondition to the prosecution of a Tax Proceeding described in Section 6.5(b) that Parent wishes to pursue (whether pursuant to clause (i) or (ii) of Section 6.5(b)), then Parent shall provide the funds necessary for such payment; provided, however, that, in the case of a Tax Proceeding pertaining to a Straddle Period, the obligation to pay such funds shall be allocated between and borne by Parent, on the one hand, and U.S. Buyer or Sweden Buyer, on the other hand, equitably in accordance with the principles of Section 6.2(d). Any payment of Tax pursuant to the provisions of this Section 6.5(d) shall be credited to the payor and

treated as a payment of Tax for all purposes of this Stock Purchase Agreement, and any refund of such payment shall be handled in a manner consistent with the provisions of Section 6.2(c).

- (e) In the event that Parent elects to control or participate in any Tax Proceedings in accordance with Section 6.5(b) then:
- (i) Parent shall keep U.S. Buyer or Sweden Buyer (as applicable) reasonably informed of the progress of such Tax Proceedings and shall promptly forward or procure to be forwarded to U.S. Buyer or Sweden Buyer copies of all material correspondence and other material written information and documentation; provided, however, that Parent may redact such correspondence, information and documentation to the extent reasonably necessary to preserve the confidentiality of commercially sensitive confidential information;
- (ii) Parent shall consult with U.S. Buyer or Sweden Buyer (as applicable) and shall endeavor in good faith to provide U.S. Buyer or Sweden Buyer (as applicable) with the opportunity to provide Parent with comments with respect to the matters described in Section 6.5(e)(i) and shall consider such comments in good faith; and
- (iii) Parent shall make no settlement or compromise of the matter to which the Tax Proceedings relate nor agree to any matter in the conduct of the Tax Proceedings the terms of which are reasonably expected to result in a material increase in any future liability to Tax of any member of the Buyer Group which is directly attributable to the terms of such settlement or compromise without the prior written approval of U.S. Buyer or Sweden Buyer (as applicable), such approval not to be unreasonably withheld, conditioned or delayed (taking into account the amount of the Tax Liability involved and the nature of the issues at stake provided that U.S. Buyer or Sweden Buyer shall not have reasonable grounds for withholding, conditioning or delaying its approval to such settlement or compromise unless the reasonably expected increase in the future liability to Tax of any member of the Buyer Group is in an amount which (taking into account in each case the time value of money) is at least equal to 125% of the cost to Parent arising as a direct result of Parent being unable to settle or compromise the matter to which the Tax Proceedings relates (being the difference between the amount which the relevant Tax Authority has indicated it is prepared to accept in settlement or compromise of the matter, and the maximum amount of the Tax Liability for which Parent would otherwise potentially be liable pursuant to this Stock Purchase Agreement, in the absence of such settlement or compromise)).
- (f) If Parent or Sweden Seller receives written notice of any Tax Proceeding which would reasonably be expected to give rise to a Tax Claim against U.S. Buyer or Sweden Buyer under Section 6.2(b), Parent shall promptly (and in any event not more than 15 days after Parent or Sweden Seller receives such written notice) notify Buyers of such Tax Proceeding (including reasonably sufficient details of such Tax Proceeding).
- (g) Subject to Section 6.5(b), U.S. Buyer or Sweden Buyer shall have the sole right to control any Tax Proceeding which would reasonably be expected to give rise to a Tax Claim against U.S. Buyer or Sweden Buyer under Section 6.2(b), provided that U.S. Buyer or Sweden Buyer (as applicable) acknowledges and agrees in writing that Parent and Sweden Seller shall not be liable to make any payment to U.S. Buyer or Sweden Buyer in respect of such Tax Proceeding or the subject matter of such Tax Proceeding, to the extent that such Liability is increased as a result of any action taken by U.S. Buyer or Sweden Buyer pursuant to this Section 6.5(g). Parent and Sweden Seller shall not, and shall procure that no member of Parent Group shall, settle any claim for Taxes for which U.S. Buyer or Sweden Buyer

may be liable pursuant to Section 6.2, without the prior written consent of U.S. Buyer or Sweden Buyer, which consent shall not to be unreasonably withheld, conditioned or delayed.

- (h) In respect of any Tax Proceeding for a Straddle Period, which is relevant both to the liability of Parent pursuant to Section 6.2(a) of this Agreement and to Tax which would be economically borne by U.S. Buyer or Sweden Buyer:
- where the greater proportion of the liability to which the Tax Proceeding relates (taking into account liabilities for the Straddle Period alone) would be borne by Parent, Parent shall have the right to control such Tax Proceedings in accordance with the provisions of Section 6.5(b), provided that Parent shall continue to keep U.S. Buyer or Sweden Buyer (as applicable) fully informed of the status and conduct of such Tax Proceeding, and shall consult with U.S. Buyer or Sweden Buyer with respect to the conduct of such Tax Proceeding. Parent shall not, and shall procure that no member of Buyer Group shall, settle any claim for Taxes which may be economically borne by U.S. Buyer or Sweden Buyer, without the prior written consent of U.S. Buyer or Sweden Buyer, which consent shall not to be unreasonably withheld, conditioned or delayed. In the event of any inconsistency between the provisions of Section 6.5(e) and this Section 6.5(h)(i), the provisions of this Section 6.5(h)(i) shall prevail;
- (ii) where the greater proportion of the liability to which the Tax Proceeding relates (taking into account liabilities for the Straddle Period alone) would be borne by U.S. Buyer or Sweden Buyer, U.S. Buyer or Sweden Buyer (as applicable) shall have the right to control such Tax Proceedings. In connection with any such Tax Proceeding, U.S. Buyer or Sweden Buyer shall continue to keep Parent fully informed of the status and conduct of such Tax Proceeding, and shall consult with Parent with respect to the conduct of such Tax Proceeding. U.S. Buyer and Sweden Buyer shall not, and shall procure that no member of Buyer Group shall, settle any claim for Taxes for which Parent may be liable (in whole or in part) pursuant to Section 6.2, without the prior written consent of Parent, which consent shall not to be unreasonably withheld, conditioned or delayed; and
- (iii) responsibility for external costs incurred by Parent, U.S. Buyer and Sweden Buyer in connection with any Tax Proceeding for a Straddle Period shall be apportioned and borne equitably between the Parties in accordance with the Parties' respective responsibilities for Tax liabilities with respect to such Straddle Period (determined in accordance with the principles of Section 6.2(d)). Payments from one Party to another in order to achieve such equitable sharing of external costs shall be made by the payor within the time period specified in Section 6.5(c) (applied without regard to which Party is payor).

6.6 Group Relief and Tax Losses

(a) Following Closing, U.S. Buyer and Sweden Buyer shall procure that, to the extent permitted by Law, each Target Company and each Company Subsidiary shall make such claims and elections and give such consents in relation to Group Relief as Parent may direct with respect to any Pre-Closing Period and the portion of any Straddle Period falling on or before the Closing Date, provided that U.S. Buyer and Sweden Buyer shall not be obliged to procure that a Target Company or Company Subsidiary take any action pursuant to this Section 6.6(a) unless such action either (i) is fully reflected in and reduces Net Working Capital or is fully reflected in and increases Net External Indebtedness, or (ii) after taking into account any payment for Group Relief pursuant to Section 6.6(b), leaves each Target

Company or Company Subsidiary in no worse position than it would have been had no such action been taken.

- (b) In the event that a Target Company or Company Subsidiary makes a claim or election or gives a consent at the direction of Parent pursuant to Section 6.6(a), the following shall apply with respect to the Group Relief effected by virtue of such claim, election or consent:
- (i) If a Target Company or Company Subsidiary which is a Group Profit Company participates in any arrangement in respect of Group Relief with a member of Parent Group which is a Group Loss Company, no payment shall be required with respect to such action. If a Target Company or Company Subsidiary which is a Group Loss Company participates in any arrangement in respect of Group Relief with a member of Parent Group which is a Group Profit Company, then, except in cases described in clause (i) of the proviso in Section 6.6(a), Parent shall procure that a payment for Group Relief shall be made to the relevant Target Company or Company Subsidiary by the relevant member of Parent Group in an amount equal to any Tax saved by or refunded to the relevant member of Parent Group as a result of such arrangements, on the date on which the Tax saved or refunded would have otherwise become due and payable.
- (ii) If Parent, on behalf of a member of Parent Group, desires to enter into an arrangement in respect of Group Relief with a Target Company or Company Subsidiary with respect to the Straddle Period, items of income, gain, loss and deduction shall be allocated between the portion of the Straddle Period ending on the Closing Date and the remaining portion of the Straddle Period based on a "closing of the books" as of the end of the Closing Date, unless otherwise required by Law.
- From the date of this Stock Purchase Agreement, Parent covenants and agrees not to take any intentional action (other than any action taken with the consent of U.S. Buyer or Sweden Buyer pursuant to and in accordance with the terms of this Stock Purchase Agreement) for the principal purpose of reducing the net operating loss carryforwards for Swedish Tax purposes of a Target Company or Company Subsidiary; provided, however, that notwithstanding the foregoing, none of the following shall be a breach of this Section 6.6(c): (A) tax-deductible group contributions by Parent Group to Sweden Company in an aggregate amount of no more than SEK 150,000,000 subsequent to the finalization of the 2008 Financial Statements, (B) actions taken by any Target Company or Company Subsidiary in the Ordinary Course of Business (provided that any tax deductible group contributions entered into between Parent Group and any Target Company or Company Subsidiary shall not be considered to be in the Ordinary Course of Business for the purposes of this Section 6.6(c)), (C) actions taken by Parent or Parent Group in connection with the NSC Reorganization or (D) actions taken as contemplated by the Transaction Documents.

6.7 Tax Cooperation

Following Closing, Parent and Buyers shall, and shall cause each member of Parent Group and Buyer Group, respectively, to:

(a) comply with the requirements of Section 6.4 and to provide reasonable assistance to the other Party in preparing and filing Tax Returns which such other Party is responsible for preparing and filing in accordance with Section 6.4;

- (b) subject to Section 6.5(e) cooperate with each other in preparing for audits of Tax Returns of a Target Company, Company Subsidiary, any member of Parent Group or any member of Buyer Group and responding to enquiries by, requests for information from, or disputes with Tax Authorities in respect of any such Tax Returns;
- (c) promptly upon receipt of a reasonable request by the other Party or a Governmental Entity for information, records or documents of a Target Company or Company Subsidiary relating to Taxes of a Target Company, Company Subsidiary, any member of Parent Group or any member of Buyer Group, make available to the other Party, and, if applicable, to the Governmental Entity all such reasonably requested information, records, and documents, provided that any Party may redact such information, records or documents to the extent necessary to preserve the confidentiality of commercially sensitive confidential information;
- (d) provide timely notice to the other Party of pending or threatened Tax audits, enquiries or assessments of a Target Company or Company Subsidiary for taxable periods for which the other Party may have Tax Liability under Section 6.2, and promptly furnish the other Party with copies of all correspondence and documents received from a Tax Authority in connection with any Tax audit, information request, enquiry or dispute with respect to such taxable period, to the extent that such materials are relevant to that other Party's liability, and provided that any Party may redact such correspondence and documents to the extent reasonably necessary to preserve the confidentiality of commercially sensitive confidential information;
- (e) promptly sign and deliver such certificates or forms as may be necessary or appropriate to establish an exemption from, or otherwise reduce Taxes described in Section 6.3, or file any Tax Return or other reports with respect to, Taxes described in Section 6.3;
- (f) promptly provide to the other Party any power of attorney or similar authorization necessary to carry out the purposes of Section 6.4 and Section 6.5.

6.8 Group Arrangements

Parent shall procure that each Target Company and Company Subsidiary will cease to be a member of or party to any relevant grouping, fiscal unity, tax sharing arrangement, tax sharing agreement or consolidation for Tax purposes (for the purposes of direct and indirect Tax, including VAT) with any member of Parent Group, no later than Closing.

6.9 BL3 Property

- (a) From and after the date of this Stock Purchase Agreement, Parent shall use, and shall cause Ford Credit Belgium to use, commercially reasonable efforts to execute the BL3 Form Leases, and to take such other actions as may be reasonably necessary for the BL3 Form Leases to become effective on or prior to Closing.
- (b) If the BL3 Form Leases are not executed on or prior to Closing, from and after Closing, Buyers shall (x) use commercially reasonable efforts to procure that each of Parent and Ford Credit Belgium shall have (i) the occupancy and access rights to that portion of the Owned Real Property referred to on Schedule B Part 3.13(a) as BL3 (the "BL3 Property") which is currently occupied by each of them, and (ii) such other rights as are set forth in the BL3 Form Leases (together with (i), the "BL3 Rights"); provided, however, that Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the stream of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy on 180 days are streamed as a second of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate such occupancy of the Buyers shall have the right to terminate shall be a shall have the right to the buyers shall be a shall be a

prior written notice, and (y) continue using commercially reasonable efforts to execute the BL3 Form Leases and to take such other actions as may be reasonably necessary for the BL3 Form Leases to become effective.

6.10 Intercompany Accounts and Agreements

- (a) Except for the Agreements set forth on Schedule K (the "Surviving Contracts") or in connection with the adjustments set forth in Section 2.4, Section 2.5 and Section 2.6, and other than as provided for pursuant to Section 6.6, all Agreements (including all clauses and provisions therein which may, by their terms or otherwise, survive or continue in effect), as well as oral agreements, courses of dealing, business practices, or other arrangements, between or among a member of Parent Group, on the one hand, and a Target Company or Company Subsidiary, on the other hand (excluding Agreements or arrangements to which a third party is also a party) (such Agreements and arrangements, the "Intercompany Agreements"), shall be terminated with effect as of Closing with no further costs or Liabilities on the part of Parent Group, on the one hand, or a Target Company or Company Subsidiary, on the other hand (except for such costs or Liabilities recoverable pursuant to Section 6.10(b)).
- (b) Following Closing, except for the Surviving Contracts or in relation to Intra-Group Trade Payables or Intra-Group Trade Receivables with respect to which the payment obligations arose prior to the Closing Date in the Ordinary Course of Business or rights or Liabilities arising under this Stock Purchase Agreement or the transactions contemplated by this Stock Purchase Agreement, no member of Buyer Group shall make any Claim against a member of Parent Group and no member of Parent Group shall make any Claim against a member of Buyer Group for any matter or event relating to an Intercompany Agreement which matter or event occurred prior to Closing.

6.11 Shared Contracts

Except to the extent otherwise provided by this Stock Purchase Agreement or the Separation Agreements or prohibited by Law or the terms of the relevant Shared Contract, with effect from Closing:

- (a) In relation to any Shared Contract entered into by a member of Parent Group:
- (i) for so long as any member of Parent Group retains any benefit pursuant to such Shared Contract, Parent shall, and shall cause the relevant members of Parent Group to, use commercially reasonable efforts to:
- (A) hold any payments, goods or other services received, to the extent related to the Business, for the benefit of U.S. Buyer or Sweden Buyer (as applicable), and, as soon as reasonably practicable following receipt, forward such payments, goods or other services to U.S. Buyer or Sweden Buyer or to such other member of Buyer Group as U.S. Buyer or Sweden Buyer may direct;
- (B) exercise all rights, powers and benefits to the extent related to the Business in accordance with Buyers' reasonable requirements notified to Parent, but without incurring any financial commitment for its own account or becoming involved in or threatening any form of legal action; and
- (C) use commercially reasonable efforts to carry out or perform its obligations to the extent relating to any business carried on by Parent Group; and

(ii) to the extent a Target Company or Company Subsidiary is not a party to such Shared Contract, U.S. Buyer or Sweden Bu	uyer
shall perform, or procure the performance of, all such Shared Contracts to the extent relating to the Business in accordance with their respective terms	and
conditions, as subcontractor for (or, to the extent not permitted, agent of) the relevant members of Parent Group, and shall indemnify Parent against all Lo	sses
or Liabilities resulting from any failure on the part of U.S. Buyer or Sweden Buyer to perform or procure performance of those obligations.	

- (b) In relation to any Shared Contract entered into by a Target Company or Company Subsidiary:
- (i) for so long as any Target Company or Company Subsidiary retains any benefit pursuant to such Shared Contract, Sweden Buyer shall, and shall cause the relevant members of Buyer Group to, use commercially reasonable efforts to:
- (A) hold any payments, goods or other services received, to the extent related to any business carried on by Parent Group, for the benefit of Parent, and, as soon as reasonably practicable following receipt, forward such payments, goods or other services to Parent or to such other members of Parent Group as Parent may direct;
- (B) exercise all rights, powers and benefits to the extent related to any business carried on by Parent Group in accordance with Parent's reasonable requirements notified to Buyers, but without incurring any financial commitment for its own account or becoming involved in or threatening any form of legal action; and
 - (C) use commercially reasonable efforts to carry out or perform its obligations to the extent related to the Business; and
- (ii) Parent or Sweden Seller shall perform, or procure the performance of, all such Shared Contracts to the extent relating to any business carried on by Parent Group in accordance with their respective terms and conditions, to the extent a member of Parent Group is not a party to such Shared Contract, as subcontractor for (or, to the extent not permitted, agent of) the relevant member(s) of Buyer Group, and shall indemnify Buyers against all Losses or Liabilities resulting from any failure on the part of Parent or Sweden Seller to perform or procure performance of those obligations.

6.12 PZEV Credits

- (a) No later than 14 days before the date by which submission is required by Law of each of the 2010 ZEV Reports to the U.S. States set forth on Schedule L, Sweden Buyer shall provide Parent with all data and information related to Sweden Company necessary to prepare the 2010 ZEV Reports. Following receipt of such data and information from Sweden Buyer, as soon as reasonably practicable, Parent shall prepare the 2010 ZEV Reports and submit such reports to the relevant Governmental Entity for each of the U.S. States set forth on Schedule L.
 - (b) The Parties shall promptly transfer PZEV credits at the conclusion of the 2010 model year as follows:
- (i) to the extent that the 2010 PZEV Credit Obligation is less than the Sweden Company PZEV Credit Shortfall for a given U.S. State, Parent shall cause to be transferred to Sweden

Company PZEV credits in an amount equal to the difference between the Sweden Company PZEV Credit Shortfall and the 2010 PZEV Credit Obligation;

- (ii) to the extent that the 2010 PZEV Credit Obligation is greater than the Sweden Company PZEV Credit Shortfall for a given U.S. State, Sweden Buyer shall cause to be transferred to Parent or its designee PZEV credits in an amount equal to the difference between the 2010 PZEV Credit Obligation and the Sweden Company PZEV Credit Shortfall; and
- (iii) to the extent that the 2010 PZEV Credit Obligation is the same as the Sweden Company PZEV Credit Shortfall for a given U.S. State, there will be no transfer of credits between Parent and Sweden Buyer for such U.S. State.
- (c) If a determination is made that Parent and Sweden Company volumes must be aggregated for 2011 model year zero emission vehicle reporting, no later than 14 days before the date by which submission is required by Law of each of the 2011 ZEV Reports to the U.S. States set forth on Schedule L, Sweden Buyer shall provide Parent with all data and information related to Sweden Company necessary to prepare the 2011 ZEV Reports. Following receipt of such data and information from Sweden Buyer, as soon as reasonably practicable, Parent shall prepare the 2011 ZEV Reports and submit such reports to the relevant Governmental Entity for each of the U.S. States set forth on Schedule L. Sweden Buyer shall cause to be transferred to Parent or its designee PZEV credits in an amount equal to Sweden Company's 2011 model year PZEV obligation in each of the U.S. States set forth on Schedule L.
- (d) Notwithstanding a determination by the relevant Governmental Entity in one or more of the U.S. States listed in Schedule L that the Sweden Company and Parent PZEV credit banks must be combined for the 2010 model year, Parent shall promptly cause to be transferred to Sweden Company at the conclusion of the final model year for which Parent and Sweden Company volumes are required to be aggregated, PZEV credits in an amount such that Sweden Company's resulting PZEV credit bank balance in each such U.S. State is equivalent to that which would have resulted under Section 6.12(b) and Section 6.12(c).

6.13 CAFE Filing

No later than December 31, 2010, Sweden Buyer shall provide Parent with all data and information related to a Target Company necessary to prepare CAFE Reports required to be filed by a member of Parent Group or a Target Company relating to the 2010 model year and changes in corporate relationships. Following receipt of such data from Sweden Buyer, as soon as reasonably practicable, Parent shall prepare and submit (or assist in preparing for submission, as the case may be) such CAFE Reports to the relevant Governmental Entity.

6.14 GHG Allowances

Sweden Buyer shall cause the operator of the Ghent Property, at no cost to Parent or any member of Parent Group, to transfer the Excess GHG Allowances to Parent on the dates set forth on Schedule V into a holding account to be designated by Parent not later than five Business Days prior to the dates specified for each transfer.

6.15 Information Barriers Protocol

With effect from Closing, U.S. Buyer, Sweden Buyer and Parent shall abide by and shall cause their respective Subsidiaries to abide by the Information Barriers Protocol set forth on Schedule I.

6.16 No Relevant Transfer

- (a) Parent, U.S. Buyer and Sweden Buyer acknowledge and agree that none of the transactions contemplated by this Stock Purchase Agreement is intended to constitute a relevant transfer under any Acquired Rights Regulations.
- (b) Notwithstanding Section 6.16(a), in the event that the contract of employment of any individual with Parent or any member of Parent Group (which does not, for the avoidance of doubt, include any Employee), takes effect or is alleged to take effect pursuant to the Acquired Rights Regulations as if originally made between a member of Buyer Group and such individual (any such individual, an "Undisclosed Individual"), then:
- (i) the relevant member of Buyer Group may, within one month of becoming aware of such contract having effect as if originally made by such member of Buyer Group, terminate such contract; and
- (ii) Parent shall indemnify the relevant member of Buyer Group against any Employment Liability incurred or suffered by it to or in relation to any such Undisclosed Individual under or in connection with his contract of employment (whether before or after Closing), and against any Employment Liabilities arising out of or in connection with the termination of the contract of employment of such Undisclosed Individual.

6.17 Employment

For the 12 months following the Closing Date, and in Canada for the 24 months following the Closing Date, except as required by Law, Buyers shall, and shall cause each Target Company and each Company Subsidiary to, provide Employees, other than those Employees to whom a Local Collective Agreement applies, with compensation and benefit plans, programs and arrangements that are no less favorable in the aggregate than those provided under the compensation and benefits plans, programs and arrangements applicable to Employees in effect immediately prior to Closing as such benefit plans, programs and arrangements may be modified as of Closing pursuant to Section 5.10, Section 5.11, Section 5.12 or Section 5.19; provided, however, that nothing in this Section 6.17 shall (a) in respect of jurisdictions other than Sweden, Belgium, the United States, the United Kingdom, the Netherlands and Germany (the "Material Jurisdictions") prevent a change in any retirement Plans from a defined benefit to a defined contribution arrangement, (b) prevent the amendment or termination of any Plan, (c) require that a member of Buyer Group or Geely Group provide or permit investment in the securities of a member of Buyer Group or Geely Group, or (d) make mandatory any discretionary incentive programs, and provided, further, that nothing in this Section 6.17 shall preclude a member of Buyer Group from terminating the employment of any Employee for any reason for which a Target Company or Company Subsidiary could have terminated such Employee prior to Closing. For the avoidance of doubt, nothing in this Section 6.17 shall create any third party beneficiary rights in any other matter.

6.18 Maintenance of Employee Benefits Plan

- (a) Except as otherwise required by Law, for the period of twelve months following Closing, Buyers shall, and shall cause each Target Company and each Company Subsidiary to:
- (i) if Buyers or a Target Company or Company Subsidiary maintain a Company Plan (a "Maintained Company Plan") following Closing, procure that such Maintained Company Plan is administered in all material respects in accordance with its terms as in effect immediately prior to Closing, subject to any amendment or termination thereof that may be permitted by Law or by the terms of such Maintained Company Plan;
- (ii) if Buyers or a Target Company or Company Subsidiary terminates a Company Plan following Closing, it being understood that any such termination must comply with the requirements of Section 6.17, and establishes a replacement Plan (a "New Plan," and such corresponding Company Plan, the "Replaced Company Plan") for purpose of each New Plan providing medical, dental, pharmaceutical or vision benefits to any Employee, (x) procure that all pre-existing condition exclusions and actively at work requirements of such New Plan are waived for such Employee and his or her covered dependents, to the extent any such exclusions or requirements were waived or were inapplicable under the Replaced Company Plan, and (y) cause any expenses eligible for reimbursement under such Replaced Company Plan incurred by such Employee and his or her covered dependents under the Replaced Company Plan to be taken into account under such New Plan for purposes of satisfying all deductible, coinsurance and maximum out of pocket requirements applicable to such Employee and his or her covered dependents for the applicable plan year as if such expenses had been incurred in accordance with such New Plan; and
- (iii) with respect to the Maintained Company Plans and the New Plans, procure that each Employee is credited with his or her service accrued prior to the Closing Date under the relevant Company Plan with his or her Target Company or Company Subsidiary employer (including predecessor or acquired entities or any other entities for which the Target Companies or Company Subsidiaries have given credit for prior service) in the Maintained Company Plan or New Plan, as the case may be, to the same extent as such Employee was entitled, immediately prior to Closing, to credit for such service under such Maintained Company Plan or applicable Replaced Company Plan, for purposes of eligibility and vesting, but excluding any credit of any purpose where service credit would result in duplication of benefits. For the avoidance of doubt, except to the extent required under the terms of a Local Collective Agreement for Represented Employees, this obligation shall not apply to any service accrued prior to the Closing Date under the Ford Retiree Medical Program.
- (b) Notwithstanding any provision to the contrary in this Stock Purchase Agreement, following Closing, Parent shall have no Liability for the payment of benefits accrued by Employees participating in any Maintained Company Plan or New Plan, whether arising before or after the Closing Date.
 - (c) For the avoidance of doubt, nothing in this Section 6.18 shall be construed as amending in any way any Company Plan.

6.19 Equity Plan Awards

- (a) Parent and Sweden Seller shall accelerate the vesting of all unvested equity awards issued under Plans and held by Employees or any directors of Target Companies or Company Subsidiaries as of the Closing Date.
- (b) Parent shall, and shall cause each member of Parent Group to withhold and account for any Tax that is to be duly paid or accounted for arising as a direct result of the vesting of any restricted stock or restricted stock unit and delivery of the underlying shares prior to, or on, the Closing Date.
- (c) Following Closing, each relevant member of Buyer Group shall withhold from Employees or any directors of Target Companies or Company Subsidiaries any Tax that is to be duly paid or accounted for as a result of stock options issued to such persons under the Plans.
- (d) Following Closing, Parent shall, and cause each relevant member of Parent Group to, as appropriate, cause all unexercised stock options issued under Plans held by Employees or any directors of Target Companies or Company Subsidiaries as of the Closing Date to have a remaining unexpired term ending on the earlier of (i) the expiration of the remaining unexpired term as of the Closing Date, or (ii) five years from the Closing Date.

6.20 Defined Contribution Plans Transfers

- (a) As soon as reasonably practicable following the receipt of (i) any approvals required by Law to be obtained from a Governmental Entity, and (ii) any consent required by Law or a Local Collective Agreement to be obtained from Employees, Parent shall, and shall cause the Company Subsidiaries in Canada and the United States, as applicable, to, transfer the accrued benefits of each Employee participating in the Pension Plan #6, the SSIP or the FRP, (each a "Parent Defined Contribution Plan") to the corresponding Replacement Canada Defined Contribution Plan or Replacement U.S. Defined Contribution Plan, or to make such alternative arrangements for accrued benefits as the Employee elects and as are allowed under the rules (including determinations as are appropriate under the rules) of the Parent Defined Contribution Plan.
- (b) Following Closing, if an Employee who is a participant in a Replacement Canada Defined Contribution Plan or Replacement U.S. Defined Contribution Plan that is a Maintained Company Plan, or in a corresponding New Plan, becomes eligible to receive a distribution under such Plan prior to the date such Plan has received a transfer of accrued benefits from the Parent Defined Contribution Plan, (i) Parent shall cause the timely distribution to such Employee of all amounts accrued under the Parent Defined Contribution Plan, and (ii) U.S. Buyer or Sweden Buyer (as applicable) shall cause the timely distribution to such Employee under the Replacement Canada Defined Contribution Plan or Replacement U.S. Defined Contribution Plan of all amounts accrued and earned by such Employees following the Closing Date.

6.21 Defined Benefit Plan in the U.S.

As soon as reasonably practicable following the receipt of (i) any approvals required by Law to be obtained from a Governmental Entity, and (ii) any consent required by Law or a Local Collective Agreement to be obtained from Employees or a collective bargaining agent of any such Employees, Parent shall cause the trustees of the Ford General Retirement Plan and the Ford UAW Retirement Plan to segregate and transfer to the Volvo U.S. Defined Benefit Plan the portion of the assets (which may be

paid in cash in Parent's sole discretion or other assets as otherwise agreed between the Parties) attributable to the accrued benefits of the U.S. Hourly Employee Pension Participants who have made the appropriate elections required by Law or a Local Collective Agreement as of the Closing Date (the "U.S. Employee Pensions Plan Transferred Assets") in accordance with Law and the terms of the U.S. Employee Pensions Plans. Simultaneously with and subject to the transfer of the U.S. Employee Pensions Plan Transferred Assets, the Volvo U.S. Defined Benefit Plan shall assume all of the obligations of the U.S. Employee Pensions Plans with respect to benefits accrued by U.S. Hourly Employee Pension Participants who have made any required elections for transfer.

6.22 Defined Benefit Plan in the Netherlands

As soon as reasonably practicable following the receipt of (i) any approvals required by Law to be obtained from a Governmental Entity, and (ii) any consent required by Law or a Local Collective Agreement to be obtained from Employees or a collective bargaining agent of any such Employees, Parent shall cause the trustee of the Ford Netherlands Pension Fund to segregate and transfer to the Volvo Netherlands Funding Vehicle the portion of the assets (which may be paid in cash in Parent's sole discretion or other assets as otherwise agreed between the Parties) allocable to the accrued benefits of the Employees who participate in the Volvo Netherlands Defined Benefit Plan and who have made appropriate elections under Law as of the Closing Date (the "Ford Netherlands Pension Fund Transferred Assets") as determined pursuant to the Ford Netherlands Pension Fund Transfer Agreement. Simultaneously with and subject to the transfer of the Ford Netherlands Pension Fund Transferred Assets, the Volvo Netherlands Funding Vehicle shall assume all of the obligations of the Ford Netherlands Pension Fund with respect to benefits accrued by Employees participating in the Volvo Netherlands Defined Benefit Plan on or prior to the Closing Date and who have made appropriate elections for such transfer.

6.23 Defined Benefit Pension Plan in Japan

- (a) During the period between the Closing and the DB Pension Split Date:
- (i) Buyers shall ensure that Volvo Cars Japan Limited ("VCJ") pays regular contributions calculated at the rate as set forth in the terms of the Joint New DB Plan and its share of the expenses of operating the Joint New DB Plan as required and charged by the trustee of the Joint New DB Plan pursuant to the Joint New DB Plan.
- (ii) Sweden Seller shall ensure that Ford Japan Limited ("FJL") pays regular contributions calculated at the rate as set forth in the terms of the Joint New DB Plan and its share of the expenses of operating the Joint New DB Plan as required and charged by the trustee of the Joint New DB Plan pursuant to the Joint New DB Plan.
- (b) On the DB Pension Split Date, Buyers and Sweden Seller shall cause VCJ and FJL to instruct the trustee of the Joint New DB Plan to segregate and transfer the assets and liabilities in the Joint New DB Plan to VCJ's individual DB Pension Plan and FJL's individual DB Pension Plan respectively in accordance with the instructions as set out in the Request for DB Asset Split set forth in Exhibit V.
- (c) As soon as reasonably practicable after the DB Pension Split Date, an actuary designated by VCJ (the "VCJ Actuary") shall calculate the Theoretical Assets Allocated to VCJ as of the DB

Pension Split Date less the assets of the Joint New DB Plan which are actually allocated to VCJ as of the DB Pension Split Date ("VCJ Amount"), pursuant to Schedule P Part 1.

- (d) As soon as reasonably practicable after the DB Pension Split Date, an actuary designated by FJL (the "FJL Actuary") shall calculate the Theoretical Assets Allocated to FJL as of the DB Pension Split Date less the assets of the Joint New DB Plan which are actually allocated to FJL as of the DB Pension Split Date ("FJL Amount"), pursuant to Schedule P Part 2.
- (e) Within 14 days after the DB Pension Split Date, VCJ Actuary and FJL Actuary shall calculate and agree on the amount of the VCJ Amount and the amount of the FJL Amount.
- (f) If the FJL Amount is a negative number, Parent (for itself and on behalf of Sweden Seller) shall pay to Sweden Buyer (for itself and on behalf of U.S. Buyer) an amount equal to the difference between zero and such negative number (plus an amount equal to interest on such payment at the Applicable Rate for the period from (but excluding) the DB Pension Split Date to (and including) the date of such payment, calculated on a daily basis), within two weeks after the date on which the VCJ Amount and the FJL Amount are agreed by VCJ Actuary and FJL Actuary, or decided by the independent actuary. Any payment pursuant to this Section 6.23(f) shall, so far as possible, be treated as an adjustment to the Purchase Price.
- (g) If the VCJ Amount is a negative number, Sweden Buyer (for itself and on behalf of U.S. Buyer) shall pay to Parent (for itself and on behalf of Sweden Seller) an amount equal to the difference between zero and such negative number (plus an amount equal to interest on such payment at the Applicable Rate for the period from (but excluding) the DB Pension Split Date to (and including) the date of such payment, calculated on a daily basis), within two weeks after the date on which the VCJ Amount and the FJL Amount are agreed by VCJ Actuary and FJL Actuary, or decided by the independent actuary. Any payment pursuant to this Section 6.23(g) shall, so far as possible, be treated as an adjustment to the Purchase Price.
- (h) Any dispute between FJL Actuary and VCJ Actuary concerning the VCJ Amount and the FJL Amount or any other matter of an actuarial nature in this Section 6.23 shall, in the absence of agreement between them, be referred to an independent actuary agreed by Parent and Buyers or, failing such agreement within 14 days of one party calling upon the other in writing so to agree, appointed by the President of the Japanese Society of Certified Pension Actuaries in Japan. Any such independent actuary shall reach his decision on the basis of the provisions of this Stock Purchase Agreement and shall act as an expert and not as an arbitrator and his decision shall (in the absence of manifest error) be final and binding upon Parent and Buyers. The charges and expenses of the independent actuary in respect of any such reference shall be borne 50% by Parent and 50% by Buyers.

6.24 Consent to Transfer; Consultation Obligations

(a) For purposes of Section 6.20, Section 6.21, Section 6.22 and Section 6.23, where consent to a transfer is required by Law or a Local Collective Agreement to be obtained from Employees or a collective bargaining agent of such Employees (the "Consenting Party"), the Consenting Party shall be provided with a maximum of two months, or such greater minimum period required by Law or a Local Collective Agreement, to decide whether to consent to the transfer.

(b) From the date of this Stock Purchase Agreement through and following Closing, Buyers shall, and shall cause each member of Buyer Group to, use commercially reasonable efforts to comply with their respective Collective Consultation Requirements.

6.25 Non-Solicitation of Certain Employees

- (a) For the 24 months following the date of this Stock Purchase Agreement, each member of Geely Group, U.S. Buyer and Sweden Buyer shall not, and U.S. Buyer and Sweden Buyer shall cause each member of Buyer Group not to and shall use commercially reasonable efforts to cause each member of Geely Group not to, whether acting alone or jointly with any other Person and whether directly or indirectly, hire from, or induce or solicit to terminate employment with, or discourage from being employed by or providing services to, Parent or any member of Parent Group, any Parent Senior Employee or Designated Engineer other than: (i) to the extent expressly permitted pursuant to a Secondment Agreement; or (ii) any employee who, on his or her own initiative, responds to any bona fide public advertisement placed by or on behalf of any member of Buyer Group and not targeted directly to any specific Parent Senior Employee or Designated Engineer.
- (b) For the 24 months following the date of this Stock Purchase Agreement, Parent shall not, and shall cause each member of Parent Group not to, whether acting alone or jointly with any other Person and whether directly or indirectly, hire from, or induce or solicit to terminate employment with, or discourage from being employed by or providing services to, a Target Company or Company Subsidiary, any Target Company Senior Employee other than: (i) to the extent expressly permitted by the Separation Agreements, or to the extent reasonably necessary to consummate the transactions contemplated by this Stock Purchase Agreement; or (ii) any employee who, on his or her own initiative, responds to any bona fide public advertisement placed by or on behalf of any member of Parent Group and not targeted directly to any specific Target Company Senior Employee.

6.26 Retained Company Name Changes

- (a) Parent shall use commercially reasonable efforts to procure that all necessary steps are taken to change the name of any company that is a member of Parent Group whose name contains the word "Volvo" so as to remove references to such word, with such name change to be effective as soon as reasonably practicable following Closing, provided, however, that Parent shall have no obligation to change the name of any company for which a member of Parent Group is granted a license to continue use of such name pursuant to the Intellectual Property License Agreement or the Vehicle Finance Separation Agreement.
- (b) Buyer Group shall use commercially reasonable efforts to procure that all necessary steps are taken to change the name of any company that is a member of Buyer Group whose name contains the word "Ford" so as to remove references to such word, with such name change to be effective as soon as reasonably practicable following Closing, provided, however, that Buyer Group shall have no obligation to change the name of any company for which the Target Company is granted a license to continue use of such name pursuant to the Intellectual Property License Agreement.

6.27 Insurance Claims

(a) To the extent that Parent shall be entitled, under the terms and conditions of any "occurrence" based insurance policy in effect on the date of this Stock Purchase Agreement or any

Replacement Policy, to coverage for losses suffered by the Target Companies or Company Subsidiaries after Closing arising out of any occurrence covered by such program or policies, Parent shall, and shall cause each relevant member of Parent Group to, use such efforts and take such actions to recover such losses on behalf of the Target Companies or Company Subsidiaries as it would use or take in conducting its own business in the ordinary course if such losses were suffered by Parent or any member of Parent Group. In the event of any dispute regarding the date of any loss or occurrence, the terms of the program or relevant policy shall govern.

(b) Buyers acknowledge and agree that, with effect from the Closing Date, but without prejudice to any accrued claims of a Target Company or Company Subsidiary and subject to the terms of the applicable policies, none of Buyers, a Target Company, or a Company Subsidiary shall have any right or interest in any policies of insurance maintained by Parent or any member of Parent Group in relation to any Target Company or Company Subsidiary prior to the Closing Date.

6.28 Directors and Officers Discharge of Liability; Indemnification

- (a) Sweden Buyer shall discharge or procure the discharge of each of the members of the board of directors of each Swedish Entity in office immediately prior to Closing from personal Liability for the period prior to and including the Closing Date at the first annual general meeting of each Swedish Entity (or of any other Target Company or Company Subsidiary, if and as applicable), held after the Closing Date, it being understood that Sweden Buyer's obligations under this Section 6.28(a) shall be subject to the condition that the third party auditors of the Swedish Entities (or of any other Target Company or Company Subsidiary) do not recommend against such discharge.
- (b) For the three years following the Closing Date, Sweden Buyer shall, and shall cause the members of Buyer Group to, indemnify and hold harmless each present or former member of the board of directors, or equivalent, or officer, or equivalent, of a Target Company or a Company Subsidiary with respect to all acts or omissions by such director or officer in his or her capacity as such, in each case to the extent required by the applicable terms of the Formation Document of such Target Company or Company Subsidiary as in effect on the date of this Stock Purchase Agreement.

6.29 Records and Assistance

- (a) For the four years following the Closing Date (or for such longer period as is required by Law for such records to be retained by the Party retaining the records), Parent and Sweden Buyer shall not, and each shall cause each member of Parent Group and Buyer Group, respectively, not to, dispose of or destroy any of the Company Records dated before Closing in their possession or control without first giving the other Party at least one month's prior written notice and the opportunity to review and, at such other Party's cost, copy any of the relevant materials.
- (b) Except to the extent such access is prohibited or restricted by Law, the terms of any confidentiality Agreement existing as of the date of this Stock Purchase Agreement, this Stock Purchase Agreement or the Separation Agreements, or to the extent such materials would reasonably be expected to violate the attorney-client privilege of a member of Buyer Group with its respective legal counsel, Sweden Buyer shall, and shall procure that each member or representative of Buyer Group shall, upon request, provide any member of Parent Group or its representatives with copies (at Parent's own cost) of such parts of the Company Records within the possession or control of any member of Buyer Group as may reasonably be required by any member of Parent Group (i) for eight years following the Closing Date, in connection with any financial, Tax or other report, return, statement, audit, or filing required by

Law, or (ii) at any time following Closing in relation to any proceedings by or against any member of Parent Group reasonably relating to the Business or the ownership, operation or control of the Business, any Target Company or any Company Subsidiary by Parent Group. Such member of Buyer Group may process documents to remove information that is not relevant to Parent Group member's request.

- (c) Where a Claim is made against a member of Parent Group reasonably relating to the Business or the ownership, operation or control of the Business, any Target Company or any Company Subsidiary by Parent Group, Sweden Buyer shall provide at Parent's own cost copies of such parts of the Company Records within the possession or control of any member of Buyer Group as may reasonably be required by any member of Parent Group, provided that Buyers shall not be required to provide any information where access to that information is prohibited or restricted by Law, the terms of any confidentiality Agreement existing at the date of this Stock Purchase Agreement, this Stock Purchase Agreement or the Separation Agreements, or to the extent such materials would reasonably be expected to violate the attorney-client privilege of a member of Buyer Group with its respective legal counsel.
- (d) Except to the extent such access is prohibited or restricted by Law, the terms of any confidentiality Agreement existing at the date of this Stock Purchase Agreement, this Stock Purchase Agreement or the Separation Agreements, or to the extent such materials would reasonably be expected to violate the attorney-client privilege of a member of Parent Group with their respective legal counsel, Parent shall, and shall procure that any member or representative of Parent Group shall, upon request, provide any member of Buyer Group or its representatives with copies (at Buyers' own cost) of such parts of the Company Records within the possession or control of any member of Parent Group as may reasonably be required by any member of Buyer Group (i) for eight years following the Closing Date, in connection with any financial, Tax or other report, return, statement, audit, or filing required by Law, or (ii) at any time following Closing in relation to any proceedings by or against any member of Buyer Group reasonably relating to the Business or the ownership, operation or control of the Business, any Target Company or any Company Subsidiary by Buyer Group. Such member of Parent Group may process documents to remove information that is not relevant to Buyer Group member's request.
- (e) Buyers shall cause Sweden Company to prepare and submit the month-end accounts in respect of the month ending prior to the Closing Date for the Target Group to Parent's financial reporting requirements, consistent with past practice.
- (f) Following Closing until the Final Closing Statement is issued, Parent shall cause Sweden Company to have access to the Dearborn-administered application of Hyperion Enterprise (HE), and to the portion of the Parent Group consolidated accounts related to the Target Group, to the extent reasonably required to prepare the Closing Statement pursuant to Schedule A-1. For the avoidance of doubt, nothing in this Section 6.29(f) shall limit Sweden Company's access to the Dearborn-administered application of Hyperion Enterprise (HE) otherwise provided for in the Transaction Documents.

6.30 Releases

(a) With effect from Closing, other than pursuant to this Stock Purchase Agreement or the Separation Agreements, U.S. Buyer and Sweden Buyer:

(i) shall procure that each relevant member of Parent Group shall be released in full from (A) all Assurances set forth on Schedule N, as soon as reasonably practicable following Closing, and (B) all other Assurances relating to a Liability of a Target Company or Company Subsidiary, as soon as reasonably practicable after becoming aware of such Assurance; and

- pending release in accordance with Section 6.30(a)(i), hereby indemnify each member of Parent Group against all Losses and Liabilities incurred by the relevant member of Parent Group with respect to any such Assurance.
 - (b) With effect from Closing, other than pursuant to this Stock Purchase Agreement or the Separation Agreements, Parent shall:
- procure that each Target Company and each Company Subsidiary shall be released in full from Assurances relating to a Liability of a member of Parent Group that is not related to the Business or the Assets, as soon as reasonably practicable after becoming aware of such Assurance: and
- pending release in accordance with Section 6.30(b)(i), indemnify each Target Company or Company Subsidiary against all Losses and Liabilities incurred with respect to such Assurance.

6.31 Wrong Pockets

- If within nine months following the Closing Date it becomes apparent that during the six months prior to the date of this Stock Purchase Agreement a Target Company or Company Subsidiary used an asset in connection with the Business which is owned by a member of Parent Group and which
 - (i) not subject to a Separation Agreement;
- (ii) not expressly contemplated to be available to a member of Buyer Group after Closing pursuant to the terms of this Stock Purchase Agreement, including Schedule B, or any other Transaction Document; and
 - (iii) exclusively used by the Target Companies or Subsidiary Companies,

Parent shall procure the transfer of all rights, title and interest in and to such asset to such Target Company or Company Subsidiary as Buyers may direct together with any Permits held by any member of Parent Group relating to that asset and such transfer shall be effected without additional consideration except to the extent that the asset should have been reflected in Net Working Capital or Net External Indebtedness in which event Sweden Buyer shall pay Parent the amount for such asset as would have been reflected;

- if within nine months following the Closing Date it becomes apparent that during the six months prior to the date of this Stock Purchase Agreement a member of Parent Group used, other than in connection with the Business, an asset which is owned by a Target Company or Company Subsidiary which is:
 - not subject to a Separation Agreement; (i)
- (ii) not expressly contemplated to be available to a member of Parent Group after Closing pursuant to the terms of any Transaction

Document; and

(iii) exclusively used by Parent Group,

U.S. Buyer or Sweden Buyer, as applicable, shall procure the transfer of all rights, title and interest in and to such asset to such member of Parent Group as Parent together with any Permits held by any member of Buyer Group relating to that asset and such transfer shall be effected without additional consideration except to the extent that the asset was reflected in Net Working Capital or reflected in Net External Indebtedness in which event Parent shall reimburse Sweden Buyer such amount as so reflected.

6.32 Replacement Vehicle Finance Arrangements

- (a) Subject to Section 6.32(b), in the event that Replacement Vehicle Finance Arrangements are not implemented in a Market (other than Germany or Switzerland) prior to Closing, Parent (or Parent shall cause the applicable member of Parent Group) and U.S. Buyer and Sweden Buyer shall enter into a transitional services Agreement for any such Market to provide vehicle finance services substantially similar to those made available in support of the Business in such Market during calendar year 2009 on terms and conditions generally consistent (to the extent applicable) with the terms and conditions of the VAB Transitional Services Agreement. For purposes of this Section 6.32, "implemented" means execution, delivery and consummation of Agreements with replacement financial services providers for Replacement Vehicle Finance Arrangements in any such Market in accordance with Section 5.15 and the terms of such Agreements.
- (b) Notwithstanding anything to the contrary in Section 6.32(a), neither Parent nor Credit Company shall have any obligation to provide transitional services, regardless of whether Replacement Vehicle Finance Arrangements are in place at Closing, in relation to the Markets and corresponding Vehicle Finance Arrangements set forth in Schedule U.
- (c) Except to the extent required under applicable Law, from the date of this Stock Purchase Agreement until the termination of the VAB Transitional Services Agreement, Parent shall not take any action which, directly or indirectly, would lead to or result in the voluntary winding up of the affairs of VAB, VAL and/or VAF (including voting in favor of a resolution to liquidate VAB, VAL and/or VAF) if such action would reasonably be expected to adversely affect the quality and standards of service provided by VAB, VAL and VAF as set forth in section 2.1.1 of the VAB Transitional Services Agreement.
- (d) From the date of this Stock Purchase Agreement until the termination of the VAB Transitional Services Agreement, neither Parent nor a member of Parent Group shall (i) cause Sweden Company, directly or indirectly, to enter into a definitive agreement with a Replacement Provider for Replacement Vehicle Finance Arrangements in Germany or Switzerland which is a Captive Automotive Finance Company or (ii) enter into an agreement to sell all or part of its equity interest in VAB, either directly or indirectly, to a Captive Automotive Finance Company.

6.33 Further Actions

From the date of this Stock Purchase Agreement through and following Closing, each of the Parties shall use commercially reasonable efforts to promptly execute, deliver and file, or shall cause to be promptly executed, delivered and filed, such documents and other papers, and shall take such further action as may be necessary, to carry out the provisions of this Stock Purchase Agreement (or any amendment to this Stock Purchase Agreement mutually agreed by the Parties) and to consummate the transactions contemplated by this Stock Purchase Agreement on or before the date that is nine months

from the date of this Stock Purchase Agreement, including the defending of any Claim, whether judicial or administrative, challenging this Stock Purchase Agreement or threatening to impair, impede or delay the transactions contemplated by this Stock Purchase Agreement.

6.34 Confidentiality

- From the date of this Stock Purchase Agreement through and following Closing, each of the Parties shall treat as confidential, and shall cause its Affiliates and Agents to treat as confidential, the existence and contents of this Stock Purchase Agreement and the Transaction Documents (which shall be treated as the Confidential Information of all Parties) as well as non-public information (including written information and information transferred orally, electronically or by any other means) obtained from any other Party or its Affiliates or Agents in connection with the negotiation and execution of the Transaction Documents and the transactions contemplated by the Transaction Documents, including any information deemed to be confidential information under the terms of any of the Separation Agreements, but excluding any information which (x) was lawfully in the possession of the receiving Party or any of its Agents (in either case as evidenced in written records) without any obligation of secrecy prior to its being received or held or (y) has become publicly available other than through the receiving Party's fault (or that of any of its Agents) ("Confidential Information"), and shall refrain from disclosing, in whole or in part to any third party, any Confidential Information except (i) as required by Law or by any Governmental Entity or securities exchange having applicable jurisdiction (in which case, unless legally restricted from doing so, the disclosing Party shall give prior notice to the other Party of its intention to disclose such information and use reasonable efforts to assist the disclosing Party in its attempts to obtain confidential treatment of such information, (ii) as required for the purpose of any arbitral or judicial proceedings arising out of or in connection with this Stock Purchase Agreement, (iii) as required by this Stock Purchase Agreement, including as required in connection with any filings for Governmental Approvals or (iv) as required or permitted under any Separation Agreement.
- (b) Notwithstanding the foregoing and any other provision in this Stock Purchase Agreement, the Parties (and each of their respective Affiliates and Agents) may disclose to any and all Persons the Tax treatment and Tax structure of the transactions contemplated by the Transaction Documents, and all opinions of any kind (including opinions or other Tax analyses) that are provided to the Parties relating to such Tax treatment and Tax structure other than the information for which nondisclosure is reasonably necessary in order to comply with any Law.

6.35 News Releases; Public Announcements

- (a) From the date of this Stock Purchase Agreement through and following Closing, except as set forth in Section 6.34, no Party may make or send a news release or other public announcement, advertisement, communication or circular disclosing the existence of the Transaction Documents or the transactions contemplated by the Transaction Documents unless such Party has first obtained the other Parties' written consent, which consent shall not be unreasonably withheld, conditioned or delayed.
- (b) The restrictions set forth in Section 6.34 shall not apply to a news release, public announcement, advertisement, communication or circular that is required by Law or pursuant to an Order of a court of competent jurisdiction, provided that the Party subject to such obligation shall, unless prohibited by Law, first inform the other Parties as to the timing and content of any such release, announcement, communication or circular and to the extent practicable allow the other Party reasonable time to comment on such release or announcement in advance of such issuance.

(c) After Closing, the Parties agree that this covenant shall not limit the ability of any Party to, without the consent of any Person (including any other Party), make any public release or announcement concerning the Business or any Target Company or Company Subsidiary (together with any historical information or results required or advisable to be included therewith). For the avoidance of doubt, nothing in this Section 6.35(c) shall relieve any Party of its obligations under Section 6.34 or the Confidentiality Agreement or any confidentiality obligations under any Separation Agreement.

6.36 Assignment of PLA

Immediately following Closing, Parent shall, and shall cause Ford Global to, execute and deliver to Sweden Company an assignment and assumption agreement substantially in the form attached to this Stock Purchase Agreement as Exhibit XXV pursuant to which Ford Global shall assign all of its rights and obligations under the PLA to Sweden Company. Parent acknowledges and agrees that if the PLA contains terms that are less favorable to Sweden Company than the Required Terms, Sweden Company shall not be bound by the PLA and Parent and its current Affiliates shall not be entitled to exercise the rights purported to be granted thereunder. In such event Sweden Company shall grant a license of the Volvo Assigned Patents (excluding the SPA Platform Patents and VED and VEP Patents) to Parent and its current Affiliates on the Required Terms.

6.37 Maintenance of Back-to-Back Financing Commitment

- (a) From the date of this Stock Purchase Agreement until such time as the Working Capital Facility is in effect and available to be drawn by Sweden Company upon Closing, Buyers shall maintain in full force and effect the Back-to-Back Financing Commitment.
- (b) If the Working Capital Facility will not be available to be disbursed upon Closing to Sweden Company, Buyers shall ensure that all filings and requests are timely made for the approval or consent required from any Governmental Entity or any party to the Entrustment Loan Agreement to permit Buyers, upon receipt of such approvals and consents to (i) draw down, from and after Closing, all funds which it could be called upon to advance pursuant to the Back-to-Back Financing Commitment to Sweden Company outside the PRC and (ii) to use, from and after Closing, such funds for purposes of fulfilling the working capital requirements of Sweden Company.
- (c) Except to the extent the Working Capital Facility is available to be disbursed to Sweden Company upon Closing, on Closing, subject to receipt of a written request from Sweden Company in accordance with the terms and conditions of the Back-to-Back Financing Commitment, Buyers shall advance to Sweden Company outside the PRC pursuant to the Back-to-Back Financing Commitment such funds as are set forth in the Working Capital Forecast up to the amount set forth in Schedule M Part 2.
- (d) Subject to any restrictions or prohibitions imposed by Law, Buyers shall provide copies to Parent of (i) any and all approvals and consents obtained pursuant to Section 6.37(b), and (ii) all final Agreements for the Working Capital Facility if the Working Capital Facility is available to be disbursed upon Closing to Sweden Company.

7. CONDITIONS PRECEDENT TO OBLIGATIONS OF PARENT AND SWEDEN SELLER

The obligations of Parent and Sweden Seller to proceed with Closing under this Stock Purchase Agreement are subject to the satisfaction of each of the following conditions precedent, unless waived in writing by Parent (in its own capacity and on behalf of Sweden Seller) in whole or in part, on or prior to the Closing Date.

7.1 Representations and Warranties

Except for the representations and warranties set forth in Section 4.1 and Section 4.2, which shall be true and complete in all respects, the representations and warranties made by U.S. Buyer and Sweden Buyer in Article 4 of this Stock Purchase Agreement shall be true and complete on and as of the Closing Date as though such representations and warranties were made on and as of such date, except in each case to the extent that those representations and warranties that are made as of a specified date shall be true and complete only as of such date, and except in all cases where the failure to be true and complete would not, individually or in the aggregate, reasonably be expected to prevent, materially delay or materially impede the ability of U.S. Buyer or Sweden Buyer to consummate the transactions contemplated by this Stock Purchase Agreement.

7.2 Governmental Approvals; Buyer Governmental Approvals

The Governmental Approvals on Schedule C Part 1 and Schedule C Part 2 and Buyer Governmental Approvals shall have been received or otherwise obtained and any waiting periods required under any of the Governmental Approvals set forth on Schedule C Part 1 and Schedule C Part 2 and Buyer Governmental Approvals shall have expired or been terminated.

7.3 Performance of Obligations

(a) U.S. Buyer and Sweden Buyer shall have performed or complied (i) in all respects with all obligations, agreements and covenants in Section 2.1(a), Section 2.3, Section 6.37 and Section 9.2 that are required by this Stock Purchase Agreement to be performed or complied with by U.S. Buyer and Sweden Buyer at or prior to the Closing Date, and (ii) in all material respects with all other obligations, agreements and covenants required to be performed or complied with by each of them under this Stock Purchase Agreement at or prior to the Closing Date.

7.4 No Law

There shall not be in effect any Law enacted, entered, promulgated, enforced or otherwise issued or other legal restraint or prohibition restraining, enjoining, preventing or making illegal the consummation of the transactions contemplated by the Transaction Documents.

7.5 Buyers Officer's Certificate

U.S. Buyer and Sweden Buyer shall have delivered to Parent a certificate, dated as of the Closing Date and executed by a duly authorized representative of U.S. Buyer and Sweden Buyer, certifying to the fulfillment of the conditions set forth in Section 7.1 and Section 7.3.

7.6 Documents prior to or at Closing

All deliveries required to be made by U.S. Buyer or Sweden Buyer to Parent prior to or at Closing shall have been made as set forth in Section 9.2.

8. CONDITIONS PRECEDENT TO OBLIGATIONS OF U.S. BUYER AND SWEDEN BUYER

The obligations of U.S. Buyer and Sweden Buyer to proceed with Closing under this Stock Purchase Agreement are subject to the satisfaction of each of the following conditions precedent, unless waived in writing by Sweden Buyer (in its own capacity and on behalf of U.S. Buyer), in whole or in part, on or prior to the Closing Date.

8.1 Representations and Warranties

Except for the representations and warranties set forth in Section 3.1(a), Section 3.2, Section 3.5 and Section 3.6, which shall be true and complete in all respects, the representations and warranties made by Parent and Sweden Seller in Article 3 of this Stock Purchase Agreement shall be true and complete on and as of the Closing Date as though such representations and warranties were made on and as of such date, except in each case to the extent that those representations and warranties that are made as of a specified date shall be true and complete only as of such date, without giving effect in any case to any supplement or amendment to Schedule B (it being understood that no action taken after the date of this Stock Purchase Agreement by Parent or its Affiliates solely in performance of the covenants or undertakings of this Stock Purchase Agreement, or as directed or consented to in writing by Sweden Buyer or U.S. Buyer under Section 5.3 or otherwise, shall constitute a breach of any representation or warranty of Parent or Sweden Seller), and except in all cases where the failure to be true and complete would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

8.2 Governmental Approvals; Buyer Governmental Approvals

The Governmental Approvals on Schedule C Part 1 and Schedule C Part 2 and Buyer Governmental Approvals shall have been received or otherwise obtained and any waiting periods required under any of the Governmental Approvals set forth on Schedule C Part 1 and Schedule C Part 2 and Buyer Governmental Approvals shall have expired or been terminated.

8.3 Performance of Obligations

Parent and Sweden Seller shall have performed or complied (i) in all respects with all obligations, agreements and covenants in Section 2.1(a) and Section 9.1 that are required by this Stock Purchase Agreement to be performed or complied with by Parent and Sweden Seller at or prior to the Closing Date and (ii) in all material respects with all other obligations, agreements and covenants required to be performed or complied with by each of them under this Stock Purchase Agreement at or prior to the Closing Date (it being understood that no action taken after the date of this Stock Purchase Agreement by Parent or its Affiliates solely in performance of the covenants or undertakings of this Stock Purchase Agreement, or as directed or consented to in writing by Sweden Buyer or U.S. Buyer under Section 5.3 or

otherwise, shall constitute a breach of any obligations, agreements or covenants of Parent or Sweden Seller).

8.4 No Company Material Adverse Effect

Since the date of this Stock Purchase Agreement, there shall not have occurred and be continuing a Company Material Adverse Effect, nor shall any event or events have occurred and be continuing that, individually or in the aggregate, would reasonably be expected to result in a Company Material Adverse Effect.

8.5 No Law

There shall not be in effect any Law enacted, entered, promulgated, enforced or otherwise issued or other legal restraint or prohibition restraining, enjoining, preventing or making illegal the consummation of the transactions contemplated by the Transaction Documents.

8.6 Parent Officer's Certificate

Parent shall have delivered to Sweden Buyer (for itself and on behalf of U.S. Buyer) a certificate, dated as of the Closing Date and executed by a duly authorized representative of Parent and Sweden Seller, certifying to the fulfillment of the conditions set forth in Section 8.1 and Section 8.3.

8.7 Documents prior to or at Closing

All deliveries required to be made by Parent or Sweden Seller to U.S. Buyer and Sweden Buyer prior to or at Closing shall have been made as set forth in Section 9.1.

8.8 U.S. EPA GHG Credits

After publication of the EPA GHG Emissions Regulations, any one or more of the following clauses (a)-(e) is true: (a) there is no projected Sweden Company GHG Credit Shortfall on the basis of Sweden Company's 2009B Cycle Plan (attached hereto as Schedule GG, the "Cycle Plan") and the volume and mix (product and powertrain) assumptions for the U.S. market attached as Schedule HH (the "Business Plan"), for each of Model Years 2012 and 2013, (b) upon Closing Sweden Company is eligible for Temporary Lead-time Allowance Alternative Standards that would eliminate the Sweden Company GHG Credit Shortfall for each of Model Years 2012 and 2013 for mobile source green house gas emissions based on the Cycle Plan and the Business Plan, (c) Parent shall have entered into a written agreement with Sweden Company to transfer and/or allocate GHG Credits to Sweden Company in an amount and at a time sufficient to cover any Sweden Company GHG Credit Shortfall, projected as of Closing, in each of the Model Years 2012 and 2013 in respect of which such Sweden Company GHG Credit Shortfall is projected and without requiring payment of substantial consideration from Sweden Company or the imposition of other onerous terms or conditions and the EPA GHG Emissions Regulations permit the transfer and/or allocation of such GHG Credits, (d) Parent shall have caused Sweden Company to implement a Vehicle compliance plan that enables Sweden Company to be in compliance with the EPA GHG Emissions Regulations for each of Model Years 2012 and 2013 (assuming constant Vehicle volume and mix (product and powertrain)) reasonably acceptable to Buyers, provided that Parent shall be responsible for any incremental cost in connection with such Vehicle compliance plan, or (e) Parent shall have agreed to provide an alternative technical, commercial or

regulatory arrangement that would enable Sweden Company to be in compliance with the EPA GHG Emissions Regulations for each of Model Years 2012 and 2013 and which is reasonably acceptable to Buyers (it being understood that Parent shall be responsible for any incremental cost in connection with such arrangement).

9. CLOSING

9.1 Deliveries by Parent and Sweden Seller

At Closing, Parent and Sweden Seller shall deliver, or shall cause to be delivered, to U.S. Buyer and Sweden Buyer each of the following:

- (a) stock certificates evidencing the Sweden Company Shares duly endorsed in favor of Sweden Buyer;
- (b) an updated copy of the shareholders' register reflecting Sweden Buyer's 100% ownership of the Sweden Company Shares;
- (c) an updated copy of the register of membership interests reflecting U.S. Buyer's 100% ownership of the U.S. Company Interest;
- (d) the LLC Interest Assignment and Assumption Agreement, duly executed by Parent;
- (e) the joint instructions for the delivery of the Goodwill Deposit duly executed by Parent and Sweden Seller pursuant to Section 2.8(b);
- (f) the aggregate amount of fees to be reimbursed to Buyers pursuant to Section 5.1(d) and Section 5.1(e) as notified by Buyers to Parent not later than two Business Days prior to the Closing Date;
- (g) to the extent not already in the possession or control of a Target Company, the Formation Documents, Bylaws, Company Records, corporate seal, share or member certificate book (with any unissued share or member certificates) and all minute books and other statutory books (which shall record all proceedings to Closing) of the Target Companies, or such equivalent items in the relevant jurisdiction as are kept by the Target Companies or which applicable Law requires to be kept, provided that in no circumstances will any of the aforementioned items or the shareholders' register or register of members interests be brought into the United Kingdom;
 - (h) the certificate described in Section 8.6;
- (i) a certificate of the Corporate Secretary of Parent certifying (A) that this Stock Purchase Agreement, the Separation Agreements and the transactions contemplated by this Stock Purchase Agreement and by the Separation Agreements have been duly authorized by Parent and each relevant member of Parent Group; and (B) that attached thereto are true, complete and correct copies of all resolutions of the board of directors (or equivalent supervisory body) of each relevant member of Parent Group authorizing the execution, delivery and performance of this Stock Purchase Agreement, Separation Agreements and the transactions contemplated by this Stock Purchase Agreement and the Separation Agreements;

- (j) a certificate certifying that Parent is not a "foreign person" as defined in section 1445 of the U.S. Code;
- (k) the written resignations of those members of the board of directors and officers, or equivalent, of each Target Company and each Company Subsidiary identified on Schedule Z, with each such resignation effective as of the Closing Date, together with such further documents, if any, as are legally required to effect such resignations in the relevant jurisdiction;
- (l) each of the Separation Agreements duly executed by the relevant member of Parent Group, Target Company or Company Subsidiary party to each Separation Agreement;
- (m) evidence of the release of the Encumbrance set forth on Schedule B Part 3.6 at or prior to Closing, in a form reasonably satisfactory to Buyers; and
 - (n) the Seller Financing Agreement duly executed by Sweden Seller.

9.2 Deliveries by U.S. Buyer and Sweden Buyer

- (a) At Closing, U.S. Buyer and Sweden Buyer shall deliver, or shall cause to be delivered, to Parent each of the following:
 - (i) the Net Initial Purchase Price pursuant to Section 2.3;
 - (ii) the joint instructions for the delivery of the Goodwill Deposit duly executed by Beijing Holdco pursuant to Section 2.8(b);
 - (iii) the LLC Interest Assignment and Assumption Agreement, duly executed by U.S. Buyer;
- (iv) the aggregate amount of fees to be reimbursed to Parent and Sweden Seller pursuant to Section 5.1(d) as notified by Parent to Buyers not later than two Business Days prior to the Closing Date;
 - (v) the certificate described in Section 7.5;
- (vi) a certificate of the Corporate Secretary of U.S. Buyer and Sweden Buyer certifying (A) that this Stock Purchase Agreement, the Separation Agreements and the transactions contemplated by this Stock Purchase Agreement and the Separation Agreements have been duly authorized by U.S. Buyer, Sweden Buyer and each other relevant member of Buyer Group, and (B) that attached thereto are true, complete and correct copies of all resolutions of the board of directors (of equivalent supervisory body) of each relevant member of Buyer Group authorizing the execution, delivery and performance of this Stock Purchase Agreement, the Separation Agreements and the transactions contemplated by this Stock Purchase Agreement and the Separation Agreements;
 - (vii) each of the Separation Agreements duly executed by the relevant Buyer Group party to each Separation Agreement;

- (viii) the Seller Financing Agreement duly executed by the relevant members of Geely Group; and
- (ix) evidence of the cash account balance of Sweden Buyer being at least RMB 9,600,000,000 (as converted into U.S. Dollars or SEK) less the amount of the Net Initial Purchase Price delivered to Parent pursuant to Section 9.2(a)(i).
- (b) At Closing, Sweden Buyer shall deliver, or shall cause to be delivered, to Sweden Company any amounts required to be delivered pursuant to Section 6.37(c).
- (c) Immediately following Closing, Sweden Buyer shall procure that each Swedish Entity holds an extraordinary general meeting of shareholders at which such Swedish Entity shall elect new board members, and acknowledge the resignation of the board members resigning from the respective boards of directors of the Swedish Entities pursuant to Section 9.1(k).
- (d) Immediately following Closing, Sweden Buyer shall deliver, or shall cause to be delivered, to Parent a copy, including a stamp of arrival, of applications for registration with the Swedish Companies Registration Office (*Bolagsverket*) regarding the resignations of the board members resigning from the respective boards of directors of the Swedish Entities pursuant to Section 9.1(k).

10. SURVIVAL OF CLAIMS; INDEMNIFICATION; REMEDIES

10.1 Survival of Claims

- Except as provided in Section 10.1(b), claims may be brought in respect of the representations and warranties set forth in this Stock Purchase Agreement, and the indemnification obligations with respect thereto, until 11:59 p.m. New York time on the date that is 18 months after the Closing Date, provided that such claims may be brought in respect of the representations and warranties set forth in Section 3.1(a), Section 3.1(c), Section 3.1(d), Section 3.2, Section 3.5, Section 3.6, Section 4.1 and Section 4.2 (such representations and warranties, the "Specified Representations"), and the indemnification obligations with respect thereto, until the expiration of the applicable statute of limitations (or, if there is no applicable statute of limitations, until 11:59 p.m. New York time on the date that is seven years after the Closing Date).
- (b) Claims may be brought in respect of the representations and warranties with respect to Taxes set forth in Section 3.12, and the covenants with respect to Taxes described in Section 6.2, Section 6.3 and Section 6.6, until 11:59 p.m. New York time on the 30th day following the later of (i) the expiration of the applicable statute of limitation, including any extension thereto, for the assessment of such Taxes and (ii) the final determination, which is not subject to appeal or with respect to which the time for appeal has expired, of any audit, examination, investigation or other proceeding relating to such Taxes.
- (c) Subject to Section 10.1(b), claims arising in respect of covenants and agreements which by their terms do not contemplate performance after the Closing Date may be brought until 11:59 p.m. New York time on the date that is 18 months after the Closing Date and claims arising in respect of covenants and agreements which by their terms contemplate performance after the Closing Date may be brought until the earlier of (i) 11:59 p.m. New York time on the 30th day following the period specified in such covenant or (ii) the time as such covenant is fully performed.

- (d) Claims arising in relation to Indemnified Product Recall Costs pursuant to Section 10.2(c) may be brought until 11:59 p.m. New York time on the date that is eight years after the Closing Date.
- (e) Claims arising in relation to Indemnified Environmental Losses pursuant to Section 10.2(d) may be brought until 11:59 p.m. New York time on the date that is 54 months after the Closing Date.
- (f) Claims arising in relation to Excluded Pensions Costs pursuant to Section 10.2(f) may be brought until 11:59 p.m. New York time on the following dates: (i) with respect to claims relating to subsections (a) and (c) of the definition of Excluded Pensions Costs, the date that is 72 months after the Closing Date, and (ii) with respect to claims relating to subsection (b) of the definition of Excluded Pensions Costs, the date that is 24 months after the Closing Date.
- (g) Notwithstanding any provision to the contrary in this Article 10, any claim for breach of a representation and warranty or covenant or agreement that is duly and properly notified in accordance with the terms of this Stock Purchase Agreement prior to the expiration of the applicable survival period specified under Section 10.1(a), Section 10.1(b), Section 10.1(c), Section 10.1(d), Section 10.1(e), or Section 10.1(f), as the case may be, will continue to survive under this Stock Purchase Agreement while such claim is reasonably pursued until (i) a final settlement Agreement is executed and delivered by the Parties, or (ii) a final Order or award in respect thereof, which is not subject to appeal or with respect to which the time for appeal has expired, has been issued by a court, arbitral tribunal, administrative agency or other applicable Governmental Entity, of competent jurisdiction.

10.2 Indemnification by Parent

Subject to the terms and conditions of this Stock Purchase Agreement, including the terms and conditions of this Article 10, after the Closing Date, Parent shall indemnify and hold harmless each member of Buyer Group and their respective direct or indirect stockholders, members, managers, officers, directors, employees, agents, successors and assigns (each, a "Buyer Indemnitee") from and against, and pay and reimburse each Buyer Indemnitee for any Losses imposed on, sustained or suffered by any Buyer Indemnitee, whether in respect of Third Party Claims or claims between the Parties resulting from, arising out of, or incurred or suffered by any Buyer Indemnitee in connection with, or otherwise with respect to:

- (a) the breach of any representation and warranty of Parent or Sweden Seller made in Article 3 of this Stock Purchase Agreement or contained in the certificate furnished to Buyers pursuant to Section 8.6 (it being understood that no action taken after the date of this Stock Purchase Agreement by Parent or its Affiliates solely in performance of the covenants or undertakings of this Stock Purchase Agreement, or as directed or consented to in writing by Sweden Buyer or U.S. Buyer under Section 5.3 or otherwise, shall constitute a breach of any representation or warranty of Parent or Sweden Seller);
- (b) a breach of any covenant or agreement of Parent or Sweden Seller set forth in this Stock Purchase Agreement (other than Section 6.2(a), to which the provisions of Section 6.2(a) shall apply) (it being understood that no action taken after the date of this Stock Purchase Agreement by Parent or its Affiliates solely in performance of the covenants or undertakings of this Stock Purchase Agreement, or as directed or consented to in writing by Sweden Buyer or U.S. Buyer under Section 5.3 or otherwise, shall constitute a breach of any obligations, agreements or covenants of Parent or Sweden Seller);

- (c) without regard to any disclosure under Schedule B, any Indemnified Product Recall Costs solely to the extent exceeding \$20,000,000 for any Product Recall with respect to a particular recall event or series of related recall events (with Parent being liable only for the amount that exceeds \$20,000,000 for any such event or series of related events);
- (d) without regard to any disclosure under Schedule B, but subject to Section 10.8, any Indemnified Environmental Losses in an amount exceeding in the aggregate \$3,000,000 incurred or suffered by any Buyer Indemnitee which, in each individual instance, exceeds \$250,000 (prior to the application of the net deductions provided in Section 10.9);
 - (e) without regard to any disclosure under Schedule B, any Indemnified JLR/AML Costs; and
 - (f) without regard to any disclosure under Schedule B, any Excluded Pensions Costs,

provided that, in each case, U.S. Buyer or Sweden Buyer makes a written claim for indemnification against Parent within the applicable survival period specified in Section 10.1, and provided, further, that to make a claim for indemnification under this Section 10.2 (a "Buyer Claim") other than with respect to a Third Party Claim to which the provisions of Section 10.7 shall apply and other than, for the avoidance of doubt, in the case of a Tax Proceeding to which the provisions of Section 6.5 shall apply, U.S. Buyer or Sweden Buyer must notify Parent in writing in reasonable detail, of the nature of such Buyer Claim including an estimate (on a without prejudice basis) of the amount of such Buyer Claim. Notwithstanding any provision to the contrary in this Stock Purchase Agreement, Sweden Seller shall not have any indemnification obligations under this Section 10.2 or otherwise under this Stock Purchase Agreement.

10.3 Limitations on Indemnification by Parent

Notwithstanding any provision to the contrary in this Stock Purchase Agreement:

- (a) Buyer Indemnitees shall be limited to only one recovery on account of any single Loss in respect of the indemnification provisions of Section 10.2 and the covenants of Section 6.2(a), even though such Loss may have resulted from the breach of more than one of the representations and warranties or covenants or agreements made by Parent or Sweden Seller in this Stock Purchase Agreement;
 - (b) the indemnification obligations provided by Section 10.2(a) shall be subject to the following limitations:
- (i) no Buyer Indemnitee shall be entitled to indemnification for Losses that do not in each individual instance exceed \$50,000 incurred by it;
- (ii) no Buyer Indemnitee shall be entitled to indemnification for individual Losses unless and until all Losses exceed, on a cumulative basis, \$9,000,000 (the "Basket Amount"), and from and after such time as the total amount of Losses under Section 10.2(a) exceeds the Basket Amount, Parent shall be liable only for the amount that exceeds the Basket Amount;
 - (iii) Parent's aggregate Liability shall not exceed \$225,000,000 (the "Cap"),

provided, however, that the limitations set forth in this Section 10.3(b) shall not apply to any Loss based on any breach of the Specified Representations;

- (c) Parent's aggregate liability under Section 6.2(a) shall not exceed an amount equal to the Purchase Price;
- (d) Parent's aggregate liability under Section 10.2(d) shall not exceed \$405,000,000;
- (e) no Buyer Indemnitee shall be entitled to indemnification for any Loss (except in the case of any claim pursuant to the covenants in Section 6.2(a), to which the provisions of Section 6.2(a) shall apply) to the extent that a reserve, provision or accrual for such Loss has been taken into account in determining the Net Working Capital and reduces Net Working Capital, or has been taken into account in determining Net External Indebtedness and increases Net External Indebtedness;
- (f) for the purposes of determining whether the limitations set forth in Section 10.3(b)(iii), Section 10.3(c), or Section 10.3(d) (as applicable) have been reached or exceeded, no account shall be taken of any amounts paid or payable pursuant to Section 12.6(d)(iii) to the extent that such amount was paid or payable or was increased solely as a result of Parent rather than Sweden Seller being the payor of such Indemnity Payment, provided, however, that this Section 10.3(f) shall not be applicable if Sweden Buyer fails to be organized under the Law of Sweden after Closing.

10.4 Indemnification by U.S. Buyer, Sweden Buyer

Subject to the terms and conditions of this Stock Purchase Agreement, including the terms and conditions of this Article 10, after the Closing Date, U.S. Buyer and Sweden Buyer shall indemnify and hold harmless each member of Parent Group and their respective direct or indirect stockholders, members, managers, officers, directors, employees, agents, successors and assigns (each, a "Parent Indemnitee") from and against, and pay and reimburse each Parent Indemnitee for any Losses imposed on, sustained or suffered by any Parent Indemnitee, whether in respect of Third Party Claims or claims between the Parties resulting from, arising out of, or incurred or suffered by any Parent Indemnitee in connection with, or otherwise with respect to:

- (a) the breach of any representation and warranty of U.S. Buyer or Sweden Buyer made in Article 4 of this Stock Purchase Agreement or contained in the certificate furnished to Parent pursuant to Section 7.5; and
- (b) a breach of any covenant or agreement of U.S. Buyer or Sweden Buyer set forth in this Stock Purchase Agreement (other than Section 6.2(b), to which the provisions of Section 6.2(b) shall apply),

provided that, in each case, Parent makes a written claim for indemnification against U.S. Buyer or Sweden Buyer within the applicable survival period specified in Section 10.1; and provided, further, that to make a claim for indemnification under this Section 10.4 (a "Seller Claim") other than with respect to a Third Party Claim to which the provisions of Section 10.7 shall apply and other than, for the avoidance of doubt, in the case of a Tax Claim pursuant to Section 6.2, to which the provisions of Section 6.5 shall apply, Parent must notify Buyers in writing in reasonable detail, of the nature of such Seller Claim including an estimate (on a without prejudice basis) of the amount of such Seller Claim.

10.5 Limitations on Indemnification by U.S. Buyer, Sweden Buyer

Notwithstanding any provision to the contrary in this Stock Purchase Agreement:

- (a) Parent Indemnitees shall be limited to only one recovery on account of any single Loss in respect of the indemnification provisions of Section 10.4 and the covenants of Section 6.2(b), even though such Loss may have resulted from the breach of more than one of the representations and warranties or covenants or agreements made by U.S. Buyer or Sweden Buyer in this Stock Purchase Agreement;
 - (b) the indemnification obligations of Section 10.4(a) shall be subject to the following limitations:
- (i) no Parent Indemnitee shall be entitled to indemnification for Losses that do not in each individual instance exceed \$50,000 incurred by it;
- (ii) no Parent Indemnitee shall be entitled to indemnification for individual Losses unless and until all Losses exceed, on a cumulative basis, the Basket Amount, and from and after such time as the total amount of Losses exceeds the Basket Amount, U.S. Buyer and Sweden Buyer shall be liable only for the amount which exceeds the Basket Amount; and
 - (iii) U.S. Buyer's and Sweden Buyer's aggregate Liability shall not exceed the Cap,

provided, however, that the limitations set forth in this Section 10.5(b) shall not apply to any Loss based on any breach of the Specified Representations; and

(c) U.S. Buyer's and Sweden Buyer's aggregate liability under Section 6.2(b) shall not exceed an amount equal to the Purchase Price.

10.6 Effect of Investigation; Waiver; No Target Company or Company Subsidiary Liability

- (a) The right to indemnification, payment in respect of Losses or other remedy based on the representations and warranties, covenants or agreements pursuant to any provision of this Stock Purchase Agreement will not be affected by any investigation conducted with respect to, or any Knowledge of Buyer or Parent acquired (or capable of being acquired), at any time, whether before or after the execution and delivery of this Stock Purchase Agreement or the Closing Date, with respect to the accuracy or inaccuracy of or compliance with, any such representation and warranty, covenant or agreement.
- (b) The waiver of any condition based on the accuracy of any representation and warranty or on the performance or compliance with any covenant or agreement will not affect the right to indemnification, payment in respect of Losses or other remedy based on such representations and warranties, covenants or agreements. Representations and warranties, covenants and agreements shall not be affected or deemed waived by reason of the fact that the Party seeking indemnification, payment in respect of Losses or other remedy knew or should have known that any representation and warranty might be inaccurate or that there was a failure to comply by the Party from whom recovery is sought with any covenant or agreement.

(c) Except as set forth on Schedule K, Parent and Sweden Seller acknowledge and agree that, upon and following the Closing, neither the Target Companies nor any Company Subsidiary shall have any liability or obligation to indemnify, save or hold harmless or otherwise pay, reimburse or make Parent or Sweden Seller whole for or on account of any indemnification claims made by any Buyer Indemnitee under this Article 10 and neither Parent nor Sweden Seller shall have any right of contribution against the Target Companies or any Company Subsidiary with respect to any such indemnification claim.

10.7 Third Party Claims

In the event that a Buyer Indemnitee or Parent Indemnitee receives notice of a Claim by a third party in respect of which indemnification may be sought under this Article 10, including in respect of penalties, fines and assessments of a Governmental Entity, and including Environmental Proceedings ("Third Party Claims"), the following shall apply:

- (a) The Person seeking indemnification with respect to such Third Party Claim pursuant to Section 10.2 or Section 10.4 (the "Indemnified Party") shall give notice within 30 Business Days (or earlier, if an earlier response shall be required in any proceedings related to a Third Party Claim) after receipt by the Indemnified Party of written notice of the Third Party Claim, including a response to an application for a temporary restraining order, preliminary injunction or other immediate or emergency relief, to the Party from whom indemnification is sought (the "Indemnifying Party") of the assertion of such Third Party Claim. Such notice shall provide reasonable detail (to the extent available) as to the nature of the Third Party Claim and if practicable on a without prejudice basis, an estimate of the amount of such Third Party Claim, and shall include copies of any notices and documents that are then in the Indemnified Party's possession and which relate to the Third Party Claim. The Indemnified Party's failure to notify the Indemnifying Party in accordance with the provisions of this Section 10.7(a) shall not result in the forfeit of any rights or claims to indemnification under this Stock Purchase Agreement with respect to such Third Party Claim or any subsequent claims relating thereto or arising in connection therewith unless, and only to the extent that, the Indemnifying Party is actually materially prejudiced by such failure.
- (b) Subject to Section 10.7(c), the Indemnifying Party shall have the right, upon notice to the Indemnified Party within 30 Business Days (or less if the nature of the Third Party Claim requires) from the date on which the Indemnifying Party received notice of the Third Party Claim, to notify the Indemnified Party of the Indemnifying Party's election to investigate, contest, defend (including the right to retain counsel reasonably acceptable to the Indemnified Party of its own choosing and to control the defense of the Third Party Claim) the Third Party Claim and, subject to Section 10.7(e) and Section 10.7(f), may, at any time after such notification to the Indemnified Party, settle the Third Party Claim.
 - (c) The Indemnifying Party will not be entitled to assume, investigate, contest, defend or settle a Third Party Claim if the Third Party Claim:
- (i) relates to or arises in connection with any criminal proceeding, action, indictment or investigation in which the Indemnified Party is a named defendant; or
- (ii) the Third Party Claim would give rise to Losses which comprise less than 50 percent of the total value of the claim giving rise to the Third Party Claim indemnifiable by the Indemnifying Party pursuant to Section 10.2(a) or Section 10.4(a).

- (d) If the Indemnifying Party chooses to defend a Third Party Claim in accordance with Section 10.7(b):
- (i) the Indemnifying Party shall be deemed to have acknowledged that such Third Party Claim is within the scope of its indemnity obligation hereunder and shall hold the Indemnified Party harmless from and against the full amount of any Losses resulting therefrom (subject to the other provisions of this Stock Purchase Agreement, including this Article 10);
- (ii) the Indemnified Party shall have the right to be represented by counsel and accountants, at its own expense, and shall be kept reasonably informed by the Indemnifying Party as to such Third Party Claim, whether or not the Indemnified Party is represented by its own counsel;
 - (iii) the Indemnifying Party shall conduct the defense or prosecution of such Third Party Claim in good faith; and
- (iv) the Indemnified Party shall allow the Indemnifying Party and its advisors (A) to investigate on a reasonable basis the matters or circumstances alleged to have given rise to the Third Party Claim, (B) to have reasonable access to, and copies of, any books, records, documents or other information relating to the matters or circumstances alleged to have given rise to the Third Party Claim and in the possession of the Indemnified Party and (C) to have reasonable access to all employees having responsibility for matters or circumstances alleged to have given rise to the Third Party Claim; provided, however, that any such books, records, documents or other information that are made available hereunder shall be held in strict confidence by the Indemnifying Party.
- (e) In the event that the Indemnifying Party fails to elect, or is not permitted, to assume the defense or prosecution of such Third Party Claim, or the Indemnifying Party fails to reasonably defend or prosecute such Third Party Claim in good faith, in accordance with this Section 10.7:
- (i) the Indemnified Party shall have the right, with counsel of its choice, to investigate, contest, defend and control (including, subject to Section 10.7(f), settle) the Third Party Claim at the sole cost and expense of the Indemnifying Party;
- (ii) the Indemnified Party shall conduct the defense or prosecution of the Third Party Claim in good faith and the Indemnifying Party will provide reasonable cooperation in the defense or prosecution of any such Third Party Claim; and
 - (iii) the Indemnified Party shall have no obligation to assume the defense or prosecution of a Third Party Claim.
- (f) Neither the Indemnifying Party nor the Indemnified Party, as the case may be, shall enter into any settlement, compromise, consent to the entry of an Order in respect of or otherwise discharge any Third Party Claim without the prior written consent of the Indemnified Party or the Indemnifying Party, respectively, which consent will not be unreasonably withheld, conditioned or delayed, provided that the consent of the Indemnified Party shall not be required if all of the following conditions are met: (A) the terms of the Order or proposed settlement include as an unconditional term thereof the giving to the Indemnified Party by the third party of a release of the Indemnified Party from all liability in respect of such Third Party Claim, (B) there is no finding or admission of any violation of Law by the Indemnified

Party (or any of its Affiliates) or any violation of the rights of any Person, (C) the settlement would not adversely effect any other claim of a similar nature that may be made against the Indemnified Party (or any of its Affiliate) and (D) the sole form of relief is monetary damages that are paid in full by the Indemnifying Party.

- (g) With respect to a Third Party Claim, after a final Order, which is not subject to appeal or with respect to which the time for appeal has expired, shall have been issued by a court, arbitral tribunal, administrative agency or other applicable Governmental Entity of competent jurisdiction, or a settlement shall have been consummated in accordance with this Article 10, or the Parties shall have arrived at a mutually-binding agreement with respect to any such Third Party Claim, the Indemnified Party shall give prompt notice to the Indemnifying Party of the amounts due and owing by the Indemnifying Party with respect to such matter, and the Indemnifying Party shall pay all of the amounts so owing in accordance with the provisions of Section 10.10.
- (h) The provisions of this Section 10.7 shall not apply with respect to a Tax Claim which shall be governed by the applicable provisions of Section 6.4 or Section 6.5, as the case may be, provided, however, that the provisions of this Section 10.7 will apply to Tax Claims described in Section 6.2(a) (iv) or Section 3.12(k). The provisions of this Section 10.7 will not apply to any Tax matter covered by Section 6.3.
- (i) Each of the Parties agrees to render to each other such assistance as it may reasonably be required of the other in order to facilitate the proper and adequate defense of a Third Party Claim.

10.8 Environmental Claims

- (a) No Buyer Indemnitee shall be entitled to indemnification under Section 10.2(d) to the extent that any Losses result from or are increased as a consequence of:
- (i) a material change of use of any Site or any part of such property after the Closing Date, except to the extent reasonably necessary for or related to the Business, provided that the Buyer Indemnitee procures that reasonable steps are taken to avoid, minimize and mitigate the relevant potential Losses;
- (ii) any material development, construction or demolition (excluding routine maintenance) in relation to any Site after the Closing Date, except to the extent reasonably necessary for or related to the Business, provided that the Buyer Indemnitee procures that reasonable steps are taken to avoid, minimize and mitigate the relevant potential Losses;
- (iii) any sale of any Site or any part thereof or the expiration of any leases in relation to such Site, except where any such sale or lease expiration will not give rise to any obligations under Environmental Law for investigative works to be undertaken in relation to that Site related to the potential presence of Hazardous Materials, and:
 - (A) the sale is to any member of the Buyer Group; or
- (B) the lease is held over, continued, extended or renewed (whether by statute, agreement or otherwise) or replaced by a new lease to the same tenant or any other member of the Buyer Group; or

EXECUTION VERSION

(0	C)	the successor in title or	successor in inte	rest of that Site agr	rees in writing	with Parent or Sw	eden Seller	(as relevant)
to prevent any of the matters lis	ted in the	e subparagraphs of Sec	tion 10.8(a) from	occurring and to	be subject to th	ne terms of this S	Section 10.8	as if it were
"Buyers" under this Stock Purcha	ase Agree	ement;						

- (iv) any action by any member of Buyer Group which causes or exacerbates contamination of soil and groundwater by Hazardous Materials at, or, under or migrating from any Sites after the Closing Date, except any action which:
- (A) a member of Buyer Group did not know, nor ought reasonably to have known, would cause or exacerbate such contamination;
- (B) is necessary to comply with the express written requirements of a formal Order issued by a Governmental Entity under Environmental Law; or
 - (C) is necessary to deal with the immediate cause of an Environmental Emergency;
- (v) the undertaking, authorizing, initiation or procuring of any intrusive investigations at or in respect of any Sites by any member of Buyer Group without the prior written consent (not to be unreasonably withheld or delayed) of Parent, except where such investigations:
- (A) are required by Environmental Law, any Material Environmental Permit, or court Order, or required to renew a Material Environmental Permit; or
 - (B) are reasonably necessary to respond to an Environmental Emergency; or
- (C) which are the minimum necessary for the purposes of development, maintenance or engineering works at any Site, provided that the Buyer Indemnitee can demonstrate that such investigations are genuinely required and that no part of its intention was or is to procure information with a view to making a claim under Section 10.2; or
- (D) which are necessary to monitor known contamination for the purpose of preventing the occurrence of an Environmental Emergency, where such monitoring would be carried out by a reasonable and prudent operator in the Buyer Indemnitee's position, acting with regard to relevant local Environmental Laws but without regard to recourse under Section 10.2;
 - (vi) any act or omission of a member of Buyer Group that is unlawful under Environmental Law;
- (vii) any Works for the purposes of preventing, containing, ameliorating or remediating any contamination of soil and groundwater by Hazardous Materials at, on, under or migrating from any Sites which go beyond the Works reasonably necessary to achieve the minimum standards that are required under Environmental Law or to address the immediate cause and control the immediate effects of an Environmental Emergency;
- (viii) the disclosure of information (or the authorization of such disclosure) by any member of Buyer Group concerning contamination or presence of Hazardous Materials to any

EXECUTION VERSION

Governmental Entity or any third party without the prior written consent (not to be unreasonably withheld or delayed) of Parent, except where such disclosure

- (A) required by Environmental Law, any Material Environmental Permit, or court Order, or required to renew a Material
- Environmental Permit; or
- (B) reasonably necessary to respond to an Environmental Emergency; or
- (C) between members of the Buyer Group and their professional advisers (provided that such members and advisers are placed under an obligation to maintain the confidentiality of such information); or
- (ix) the passing of, or any change in, or entry into force of, after the Closing Date, any Law of any Governmental Entity or the requirements of any environmental Permit obtained after the date of this Stock Purchase Agreement, or any change after the date of this Stock Purchase Agreement of any generally accepted judicial interpretation of Environmental Law.
- At its sole discretion and having notified Buyer in writing, Parent may assume conduct of any Works relevant to any matter to which Section 10.2(d) may apply, and Buyers shall take such reasonable steps as Parent may reasonably require to enable such assumption by Parent.
- In the event Parent assumes conduct of any Works, Parent shall (subject to appropriate arrangements to maintain confidentiality and privilege):
 - use commercially reasonable efforts to reduce any Indemnified Environmental Losses; (i)
 - (ii) consult in advance with Buyers in relation to such Works;
 - promptly provide regular reports to Buyers regarding the progress of such Works; (iii)
- permit Buyers to attend any relevant hearing, site visit or meeting with any third party, including any Governmental Entity, in relation to such Works;
- permit Buyers to be involved in discussion with the relevant consultants or contractors concerning the scope, nature and extent of any Works to be carried out and to attend any meetings with those consultants or contractors;
- comply with all reasonable requests of Buyers in relation to such Works (provided that if any such requests would increase the cost of the Works, Parent need not comply with such requests unless the Buyer agrees in writing to be liable for any such increase in cost), including any requests for the provision of copies of correspondence or documents held by Parent which are material to the Works (except to the extent that such material is legally privileged or disclosure would be prohibited by Environmental Law or the rules or Orders of a court); and

- (vii) not make any admission of liability or agree any settlement in relation to such Works without prior consultation with Buyers.
- (d) In the event Parent assumes conduct of any Works, Buyers shall, in relation to any such Works, cooperate with Parent, including providing to Parent all relevant information and access to personnel, premises, chattels, documents and records as Parent may reasonably request.
- (e) In the event Parent assumes conduct of any Works, Parent must make all reasonable efforts to avoid adverse effects on the carrying on of the business of the Buyers (including the Business), including business interruption.
- (f) Unless and until Parent elects to assume conduct of any Works in accordance with this Section 10.8, the provisions in Sections 10.8(c) and Section 10.8(e) shall apply to Buyers as if the word "Parent" were replaced in such provisions with the word "Buyers" and the word "Buyers" were replaced with the word "Parent."

10.9 Adjustment of Losses

The amount of any Loss for which indemnification is provided under this Article 10 or Section 6.2 shall be net of (a) any amounts recovered by the Indemnified Party pursuant to any indemnification by, or indemnification agreement with, any third party, and, (b) any insurance proceeds and all other cash receipts or sources of reimbursement actually paid to the Indemnified Party or any of its Affiliates in connection with the facts giving rise to the right of indemnification (net of any increase in premium payable by an Indemnified Party under any such insurance due to any such Loss). If, other than in the case of a claim pursuant to the covenants in Section 6.2(a), to which the provisions of Section 6.2(e) shall apply, the Indemnified Party receives any Relief or other Tax benefit attributable to such Loss, an amount equal to the decrease in Tax attributable to such Relief or other Tax benefit (net of any Tax payable on the amount received or that would have been payable but for the use of set off of any Relief) shall reduce the amount of Loss for which indemnification is provided or, if indemnification has already been provided, the Indemnified Party shall pay the Indemnifying Party such amount promptly following the date on which such Relief or other Tax benefit results in a decrease in liability to Tax. If at any time an Indemnified Party recovers from a third party in respect of a Loss subject to indemnification hereunder after payment has been made by the Indemnifying Party in full discharge of the liability giving rise to such payment, then the Indemnified Party shall reimburse the Indemnifying Party for amounts the Indemnifying Party has paid to the Indemnified Party in respect of such Loss to the extent the aggregate amounts received by the Indemnified Party from such third party (less any Tax suffered thereon, except to the extent that the payment made by the Indemnified Party to the Indemnifying Party pursuant to this sentence of Section 10.9 (the "Third Party Refund Payment") results in a decrease in liability to Tax of the Indemnified Party which would not have arisen but for the making of the Third Party Refund Payment. For the avoidance of doubt, where the Third Party Refund Payment results in the cancellation or reduction of the Tax which would otherwise be suffered on the payment received by the Indemnified Party from the relevant third party, such cancellation or reduction shall be a decrease in liability to Tax of the Indemnified Party for the purposes of the previous sentence) and amounts received by the Indemnified Party from the Indemnifying Party (less any Tax suffered on that payment) in respect of such Loss exceed the amount of such Loss for which the Indemnifying Party is responsible under this Article 10.

10.10 Payments

Indemnification shall be payable pursuant to Section 6.2(a), Section 6.2(b), Section 10.2 and Section 10.4, upon the resolution of such claim for indemnification by mutual Agreement between Parent and Sweden Buyer or the issuance of a final Order, which is not subject to appeal or with respect to which the time for appeal has expired, by a court having jurisdiction over the Parties and the subject matter of such claim for indemnification, or within ten Business Days of the issuance of a final arbitral award, unless within such ten Business Day period a judicial proceeding to confirm or vacate such award is filed, and in such event within ten Business Days of the issuance of a final Order of a court, administrative agency or other applicable Governmental Entity of competent jurisdiction in such proceeding that is not subject to appeal, provided that in respect of any Third Party Claim the defense of which was assumed by the Indemnifying Party any payment will be made directly by the Indemnifying Party, on behalf of the Indemnified Party, in accordance with any settlement Agreement or final Order or award in connection with such Third Party Claim.

10.11 Treatment of Payments

All payments made by Parent to U.S. Buyer or Sweden Buyer for breach of any warranty or pursuant to any indemnity or covenant to pay (including pursuant to this Article 10 or Section 6.2 and Section 6.3) shall be treated as adjustments to the Purchase Price, unless such treatment is inconsistent with Law. Except as otherwise provided in Section 6.2(a)(v) with respect to certain Deemed Tax Liabilities and except pursuant to any breach of the warranty set out in Section 3.12(v) or the covenants set out in Section 6.6, no Buyer Indemnitee shall be indemnified or reimbursed for any reduction of, or utilization of, any net operating loss carry forward, tax credit carry forward or similar Tax attribute or any Tax liability attributable to such reduction or utilization.

10.12 Mitigation; Exclusive Remedy

- Each applicable Indemnified Party shall take all commercially reasonable steps to mitigate any amounts payable with respect to Losses such Indemnified Party may incur with respect to any circumstances giving rise to a claim for indemnification under Section 6.2(a)(iv), Section 6.3, Section 10.2 (including any amounts payable pursuant to Section 12.6) or Section 10.4 (including any amounts payable pursuant to Section 12.6); provided that this Section 10.12 shall not obligate any Indemnified Party to take commercially reasonable steps to mitigate any Losses such Indemnified Party may incur with respect to any circumstances giving rise to a claim pursuant to Section 6.2 other than a Claim described in Section 6.2(a)(iv). In the event that any such Indemnified Party fails to take such commercially reasonable steps to mitigate any Losses, then, notwithstanding any provision to the contrary in this Stock Purchase Agreement, such Indemnifying Party shall not be required to indemnify any Person for any Loss that would reasonably be expected to have been avoided if such Indemnified Party had taken such steps, provided that nothing in this Section 10.12(a) shall oblige an Indemnified Party to take any action which would be unduly onerous or materially prejudicial to it or its business (including the Business) and the application of this Section 10.12(a) shall not cause an Indemnified Party to suffer as a result of failure to take such unduly onerous or materially prejudicial action.
- (b) Except as set forth in Section 6.10 and Section 6.15, the provisions set forth in this Article 10, as well as Section 6.2, Section 6.4, Section 6.5, Section 6.6 and Section 12.13, constitute the exclusive remedy, following Closing, of Parent, Sweden Seller, U.S. Buyer and Sweden Buyer with respect to this Stock Purchase Agreement, including for any breach, or alleged breach, of any representation and warranty or covenant or agreement in this Stock Purchase Agreement, provided,

however, that nothing in this Section 10.12(b) shall limit the right of any Party to seek specific performance or injunctive relief, pursuant to Section 12.13 or otherwise, in connection with any breach, or an alleged or threatened breach, by another Party of its obligations under this Stock Purchase Agreement, whether before or following Closing, and, provided, further, that each Party acknowledges and agrees that in no event shall either Party or any of its Affiliates, directors, officers, employees, or representatives be liable to indemnify, reimburse or covenant to pay the other Party in any respect pursuant to this Stock Purchase Agreement, except (i) as provided for in this Article 10 and Section 6.2, Section 6.3, Section 6.4, Section 6.5, Section 6.6, Section 6.10 and Section 6.15 or (ii) in the case of actual fraud or willful misconduct, where the Party against whom such actual fraud or willful misconduct was committed shall have all rights and remedies available under this Stock Purchase Agreement and at Law, in equity or otherwise against the Party that committed such actual fraud or willful misconduct. The foregoing shall not limit the rights of the Parties under Section 12.5 or Section 12.19.

11. TERMINATION

11.1 Termination

This Stock Purchase Agreement may be terminated and the transactions contemplated by this Stock Purchase Agreement abandoned at any time before Closing:

- (a) by the mutual written consent of the Parties;
- (b) upon written notice by U.S. Buyer or Sweden Buyer at any time, provided that each of U.S. Buyer and Sweden Buyer is not then in material breach of any representation and warranty or covenant or agreement under this Stock Purchase Agreement, if there shall have been a breach of any representation and warranty made in this Stock Purchase Agreement by Parent or Sweden Seller or a breach or failure to perform in any respect any covenant or agreement contained in this Stock Purchase Agreement by Parent or Sweden Seller, and such breach or failure to perform prevents a condition in Article 8 from being satisfied and has not been cured within 90 days following delivery of written notice thereof to Parent and Sweden Seller;
- (c) upon written notice by Parent or Sweden Seller at any time, provided that each of Parent and Sweden Seller is not then in material breach of any representation and warranty or covenant or agreement under this Stock Purchase Agreement, if there shall have been a breach of any representation and warranty made in this Stock Purchase Agreement by U.S. Buyer or Sweden Buyer or a breach or failure to perform in any respect any covenant or agreement contained in this Stock Purchase Agreement by U.S. Buyer or Sweden Buyer, and such breach or failure to perform prevents a condition in Article 7 from being satisfied and has not been cured within 90 days following delivery of written notice thereof to U.S. Buyer and Sweden Buyer;
- (d) upon written notice by U.S. Buyer or Sweden Buyer if a Company Material Adverse Effect shall have occurred and be continuing (or would reasonably be expected to occur and continue) which has not been cured within 90 days following delivery of written notice thereof to Parent and Sweden Seller;
- (e) upon written notice by Parent or Sweden Seller, on the one hand, or by U.S. Buyer or Sweden Buyer, on the other hand, in the event that any Governmental Entity shall have issued an Order or taken any other action preventing or prohibiting Closing and such Order or other action shall have

become final without possibility of appeal; provided, however, that the Party so requesting termination shall have (i) used its commercially reasonable efforts to resist, resolve, annul, quash or lift such Order and (ii) complied with such requesting Party's obligations under Section 5.1;

- (f) upon written notice by Parent and Sweden Seller, on the one hand, or by U.S. Buyer and Sweden Buyer, on the other hand, if Closing has not occurred on or before the date that is nine months after the date of this Stock Purchase Agreement (the "Outside Date"), unless the Closing shall not have occurred due to the failure of the Party so requesting termination, or of its Affiliates, to perform or comply with any of the obligations, covenants, agreements or conditions of the Stock Purchase Agreement to be performed or complied with by it prior to Closing; or
- (g) upon written notice by Parent and Sweden Seller if all of the conditions in Article 8 have been satisfied (other than those conditions which are satisfied by action taken at Closing) and Buyers fail to pay the full amount of the Net Initial Purchase Price pursuant to Section 9.2(a) on the date which would otherwise be the Closing Date as contemplated by this Stock Purchase Agreement.

11.2 Effect of Termination

- (a) In the event that this Stock Purchase Agreement is terminated as provided in Section 11.1, this Stock Purchase Agreement shall forthwith become null and void and none of the Parties shall have any Liability hereunder or with respect to the transactions contemplated by this Stock Purchase Agreement, except that (i) the rights and obligations of the Parties and their Affiliates provided in the Confidentiality Agreement, the Escrow Agreement and the provisions of Article 1, Section 6.25, Section 6.35, this Section 11.2 and Article 12 shall not be extinguished but shall survive such termination and shall remain in full force and effect and (ii) nothing in this Section 11.2 shall be deemed to release or relieve any Party from any Liability for fraud or any material and intentional breach of any representation and warranty or covenant or agreement of this Stock Purchase Agreement. For the avoidance of doubt, the Parties shall have any and all remedies to enforce such surviving obligations as provided for at Law or in equity, including specific performance.
 - (b) In the event that this Stock Purchase Agreement is terminated:
- (i) by Parent or Sweden Seller as provided in Section 11.1(c), provided that, for purposes of this Section 11.2(b)(i), the 90 day cure period referenced in Section 11.1(c) shall be deemed to be the earlier of the expiration of such 90 day cure period and the Outside Date; or
- (ii) by any Party as provided in Section 11.1(e) if the Governmental Entity issuing an Order or taking other action preventing or prohibiting Closing is a Chinese Governmental Entity;
- (iii) by any Party as provided in Section 11.1(f) if all the conditions in Article 8 have been satisfied (other than those conditions which are satisfied by action taken at Closing) except for (A) the condition set forth in Section 8.2 to the extent such condition has not been satisfied because of a failure to receive any Buyer Governmental Approvals or Governmental Approvals required to be received from a Chinese Governmental Entity, or (B) the condition set forth in Section 8.5 to the extent such condition has not been satisfied because of the issuance of any Law or other legal restraint or prohibition by a Chinese Governmental Entity; or

(iv) by Parent and Sweden Seller as provided in Section 11.1(g),

then, in each case, Parent shall receive the Goodwill Deposit (together with any interest accrued thereon) and Parent shall, and Sweden Buyer shall cause Beijing Holdco to, promptly execute and deliver to Escrow Agent, in accordance with the terms of the Escrow Agreement, a joint instruction to immediately release the Goodwill Deposit (together with any interest accrued thereon) to Parent by wire transfer of immediately available funds.

(c) In the event that this Stock Purchase Agreement is terminated pursuant to Section 11.1(a), Section 11.1(b), Section 11.1(d), Section 11.1(e) (except under the circumstances described in Section 11.2(b)(iii)) or Section 11.1(f) (except under the circumstances described in Section 11.2(b)(iii)), then Sweden Buyer shall receive the Goodwill Deposit (together with any interest accrued thereon) and Parent shall, and Sweden Buyer shall cause Beijing Holdco to, promptly execute and deliver to Escrow Agent in accordance with the terms of the Escrow Agreement a joint instruction to immediately release the Goodwill Deposit to Sweden Buyer by wire transfer of immediately available funds.

12. MISCELLANEOUS

12.1 Notices

- (a) All notices, demands, requests and other communications to any Party as set forth in, or in any way relating to the subject matter of this Stock Purchase Agreement, must be in legible writing in the English language delivered by personal delivery, facsimile, email transmission or prepaid overnight courier using an internationally recognized courier service and shall be effective upon receipt, which shall be deemed to have occurred:
 - (i) at the time and date of personal delivery;
- (ii) if sent by facsimile or email transmission, at the time and on the date indicated on a confirmation of successful transmission page relating to such facsimile transmission or at the time and on the date of such email transmission;
 - (iii) at the time and on the date of delivery if delivered by courier as confirmed in the records of such courier service; or
 - (iv) at such time and date as delivery by personal delivery or courier is refused by the addressee upon presentation,

in each case provided that if such receipt occurred on a non-Business Day or after 5.00 p.m. local time in the place of receipt on a Business Day, then notice shall be deemed to have been received at the start of the next following Business Day and provided, further, that where any notice, demand, request or other communication is provided by any Party by email in accordance with this Section 12.1, such Party shall also provide a copy of such notice, demand, request or other communication by using one of the other methods set out in this Section 12.1.

All such notices, demands, requests and other communications shall be addressed as follows, or to such other address, number or email address as a Party may designate:

if to Parent or Sweden Seller:

Ford Motor Company One American Road - Suite 1134 Dearborn, Michigan 48126-2798 United States Facsimile: +1.313.248.8713 Email: psherry@ford.com

Attention: Corporate Secretary

with a copy (which shall not constitute notice) to:

Hogan & Hartson LLP 555 Thirteenth Street, NW Washington, D.C. 20004 United States Facsimile: +1.202.637.5910 Email: wjcurtin@hhlaw.com Attention: William J. Curtin, III, Esq.

if to U.S. Buyer or Sweden Buyer:

Geely Sweden AB c/o. Zhejiang Geely Holding Co., Ltd No 1760, Jiangling Road, Hangzhou, PRC

Facsimile: +86 571 2811 6078 Email: shenh@geely.com Attention: Mr. Shen Hui

with a copy (which shall not constitute notice) to:

Freshfields Bruckhaus Deringer LLP 65 Fleet Street London EC4Y 1HS

England

Facsimile: +44 20 7832 7001

christopher.bown@freshfields.com Email:

Attention: Christopher Bown

12.2 Amendment

Any provision of this Stock Purchase Agreement may be amended only if such amendment is in writing and executed and delivered by all of the Parties; provided, however, that following any assignment by Sweden Seller pursuant to Section 12.4(a), any provision of this Stock Purchase Agreement may be amended only if such amendment is in writing and executed and delivered by Parent,

U.S. Buyer and Sweden Buyer. Any amendment to this Stock Purchase Agreement shall be in the English language.

12.3 Waivers

- (a) Any provision of this Stock Purchase Agreement may be waived only if such waiver is in writing and executed and delivered by the Party against whom enforcement of the waiver is sought, and such writing expressly references the provision of this Stock Purchase Agreement that is waived.
- (b) No failure or delay by any Party in exercising any right, power or privilege under this Stock Purchase Agreement shall operate as a waiver of any such right, power or privilege nor shall any single or partial exercise of any such right, power or privilege preclude any other or further exercise of the same right, power or privilege or the exercise of any other right, power or privilege.
- (c) No waiver by any Party to this Stock Purchase Agreement of any breach of any representation and warranty or covenant or agreement hereunder or any failure or delay in exercising any right, power or privilege by any Party to this Stock Purchase Agreement with respect to any representation and warranty or covenant or agreement hereunder, in each case whether intentional or not, will be deemed to extend to any independent breach of any such representation and warranty or covenant or agreement hereunder or affect in any way any rights arising by virtue of any such breach.

12.4 Successors and Assigns

- (a) Following Closing, Sweden Seller shall have the right at any time to assign its rights and obligations under this Stock Purchase Agreement to Parent, provided that Parent shall execute and deliver to Buyers an assignment and assumption agreement substantially in the form attached to this Stock Purchase Agreement as Exhibit III and provided, further, that no such assignment shall make greater, enlarge, alter, modify or otherwise change any obligation or Liability of Buyers under this Stock Purchase Agreement or to any member of Parent Group or Parent Indemnitee. The Parties acknowledge and agree that upon such assignment by Sweden Seller, notwithstanding any provision to the contrary in this Stock Purchase Agreement, Sweden Seller shall have no Liabilities of any nature whatsoever, whether current, accrued or contingent, under this Stock Purchase Agreement.
- (b) Following Closing, each of Sweden Buyer and U.S. Buyer shall have the right at any time to assign its rights and obligations under this Stock Purchase Agreement to any member of Buyer Group, provided that no such assignment shall relieve Sweden Buyer or U.S. Buyer of any obligations or Liability under this Stock Purchase Agreement or make greater, enlarge, alter, modify or otherwise change any obligation or Liability of Parent or Sweden Seller under this Stock Purchase Agreement or to any member of Buyer Group or Buyer Indemnitee.
- (c) Except as set forth in Section 12.4(a) and Section 12.4(b), or as otherwise explicitly set forth in this Stock Purchase Agreement, no Party may assign, delegate or otherwise transfer (whether by merger, sale of stock, sale of substantially all the assets or business, by operation of law or otherwise) any of its rights or obligations under this Stock Purchase Agreement without the prior written consent of the other Parties to this Stock Purchase Agreement. Any purported assignment in violation of this Section 12.4 shall be null and void, and the Parties agree that no trust (constructive or otherwise) shall arise as a consequence and the assignor shall not be deemed to hold the benefit of this Stock Purchase Agreement, or any right under this Stock Purchase Agreement, as trustee for the purported assignee, transferee, delegate or other Person as a result.

(d) Subject to the terms and conditions of Section 12.4(a), Section 12.4(b) and Section 12.4(c), the provisions of this Stock Purchase Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns.

12.5 Payments

- (a) Except as otherwise specified in this Stock Purchase Agreement, any payment to be made pursuant to this Stock Purchase Agreement by any member of Buyer Group shall be made to Parent (for itself or as agent for the relevant member of Parent Group, as applicable) and Parent agrees to pay each member of Parent Group that part of each payment to which it is entitled, and any payment to be made pursuant to this Stock Purchase Agreement by any member of Parent Group shall be made to the Sweden Buyer (for itself or as agent for the relevant member of Buyer Group or Geely Group, as applicable) and Sweden Buyer agrees to pay each member of Buyer Group or Geely Group that part of each payment to which it is entitled.
- (b) Payment under this Section 12.5 shall be in immediately available funds by electronic transfer on the due date for payment. Receipt by Parent or Sweden Seller, as applicable, of the amount due shall be an effective discharge of the relevant payment obligation. Parent and Sweden Seller acknowledge and agree that no member of Buyer Group shall have, owe or incur any Liabilities to any member of Parent Group in respect of amounts so paid to Parent on behalf of any member of Parent Group and such payments to Parent shall discharge all Liabilities owing by the such relevant member of Buyer Group. U.S. Buyer and Sweden Buyer acknowledge and agree that no member of Parent Group shall have, owe or incur any Liability to any member of Buyer Group in respect of amounts so paid to Sweden Buyer on behalf of any member of Buyer Group and such payments to Sweden Buyer shall discharge all Liabilities owing by the relevant member of Parent Group.
- (c) If any sum due for payment in accordance with this Stock Purchase Agreement is not paid on the due date for payment, the person in default shall pay Default Interest on that sum from, but excluding, the due date to, and including, the date of actual payment.

12.6 Deductions from Payments; Gross-up

- (a) Except as set forth in Section 2.4(d) and Section 2.5(d), or otherwise in this Stock Purchase Agreement or as required by Law, any payment by a Party required to make such payment under this Stock Purchase Agreement (for purpose of this Section 12.6, the "payer") to or as directed by any other Party (for purpose of this Section 12.6, the "payee") shall be made in full without any set-off, restriction or condition, whether for or on account of any counterclaim or otherwise, and without deduction or withholding of any kind, other than any deduction or withholding required by Law.
- (b) If any deduction or withholding of Taxes is required by Law, the payer shall (i) make such deduction or withholding and (ii) pay the full amount deducted or withheld to the relevant Governmental Entity, in accordance with Law.
- (c) The Parties acknowledge and agree, for the avoidance of doubt, that where there is a deduction or withholding of Taxes required by Law from a payment under this Stock Purchase Agreement (other than an Indemnity Payment, to which Section 12.6(d) below will apply), the payee shall be deemed to have received the gross amount, before any such deduction or withholding.

- (d) Where a payer makes a payment in satisfaction of a liability arising in respect of any representation, warranty, covenant or agreement under the indemnification provisions of this Stock Purchase Agreement (including, for the avoidance of doubt, under the provisions of Section 6.2(a), Section 6.2(b), Section 10.2 and Section 10.4 and excluding, for the avoidance of doubt, the Purchase Price and any payments made pursuant to Section 2.4, Section 2.5 and Section 2.6) (an "**Indemnity Payment**"), then the following provisions shall apply:
- (i) if the payer makes a deduction or withholding required by Law from an Indemnity Payment, the payer shall make the deduction or withholding and any payment required to be made in connection with it within the time allowed by the relevant Law;
- (ii) the sum due from the payer shall be increased to the extent necessary to ensure that, after the making of any such deduction or withholding, the payee receives a sum equal to the sum it would have received had no deduction or withholding been made;
- (iii) if an Indemnity Payment is subject to Tax in the hands of the payee then the sum due shall be increased to the extent necessary to ensure that after payment of the Tax the Payee receives and retains a net sum equal to the sum it would have received had the payment not been subject to Tax; and
- (iv) the payee shall use its reasonable efforts to identify and apply any Tax credit or refund arising as a result of any deduction, withholding or increased payment referred to in paragraph (i), (ii) or (iii) above and shall pay to payer an amount (not exceeding the amount of any additional payment made under paragraph (ii) or (iii), as relevant, as a result of such withholding, deduction or payment) equal to any refund or reduction in Tax obtained as a result net of any Tax payable on the amount received, as and when such refund is received by it or the amount of Tax so reduced would otherwise have been payable.
- (e) For the avoidance of doubt, the provisions of this Section 12.6 shall not apply to payments made pursuant to the Seller Financing Agreement; such payments shall be governed by the provisions of the Seller Financing Agreement.

12.7 Currency Indemnity

If any sum due from one Party (for purposes of this Section 12.7, the "payer") to another Party (for purposes of this Section 12.7, the "payee") under this Stock Purchase Agreement or pursuant to any final Order relating to this Stock Purchase Agreement, which is not subject to appeal or with respect to which the time for appeal has expired, is required to be converted from the currency in which such sum is payable (the "Payment Currency") under this Stock Purchase Agreement or pursuant to such Order into another currency (the "Enforcement Currency") for the purposes of:

- (a) making or filing a Claim or proof against the payer;
- (b) obtaining an Order in any court or other tribunal of competent jurisdiction; or
- (c) enforcing any Order given or made relating to this Stock Purchase Agreement,

the payer shall indemnify and hold harmless the payee from and against any Loss suffered or incurred as a result of any difference between: (i) the rate of exchange used for such purpose to convert the sum in question from the Payment Currency into the Enforcement Currency and (ii) the rate of exchange at which the payee would in the Ordinary Course of Business, acting in a reasonable manner and in good faith, purchase the Payment Currency with the Enforcement Currency upon receipt of a sum paid to such payee in satisfaction, in whole or in part, of any such Claim, proof or Order. The expression "rate of exchange" includes any usual and customary premiums and costs of exchange payable in connection with the purchase of, or conversion into, the Enforcement Currency.

12.8 Entire Agreement

This Stock Purchase Agreement, along with the Confidentiality Agreement and the Separation Agreements, as well as any other document executed by or among the Parties necessary to consummate the transactions contemplated by this Stock Purchase Agreement (and any schedules, exhibits or other annexes of each), constitute the entire Agreement among the Parties with respect to the subject matter of this Stock Purchase Agreement and supersede all prior agreements, commitments, understandings and negotiations, both written and oral, among the Parties with respect to the subject matter of this Stock Purchase Agreement. Each Party acknowledges and agrees that, in entering into this Stock Purchase Agreement, it has not relied on, and shall have no right or remedy in respect of, any statement, representation, assurance, warranty (whether made negligently or innocently) other than as expressly set forth in this Stock Purchase Agreement, provided, however, that nothing in this Section 12.8 shall limit or exclude any liability (including liability under any other Agreement entered into between the Parties and their respective Affiliates, including this Stock Purchase Agreement) for actual fraud or willful misconduct.

12.9 Schedules; Exhibits; Appendices Preamble; Recitals

All Schedules, all Exhibits, all Appendices, the preamble and all recitals to this Stock Purchase Agreement, and all terms and conditions contained therein, will be construed with, and as integral parts of, this Stock Purchase Agreement to the same extent as if they were set forth verbatim in this Stock Purchase Agreement.

12.10 Headings; Interpretation; Governing Language

- (a) The table of contents, headings and sub-headings are for ease of reference only and shall not be taken into consideration in the interpretation or construction of this Stock Purchase Agreement.
 - (b) Unless the content otherwise provides, in this Stock Purchase Agreement:
- (i) words (including the definitions in Appendix 1) importing the singular shall include the plural and vice versa and any reference to any gender shall include the other genders;
- (ii) any reference to the "preamble," the "recitals," an "Article," a "Section," a "subsection," a "Part," a "paragraph," a "clause," a "provision," a "Schedule," an "Appendix" or an "Exhibit" shall be construed as a reference to the preamble, the recitals, an article, a section, a subsection, a part, a paragraph, a clause, a provision of, or a schedule, appendix or exhibit to, this Stock Purchase Agreement or Schedule, Appendix or Exhibit hereto;

- (iii) words denoting gender shall include both genders as the context requires;
- (iv) any reference to this Stock Purchase Agreement or any other agreement or document shall be construed as a reference to that agreement or document as it may have been, or may from time to time be, amended, varied, novated, replaced or supplemented;
- (v) any reference to "day" means any calendar day, any reference to "month" means a calendar month and any reference to "year" means a calendar vear;
- (vi) general words introduced by the word "other" shall not be given a restrictive meaning by reason of the fact that they are preceded by words indicating a particular class of acts, matters or things and any phrase introduced by the words "include," "including" or "in particular" or any similar words or expression shall be construed as illustrative and shall not be given a restrictive meaning by reason of the fact that they are followed by particular examples intended to be embraced by the general words; and
- (vii) any reference to a statute or enactment shall be construed as a reference to such statute or enactment as it may have been, or may from time to time be (with or without modification), amended or re-enacted, and any subordinate legislation made under the statute or enactment and any reference to any statutory reference of the law of the State of New York or legal term for any action, remedy, method of judicial proceeding, legal document, legal status, court, official, or any legal concept or thing shall in respect of any jurisdiction (other than the State of New York) be deemed to include what most nearly approximates in that jurisdiction to the State of New York statutory reference or legal term.
- (c) The English language shall be the definitive and controlling text of this Stock Purchase Agreement and any amendments to this Stock Purchase Agreement, notwithstanding the translation of this Stock Purchase Agreement into any other language.

12.11 Counterparts; Effectiveness

This Stock Purchase Agreement may be signed in any number of counterparts, each of which shall be deemed an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. To facilitate execution, this Stock Purchase Agreement may be executed through the use of facsimile transmission or by email transmission in portable document format ("PDF"), and a counterpart of this Stock Purchase Agreement that contains the facsimile signature of a Party or a signature of a Party in PDF, which counterpart has been transmitted by facsimile or electronic mail transmission to the other Party at its facsimile number or email address set out in Section 12.1, shall constitute an executed counterpart of this Stock Purchase Agreement. In each case, original copies will subsequently be promptly delivered in accordance with Section 12.1 by commercial overnight delivery service. This Stock Purchase Agreement to each other Party in accordance with Section 12.1.

12.12 Severability

(a) Any provision of this Stock Purchase Agreement (or any portion thereof) that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of

such prohibition or unenforceability without invalidating the remaining provisions of this Stock Purchase Agreement (or the remaining portion thereof) or affecting the validity or enforceability of such provision (or the remaining portion thereof) in any other jurisdiction.

(b) To the extent any provision of this Stock Purchase Agreement (or any portion thereof) is determined to be prohibited or unenforceable in any jurisdiction, each Party shall negotiate in good faith to amend this Stock Purchase Agreement to substitute one or more valid, legal and enforceable provisions that, insofar as practicable, implement the purposes and intent of the prohibited or unenforceable provision (or portion thereof).

12.13 Specific Performance; Injunctive Relief

- (a) Except as provided in Section 12.13(b) notwithstanding any other provision to the contrary in this Stock Purchase Agreement, each of the Parties acknowledges and agrees that irreparable damage would occur if any provision of this Stock Purchase Agreement were not performed in accordance with the terms of this Stock Purchase Agreement and that the Parties shall be entitled in (i) any arbitration or (ii) any judicial proceeding for interim, provisional or conservatory measures brought pursuant to Section 12.19, to injunctive relief and specific performance to prevent breaches of this Stock Purchase Agreement and to enforce specifically the provisions of this Stock Purchase Agreement, without proof of irreparable injury, in addition to any other remedy available under this Stock Purchase Agreement.
- (b) In the event (i) this Stock Purchase Agreement is terminated and (ii) Parent receives the Goodwill Deposit pursuant to the Escrow Agreement, receipt of the Goodwill Deposit shall be the sole and exclusive remedy of Parent and Sweden Seller against the members of Buyer Group for any Loss suffered in connection with this Stock Purchase Agreement, except in relation to provisions which survive termination of this Stock Purchase Agreement pursuant to Section 11.2(a). The Parties agree that in the event that this Stock Purchase Agreement is terminated under the circumstances where the Goodwill Deposit is payable to Parent, it will be difficult or impossible to calculate the actual damages. Accordingly, the Parties have agreed that the payment of the Goodwill Deposit is intended to be a reasonable estimate of the damages that would arise if Closing did not occur for the reasons set forth in Section 11.2(b).
- (c) Except as provided in Section 12.13(b) or otherwise provided in this Stock Purchase Agreement, any and all remedies expressly conferred by this Stock Purchase Agreement upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred by this Stock Purchase Agreement, or by law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy.

12.14 Expenses

Except as otherwise provided in Section 12.19 or otherwise in this Stock Purchase Agreement or as expressly stated in any Separation Agreement, all costs and expenses, including legal and accounting fees and disbursements, incurred in connection with this Stock Purchase Agreement and the transactions contemplated by this Stock Purchase Agreement shall be borne by the Party incurring such costs and expenses.

12.15 No Brokers

- (a) Except as set forth on Schedule O, each of the Parties represents and warrants to the other Parties that such Party has not engaged any broker, finder or agent in connection with the transactions contemplated by the Transaction Documents, and has not incurred, and will not incur, any Liability to any broker, finder or agent for any brokerage fees, finders' fees or commissions with respect to the transactions contemplated by the Transaction Documents.
- (b) Each of the Parties shall indemnify, defend and hold harmless the other Parties from and against any and all Claims asserted against such Parties for any such fees or commissions by any Person purporting to act or to have acted for or on behalf of the indemnifying Party.

12.16 No Third Party Beneficiaries

Except as set forth in Section 6.28, Section 10.2 or Section 10.4 in this Stock Purchase Agreement, the covenants or agreements set forth in this Stock Purchase Agreement shall not confer any rights or remedies upon any Person other than the Parties and their respective successors and permitted assigns.

12.17 Representation by Counsel; Mutual Drafting

- (a) Each of the Parties acknowledges that such Party has been represented by counsel with respect to the negotiation and execution of this Stock Purchase Agreement and in connection with the transactions contemplated by this Stock Purchase Agreement.
- (b) Each of the Parties acknowledges that (i) such Party has participated in the drafting of this Stock Purchase Agreement and that this Stock Purchase Agreement is the result of extensive negotiations with the other Party and (ii) this Stock Purchase Agreement shall be interpreted without reference to any rule or precept of Law that states that any ambiguity in a document be construed against the drafter.

12.18 Governing Law

This Stock Purchase Agreement, the rights and obligations of the Parties and all Disputes shall be construed in accordance with and governed by the internal Law of the State of New York, without regard to the choice of law provisions thereof (other than section 5-1401 of the New York General Obligations Law).

12.19 Dispute Resolution

- (a) Subject to the rights of the Parties to seek specific performance and injunctive relief pursuant to Section 12.19(e) and Section 12.13, all Disputes shall be finally settled by binding arbitration as follows:
- (i) the Dispute shall be referred to and finally resolved under the Rules of Arbitration of the International Chamber of Commerce then prevailing (the "ICC Rules") by three arbitrators appointed as the Arbitral Tribunal in accordance with such rules;

- (ii) the seat, or legal place, of the arbitral proceedings shall be **London, England**;
- (iii) the language to be used in the arbitral proceedings shall be English;
- (iv) the Arbitral Tribunal shall be empowered to grant either Party summary judgment based on documentary evidence or, if a hearing shall be required, such hearing shall be held as soon as reasonably practicable after the issuance by the Arbitral Tribunal of its Terms of Reference; and
 - (v) all rulings of the Arbitral Tribunal shall be in writing and shall be delivered to the Parties.
- (b) The costs of the arbitration and any proceeding in court to confirm, enforce or to vacate any Award or to obtain specific performance or injunctive relief as provided in Section 12.13 and Section 12.19(e), including actual attorneys' fees and costs, shall be borne by the unsuccessful Party. The costs of the arbitration shall be fixed as part of the Arbitral Tribunal's Award.
- (c) In order to facilitate the comprehensive resolution of related disputes, and upon request of any party to the arbitration proceeding, the Arbitral Tribunal may consolidate the arbitration proceeding with any other arbitration proceeding relating to this Stock Purchase Agreement, to any other Transaction Documents or to the Escrow Agreement and involving any of the Parties hereto, any other parties to the Transaction Documents or any parties to the Escrow Agreement. If more than one Arbitral Tribunal has been appointed, the Arbitral Tribunal appointed pursuant to this Stock Purchase Agreement shall have the power to determine whether the proceedings should be consolidated, and if that Arbitral Tribunal decides to consolidate proceedings, the proceedings shall be heard by that Arbitral Tribunal alone. If more than one Arbitral Tribunal has been appointed but no Arbitral Tribunal has been appointed pursuant to this Stock Purchase Agreement, then the Arbitral Tribunal appointed in the first-filed of such proceedings shall have the power to determine whether the proceedings should be consolidated, and if that Arbitral Tribunal decides to consolidate proceedings shall be heard by that Arbitral Tribunal alone. The Arbitral Tribunal shall not consolidate such arbitrations unless it determines that (i) there are issues of fact or law common to the two proceedings so that a consolidated proceeding would be more efficient than separate proceedings, and (ii) no Party would be prejudiced as a result of such consolidation through undue delay or otherwise. For the avoidance of doubt, in the event of different rulings on this question by the Arbitral Tribunal constituted under this Stock Purchase Agreement and the Arbitral Tribunal constituted under any other Transaction Document, the ruling of the Arbitral Tribunal appointed in the first-filed of such proceedings shall control. In the case of a consolidated proceeding where no Arbitral Tribunal has been appointed, the Arbitral Tribunal shall be appointed in accordance wit
- (d) Any Award may be entered as a judgment in, and enforced by, any court of competent jurisdiction and shall be final and binding upon the Parties. Each of the Parties knowingly, voluntarily and intentionally waives any and all rights it may have to a trial by jury with respect to any litigation instituted pursuant to Section 12.13 or Section 12.19(e) or to compel arbitration pursuant to this Section 12.19 or to confirm, enforce, vacate, modify or correct an Award. Each of the Parties acknowledges and agrees that any Party may effect valid service of process in any arbitration or judicial proceeding by delivering the Request for Arbitration, or any judicial process or notice, by utilizing the notice provisions set forth in Section 12.1.

- (e) Nothing in this Section 12.19 shall prevent any Party from applying to any competent judicial or arbitral authority for interim, provisional or conservatory measures. The application of a Party to a judicial authority for such measures or for the implementation of any such measures ordered by an Arbitral Tribunal shall not be deemed to be an infringement or a waiver of the arbitration agreement and shall not affect the relevant powers reserved to the Arbitral Tribunal. Each Party irrevocably consents to the jurisdiction of the federal and state courts located in the State and City of New York for purposes of any such action and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue for any such action in any such court or any objection that any such action brought in any such court has been brought in an inconvenient forum.
- (f) Capitalized terms used in this Section 12.19 but not defined in this Stock Purchase Agreement shall have the meaning assigned to them in the ICC Rules.

12.20 Waiver of Sovereign Immunity

To the fullest extent permitted by Law, each of U.S. Buyer and Sweden Buyer unconditionally and irrevocably waives generally all immunity on the grounds of sovereignty or other similar grounds, and acknowledges and agrees that it will not claim any such immunity with respect to, (a) the giving of any relief by way of injunction or order for specific performance or for the recovery of land or property, and (b) the issue of any process against its property for the enforcement of a judgment or, in any action *in rem*, for the arrest, detention or sale of any of its property.

12.21 Waiver of Jury Trial

WITHOUT IN ANY WAY DEROGATING FROM, WAIVING, OR LIMITING THE AGREEMENT TO ARBITRATE CONTAINED IN SECTION 12.19, EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING OUT OF OR RELATED IN ANY WAY WHATSOEVER (WHETHER IN CONTRACT, TORT OR OTHERWISE) TO THIS STOCK PURCHASE AGREEMENT OR THE OTHER TRANSACTIONS CONTEMPLATED BY THIS STOCK PURCHASE AGREEMENT.

[Remainder of this page intentionally left blank: signature page follows.]

IN WITNESS WHEREOF, each of the Parties has caused this Stock Purchase Agreement to be duly executed and delivered by their respective authorized representatives

FORD MOTOR COMPANY

By: /s/ L.W.K. Booth

Name: L.W.K. Booth

Title: Executive Vice President and Chief Financial Officer

VOLVO PERSONVAGNAR HOLDING AB (SWEDEN)

By: /s/ L.W.K. Booth

Name: L.W.K. Booth

Title: Authorized Representative

MINTIME NORTH AMERICA, LLC

By: /s/ Li Shufu

Name: Li Shufu

Title: Authorized Representative

GEELY SWEDEN AB

By: /s/ Li Shufu

Name: Li Shufu
Title: Chairman

APPENDIX 1 – DEFINITIONS

- "2008 Financial Statements" has the meaning set forth in Section 3.8(a).
- "2009 MY Volvo Credits" means any GHG Credits generated by sales of Volvo branded vehicles with respect to Model Year 2009.
- "2010 MY Volvo Credits" means any GHG Credits generated by sales of Volvo branded vehicles with respect to Model Year 2010 prior to Closing.
- "2010 PZEV Credit Obligation" means the actual number of PZEV credits submitted to the U.S. States listed in Schedule L by Parent in relation to the Vehicles for the 2010 model year.
- "2010 ZEV Reports" means filings with the U.S. States listed in Schedule L for the 2010 model year to establish compliance with the 2010 vehicle emissions requirements in each such U.S. State.
- "2011 ZEV Reports" means filings with the U.S. States listed in Schedule L for the 2011 model year to establish compliance with the 2011 vehicle emissions requirements in each such U.S. State.
- "AB Volvo License Agreement" means the AB Volvo Trademarks License Agreement between Volvo Trademark Holding AB and AB Volvo, effective on February 28, 1999.
- "Acquired Rights Regulations" means the Acquired Rights Directive 2001/23/EC together with the Laws adopted thereunder by the Member States of the European Union.
- "Acquisition Financing" has the meaning set forth in Section 4.4(a).
- "Actual Tax Liability" means a liability of a Target Company or Company Subsidiary to make or suffer an actual payment of Tax.
- "Additional Governmental Approvals" has the meaning set forth in Section 5.1(a)(iii).
- "Additional Governmental Approvals Filing Deadline" has the meaning set forth in Section 5.1(d).
- "Affiliate" means a Person that directly or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with, or is a Subsidiary of the Person specified.
- "Agent" has the meaning set forth in Section 5.17.
- "Agreement" means a contract, instrument, lease, license, indenture, bond, warranty, promissory note or other legally binding arrangement.
- "AMLL Agreement" means any Agreement entered into in connection with Parent's sale of the Aston Martin business and to which a Target Company or Company Subsidiary is a party.
- "Applicable Rate" means LIBOR (being the display rate per annum of the offered quotation for deposits in U.S. Dollars for a period of one month which appears on the appropriate page of the Reuters Screen at or about 11:00 a.m. New York time on the Closing Date) plus 2%.

- "Arbitral Tribunal" has the meaning set forth in the ICC Rules.
- "Asset" means an asset that is used in and necessary for the operation of the Business.
- "Assurances" means all guarantees, indemnities, warranties, representations, covenants, undertakings or commitments of any nature whatsoever:
 - (a) given to a third party by a Target Company or Company Subsidiary with respect to a Liability of a member of Parent Group; or
 - (b) given to a third party by a member of Parent Group with respect to a Liability of a Target Company or Company Subsidiary.
- "Award" has the meaning set forth in the ICC Rules.
- "Back-to-Back Financing Commitment" means the Agreement by and between Zhejiang Geely Holding Group Co., Ltd. and Sweden Company, dated as of the date of this Stock Purchase Agreement and attached to this Stock Purchase Agreement as Exhibit XXVII.
- "Basket Amount" has the meaning set forth in Section 10.3(b)(ii).
- "Beijing Holdco" means Beijing Geely Wan Yuan Investment Co., Ltd, a company organized under the Law of China.
- "Bid Price" means \$1,800,000,000.
- "Binding Acquisition Financing Commitment Letter" has the meaning set forth in Section 4.4(a).
- "BL3 Form Leases" means lease documents relating to the BL3 Property in the forms attached to this Stock Purchase Agreement as Exhibit XXII.
- "BL3 Property" has the meaning set forth in Section 6.9(b).
- "BL3 Rights" has the meaning set forth in Section 6.9(b).
- "Business" means the business of designing, engineering, testing, manufacturing, marketing, selling and distributing Vehicles and replacement and service parts and components and assemblies for such Vehicles throughout the world as operated by the Target Companies and Company Subsidiaries as of the date of this Stock Purchase Agreement.
- "Business Day" means a day, other than a Saturday or a Sunday, on which banks generally are open for business in London, New York, Stockholm and Beijing.
- "Business Plan" has the meaning set forth in Section 8.8.
- "Buyer Claim" has the meaning set forth in Section 10.2.
- "Buyer Governmental Approvals" has the meaning set forth in Section 3.3(a).

- "Buyer Group" means U.S. Buyer and Sweden Buyer and, after the Closing Date, each Target Company and each Company Subsidiary.
- "Buyer Indemnitee" has the meaning set forth in Section 10.2.
- "Buyer's Relief" means:
- (a) any Relief arising to any Target Company or Company Subsidiary in respect of a Pre-Closing Period or the portion of any Straddle Period ending on and including Closing which was included as an asset in the Final Closing Statement and was taken into account as an increase in Net Working Capital or as a decrease in Net External Indebtedness;
- (b) any Relief arising to any Target Company or Company Subsidiary in respect of an event occurring or period commencing after Closing, including the portion of a Straddle Period falling after Closing; and
- (c) any Relief arising to any member of Buyer Group (other than any Target Company or Company Subsidiary) or Geely Group (other than any Target Company or Company Subsidiary).
- "Buyers" has the meaning set forth in the preamble to this Stock Purchase Agreement.
- "Bylaws" means, for a Person (other than an individual), the bylaws or limited liability company operating agreement, or equivalent document, as applicable, of that Person.
- "CAFE Reports" means the automotive fuel economy reports required by Law to be submitted to the U.S. Department of Transportation's National Highway Traffic Safety Administration pursuant to Title 49 of the U.S. Code of Federal Regulations at parts 534 and 537.
- "Camshaft Supply Agreement" means the Camshaft Supply Agreement to be entered into between certain members of Parent Group and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XV.
- "Cap" has the meaning set forth in Section 10.3(b)(iii).
- "Captive Automotive Finance Company" means a business engaged in the provision of automotive finance products and services with respect to new automobiles manufactured by such business or manufactured by an Affiliate of such business.
- "CFMA" means Changan Ford Mazda Automobile Co., Ltd, a Sino-foreign joint venture having its principal office at No. 1 Changfu West Road, Yuan Yang Town, New Northern Zone, Chongqing, PRC.
- "Chinese Governmental Entity" means any Governmental Entity of the PRC.
- "Claim" means a demand, claim, action, cause of action, cease-and-desist letter, suit, arbitration proceeding, subpoena or investigation (in each case whether civil, criminal, administrative, regulatory or otherwise), in law or in equity.
- "Clearance Statements" has the meaning set forth in Section 5.20(a).

- "Closing" means the consummation of the sale and purchase of the Sweden Company Shares and the U.S. Company Interest, the execution and delivery of the Intellectual Property Agreements and the delivery of such other documents and Agreements pursuant to Article 9.
- "Closing Date" has the meaning set forth in Section 2.7(a).
- "Collective Consultation Requirements" means all information, consultations or notifications to be made to or with any Employees or any Represented Employees or any Employee representative body in connection with the transactions contemplated by this Stock Purchase Agreement required either (a) by Law, or (b) pursuant to a relevant Agreement entered into prior to the date of this Stock Purchase Agreement with an Employee representative body.
- "Company Balance Sheet" has the meaning set forth in Section 3.8(a).
- "Company Financial Statements" has the meaning set forth in Section 3.8(a).
- "Company Intellectual Property" means (i) the Company Registered Intellectual Property, (ii) any material unregistered Intellectual Property Rights owned by any Target Company or Company Subsidiary, and (iii) Non-Controlled IP.
- "Company Material Adverse Effect" means any event, occurrence, fact, condition or change that, individually or in the aggregate, causes a material adverse effect on the financial condition, results of operations, or assets (tangible or intangible) of the Business, taken as a whole or the ability of Parent or Sweden Seller to consummate the transactions contemplated hereby; provided, however, that a "Company Material Adverse Effect" shall be deemed not to include any adverse effect that is caused by, or results from, any of the following: (a) changes in conditions, including economic, financial, market or political conditions, of the industry in which each Target Company and each Company Subsidiary operates, but only if such change does not materially disproportionately adversely affect the Target Company or Company Subsidiary compared to similar participants in the business of manufacturing and selling automobiles; (b) changes in interest rates, exchange rates or securities or commodity prices; (c) changes in Law or U.S. GAAP or Swedish GAAP or any interpretation thereof, but only to the extent such change does not materially disproportionately adversely affect the Target Company or Company Subsidiary compared to similar participants in the business of manufacturing or selling automobiles; (d) acts of terrorism or war, whether or not declared, or weather or other natural disasters; (e) normal seasonal changes in the results of operations of the Business; (f) any investigation, inquiry or similar action by a Governmental Entity arising or resulting from the transactions contemplated by this Stock Purchase Agreement; (g) a loss of customers, vendors or partners or others having a relationship with the Business as a consequence of the announcement of the execution of this Stock Purchase Agreement; or (h) actions or others having a relationship with the Business as a consequence of the announcement of the execution of this Stock Purchase Agreement; or (h) actions or others having a relationship with the Business as
- "Company Plan" means a Plan sponsored or maintained by a Target Company or Company Subsidiary.
- "Company Records" means books, accounts, reports, Tax Returns and records relating to the Business maintained during the period prior to Closing, other than data relating to an Excluded Contract.

- "Company Registered Intellectual Property" means all registered Intellectual Property Rights owned by any Target Company or Company Subsidiary.
- "Company Subsidiaries" means all of the direct and indirect Subsidiaries of the Target Companies.
- "Compulsory License" means a license in Intellectual Property Rights required to be granted under Law and to which permission is not required, provided that a fee is paid by the licensee.
- "Confidential Information" has the meaning set forth in Section 6.34(a).
- "Confidentiality Agreement" means the Confidentiality Agreement between Zhejiang Geely Holding Group Co., Ltd. and Parent, dated January 24, 2009 and attached to this Stock Purchase Agreement as Exhibit IV.
- "Conrods and Brake Discs Supply Agreement (Floby)" means the Conrods and Brake Disc Supply Agreement (Floby) to be entered into between certain members of Parent Group and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XIII.
- "Consent Contract" means any Shared Contract or Excluded Contract entered into by a member of Parent Group that is a Material IP Settlement Agreement or a Material Third Party License In.
- "Consenting Party" has the meaning set forth in Section 6.24(a).
- "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through ownership of voting securities, by Agreement, as trustee or otherwise. The terms "Controlled by" and "under common Control with" shall be construed accordingly.
- "Credit Company" means Ford Motor Credit Company LLC, a Delaware limited liability company whose executive office is at One American Road, Dearborn, Michigan 48121-1899, USA.
- "Credit Insurance" means those insurance products as may be provided to Credit Insurance Customers by Third Party Suppliers, including motor vehicle insurance and payment protection insurance.
- "Credit Insurance Customers" means those persons who enter into an Agreement with a Third Party Supplier for the provision of Credit Insurance.
- "Customs Duties" means customs or excise duty, import duty, anti-dumping duty, and any other turnover or import Taxes, other than VAT.
- "Cycle Plan" has the meaning set forth in Section 8.8.
- "DB Pension Plan" means a defined benefit pension plan established pursuant to the Defined Benefits Pension Law of Japan (Law No. 50 of 2001, as amended).
- "DB Pension Split Date" means the date as soon as reasonably practicable following the day VCJ and FJL have received in connection with the separation of the Joint New DB Plan (a) any approvals required by Law to be obtained from a Governmental Entity, and (b) any consent required by Law or a Local Collective Agreement to be obtained from employees of FJL and/or VCJ or a collective bargaining agent of any such employees.

"Dealers" means, collectively, dealers appointed or approved by a Target Company or Company Subsidiary as approved distributors of the Products;

"Deemed Tax Liability" means the deemed amount of Tax resulting from:

- (a) the loss of any Relief falling within paragraph (a) of the definition of "Buyer's Relief" or
- (b) the use or set-off of any Buyer's Relief in circumstances where, but for such use or set-off, a Target Company or Company Subsidiary would have had an Actual Tax Liability in respect of which U.S. Buyer or Sweden Buyer would have been entitled to payment from Parent or Sweden Seller pursuant to Section 6.2(a)(i) or Section 6.2(a)(ii),

where the deemed amount of Tax is determined as follows:

- (x) where the Relief that is the subject of the loss or which is used or set-off as mentioned in those paragraphs is a deduction from or offset against Tax, the Tax Liability shall be the amount of that Relief so lost, used or set-off;
- (y) where the Relief that is the subject of the loss or which is used or set-off as mentioned in those paragraphs is a deduction from or offset against income, profits or gains, the Tax Liability shall be, in the case of a Relief which is used or set-off, the amount of Tax saved thereby and, in the case of a Relief which is lost, the amount of Tax which but for such loss would have been saved by virtue of the Relief so lost, ignoring for this purpose the effect of Reliefs (other than deductions in computing profits for the purpose of Tax) arising in respect of an event occurring or period commencing after Closing (including the portion of a Straddle Period falling after Closing); and
- (z) where the Relief that is the subject of the loss or setting off is a repayment of Tax, the Tax Liability shall be the amount of the repayment that would have been obtained but for the loss, use or setting off.
- "Default Interest" means LIBOR (being the display rate per annum of the offered quotation for deposits in U.S. Dollars for a period of one month which appears on the appropriate page of the Reuters Screen at or about 11:00 a.m. New York time on the first Business Day on which payment of the sum under this Stock Purchase Agreement was due but not paid) plus 3% per anum, calculated on a daily basis.
- "Design, Development and Research Cooperation Agreement" means the Design, Development and Research Cooperation Agreement to be entered into between the Ford Entities and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit X.

"Designated Employee" has the meaning set forth in Exhibit XVII.

"Designated Engineer" means an employee of Parent or a member of Parent Group that is an engineer or other similar technical professional and who is employed in the PRC, in each case whose salary grade is at or above General Salary Range 7 (GSR7) as designated in Parent's employment policies (or the equivalent grade in any successor policies as notified to Buyers from time to time).

"Discontinued Business Liability" has the meaning set forth in Section 3.8(h).

- "Dispute" means a dispute, controversy or claim arising out of, relating to or in connection with this Stock Purchase Agreement (whether in contract, tort or otherwise) including any question regarding its interpretation, existence, validity or termination.
- "Dispute Resolution Agreement" means the Intellectual Property Dispute Resolution Agreement to be entered into by and among each of Parent and certain of its Affiliates, Zhejiang Geely Holding Group Co., Ltd and certain of its Affiliates and Sweden Company in the form attached to this Stock Purchase Agreement as Exhibit VII.
- "Divestitures" has the meaning set forth in Section 5.1(g).
- "Documented Plans" means (a) Volvo's 2009B Vehicle Cycle Plan included in schedule C of the Intellectual Property License Agreement; (b) Volvo's 2009B Technology Cycle Plan included in schedule C of the Intellectual Property License Agreement; and (c) the Information Technology Agreement and the DDR Agreement.
- "Due Diligence Materials" means any information relating to the Business set forth in (a) management presentations, (b) vendor assist data books, (c) materials contained in the Legal Data Room, including any replacement cost estimates for services that are currently provided or made available by Parent or its Affiliates to the Business, (d) "break-out" or "expert" sessions, (e) responses to questions submitted by or on behalf of U.S. Buyer, Sweden Buyer, their respective Affiliates or their respective representatives, whether orally or in writing, or (f) other materials prepared by or on behalf of Parent or Sweden Seller in connection with the transactions contemplated by this Stock Purchase Agreement and made available to U.S. Buyer, Sweden Buyer, their respective Affiliates or their respective representatives.
- "DW 10 b/c Supply Agreement" means the DW 10 b/c Supply Agreement to be entered into between certain members of Parent Group and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XIV.
- "Embedded NSC" means the business conducted by Parent exclusively relating to the Business in the jurisdictions set forth on Schedule F.
- "Employee" means each current employee of a Target Company or Company Subsidiary, except where a former Employee is specified.
- "Employment Liability" means any Loss relating to, or arising out of, any Claim in respect of an Undisclosed Individual or its representative made in connection with any act or omission, or alleged act or omission, by a member of Parent Group or Buyer Group, as a result of application of the Acquired Rights Regulations.
- "Encumbrance" means a mortgage, deed of trust, deed to secure debt, pledge, lien, claim, security or other interest, conditional and installment sale agreement or other title retention agreement, lease, sublease, option to purchase or lease real property, easement, covenant, right of way, claim, restriction, charge or other encumbrance of any kind.
- "Enforcement Currency" has the meaning set forth in Section 12.7.
- "Engines and Stampings Supply Agreement (Parent to Sweden Company)" means the Engines and Stampings Supply Agreement (Parent to Sweden Company) to be entered into between certain members

of Parent Group and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XI.

- "Engines and Stampings Supply Agreement (Sweden Company to Parent)" means the Engines and Stampings Supply Agreement (Sweden Company to Parent) to be entered into between certain members of Parent Group and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XII
- "Entrustment Loan Agreement" has the meaning set forth in Exhibit XXVII.
- "Environmental Consultants" means Environmental Resources Management (ERM), Golder Associates, Inc., and their local subcontractors retained by Parent to carry out Phase I environmental site assessments in connection with the transactions contemplated by this Stock Purchase Agreement.
- "Environmental Emergency" means any sudden, unforeseen event after the Closing Date that U.S. Buyer or Sweden Buyer reasonably determines requires immediate remediation to prevent the contamination of, or exacerbation of contamination of, soil or groundwater by Hazardous Materials at, on, under or migrating from a facility of a Target Company or Company Subsidiary, in view of the likelihood of risk of significant harm to the environment or to human health.
- "Environmental Law" means Law in effect on the date of this Stock Purchase Agreement relating to the pollution or the protection of the environment or the generation, production, installation, use, storage, treatment, transportation, release or disposal of Hazardous Materials.
- "Environmental Proceedings" means any criminal, civil, judicial, administrative or regulatory proceedings, suit or action against any member of Buyer Group or Geely Group by any Governmental Entity or third party under Environmental Law.
- "EPA GHG Emissions Regulations" means the final regulations promulgated by the U.S. Environmental Protection Agency for the reduction of greenhouse gas emissions from passenger cars, light-duty trucks, and medium-duty passenger vehicles for model years 2012-2016, flowing from the U.S. EPA's proposal to codify regulations at 40 CFR parts 86 and 600 in its Notice of Proposed Rulemaking published on September 28, 2009, and published at 74 Fed. Reg. 49454, as may at the relevant time have been further described, supplemented or varied in any subsequent final regulations on light-duty vehicle greenhouse gas emissions.
- "Equity Interest" means any capital stock, membership interest, joint venture interest or other equity interest of a Target Company or Company Subsidiary or any other securities convertible into or exchangeable or exercisable for such interests, or options, warrants or other rights to subscribe for or purchase such interests (including any phantom interest) of a Target Company or Company Subsidiary.
- "Escrow Agent" has the meaning set forth in Section 2.8(a).
- "Escrow Agreement" means the Escrow Agreement between Parent, Beijing Holdco and Escrow Agent executed and delivered as of the date of this Stock Purchase Agreement.
- "Estimated Intra-Group Balances" (which can be a positive or negative number) means the Estimated Intra-Group Receivables minus the Estimated Intra-Group Payables.

- "Estimated Intra-Group Payables" means Parent's good faith estimate of the Intra-Group Payables, estimated by Parent and notified to Buyers pursuant to Section 2.2 and, to the extent any Estimated Intra-Group Payable is expressed in a currency other than U.S. Dollars, the corresponding amount shall be calculated by converting such amount into U.S. Dollars at the same exchange rate used to calculate balances in that currency included in Estimated Net Working Capital and Estimated Net External Indebtedness, or in the absence of balances in such currency, at the relevant exchange rate as at the same date foreign currency balances more generally are translated in calculating Estimated Net Working Capital and Estimated Net External Indebtedness.
- "Estimated Intra-Group Receivables" means Parent's good faith estimate of the Intra-Group Receivables, estimated by Parent and notified to Buyers pursuant to Section 2.2 and, to the extent any Estimated Intra-Group Receivable is expressed in a currency other than U.S. Dollars, the corresponding amount shall be calculated by converting such amount into U.S. Dollars at the same exchange rate used to calculate balances in that currency included in Estimated Net Working Capital and Estimated Net External Indebtedness, or in the absence of balances in such currency, at the relevant exchange rate as at the same date foreign currency balances more generally are translated in calculating Estimated Net Working Capital and Estimated Net External Indebtedness.
- "Estimated Net External Indebtedness" means Parent's good faith estimate of the Net External Indebtedness, estimated by Parent and notified to Buyers pursuant to Section 2.2.
- "Estimated Net Working Capital" means Parent's good faith estimate of Net Working Capital estimated by Parent and notified to Buyers pursuant to Section 2.2.
- "Estimated Net Working Capital Adjustment" means the following amount (as applicable): (a) the amount by which the Estimated Net Working Capital is greater than the Specified Net Working Capital (in which event, the Initial Purchase Price shall be increased by such amount) or (b) the amount by which the Estimated Net Working Capital is less than the Specified Net Working Capital (in which event, the Initial Purchase Price shall be decreased by such amount).
- "Estimated Pension Adjustment" means the amount determined as set forth in annex 2 to Schedule A-2.
- "EU Data Protection Directive" means the Data Protection Directive 95/46/EC together with the Laws adopted thereunder by the Member States of the European Union.
- "European Commission Filing Deadline" has the meaning set forth in Section 5.1(c).
- "Excess GHG Allowances" means any GHG Allowances that are not required to be surrendered to Governmental Entities for compliance purposes in respect of the Ghent Property for each credit year set forth in Schedule V, up to a yearly maximum of the amounts set forth in Schedule V.
- "Excluded Contracts" means the Agreements set forth on Schedule X, as may be updated by Parent prior to Closing to be consistent with the Transitional Services Agreement.
- "Excluded Pensions Costs" means the actual Liability of a Target Company or Company Subsidiary to contribute to or in respect of any retirement benefit arrangement (other than any New Plan or any Maintained Company Plan (including the Volvo Car UK Pension Scheme)) in relation to which a Target Company or a Company Subsidiary has, prior to the Closing Date, been an associate or connected person of an employer (for purposes of the UK Pensions Act) under (a) a contribution notice issued to any Target Company or Company Subsidiary within six years after the Closing Date under section 38 of the UK

Pensions Act in respect of any acts, deliberate failures to act occurring, or knowing assistance given by any of Target Company or Company Subsidiary, prior to the Closing Date, (b) a financial support direction issued to any of Target Company or Company Subsidiary within two years after the Closing Date under section 43 of the UK Pensions Act, or (c) a contribution notice issued to any Target Company or Company Subsidiary within six years after the Closing Date under section 47 of the UK Pensions Act in respect of non-compliance with a financial support direction referred to in the preceding subsection (b).

"Excluded Taxes" has the meaning set forth in Section 6.2(a)(vi).

"Exhibit" means an exhibit attached to this Stock Purchase Agreement, as the same may be amended and supplemented from time to time after the date of this Stock Purchase Agreement and prior to Closing in accordance with the terms of this Stock Purchase Agreement.

"Extended Deadline" has the meaning set forth in Section 5.1(a)(ii).

"Filing Deadline" has the meaning set forth in Section 5.1(a).

"Final Closing Statement" has the meaning set forth in Schedule A-1.

"Financing Plans" means, collectively, the various wholesale and retail (both retail installment credit financing and leasing) financing plans provided to Dealers and Target Company Customers by Credit Company, including, for the avoidance of doubt, plans structured such that Credit Company does not make such financing directly available to the Dealers and/or Target Company Customers but instead acts in an undisclosed capacity and/or as Target Company's receivables collection agent and those plans where Target Company pays Credit Company's financing costs.

"FJL" has the meaning set forth in Section 6.23(a)(ii).

"FJL Actuary" has the meaning set forth in Section 6.23(d).

"FJL Amount" has the meaning set forth in Section 6.23(d).

"Floor Plan Finance" means, collectively, the following finance arrangements: (a) wholesale vehicle stocking financing facilities (howsoever structured, including sale or return, cash enhancement, bailment, display and consignment structures and those where Credit Company retains title to relevant Vehicles or related Vehicle registration or other documentation until Dealer payment) for new Vehicles, including Vehicles stored from time to time in holding compounds operated by a Target Company or Company Subsidiaries or Third Parties on behalf of Dealers or Target Companies or Company Subsidiaries or the demonstrator Vehicles which Credit Company finances through a new wholesale credit facility; (b) wholesale vehicle financing facilities (howsoever structured, including overdrafts and individually financed vehicles) for used vehicles, including any make or model of vehicle; (c) any Financing Plan provided to Dealers by Credit Company which involves the selling or leasing of any make or model of vehicle by Credit Company to Dealers for use in their business, which include demonstrator vehicle, courtesy vehicle and company car Financing Plans; (d) parts and accessories financing facilities provided by Credit Company to Dealers; (e) any Financing Plan whereby Credit Company provides financing to Dealers to facilitate their purchase or lease of equipment or dealer signage; and (f) receivables created under any Financing Plan provided by Credit Company which has the net effect of extending deferred payment terms and/or leasing terms to Dealers for their purchase or lease of those vehicles defined in subsections (a) and (b) of this definition and in addition vehicle parts and

accessories, equipment and dealer signage, which is not structured as a direct finance facility to Dealers, including but not limited to undisclosed receivables factoring arrangements, arrangements where Credit Company acts as collection agent of the Business and arrangements where a Target Company or Company Subsidiary and not the Dealers pays Credit Company's financing costs.

"Ford Credit Belgium" means FCE Bank plc, Hunderenveldlaan 10, B-1082 Brussels, Belgium, a branch of FCE Bank plc, a company incorporated under the laws of England and Wales.

"Ford Credits" means any GHG Credits generated by sales of vehicles by Parent Group (other than Volvo branded vehicles) with respect to Model Year 2009 and/or Model Year 2010 prior to Closing.

"Ford Entities" means Parent, Ford Global, Ford Germany and Ford UK.

"Ford Germany" means Ford-Werke GmbH, a company incorporated under the Law of Germany.

"Ford Global" means Ford Global Technologies, LLC, a Delaware limited liability company.

"Ford Limited License IT" has the meaning set forth in the Intellectual Property License Agreement.

"Ford Netherlands Pension Fund Transfer Agreement" means the Agreement between Parent and the board of the Ford Netherlands Pension Fund in substantially the form attached to this Stock Purchase Agreement as Exhibit XXIII.

"Ford Netherlands Pension Fund Transferred Assets" has the meaning set forth in Section 6.22.

"Ford Retiree Medical Program" has the meaning set forth in Section 5.19(b).

"Ford Secondment Agreement means the Secondment Agreement to be entered into between Parent and Sweden Company in relation to Ford employees, in the form attached to this Stock Purchase Agreement as Exhibit XVII.

"Ford UK" means Ford Motor Company Limited, a company incorporated under the Law of England and Wales.

"Foreign Real Property Leases" means all leases, subleases and occupancy agreements, together with any amendment thereto, with respect to real property located outside of Sweden, in which a Target Company or Company Subsidiary is a lessee or has occupancy rights, except for those that are not, individually or in the aggregate, material to the operation of the Business.

"Formation Document" means, for a Person (other than an individual), the certificate of incorporation, certificate of registration, articles of incorporation, certificate of formation, certificate of partnership, partnership agreement or equivalent document establishing such Person's legal existence.

"FRP" has the meaning set forth in Section 5.11(c).

"FSL" means a full service leasing product (which includes, inter alia, the provision to the lessee of vehicle maintenance services) which involves a leasing contract being entered into between an FSL Customer and a Third Party Supplier in respect of a Target Company Vehicle.

- "FSL Customers" means those persons who lease a Target Company Vehicle from a Third Party Supplier by way of an FSL.
- "Geely Group" means Zhejiang Geely Holding (Group) Co., Ltd. and Geely Automobile Holdings Co. Limited and their respective direct and indirect Subsidiaries.
- "Gemini Engine IP" has the meaning set forth in the Intellectual Property License Agreement.
- "German IP Pledge Agreement" means the German IP Pledge Agreement to be entered into between a members of Parent Group, on the one hand, and Sweden Company, on the other hand, in the form attached to this Stock Purchase Agreement as Exhibit XXVI.
- "Ghent Property" means the manufacturing facilities located at (a) John Kennedylaan 25, 9000 Ghent, Belgium, and (b) Skaldenstraat, 9000 Ghent, Belgium.
- "GHG Allowances" means the greenhouse gas allowances allocated by Governmental Entities to the Ghent Property pursuant to European Union Directive 2003/87/EC together with the Laws adopted thereunder by the Member States of the European Union.
- "GHG Credits" means units used to determine compliance with the green house gas emissions standards established by U.S. Environmental Protection Agency's light-duty vehicle greenhouse gas emissions standards program discussed in the EPA GHG Emissions Regulations.
- "Goodwill Deposit" has the meaning set forth in Section 2.8(a).
- "Governmental Approvals" has the meaning set forth in Section 3.3(a).
- "Governmental Entity" means any local, state, regional, national, supra-national or foreign administrative agency, authority, arbitral tribunal, court, commission, department, registry, division or other governmental, quasi-governmental or government accredited institution, tribunal or regulatory body.
- "Group Loss Company" is a company with losses, which is (or, at the relevant time, was) a member of a group of companies for relevant Tax purposes.
- "Group Profit Company" is a company with profits, which is (or, at the relevant time, was) a member of a group of companies for relevant Tax purposes.
- "Group Relief" means any transfer, sharing or consolidation of profits or losses within a group of companies or other relevant tax relationship for the purposes of any Taxes including any profit and loss transfer agreement, group contribution agreement and any Tax consolidation pursuant to any provision of any taxation statute in any jurisdiction.
- "Hazardous Materials" means any waste, substance, radiation, or material (whether solid, liquid or gas) which are defined as a "pollutant," "contaminant," "hazardous material," "hazardous waste," "hazardous substance," "toxic substance," "radioactive material," "solid waste," or other similar designation in, or otherwise subject to regulation under, Law, or which, alone or in combination with other substances, has been determined by a Governmental Entity to cause harm to the environment. The term Hazardous Materials shall only include asbestos, lead based paint or other Hazardous Materials that are part of,

affixed to or contained in, and result primarily in exposure within buildings, structures or equipment to the extent that such materials are not managed in accordance with Law.

- "ICC Rules" has the meaning set forth in Section 12.19(a)(i).
- "Improvement" means a development, improvement, enhancement, modification, or derivative work.
- "Inconsistency" has the meaning set forth in Section 6.4(c)(iii)(D).
- "Indemnified Environmental Losses" means any Losses, whether or not resulting from a Third Party Claim, arising out of or relating to:
- (a) any Hazardous Materials present at, on or under any Site at the Closing Date, and any such Hazardous Materials migrating from any Site at any time; or
- (b) any breach of, or failure by Parent or Sweden Seller to comply with Environmental Law or a Material Environmental Permit or failure to cause a Target Company or Company Subsidiary to comply with Environmental Law or a Material Environmental Permit prior to the Closing Date,
- in each case, incurred by any Buyer Indemnitee under Environmental Law as a result of:
- (x) Works required to prevent the commencement of Environmental Proceedings (provided that such Works would be carried out by a reasonable and prudent operator in the Buyer Indemnitee's position, acting without regard to recourse under Section 10.2);
- (y) receipt by Buyers of a written request for remediation pursuant to Environmental Law from a Governmental Entity authorized to enforce such Environmental Law; or
 - (z) a resolution of Environmental Proceedings.
- "Indemnified JLR/AML Costs" means any and all Liabilities incurred by a Target Company or Company Subsidiary relating to or arising out of the business of designing, engineering, testing, manufacturing, marketing, selling and distributing Jaguar, Land Rover or Aston Martin vehicles and replacement parts and service parts and components and assemblies for such vehicles, except to the extent arising under (a) any Agreement to which a Target Company or Company Subsidiary (or their respective successors or assigns) is a party or (b) the Separation Agreements.
- "Indemnified Party" has the meaning set forth in Section 10.7(a).
- "Indemnified Product Recall Costs" means the actual direct costs and expenses resulting from or arising out of a Product Recall for Vehicles sold prior to the Closing Date, after deducting from such costs and expenses any reimbursement or other amounts paid, credited or otherwise provided in connection with any such Product Recall by suppliers, sub-suppliers or other third parties.
- "Indemnifying Party" has the meaning set forth in Section 10.7(a).
- "Indemnity Payment" has the meaning set forth in Section 12.6(d).
- "Independent Tax Accountant" has the meaning set forth in Section 6.4(c)(iii)(A).

- "Industrial Action" means any strike, work stoppage, slow-down, rule-book slow-down, work-to-rule, picketing or lock-out by or affecting Employees.
- "Information Barriers Protocol" means the Information Barriers Protocol set forth in Schedule I.
- "Initial Purchase Price" means an amount equal to (a) the Bid Price, minus (b) the amount of the Estimated Net External Indebtedness, plus (c) the amount of the Estimated Intra-Group Balances minus (d) the amount of the Estimated Pension Adjustment and plus or minus, as applicable, the amount of the Estimated Net Working Capital Adjustment.
- "Insurance Policy" means a material policy of insurance in relation to a Target Company or Company Subsidiary and with respect to their current operations, assets, properties, directors, officers and Employees, whether taken out by a Target Company or Company Subsidiary or by another Person, currently in effect, or with respect to which a claim may currently be validly made.
- "Intellectual Property Agreements" means, collectively, the Intellectual Property License Agreement, the Intellectual Property Assignment Agreement, and the Dispute Resolution Agreement.
- "Intellectual Property Assignment Agreement" means the Intellectual Property Assignment Agreement to be entered into among the Ford Entities, on the one hand, and Sweden Company, on the other hand, in the form attached to this Stock Purchase Agreement as Exhibit IX.
- "Intellectual Property License Agreement" means the Intellectual Property License Agreement to be entered into among the Ford Entities, on the one hand, and Sweden Company, on the other hand, in the form attached to this Stock Purchase Agreement as Exhibit VIII.
- "Intellectual Property Rights" means patents, rights in inventions, utility models, works of authorship, data, designs, databases, software, topography and mask works, know-how, trade secrets, confidential information, trademarks, service marks, trade, business and domain names, logos, trade dress, assumed names, get-up and goodwill, along with copyrights, moral rights and any other intellectual property or proprietary rights, in each case whether registered or unregistered and including all applications and registrations for and renewals, continuations, continuations in part, divisionals or extensions of such rights, and all similar or equivalent rights or forms of protection in any jurisdiction.
- "Intercompany Agreements" has the meaning set forth in Section 6.10(a).
- "Interim Financial Statements" means: (a) the consolidated balance sheet of Target Companies and Company Subsidiaries as of October 31, 2009, (b) the consolidated income statements of Target Companies and Company Subsidiaries for the ten-month period ended October 31, 2009, and (c) the consolidated operating cash flow statements of Target Companies and Company Subsidiaries for the ten-month period ended October 31, 2009, all as set forth in Schedule DD.
- "Intra-Group Balances" (which may be a positive or negative number) means the Intra-Group Receivables minus the Intra-Group Payables, as set forth in the Final Closing Statement in accordance with Schedule A-1.
- "Intra-Group Balances Price Adjustment" means the amount, if any, payable by U.S. Buyer or Sweden Buyer to Parent or Sweden Seller or, as applicable, by Parent or Sweden Seller to U.S. Buyer or Sweden Buyer pursuant to Section 2.5(b)(i).

"Intra-Group Payables" means the amounts that are, as of the Measuring Time, owed by a Target Company or Company Subsidiary to any member of Parent Group including any interest accrued thereon but excluding Intra-Group Trade Payables, as set forth in the Final Closing Statement in accordance with Schedule A-1 and, to the extent any Intra-Group Payable is expressed in a currency other than U.S. Dollars, the corresponding amount shall be calculated by converting such amount into U.S. Dollars using the exchange rate required by Schedule A-1 to be used in translating the relevant currency into U.S. Dollars for the purposes of preparing the Final Closing Statement.

"Intra-Group Receivables" means the amounts that are, as of the Measuring Time, owed by any member of Parent Group to a Target Company or Company Subsidiary, including any interest accrued thereon but excluding Intra-Group Trade Receivables, as set forth in the Final Closing Statement in accordance with Schedule A-1 and, to the extent any Intra-Group Receivable is expressed in a currency other than U.S. Dollars, the corresponding amount shall be calculated by converting such amount into U.S. Dollars using the exchange rate required by Schedule A-1 to be used in translating the relevant currency into U.S. Dollars for the purposes of preparing the Final Closing Statement.

"Intra-Group Trade Payables" means the amounts that are, as of the Measuring Time, owed by a Target Company or Company Subsidiary to any member of Parent Group, relating to goods and services provided in the Ordinary Course of Business (which, for the avoidance of doubt, shall not include financing or licensing related payables).

"Intra-Group Trade Receivables" means amounts that are, as of the Measuring Time, owed by any member of Parent Group to a Target Company or Company Subsidiary relating to goods and services provided in the Ordinary Course of Business (which, for the avoidance of doubt, shall not include financing or licensing related payables).

"IP Contracts" means:

- (a) any Material Third Party License In or Material Third Party License Out;
- (b) the Volvo Trademark Holding License Agreement, the Volvo Trademark Holding Shareholders' Agreement or the AB Volvo License Agreement;
 - (c) any Material IP Settlement Agreement; and
 - (d) any material Third Party IT Agreement to which a Target Company or Company Subsidiary is a party.
- "IP Settlement Agreements" has the meaning set forth in Section 3.14(q).
- "IT Agreement" means the Information Technology Transitional Services Agreement to be entered into between Parent and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit VI.
- "IT Applications" means operating systems, middleware, applications and data that are connected to, or running on, IT Infrastructure and any other information technology hardware and software that is not included in IT Infrastructure, and shall include software interfaces and source code.
- "IT Infrastructure" means information technology infrastructure including any platform hardware, mainframes, servers, storage devices and networks and network equipments such as hubs, routers and

repeaters for LAN, WAN and wireless connections but excluding networks generally available to the public.

- "IT Permitted Encumbrance" means (a) Agreements entered into in the Ordinary Course of Business granting non-exclusive licenses to third parties in the Parent IT System or Target Companies IT System; (b) leases, title retention agreements, and other financing arrangements; (c) liens (including mechanics', carriers', workers', repairers' and other similar liens, but excluding Tax liens) arising or incurred in the Ordinary Course of Business relating to obligations as to which there is no default on the part of a Target Company or Company Subsidiary, as applicable, or the validity or amount of which is being contested in good faith by appropriate proceedings, and only to the extent that such liens do not or would not reasonably be expected to, individually or in the aggregate, materially impair the continued use and operation of the asset to which such liens relate in the Ordinary Course of Business; and (d) security interests arising by operation of Law.
- "IT Steering Committee" has the meaning given to the term "Steering Committee" in the IT Agreement.
- "IT Systems" means, together, the Target Companies IT System, the Parent IT System and all other IT Applications and IT Infrastructure used by a Target Company or a Company Subsidiary as at Closing.
- "IT Transition Plan" has the meaning given to the term "Transition Plan" in the IT Agreement.
- "JLR Agreement" means any Agreement entered into in connection with Parent's sale of the Jaguar and Land Rover businesses and to which a Target Company or Company Subsidiary is a party.
- "Joint New DB Plan" means a defined benefit pension plan established pursuant to the Defined Benefits Pension Law of Japan (Law No. 50 of 2001, as amended) jointly established and managed by VCJ and FJL.
- "Knowledge of Buyer" means of a fact, event, circumstance or occurrence known by the individuals listed on Schedule G after making reasonable inquiry.
- "Knowledge of Parent" means of a fact, event, circumstance or occurrence known by the individuals listed on Schedule D after making reasonable inquiry.
- "Law" means any applicable law, common law, statute, directive, ordinance, rules of any stock exchange or Governmental Entity, regulation or Order.
- "Leadership Level 3 Employee" means an employee of Parent or a member of Parent Group whose salary grade is at Leadership Level 3 as designated in Parent's employment policies (or the equivalent grade in any successor policies as notified to Buyers from time to time).
- "Leadership Level 5 Employee" means an employee of Parent or a member of Parent Group whose salary grade is at Leadership Level 5 as designated in Parent's employment policies (or the equivalent grade in any successor policies as notified to Buyers from time to time).
- "Lease Finance" means, collectively, any and all lease and/or trade cycle management contracts relating to Vehicles entered into between Credit Company and a Target Company Customer and which are originated or purchased by Credit Company by way of Volvo Finance.
- "Leased Foreign Real Property" has the meaning set forth in Section 3.13(b).

- "Leased Local Real Property" has the meaning set forth in Section 3.13(b).
- "Leased Real Property" means the Leased Local Real Property together with the Leased Foreign Real Property.
- "Legal Data Room" means the IntraLinks workspace site entitled "Project Delta Data Room" as made available to Buyer Group, Geely Group and their respective representatives.
- "Liabilities" means all liabilities, debts, duties and obligations of every description, whether deriving from Agreement, Claim, Order or Law, actual or contingent, determined or determinable, asserted or unasserted, ascertained or unascertained or unmatured and whether owed or incurred severally or jointly, or as principal or surety.
- "LLC Interest Assignment and Assumption Agreement" means the Limited Liability Company Interest Assignment and Assumption Agreement substantially in the form attached to this Stock Purchase Agreement as Exhibit II.
- "Local Collective Agreements" has the meaning set forth in Section 3.17(b).
- "Local GAAP" means the generally accepted accounting principles adopted to prepare the relevant accounts of a Person.
- "Local Real Property Leases" means all leases, subleases and occupancy agreements, together with any amendments thereto, with respect to real property located in Sweden, in which a Target Company or Company Subsidiary is a lessee or has occupancy rights, except for those that are not, individually or in the aggregate, material to the operation of the Business.
- "Loss" means any cost, damage, disbursement, expense, Liability, loss, deficiency, penalty or settlement, including reasonable legal, accounting and other professional fees and expenses and amounts paid in settlement, that are actually incurred or suffered by the specified Person; but specifically excluding (a) any consequential, exemplary or punitive damage or other special damage, provided that such exclusions shall not apply to claims for indemnification with respect to Third Party Claims and there shall be no limit or restriction in any manner of any rights or remedies which any member of Buyer Group or Geely Group has based on fraud, or (b) any reduction of, or utilization of, any net operating loss carryforward, tax credit carryforward or similar Tax attribute or any Tax liability attributable to such reduction or utilization, except as otherwise provided in Section 6.2(a)(v) with respect to certain Deemed Tax Liabilities and except where such reduction or utilization is in breach of the warranty set out in Section 3.12(v) or the covenants set out in Section 6.6.
- "Maintained Company Plan" has the meaning set forth in Section 6.18(a)(i).
- "Market" means Belgium, Brazil, Germany, Ireland, Italy, Mexico, Spain, and Switzerland.
- "Material Contracts" has the meaning set forth in Section 3.11(a).
- "Material Environmental Permits" means all Permits required under Environmental Law for a Target Company or Company Subsidiary to operate its respective business, except for such Permits the absence of which would not, individually or in the aggregate, reasonably be expected to materially impair the operation of the Business or result in a significant fine.

- "Material IP Settlement Agreement" means any IP Settlement Agreement that is (a) material to the Business, or (b) if breached, terminated or not renewed would reasonably be expected to materially impair the operation of the Business.
- "Material Jurisdictions" has the meaning set forth in Section 6.17.
- "Material Non-Environmental Permits" has the meaning set forth in Section 3.27(c).
- "Material Real Property Improvements" means the improvements, alterations and additions that are not deemed personal property by the Law of the applicable jurisdiction and that are (a) located on, and forming part of, the Owned Real Property, or (b) located on the Leased Real Property, to the extent a Target Company or Company Subsidiary is responsible for such improvements pursuant to the Real Property Leases, and that are material individually or in the aggregate to the operation of the Business; but specifically excluding, with respect to Owned Local Real Property, industrial assets that are subject to a pending or granted application under chapter 24 section 1 of the Swedish Real Property Code.
- "Material Third Party Licenses In" means all Third Party Licenses In that are material to the operation of the Business.
- "Material Third Party Licenses Out" means all Third Party Licenses Out that are material to the operation of the Business.
- "Measuring Time" means 11:59 p.m. local time on the last day of the month preceding the Closing Date.
- "Member State Filing Deadline" has the meaning set forth in Section 5.1(c).
- "Model Year" shall have the meaning by which such term is defined or understood (as the case may be) in the EPA GHG Emissions Regulations..
- "Net External Indebtedness" means the net amount of assets and liabilities of the Target Companies and Company Subsidiaries (including, where applicable, joint ventures in accordance with Schedule A-1) to be included in Net External Indebtedness as set forth in the Final Closing Statement in accordance with Schedule A-1, provided that if the resulting amount is a net liability then Net External Indebtedness shall be a positive number, and if the resulting amount is a net asset then Net External Indebtedness shall be a negative number.
- "Net External Indebtedness Adjustment" means the amount, if any, payable by Sweden Buyer to Parent or, as applicable, by Parent to Sweden Buyer pursuant to Section 2.5(b)(ii).
- "Net Initial Purchase Price" has the meaning set forth in Section 2.3.
- "Net Working Capital" means the net amount of assets and liabilities (treating assets as positive numbers and liabilities as negative numbers) of the Target Companies and Company Subsidiaries to be included in Net Working Capital as set forth in the Final Closing Statement in accordance with Schedule A-1, provided that if the resulting amount is a net asset then Net Working Capital shall be a positive number and if the resulting amount is a net liability Net Working Capital shall be a negative number.
- "Net Working Capital Adjustment" means the amount, if any, payable by Sweden Buyer to Parent or Parent to Sweden Buyer pursuant to Section 2.5(b) (iii).

- "New Plan" has the meaning set forth in Section 6.18(a)(ii).
- "Non-Controlled Company" means any entity of which any Target Company or Company Subsidiary owns, directly or indirectly, or controls (through agreement or otherwise) 50% or less but more than 40% of any outstanding securities or other ownership interests, including Volvo Trademark Holding AB.
- "Non-Controlled IP" means any Intellectual Property Rights which are owned by any Non-Controlled Company and which are (i) in and to the Volvo brand ("Non-Controlled Volvo Brand IP"), or (ii) material to the Business.
- "Non-Controlled IP Knowledge of Parent" means of a fact, event, circumstance or occurrence actually known by the individuals listed on Schedule FF after making reasonable inquiry.
- "Non-Controlled Volvo Brand IP" has the meaning set forth in the definition of Non-Controlled IP in this Appendix 1.
- "Non-Production Agreements" means the Purchasing Agreements of a Target Company or Company Subsidiary other than the Production Agreements, relating to payment or receipt by a Target Company or Company Subsidiary of more than \$5,000,000 per year.
- "NSC Reorganization" means the transfer of the Embedded NSCs to Sweden Company, a Company Subsidiary or a third party importer as set forth on Schedule F in accordance with the Separation Arrangements.
- "NSC Transitional Services Agreement" means the NSC Transitional Services Agreement to be entered into between Parent and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XIX.
- "Order" has the meaning set forth in Section 3.2(d)(i).
- "Ordinary Course of Business" means the ordinary course of business consistent with past practices.
- "Outside Date" has the meaning set forth in Section 11.1(f).
- "Owned Foreign Real Property" means all real property located outside of Sweden that is owned by a Target Company or Company Subsidiary, except for such real property that is not, individually or in the aggregate, material to the operation of the Business.
- "Owned Local Real Property" means all real property located in Sweden that is owned by a Target Company or Company Subsidiary, except for such real property that is not, individually or in the aggregate, material to the operation of the Business.
- "Owned Real Property" means the Owned Local Real Property together with the Owned Foreign Real Property.
- "Parent" has the meaning set forth in the preamble to this Stock Purchase Agreement.
- "Parent Assigned Intellectual Property" means the Intellectual Property Rights that are to be assigned to Sweden Company under the Intellectual Property Assignment Agreement, provided however, for representations and warranties provided in Section 3.14 and given as of the date of execution and delivery

of this Stock Purchase Agreement, this definition shall be limited to those Intellectual Property Rights in existence at such time.

- "Parent Defined Contribution Plan" has the meaning set forth in Section 6.20(a).
- "Parent Group" means Parent and any Affiliates of Parent, other than Target Companies and the Company Subsidiaries.
- "Parent Indemnitee" has the meaning set forth in Section 10.4.
- "Parent IT System" means IT Applications and IT Infrastructure that are (a) owned by a member of the Parent Group or provided to a member of the Parent Group or a Target Company or Company Subsidiary pursuant to a third party Agreement under which a member of Parent Group is a party; (b) used by a Target Company or a Company Subsidiary (or any of their respective suppliers, dealers or customers in or in relation to one or more Vehicles) as at Closing; and (c) identified in Schedule W.
- "Parent IT Transition Services" means the provision of the Parent IT System, along with technical assistance, support, maintenance and other services relating to the operation and installation of the Parent IT System in accordance with the terms of the IT Agreement.
- "Parent Licensed Intellectual Property" means the Intellectual Property Rights that are to be licensed to Sweden Company under the Intellectual Property License Agreement; provided however, for representations and warranties provided in Section 3.14 and given as of the date of execution and delivery of this Stock Purchase Agreement, this definition shall be limited to those Intellectual Property Rights that are in existence at such time.
- "Parent Senior Employee" means an employee of Parent or a member of Parent Group whose grade is at or above that of a Leadership Level 5 Employee.
- "Parent Tax Group" means Parent and any other company or companies (other than any Target Company or Company Subsidiary) which either are or become after Closing, or have within the six years ending at Closing been, treated as members of the same group as, or otherwise connected or associated in any way with, Parent for any Tax purpose.
- "Party" has the meaning set forth in the preamble to this Stock Purchase Agreement.
- "Payment Currency" has the meaning set forth in Section 12.7.
- "PDF" has the meaning set forth in Section 12.11.
- "Pension Plan #3" has the meaning set forth in Section 5.10(a).
- "Pension Plan #6" has the meaning set forth in Section 5.10(b).
- "Permit" means a certificate, franchise, license, permit, order, concession, authorization, registration, declaration, filing, approval or similar right granted by a Governmental Entity.

"Permitted Encumbrances" means:

- (a) the documents and matters set forth on Schedule E and certain matters reflected in the Company Financial Statements and set forth in the documents listed on Schedule E, copies of which documents have been made available to Buyers; together with
- (b) (i) statutory liens for current Taxes not yet due, payable or delinquent, or which may be paid without interest or penalties, or the validity or amount of which is being contested in good faith by appropriate proceedings, and for which contested amount adequate reserves are maintained on the Company Financial Statements, (ii) liens (including mechanics', carriers', workers', repairers' and other similar liens, but excluding Tax liens) arising or incurred in the Ordinary Course of Business relating to obligations as to which there is no default on the part of a Target Company or Company Subsidiary, as applicable, or the validity or amount of which is being contested in good faith by appropriate proceedings, and only to the extent that such liens do not or would not reasonably be expected to, individually or in the aggregate, materially impair the continued use and operation of the asset to which such liens relate in the Ordinary Course of Business, (iii) pledges, deposits or other liens securing the performance of bids, trade contracts (other than for borrowed money), leases or statutory obligations (including workers' compensation, unemployment insurance or other social security legislation, but excluding Tax liens) that are incurred in the Ordinary Course of Business and none of which, individually or in the aggregate, materially impair the continued use and operation of the asset to which such liens relate in the Ordinary Course of Business, (iv) Encumbrances imposed by Law, including zoning, entitlement, conservation restriction and other land use and environmental regulations by a Governmental Entity, none of which, individually or in the aggregate, materially impair the current occupancy or current use of such real property, (vi) minor survey exceptions and matters as to real property, as applicable, (v) all covenants, conditions, restrictions, utility easements, easements, eacroachments, joint facilities, charges, rights of way, other liens and any similar matter set forth in any state, local or munic

"Person" means an individual, association, a corporation, a partnership, a limited liability company, a limited liability partnership, a trust or any other entity or organization, including a Governmental Entity.

"Phase I Environmental Site Assessment" means the Phase I environmental site assessments prepared by the Environmental Consultants and made available to Buyer Group and Geely Group.

"PLA" has the meaning set forth in Section 5.23.

"Plan" means any employee benefit scheme, plan, program, policy, practice, or other arrangement providing benefits to any current or former Employee or any beneficiary or dependent thereof that is sponsored or maintained by Parent, Sweden Seller, a Target Company or Company Subsidiary or to which Parent, Sweden Seller, a Target Company or Company Subsidiary contributes or is obligated to

contribute, including any pension, retirement, supplemental retirement, deferred compensation, bonus, incentive, vacation, insurance, stock purchase, stock option, equity award, severance, employment, change of control or fringe benefit plan, program or Agreement.

"PRC" means the People's Republic of China which, for the purposes of this Stock Purchase Agreement, excludes Hong Kong Special Administrative Region, Macau Special Administrative Region and the territory of Taiwan.

"Pre-Closing Period" has the meaning set forth in Section 6.2(a)(i).

"Products" means Vehicles and Vehicle parts and accessories.

"Product Recall" means a vehicle repair campaign (whether voluntary or required by a Governmental Entity) to address product safety defects or non-compliances or emissions defects or non-compliances where a Target Company or Company Subsidiary, their respective importer, national sales company or other market representative has reported the defect to a Governmental Entity and has directly notified each affected customer to deliver the vehicle in question to an authorized service provider for repair, and where the subject repair was performed at no cost to the customer. For the avoidance of doubt, "Product Recalls" shall not include any campaigns or other actions to honor or otherwise satisfy any Vehicle warranty, extended warranty, extended service plan or similar plan or program.

"Product Regulatory Requirement" means a mandatory legal standard, certification or approval related to motor vehicle safety, emissions or the environment, that must be met or obtained by a Target Company or Company Subsidiary in order for Vehicles to be legally sold or imported in countries where a Target Company or Company Subsidiary sells or imports Vehicles.

"Product Supply Agreements" means all of the following Agreements: (a) the Agreement for the supply of engines and stampings dated as of the Closing Date among Ford Motor Company Limited, Ford-Werke GmbH, Ford España, S.L. and Sweden Company; (b) the Agreement for the supply of engines and stampings dated as of the Closing Date among Sweden Company, Ford Motor Company Limited and Ford España, S.L.; (c) the Agreement for the supply of conrods and brake discs dated as of the Closing Date among Sweden Company, Ford-Werke GmbH, and Ford España, S.L.; (d) the Agreement for the supply of conrods and brake discs dated as of the Closing Date among Sweden Company, Ford Motor Company Limited and Ford-Werke GmbH; and (e) the Agreement for the supply of camshafts dated as of the Closing Date among Sweden Company, Ford Motor Company Limited and Ford-Werke GmbH.

"Production Agreements" means the purchasing agreements of a Target Company or Company Subsidiary, as applicable, relating specifically to vehicle and component manufacturing and production, relating to payment or receipt by a Target Company or Company Subsidiary of more than \$20,000,000 per year.

"Production Goods and Services" means goods and services including (a) production and service parts, components, assemblies and accessories, (b) raw materials, (c) tooling and (d) design, engineering or other services.

"Production Purchasing Global Terms and Conditions" means the primary contract terms and conditions that apply to the purchase of Production Goods and Services which in each case shall for the purposes of this Stock Purchase Agreement include applicable supplemental terms and conditions issued to address special product or local market requirements and applicable purchase orders that specify price and other key terms.

"Project Company" means (a) the Jiading Project Co. (as defined in recital (B) of the Framework Agreement on Geely's Shanghai Project by and among Management Committee of Shanghai Jiading Industrial Park, Shanghai Jiading Industrial Park Development (Group) Co., Ltd., Zhejiang Geely Holding Group Co., Ltd., and Mr. Li Shufu, dated February 9, 2010), (b) the Daqing Volvo Company (as described in paragraph 3.1 of the Investment Cooperation Agreement by and between Daqing State Owned Assets Operating Co., Ltd. and Zhejiang Geely Holding (Group) Co., Ltd., dated February 7, 2010), (c) any company formed or otherwise set up to own or operate the Zhangjiakou base (as described in the Geely Group Zhangjiakou Investment Project Cooperation Agreement by and among People's Government of Zhangjiakou Municipality, Zhangjiakou Tongtai Highway Group Co., Ltd. and Zhejiang Geely Holding (Group) Co., Ltd., dated March 18, 2010), and (d) any other company Controlled by Zhejiang Geely Holding (Group) Co., Ltd. which is formed or otherwise set up to develop, manufacture or assemble Volvo Branded Vehicles in China, or which is engaged in the development, manufacture, or assembly of Volvo Branded Vehicles in China.

"Proposal" has the meaning set forth in Section 5.17.

"PSA" means PSA S.A., a company incorporated under the Laws of France.

"PSA Group" means PSA and its Affiliates.

"Purchase Price" has the meaning set forth in Section 2.1(b).

"Purchasing Agreements" means the Production Agreements and the Non-Production Agreements.

"PZEV" means partial zero emissions vehicle as defined by the California Air Resources Board.

"Real Property Leases" means the Local Real Property Leases together with the Foreign Real Property Leases.

"Relevant Employees" means those employees (including temporary, agency and permanent employees) and consultants of a member of Parent Group, a member of Buyer Group, or a member of Geely Group who would reasonably be expected to have access to Restricted Information following Closing.

"Relief" includes, unless the context otherwise requires, any allowance, credit, deduction, exemption or set-off in respect of any Tax or relevant to the computation of any income, profits or gains for the purposes of any Tax, or any repayment of or saving of Tax (including any interest in respect of Tax) and any reference to the "use" or "set-off" of relief shall be construed accordingly and shall include use or set-off in part.

"Reorganization" means the Separation Arrangements, and the transactions contemplated by them, including the NSC Reorganization, as well as the transfer, assignment, sale or licensing of any Intellectual Property Rights and any other transactions carried out or Agreements entered into in preparation for or in connection with the sale of the U.S. Company Interest and Sweden Company Shares.

"Replaced Company Plan" has the meaning set forth in Section 6.18(a)(ii).

"Replacement Canada Defined Contribution Plans" has the meaning set forth in Section 5.10(b).

"Replacement Policies" has the meaning set forth in Section 5.13(a).

- "Replacement Provider" means the third party or third parties appointed by a VAB Manufacturer to provide the Services in substitution for the VAB Suppliers.
- "Replacement U.S. Defined Contribution Plans" has the meaning set forth in Section 5.11(c).
- "Replacement Vehicle Finance Arrangements" has the meaning set forth in Section 5.15.
- "Represented Employees" has the meaning set forth in Section 3.17(b).
- "Request for Arbitration" has the meaning set forth in the ICC Rules.
- "Required Terms" has the meaning set forth in Section 5.23.
- "Restricted Information" means confidential information of a member of Parent Group, on the one hand, and a member of Buyer Group or a member of Geely Group, on the other hand, (or confidential information of a third party, including the PSA Group, that is in the possession of a member of Parent Group, or a member of Buyer Group or a member of Geely Group) that includes: (a) prices or pricing plans in relation to the supply of vehicles or parts to downstream customers of vehicles or parts, (b) margins or margin targets in relation to the supply of vehicles or parts to downstream customers, (c) raw material, component and manufacturing costs which would not normally be disclosed or made accessible as between competitors, (d) plant capacity, capacity utilization and allocation of capacity to different products which would not normally be disclosed or made accessible as between competitors, (e) product development and launch plans and activities (other than with respect to products jointly developed pursuant to a Separation Agreement) or (f) any other information that reveals current or future business plans or anticipated commercial conduct of a member of Buyer Group, a member of Geely Group or a member of Parent Group, or of a third party that is in the possession of a member of Buyer Group, a member of Geely Group or a member of Parent Group, on the one hand, or by or to a member of Buyer Group, to the extent that the disclosure of such information by or to a member of Parent Group, on the one hand, or by or to a member of Buyer Group or a member of Geely Group, on the other hand, would give rise to an infringement of competition Law, but specifically excluding (x) information as to the total price paid for materials comprised in the goods (at the agreed cost disclosure level) to any customer of the goods provided pursuant to the Separation Agreements; and (y) information as to the actual capacity reserved by a customer where a customer has an obligation to share the available capacity with the supplier pursuan
- "Retirement Benefits" means any pension, lump sum, gratuity or similar benefit (including post-retirement medical, dental and welfare benefits and termination indemnity (or seniority) payments) payable or prospectively or contingently payable on or following retirement, leaving service, invalidity and death.
- "Retail Finance" means, collectively, any and all retail installment credit contracts relating to Vehicles entered into between a member of Parent Group and a Target Company Customer and which are originated or purchased by a member of Parent Group by way of Volvo Finance.
- "RMB" denotes the lawful currency of the PRC.
- "S40 TLC" means the Volvo S40 Vehicle Technology Licensing Contract among Sweden Company, Ford Global and CFMA.

- "S80 TLC" means the Volvo S80 LWB Vehicle Technology Licensing Contract among Sweden Company, Ford Global and CFMA.
- "Schedule" means a schedule attached to this Stock Purchase Agreement, as the same may be amended or supplemented from time to time after the date of this Stock Purchase Agreement and prior to Closing in accordance with the terms of this Stock Purchase Agreement.
- "Secondment Agreements" means the Ford Secondment Agreement and the Volvo Secondment Agreement.
- "Seller Claim" has the meaning set forth in Section 10.4.
- "Seller Financing Agreement" means the Seller Financing Agreement by and between Sweden Seller and the relevant member of Geely Group, substantially in the form attached hereto as Exhibit XXVIII and any financing documents to be delivered in connection therewith.
- "Seller Financing Amount" means \$200,000,000.
- "SEK" denotes the lawful currency of Sweden.
- "Separation Agreements" means the Agreements to be entered into by certain members of Parent Group, on the one hand, and certain Target Companies or Company Subsidiaries, on the other hand, in the forms attached to this Stock Purchase Agreement as Exhibit VI, Exhibit VII, Exhibit VIII, Exhibit IX, Exhibit X, Exhibit XI, Exhibit XII, Exhibit XIII, Exhibit XIV, Exhibit XVI, Exhibit XVII, Exhibit XVIII, Exhibit XIX, Exhibit XXI and Exhibit XXVI.
- "Separation Arrangements" means, in relation to an Embedded NSC, as applicable (a) formation of a new entity in the jurisdiction listed for such Embedded NSC, (b) acquisition by Sweden Company or a Company Subsidiary of the shares of such new entity, (c) establishment of third party contractual arrangements and applications for, and receipt of, Permits, licenses and registrations, (d) transfer of assets and assumption of Liabilities, (e) separation of properties, (f) separation of employee pensions and benefits arrangements and (g) all other actions taken by a member of Parent Group, a Target Company or Company Subsidiary in connection with (a) through (f) above.
- "Separation IP" means the Intellectual Property created, developed or invented by, or on behalf of, a Party or its Affiliates after Closing (a) in materials, documentation and other items, (b) in making those materials, documentation or other items, and (c) in any process used in making those materials, documentation or other items, in each case with respect to those materials, documentation or other items that are required to be delivered by that party or its relevant Affiliate pursuant to any of the Development Agreements.
- "Services" means the respective services listed in Section 2.1.4 of the VAB Transitional Services Agreement.
- "Shanghai Holdco" means Shanghai Geely Zhao Yuan International Investment Co., Ltd., a company organized under the Law of China.
- "Shared Contract" means an Agreement, excluding any Excluded Contract, with a third party to which (a) a member of Parent Group is a party and pursuant to which a Target Company or Company Subsidiary is a party or receives a benefit or has an obligation to perform, or (b) a Target Company or Company

Subsidiary is a party and pursuant to which a member of Parent Group is a party or receives a benefit or has an obligation to perform.

"Sites" means any real property (a) owned, leased or occupied by a Target Company or Company Subsidiary as of the Closing Date, or (b) owned, leased or occupied by a Target Company or a Company Subsidiary during the five-year period prior to the Closing Date.

"Special GHG Credits" has the meaning set forth in Section 5.22(a).

"Specified Net Working Capital" means a negative amount of \$1,343,000,000.

"Specified Representations" has the meaning set forth in Section 10.1(a).

"SSIP" has the meaning set forth in Section 5.10(b).

"Statutory Audit Companies" means the Target Companies and Company Subsidiaries set forth on Schedule Q.

"Stock Purchase Agreement" has the meaning set forth in the preamble to this Stock Purchase Agreement.

"Straddle Period" has the meaning set forth in Section 6.2(a)(i).

"Subsidiary" means a corporation, partnership, joint venture or other entity of which any Person (a) owns, directly or indirectly, or controls (through agreement or otherwise) more than 50% of the outstanding securities or other interests the holders of which are generally entitled to vote for the election of the board of directors or other governing body or (b) has the right to appoint or remove a majority of the board of directors or other governing body.

"Surviving Contracts" has the meaning set forth in Section 6.10(a).

"Sweden Buyer" has the meaning set forth in the preamble to this Stock Purchase Agreement.

"Sweden Company" means Volvo Car Corporation, a corporation organized under the Law of Sweden.

"Sweden Company GHG Credit Shortfall" means, in respect of a Model Year, Target Group's GHG Credit shortfall for Vehicles in that Model Year under the EPA GHG Emissions Standards, reduced by (i) the Special GHG Credits (without duplication in the calculation of such shortfall), (ii) any GHG Credits generated, or projected to be generated, by Target Group in prior or subsequent Model Years that are eligible under the EPA GHG Emissions Regulation to be applied by a member of Target Group to the applicable credit balance for the Model Year including any A/C leakage credits; and (iii) any portion of such actual GHG Credit shortfall under the EPA GHG Emissions Standards that is attributable to actual deviations from the Cycle Plan and the Business Plan to the extent of the sales plan contained therein.

"Sweden Company Management Team" means Steven Armstrong, Olle Axelson, Bengt Banck, Bernt Ejbyfeldt, Magnus Hellsten, Magnus Jonsson, Gerry Keaney, Stephen Odell, Stuart Rowley, Björn Sällström and Elisabet Wenzlaff.

- "Sweden Company PZEV Credit Shortfall" means the total number of additional PZEV credits estimated as of the date of this Stock Purchase Agreement to be required by Sweden Company for the 2010, 2011, 2012, 2013, 2014 and 2015 model years as set forth on Schedule L.
- "Sweden Company Shares" has the meaning set forth in the recitals to this Stock Purchase Agreement.
- "Sweden Seller" has the meaning set forth in the preamble to this Stock Purchase Agreement.
- "Swedish Entities" means Sweden Company and each Company Subsidiary incorporated under the Law of Sweden.
- "Swedish GAAP" means Swedish generally accepted accounting principles.
- "Target Companies" means U.S. Company and Sweden Company.
- "Target Companies IT System" means IT Applications and IT Infrastructure that are (a) owned by a Target Company or Company Subsidiary or provided to a member of Parent Group or a Target Company or Company Subsidiary pursuant to a third party agreement under which a Target Company or Company Subsidiary is a party; (b) used by Parent Group at Closing; and (c) to the Knowledge of Parent, listed in all material respects on Schedule T, as such Schedule may be updated between the date of this Stock Purchase Agreement and Closing.
- "Target Companies IT Transition Services" means the provision of the Target Companies IT System, along with technical assistance, support, maintenance and other services relating to the operation and installation of the Target Companies IT System in accordance with the terms of the IT Agreement.
- "Target Company Customers" means those obligors that finance or lease a Vehicle by way of Volvo Finance and for the avoidance of doubt such obligors shall include FSL Customers, Credit Insurance Customers and any Target Company Customers that enter into fleet and daily rental arrangements (including where such arrangements are negotiated directly with Target Company).
- "Target Company Senior Employee" means an Employee whose salary grade is at or above that of a Leadership Level 5 Employee.
- "Target Group" means the Target Companies and the Company Subsidiaries.
- "Tax" or "Taxes" means all taxes of any kind, including all national, state, provincial, territorial, municipal or local income, profits, franchise, gross receipts, environmental, customs, import, export, net worth, sales, use, goods and services, value added, ad valorem, withholding, employment, social security, disability, occupation, pension, real property, tangible and intangible personal property, stamp, transfer, conveyance, severance, production, excise and other taxes, withholdings, duties, levies, imposts and other similar charges and assessments (including any and all fines, penalties, interest and other additions attributable to or otherwise imposed on or with respect to any such taxes, charges, fees, levies or other assessments and interest thereon) imposed by or on behalf of a Governmental Entity.
- "Tax Authority" means any Governmental Entity competent to impose any Tax, or assess or collect any Tax.
- "Tax Claim" means any claim pursuant to Section 6.2 or any claim for a breach of any of the warranties in Section 3.12 or for a breach of the covenants and undertakings set out in Section 6.6 (including, for the

avoidance of doubt, a claim for indemnification in respect of the breach of such warranties pursuant to Section 10.2(a)).

"Tax Liability" means:

- (a) an Actual Tax Liability; or
- (b) a Deemed Tax Liability.

"Tax Proceeding" has the meaning set forth in Section 6.5(a).

"Tax Refund Payment" has the meaning set forth in Section 6.2(e).

"Tax Return" means any return, declaration, report, claim for refund, information return or statement required to be supplied to any Tax Authority in relation to any Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Temporary Lead-time Allowance Alternative Standards" are the final regulations, if any, promulgated by the U.S. Environmental Protection Agency with respect to optional interim fleet average green house gas emissions standards, flowing from the U.S. Environmental Protection Agency's proposal to codify such regulations at 40 C.F.R. § 86.1818-12(e) in its Notice of Proposed Rulemaking published on September 28, 2009, and published at 74 Fed. Reg. 49454, as may be further described in EPA GHG Emissions Regulations.

"Terms of Reference" has the meaning set forth in the ICC Rules.

"Theoretical Assets Allocated to VCJ" means the assets as calculated pursuant to Schedule P Part 1.

"Theoretical Assets Allocated to FJL" means the assets as calculated pursuant to Schedule P Part 2.

"Third Party Claim" has the meaning set forth in Section 10.7.

"Third Party IT Agreements" means all contracts or other agreements with third parties relating to the IT Systems, including software licenses and support agreements.

"Third Party Licenses In" means all licenses granted by third parties to a member of Parent Group, a Target Company or a Company Subsidiary of Intellectual Property Rights owned by or licensed to such third parties and which are used by a Target Company or Company Subsidiary (except for (a) licenses for off-the-shelf software, publications or newsletters, (b) IP Settlement Agreements, and (c) purchase orders), provided that "third party" shall include any entity of which any Target Company owns, directly or indirectly, or controls (through agreement or otherwise) 40% or less of any outstanding securities or other ownership interests.

"Third Party Licenses Out" means all licenses granted by a member of Parent Group, a Target Company or a Company Subsidiary to third parties of Company Intellectual Property or Parent Assigned Intellectual Property (except for (a) licenses for off-the-shelf software, publications or newsletters, (b) IP Settlement Agreements, and (c) purchase orders), provided that "third party" shall include any entity of which any Target Company owns, directly or indirectly, or controls (through agreement or otherwise) 40% or less of any outstanding securities or other ownership interests.

- "Third Party Refund Payment" has the meaning set forth in Section 10.9.
- "Third Party Supplier" means a Third Party with which Credit Company has a contractual arrangement for such Third Party to provide FSL or Insurance by way of Volvo Finance.
- "Tooling Agreement" means the Tooling Agreement to be entered into between certain Parent Group Entities and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XVI.
- "Total Pension Adjustment" means the Total Pension Adjustment calculated in accordance with Schedule A-2.
- "Transaction Documents" means this Stock Purchase Agreement, the Separation Agreements and the Seller Financing Agreement.
- "Transaction Financial Statements" means the Dearborn-administered Hyperion Enterprise (HE) consolidated accounts in respect of the Target Companies and the Company Subsidiaries as of the year ended December 31, 2008 set forth on Schedule BB.
- "Transitional Services Agreement" means the Transitional Services Agreement to be entered into between Parent and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XVIII.
- "Transfer Taxes" has the meaning set forth in Section 6.3.
- "UK Pensions Act" means the United Kingdom Pensions Act 2004.
- "Undertakings" has the meaning set forth in Section 5.1(g).
- "Undisclosed Individual" has the meaning set forth in Section 6.16(b).
- "U.S." and "United States" mean the United States of America.
- "U.S. Buyer" has the meaning set forth in the preamble to this Stock Purchase Agreement.
- "U.S. Code" means the U.S. Internal Revenue Code of 1986, as amended.
- "U.S. Company" means Volvo Cars of North America, LLC, a Delaware limited liability company.
- "U.S. Company Interest" has the meaning set forth in the recitals to this Stock Purchase Agreement.
- "U.S. Dealer Agreement" means any dealer Agreement, sales Agreement, retailer Agreement, franchise Agreement, letter of intent, or similar Agreement, including any amendments, addenda, riders or other writings by and between U.S. Company, or its predecessors, on the one hand, and any dealers or prospective dealers in the United States, on the other hand, relating to the sale or distribution of Vehicles.
- "U.S. Dollars," "USD" and "\$" each denote the lawful currency of the U.S.
- "U.S. Employee Pension Participants" has the meaning set forth in Section 5.11(a).
- "U.S. Employee Pensions Plans" has the meaning set forth in Section 5.11(a).

- "U.S. Employee Pensions Plan Transferred Assets" has the meaning set forth in Section 6.21.
- "U.S. GAAP" means U.S. generally accepted accounting principles.
- "U.S. Hourly Employee Pension Participants" has the meaning set forth in Section 5.11(b).
- "VAB" means Volvo Auto Bank Deutschland GmbH, a German corporation.
- "VAB Manufacturer" means Volvo Car Germany GmbH or Volvo Automobile (Schweiz) AG.
- "VAB Supplier" means Volvo Car Holding Germany GmbH, VAB, VAL and VAF
- "VAB Transitional Services Agreement" means the VAB Services Agreement to be entered into between certain Credit Company Affiliates and certain Target Companies and Company Subsidiaries, in the form attached to this Stock Purchase Agreement as Exhibit XXI.
- "VAF" means Volvo Auto Finanz-Service Deutschland GmbH, a German corporation.
- "VAL" means Volvo Auto Leasing Deutschland GmbH, a German corporation.
- "VAT" means value added tax chargeable under or pursuant to the EC Council Directive 2006/112/EC on the common system of value added tax or any similar sales, purchase or turnover tax chargeable outside the European Union, including any sales tax imposed by any state or local government within the United States.
- "VCJ" has the meaning set forth in Section 6.23(a)(i).
- "VCJ Actuary" has the meaning set forth in Section 6.23(c).
- "VCJ Amount" has the meaning set forth in Section 6.23(c).
- "Vehicle" means a vehicle manufactured or distributed by a Target Company or Company Subsidiary.
- "Vehicle Finance Arrangements" means collectively, Floor Plan Finance, Lease Finance and Retail Finance.
- "Vehicle Finance Separation Agreement" means the Vehicle Finance Separation Agreement to be entered into between Credit Company and Sweden Company, in the form attached to this Stock Purchase Agreement as Exhibit XX.
- "Volvo Assigned Patents" has the meaning set forth in the Intellectual Property License Agreement.
- "Volvo Branded Vehicles" has the meaning set forth in the Intellectual Property License Agreement.
- "Volvo Finance" means collectively the Volvo finance trade brand names set forth in Schedule 4 of the Vehicle Finance Services Agreement.
- "Volvo Netherlands Defined Benefit Plan" has the meaning set forth in Section 5.12(a).
- "Volvo Netherlands Funding Vehicle" has the meaning set forth in Section 5.12(a).

- "Volvo Secondment Agreement" means the Secondment Agreement to be entered into between Parent and Sweden Company in relation to Employees, in the form attached to this Stock Purchase Agreement as Exhibit XVII.
- "Volvo Trademark Holding License Agreement" means the Volvo Cars Trademarks License Agreement between Volvo Trademark Holding AB and Sweden Company, effective on February 28, 1999.
- "Volvo Trademark Holding Shareholders' Agreement" means the Shareholders' Agreement, by and among Volvo Trademark Holding AB, AB Volvo, Sweden Company and Parent, effective on March 31, 1999.
- "Volvo U.S. Defined Benefit Plan" has the meaning set forth in Section 5.11(b).
- "Working Capital Facility" has the meaning set forth in Section 5.18(d).
- "Working Capital Forecast" has the meaning set forth in Section 5.24.
- "Works" means any of the following, to the extent required by Environmental Law:
 - (a) inspecting, investigating, assessing, sampling or monitoring measures in relation to a Hazardous Material;
 - (b) carrying out of measures to prevent, treat, abate, remove, remediate, control or contain the presence, escape effect or potential effect of a Hazardous Material; and
 - (c) restoring of the environment affected by any Hazardous Material.

Executive Compensation Recoupment Policy

On March 26, 2010, the Compensation Committee of the Board of Directors of Ford Motor Company formally adopted a policy of recoupment of compensation in certain circumstances. The purpose of this policy is to help ensure executives act in the best interests of the Company. The policy requires all Company officers to repay or return cash bonuses and/or equity awards in the event: (i) the Company issues a material restatement of its financial statements and where the restatement was caused by an employee's intentional misconduct; (ii) the employee was found to be in violation of non-compete provisions of any plan or agreement; or (iii) the employee has committed ethical or criminal violations. The Compensation Committee will consider all relevant factors and exercise business judgment in determining any appropriate amounts to recoup and has the discretion to determine the timing and form of recoupment. The policy will apply to the Annual Incentive Compensation Plan beginning with the 2010 performance period and equity awards beginning with grants made in 2011.

Amendment Dated May 5, 2010 to the Consulting Agreement between Ford Motor Company and Edsel B. Ford II

On May 5, 2010, Ford Motor Company and Edsel B. Ford II, a director of and consultant to the Company, agreed to amend Mr. Ford's January 1999 consulting agreement to provide that, effective beginning with the third calendar quarter of 2010 and for the remainder of 2010, the payment of the consulting fee of \$125,000 per calendar quarter shall be paid in cash instead of in restricted shares of Company common stock. For each year thereafter that the consulting agreement remains in effect, the Company and Mr. Ford will agree prior to December 31 of each year whether the subsequent yearly fee will be paid in cash or restricted stock.

FORD MOTOR COMPANY AND SUBSIDIARIES

CALCULATION OF RATIO OF EARNINGS TO COMBINED FIXED CHARGES (a) (in millions)

	First Quarter 2010	2009	2008	2007	2006	2005
Earnings Income/(Loss) before income taxes and cumulative effects of changes in accounting principles (b) Less: Equity in net (income)/loss of affiliated companies included in income/(loss) before	\$ 2,135	\$2,599	\$(14,895)	\$ (4,286)	\$(15,490)	\$ 633
income taxes Adjusted income/(loss) Adjusted fixed charges (c) Earnings/(Losses)	(142) 1,993 1,862 \$ 3,855	2,263 7,395	(382) (15,277) 10,518 \$ (4,759)	(4,925) 11,753	(599) (16,089) 9,439 \$ (6,650)	151 9,240
Combined Fixed Charges Interest expense (d) Interest portion of rental expense (e)	<u> </u>	\$6,818 258	\$ 9,787 322	\$11,036 344	\$ 8,804 327	\$8,452 511
Total combined fixed charges Ratios Ratio of earnings to combined fixed charges	\$ 1,765 2.2	\$7,076 1.4	\$ 10,109 (f)	\$11,380 (f)	\$ 9,131 (f)	1.0

- (a) Discontinued operations are excluded from all amounts. There were no preferred stock dividends in the periods displayed.
- (b) Income/(Loss) before income taxes includes equity income/(loss) from unconsolidated subsidiaries.
- (c) Combined fixed charges, as shown above, adjusted to exclude capitalized interest, and to include dividends from affiliated companies as well as amortization of capitalized interest. (Capitalized interest (in millions): 2010 Q1 \$6; 2009 \$28; 2008 \$50; 2007 \$47; 2006 \$53; 2005 \$65)
- (d) Includes interest, as shown on our statement of operations, plus capitalized interest.
- (e) One-third of all rental expense is deemed to be interest.
- (f) Earnings/(Losses) were inadequate to cover fixed interest charges by (in billions): 2008 \$14.9; 2007 \$4.6; 2006 \$15.8.

May 7, 2010

Securities and Exchange Commission 100 F Street, N.E. Washington, DC 20549

RE: Ford Motor Company Registration Statement Nos. 33-39402, 33-54348, 33-55847, 33-62227, 333-02735, 333-20725, 333-31466, 333-46295, 333-47733, 333-56660, 333-57596, 333-58697, 333-65703, 333-71380, 333-74313, 333-85138, 333-87619, 333-104063, 333-113584, 333-123251, 333-138819, 333-138821, 333-149453, 333-149456, 333-153815, 333-156630, 333-156631, and 333-157584, 333-162992, 333-162993, and 333-165100 on Form S-8 and Nos. 333-151355 on Form S-3.

Commissioners:

We are aware that our report dated May 7, 2010 on our review of interim financial information of Ford Motor Company (the "Company") for the three-month periods ended March 31, 2010 and 2009 and included in the Company's quarterly report on Form 10-Q for the quarter ended March 31, 2010 is incorporated by reference in the aforementioned Registration Statements.

Very truly yours,

/s/ PricewaterhouseCoopers LLP

PricewaterhouseCoopers LLP Detroit, Michigan

CERTIFICATION

I, Alan Mulally, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q for the quarter ended March 31, 2010 of Ford Motor Company;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to
 make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the
 period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 7, 2010 /s/ Alan Mulally
Alan Mulally

President and Chief Executive Officer

CERTIFICATION

I, Lewis Booth, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q for the quarter ended March 31, 2010 of Ford Motor Company;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to
 make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period
 covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 7, 2010 /s/ Lewis Booth L.W.K. Booth

Executive Vice President and Chief Financial Officer

FORD MOTOR COMPANY

CERTIFICATION OF CHIEF EXECUTIVE OFFICER

I, Alan Mulally, President and Chief Executive Officer of Ford Motor Company (the "Company"), hereby certify pursuant to Rule 13(a)-14(b) or 15d-14(a) of the Securities Exchange Act of 1934, as amended, and Section 1350 of Chapter 63 of Title 18 of the United States Code that to my knowledge:

- 1. The Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2010, to which this statement is furnished as an exhibit (the "Report"), fully complies with the requirements of Rule 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in this Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 7, 2010 /s/ Alan Mulally

Alan Mulally

President and Chief Executive Officer

FORD MOTOR COMPANY

CERTIFICATION OF CHIEF FINANCIAL OFFICER

I, Lewis Booth, Executive Vice President and Chief Financial Officer of Ford Motor Company (the "Company"), hereby certify pursuant to Rule 13a-14(b) or Rule 15d-14(a) of the Securities Exchange Act of 1934, as amended, and Section 1350 of Chapter 63 of Title 18 of the United States Code that to my knowledge:

- 1. The Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2010, to which this statement is furnished as an exhibit (the "Report"), fully complies with the requirements of Rule 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in this Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 7, 2010 /s/ Lewis Booth

L.W.K. Booth Executive Vice President and Chief Financial Officer